In the Supreme Court of the United States

DIAMOND ALTERNATIVE ENERGY, LLC, ET AL., PETITIONERS,

v.

ENVIRONMENTAL PROTECTION AGENCY, ET AL.

ON PETITION FOR A WRIT OF CERTIORARI TO THE UNITED STATES COURT OF APPEALS FOR THE DISTRICT OF COLUMBIA CIRCUIT

PETITION FOR A WRIT OF CERTIORARI

ERIC D. MCARTHUR SIDLEY AUSTIN LLP 1501 K Street NW Washington, DC 20005

MICHAEL BUSCHBACHER JARED M. KELSON BOYDEN GRAY PLLC 801 17th Street NW Suite 350 Washington, DC 20006 JEFFREY B. WALL
Counsel of Record
MORGAN L. RATNER
ZOE A. JACOBY
SULLIVAN & CROMWELL LLP
1700 New York Avenue NW
Suite 700
Washington, DC 20006
(202) 956-7660
wallj@sullcrom.com

LESLIE B. ARFFA SULLIVAN & CROMWELL LLP 125 Broad Street New York, NY 10004

(Additional counsel on signature page)

QUESTIONS PRESENTED

Section 209(a) of the Clean Air Act generally preempts States from adopting emission standards for new motor vehicles. 42 U.S.C. § 7543(a). But under Section 209(b) of that Act, EPA may grant California—and only California—a waiver from federal preemption to set its own vehicle-emission standards. Before granting a preemption waiver, EPA must find that California "need[s]" its own emission standards "to meet compelling and extraordinary conditions." *Id.* § 7543(b)(1)(B).

In 2022, EPA granted California a waiver to set its own standards for greenhouse-gas emissions and to adopt a zero-emission-vehicle mandate, both expressly intended to address global climate change by reducing California vehicles' consumption of liquid fuel. Fuel producers challenged EPA's waiver as contrary to the text of Section 209(b). The D.C. Circuit rejected the challenge without reaching the merits, concluding that fuel producers' injuries were not redressable because they had not established that vacating EPA's waiver would have *any* effect on automakers.

The questions presented are:

- 1. Whether a party may establish the redressability component of Article III standing by relying on the coercive and predictable effects of regulation on third parties.
- 2. Whether EPA's preemption waiver for California's greenhouse-gas emission standards and zero-emission-vehicle mandate is unlawful.

PARTIES TO THE PROCEEDING

Petitioners are Diamond Alternative Energy, LLC, American Fuel & Petrochemical Manufacturers, Clean Fuels Development Coalition, Diamond Alternative Energy, LLC, Domestic Energy Producers Alliance, Energy Marketers of America, ICM, Inc., Illinois Corn Growers Association, Iowa Soybean Association, Kansas Corn Growers Association, Michigan Corn Growers Association, Minnesota Soybean Growers Association, Missouri Corn Growers Association, National Association of Convenience Stores, South Dakota Soybean Association, and Valero Renewable Fuels Company, LLC.

Respondents are the Environmental Protection Agency and Michael S. Regan, in his official capacity as Administrator of the Environmental Protection Agency.

Other petitioners in the court of appeals were the States of Ohio, Alabama, Arkansas, Georgia, Indiana, Kansas, Kentucky, Louisiana, Mississippi, Missouri, Montana, Nebraska, Oklahoma, South Carolina, Texas, Utah, and West Virginia.

Intervenors on behalf of respondents in the court of appeals were Ford Motor Company, Volkswagen Group of America, Inc., American Honda Motor Co., Inc., BMW of North America, LLC, Volvo Car USA LLC, New York Power Authority, National Grid USA, Calpine Corporation, Advanced Energy Economy, Power Companies Climate Coalition, National Coalition for Advanced Transportation, State of Washington, District of Columbia, State of New Jersey, State of Maine, State of Hawaii, State of Illinois, State of

Maryland, State of Colorado, State of Nevada, State of New York, State of Connecticut, State of Vermont, State of Rhode Island, State of North Carolina, State of California, State of New Mexico, State of Minnesota, State of Delaware, State of Oregon, City of New York, Commonwealth of Pennsylvania, Commonwealth of Massachusetts, City of Los Angeles, Clean Air Council, Natural Resources Defense Council, Public Citizen, Center for Biological Diversity, Environmental Defense Fund, Sierra Club, National Parks Conservation Association, Union of Concerned Scientists, Conservation Law Foundation, and Environmental Law and Policy Center.

RULE 29.6 DISCLOSURE STATEMENT

Petitioner Diamond Alternative Energy, LLC, is a Delaware limited liability company that manufactures biomass-derived liquid fuels. It is a wholly owned direct subsidiary of Valero Energy Corporation, a Delaware corporation whose common stock is publicly traded on the New York Stock Exchange under the ticker symbol VLO.

Petitioner American Fuel & Petrochemical Manufacturers is a national trade association that represents American refining and petrochemical companies. The Association has no parent corporation, and no publicly held corporation has a 10% or greater ownership in it.

Petitioner Clean Fuels Development Coalition is a business league organization established in a manner consistent with Section 501(c)(6) of the Internal Revenue Code. Established in 1988, the Coalition works with auto, agriculture, and biofuel interests in support of a broad range of energy and environmental programs. It has no parent companies, and no publicly held company has a 10% or greater ownership interest in the Coalition.

Petitioner Domestic Energy Producers Alliance is a non-profit, nonstock corporation organized under the laws of the State of Oklahoma. The Alliance has no parent corporation, and no publicly held company owns 10% or more of its stock.

Petitioner Energy Marketers of America is a federation of 47 state and regional trade associations representing energy marketers throughout the United States. It is incorporated under the laws of the Commonwealth of Virginia, has no parent corporation, and

no publicly held corporation has a 10% or greater ownership in it.

Petitioner ICM, Inc. is a Kansas corporation that is a global leader in developing biorefining capabilities, especially for the production of ethanol. It is a wholly owned subsidiary of ICM Holdings, Inc., and no publicly held company has a 10% or greater ownership interest in ICM Holdings, Inc.

Petitioner Illinois Corn Growers Association is an agricultural organization. It has no parent companies, and no publicly held company has a 10% or greater ownership interest in it.

Petitioner Iowa Soybean Association is a non-profit trade association within the meaning of D.C. Circuit Rule 26.1(b). Its members are soybean farmers and supporters of the agriculture and soybean industries. It operates for the purpose of promoting the general commercial, legislative, and other common interests of its members. The Iowa Soybean Association does not have a parent company, it has no privately or publicly held ownership interests, and no publicly held company has ownership interest in it.

Petitioner Kansas Corn Growers Association is an agricultural organization. It has no parent companies, and no publicly held company has a 10% or greater ownership interest in it.

Petitioner Michigan Corn Growers Association is an agricultural organization. It has no parent companies, and no publicly held company has a 10% or greater ownership interest in it.

Petitioner the Minnesota Soybean Growers Association is a non-profit trade association. Its members are soybean farmers, their supporters, and members of

soybean industries. It operates for the purpose of promoting the general commercial, legislative, and other common interests of its members. The Minnesota Soybean Growers Association is a not-for-profit corporation that is not a subsidiary of any corporation and that does not have any stock which can be owned by a publicly held corporation.

Petitioner Missouri Corn Growers Association is an agricultural organization. It has no parent companies, and no publicly held company has a 10% or greater ownership interest in it.

Petitioner National Association of Convenience Stores is an international trade association that represents both the convenience and fuel retailing industries with more than 1,300 retail and 1,600 supplier company members. The United States convenience industry has more than 152,000 stores across the country, employs 2.74 million people, and had more than \$859 billion in sales in 2023, of which more than \$532 billion were fuel sales. The Association has no parent corporation, and no publicly held corporation has a 10% or greater ownership interest in it.

Petitioner the South Dakota Soybean Association is a non-profit trade association. Its members are soybean farmers, their supporters and members of soybean industries. It operates for the purpose of promoting the general commercial, legislative, and other common interests of its members. The South Dakota Soybean Association is a not-for-profit corporation, is not a subsidiary of any corporation, and does not have any stock which can be owned by a publicly held corporation.

Petitioner Valero Renewable Fuels Company, LLC, a Texas limited liability company that manufactures ethanol, is a wholly owned direct subsidiary of Valero Energy Corporation.

RELATED PROCEEDINGS

United States Court of Appeals (D.C. Cir.):

Ohio v. EPA, No. 22-1081 (Apr. 9, 2024)

Iowa Soybean Association v. EPA, No. 22-1083 (Apr. 9, 2024)

American Fuel & Petrochemical Manufacturers v. EPA, No. 22-1084 (Apr. 9, 2024)

Clean Fuels Development Coalition v. EPA, No. 22-1085 (Apr. 9, 2024)

TABLE OF CONTENTS

Page
Introduction1
Opinions below5
Jurisdiction5
Statutory provisions involved5
Statement of the case5
A. Legal background5
B. Regulatory background7
C. Proceedings below10
Reasons for granting the petition13
I. The D.C. Circuit's standing decision warrants this Court's review
A. The decision below is wrong15
B. The decision below creates a conflict among the courts of appeals21
C. The question presented is important and warrants review in this case24
II. This Court should also reach the merits and vacate the waiver
A. EPA's decision is wrong27
1. Global climate change is not an "extraordinary" California condition within the meaning of Section 209

2.	California does not "need" its own emission standards to "meet" climate-change	
	conditions	30
3.	EPA's whole-program approach is wrong	32
4.	Clear-statement rules favor petitioners'	
	reading	33
	ne question presented is important, is implicat	ted
	the States' petition, and repeatedly evades	
re	view	35
Conclusi	on	37
Appendi	x A - Opinion of the court of appeals	
(Apr.	9, 2024)	1a
Appendi	x B - Decision of the Environmental Protection	on
Agen	cy (Mar. 14, 2022)	50a
Appendi	x C - Relevant statutory provisions2	286a

TABLE OF AUTHORITIES

Page(s)
Cases:
Bennett v. Spear, 520 U.S. 154 (1997)4, 16, 18
Bolln v. Nebraska, 176 U.S. 83 (1900)33
California v. Texas, 593 U.S. 659 (2021)17
Chamber of Commerce v. EPA, 642 F.3d 192 (D.C. Cir. 2011)8
Corner Post, Inc. v. Board of Governors of the Fed. Rsrv. Sys.,
603 U.S (2024)16
Davis v. FEC, 554 U.S. 724 (2008)21
Department of Commerce v. New York, 588 U.S. 752 (2019)4, 16, 17
Energy Future Coalition v. EPA, 793 F.3d 141 (D.C. Cir. 2015)4, 12, 16, 23, 24
FDA v. Alliance for Hippocratic Medicine, 602 U.S (2024)16, 17
Ford Motor Co. v. EPA, 606 F.2d 1293 (D.C. Cir. 1979)29
General Land Office v. Biden, 71 F.4th 264 (5th Cir. 2023)22
Gregory v. Ashcroft, 501 U.S. 452 (1991)34
Lujan v. Defenders of Wildlife, 504 U.S. 555 (1992)15, 16
(XI)

XII

Cases—Continued:
Massachusetts v. EPA,
549 U.S. 497 (2007)15, 17, 18
Motor & Equip. Mfrs. Ass'n v. EPA,
627 F.2d 1095 (D.C. Cir. 1979)6
Motor Vehicle Mfrs. Ass'n of the U.S., Inc. v. New
$York\ State\ Dep't\ of\ Env't\ Conservation,$
17 F.3d 521 (2d Cir. 1994)27
$Murthy \ v. \ Missouri,$
603 U.S (2024)18, 19
NRDC v. $NHTSA$,
894 F.3d 95 (2d Cir. 2018)22
Skyline Wesleyan Church v. California Dep't of
Managed Health Care,
968 F.3d 738 (9th Cir. 2020)23
Solid Waste Agency of N. Cook Cty. v. U.S. Army
$Corps\ of\ Eng$ 'rs,
531 U.S. 159 (2001)34
South Carolina v. Katzenbach,
383 U.S. 301 (1964)34
United States v. Sanchez-Gomez,
584 U.S. 381 (2018)26
United States v. Washington,
596 U.S. 832 (2022)26
$Utility\ Air\ Regul.\ Grp.\ v.\ EPA,$
573 U.S. 302 (2014)34
$Uzuegbunam\ { m v.}\ Preczewski,$
592 U.S. 279 (2021)15
$West\ Virginia\ { m v.}\ EPA,$
597 U.S. 697 (2022)3, 26, 34

XIII

Cases—Continued:
Wieland v. Department of Health & Human
Services,
793 F.3d 949 (8th Cir. 2015)23
Statutes:
28 U.S.C. § 1254(1)5
42 U.S.C.
§ 74087
§ 74097
§ 7410
§ 75077, 28, 286a
§ 7543(a)
§ 7543(b)I, 1, 6, 7, 27, 28, 32, 33, 287a
45 U.S.C. § 7521(a)(2)32
Regulations:
38 Fed. Reg. 10,235 (Apr. 26, 1973)7
49 Fed. Reg. 18,887 (May 3, 1984)6
59 Fed. Reg. 48,557 (Sept. 22, 1994)7
73 Fed. Reg. 12,156 (Mar. 6, 2008)7
74 Fed. Reg. 32,755 (July 8, 2009)8
78 Fed. Reg. 2,112 (Jan. 9, 2013)8, 9
84 Fed. Reg. 51,310 (Sept. 27, 2019)9, 29, 30,
31, 32, 33
86 Fed. Reg. 7,037 (Jan. 25, 2021)10
86 Fed. Reg. 74,434 (Dec. 30, 2021)35
87 Fed. Reg. 14,332 (Mar. 14, 2022)5, 10, 32
87 Fed. Reg. 25,710 (May 2, 2022)35
σ ,

XIV

Other authorities:
113 Cong. Rec. 30,948 (Nov. 2, 1967)6
Advanced Clean Cars Waiver Request (May 2012), https://www.regulations.gov/document/EPA-HQ-OAR-2021-0257-00069
American Heritage Dictionary (1st ed. 1969)31
California Air Resources Board, States that Have Adopted California's Vehicle Regulations (June 2024)10
California Air Resources Board, Public Hearing to Consider the Proposed Advanced Clean Cars II Regulations (Apr. 12, 2022)36
Coral Davenport et al., California to Ban the Sale of New Gasoline Cars, N.Y. Times (Aug. 24, 2022)35
H.R. Rep. No. 90-728 (1967)6, 29
Webster's New International Dictionary (3d ed.
1961)

In the Supreme Court of the United States

No.

DIAMOND ALTERNATIVE ENERGY, LLC, ET AL., PETITIONERS,

v.

ENVIRONMENTAL PROTECTION AGENCY, ET AL.

ON PETITION FOR WRIT OF CERTIORARI TO THE UNITED STATES COURT OF APPEALS FOR THE DISTRICT OF COLUMBIA CIRCUIT

PETITION FOR A WRIT OF CERTIORARI

INTRODUCTION

Section 209 of the Clean Air Act is a remarkable provision. It starts off modestly enough: because both motor vehicles and air pollution often cross state borders, Section 209(a) entrusts EPA to enact nationwide vehicle-emission standards and preempts States from adopting their own. 42 U.S.C. § 7543(a); see id. § 7410. But then Section 209(b) does something unusual: it allows EPA to grant a waiver from preemption to a single State—California—to enforce its own vehicle-emission standards. To escape preemption, California must demonstrate to EPA that it "need[s]" its own emission standards "to meet compelling and extraordinary conditions." *Id.* § 7543(b)(1)(B). Other States

may not establish their own standards, but they may elect to follow EPA's or California's.

As a concurrent challenge of 17 States explains, there are serious constitutional concerns with a statute that allows only California to act as a junior-varsity EPA. Still, for its first several decades, Section 209 worked more or less as intended. EPA granted preemption waivers for California to tackle local problems like smog in the Los Angeles basin, where the pollution was both generated by and felt by Californians. But all sensibility stopped in 2009, when California began claiming that Section 209 authorized it to set standards to curb greenhouse gases in an effort to tackle global climate change. For the next decade, EPA flipflopped under each new Presidential Administration about whether it could grant California such a waiver. In this latest order, EPA has again taken the position that Section 209 permits California to operate as a quasi-federal regulator on global climate change.

That vaunted role cannot be squared with the text, structure, or purpose of Section 209. First, climate change is not an "extraordinary" condition within California, and thus within the meaning of Section 209(b)(1)(B), because it is a global issue that is not localized to California. Second, California does not "need" its own emission standards to "meet" global climate-change conditions, as its standards would have no discernable effect on climate-change-related conditions in the State (or anywhere else). Global climate change is, in short, not the kind of California-specific condition that Section 209(b) carves out from preemption.

The question whether California may set greenhouse-gas emission standards for itself and other

States is undeniably major. California has mandated 100% electric vehicles by 2036, and is forcing electrification of the country's vehicle fleet. And the Executive Branch intends exactly that. The National Highway Traffic Safety Administration and EPA have set their own fuel-economy and emission standards that impose de facto electric-vehicle mandates in violation of their governing statutes. The companion challenges to those rules are awaiting decision by the D.C. Circuit. See Texas v. EPA, No. 22-1031 (filed Feb. 28, 2022); Natural Res. Def. Council v. NHTSA, No. 22-1080 (filed May 11, 2022). But by granting California the authority to outstrip even those mandates, EPA has put in place a backstop to the unlawful federal rules. Simply put, the waiver and authority claimed here are key parts of a coordinated agency strategy to convert the Nation from liquid-fuel-powered vehicles to electric vehicles. If that strategy seems familiar, it should. See West Virginia v. EPA, 597 U.S. 697 (2022).

Despite the issue's importance, the D.C. Circuit has avoided the merits of EPA's interpretation for over 15 years. It has met each challenge to EPA's waiver determination with a different jurisdictional barrier. This time, the D.C. Circuit dodged the issue under the guise of redressability. Petitioners are entities and associations of entities that produce or sell liquid fuels and the raw materials used to make them. Facing a regulation designed to reduce the demand for their products, petitioners' standing is self-evident. They also introduced unrebutted standing declarations about how California's standards will increase the sales of electric vehicles and reduce the consumption of liquid fuel—which, after all, is their express purpose. See App., infra, 19a-20a. But in a per curiam opinion, the

D.C. Circuit held that petitioners had not established redressability because they had not submitted evidence, such as affidavits from third-party automakers, showing precisely how automakers would adjust their production or prices during the four years of the waiver. *Id.* at 30a.

The D.C. Circuit's standing decision is wrong and warrants this Court's review. Under precedents of this Court and even the D.C. Circuit, "remov[ing] a regulatory hurdle" to the sale of a challenger's product suffices to establish redressability. Energy Future Coalition v. EPA, 793 F.3d 141, 144 (D.C. Cir. 2015) (Kavanaugh, J.); see Bennett v. Spear, 520 U.S. 154, 169 (1997). At a minimum, challengers to governmental action can rely on the "predictable effect" of regulation on third parties to establish causation and redressability. Department of Commerce v. New York, 588 U.S. 752, 768 (2019). Where, as here, the entire point of the challenged regulation is to reduce consumption of the plaintiffs' products, a decision vacating the regulation would obviously provide redress. Other courts of appeals applying these principles have thus found standing without relying on the sort of affidavits that the court below demanded. Left uncorrected, the decision below will impose an often insurmountable barrier to challenges in the court of appeals that is the primary home for administrative litigation.

This Court should also reach the merits of petitioners' challenge, and finally decide whether EPA has the authority to grant California a preemption waiver to address global climate change. The challenged California standards are in effect only through model year 2025, so merely correcting the D.C. Circuit's standing mistake may mean that no court considers the merits

before the waiver expires. And granting review of both questions presented is also the most reasonable way to resolve the States' concurrent challenge to the decision below. The States challenged EPA's waiver as unconstitutional because Section 209(b) violates the principle of equal sovereignty, and the court below rejected that constitutional argument on the merits. App., *infra*, 49a. Granting review of the merits would allow the Court to consider whether EPA's waiver is statutorily authorized before needing to reach the States' constitutional questions, raised in a separate petition. The Court should grant both this petition and the States', and reverse.

OPINIONS BELOW

The opinion of the court of appeals (App., *infra*, 1a-49a) is reported at 98 F.4th 288. The EPA order under review (App., *infra*, 50a-285a) is available at 87 Fed. Reg. 14,332.

JURISDICTION

The court of appeals entered judgment on April 9, 2024. This Court has jurisdiction under 28 U.S.C. § 1254(1).

STATUTORY PROVISIONS INVOLVED

Relevant statutory provisions are reproduced in the appendix to this petition. App., *infra*, 286a-290a.

STATEMENT OF THE CASE

A. Legal Background

Title II of the Clean Air Act makes EPA the Nation's primary regulator of motor-vehicle emissions.

42 U.S.C. § 7410. To effectuate a (mostly) uniform federal emissions regime, Section 209(a) of the Act prohibits States from "adopt[ing] or attempt[ing] to enforce any standard relating to the control of emissions from new motor vehicles." *Id.* § 7543(a). This preemption provision prevents "an anarchic patchwork of federal and state regulatory programs." *Motor & Equip. Mfrs. Ass'n* v. *EPA*, 627 F.2d 1095, 1109 (D.C. Cir. 1979).

Congress allowed only one exception to Section 209(a)'s broad preemption provision: Section 209(b), which authorizes EPA to "waive" preemption for California, under limited circumstances. 42 U.S.C. § 7543(b). Congress granted California this special status because of the State's "unique problems" with smog and other local issues caused by criteria pollutants like ozone and particulate matter. H.R. Rep. No. 90-728, at 22 (1967). California's atypical "geography and prevailing wind patterns," together with its unusually large number of vehicles, made smog a more persistent problem there than elsewhere. 49 Fed. Reg. 18,887, 18,890 (May 3, 1984) (citing 113 Cong. Rec. 30,948 (Nov. 2, 1967)).

Congress limited California's ability to separately regulate emissions in several ways. California may apply for a waiver only if it "determines that [its own] State standards will be, in the aggregate, at least as protective of public health and welfare as applicable Federal standards." 42 U.S.C. § 7543(b)(1). EPA must then evaluate the application and deny a waiver if, among other things, it finds that California's protectiveness determination is "arbitrary and capricious," id. § 7543(b)(1)(A); or if California "does not need such

State standards to meet compelling and extraordinary conditions," *id.* § 7543(b)(1)(B).

In 1977, Congress amended the Clean Air Act to allow any other State to "adopt and enforce" California standards "for which a waiver has been granted," so long as the State has a federally approved plan to attain the air-quality standards EPA sets for criteria pollutants. 42 U.S.C. §§ 7507, 7408(a), 7409. The upshot of this unusual preemption system is that EPA sets nationwide emission standards; California may in limited circumstances set more stringent ones in California; and other States may either apply EPA's standards or adopt California's, but may not set their own.

B. Regulatory Background

1. For decades, California used its waiver authority as Congress intended, to set emission standards to combat local air-quality problems like smog. See, e.g., 38 Fed. Reg. 10,235, 10,318 (Apr. 26, 1973); 59 Fed. Reg. 48,557, 48,626 (Sept. 22, 1994). In recent years, however, California has sought to transform this exception for tackling localized pollution into a tool for addressing global climate change through forced electrification of its vehicle fleet.

That effort initially failed. In 2008, EPA denied California's first application for a waiver allowing it to set emission standards to address climate change. EPA determined that Section 209(b)'s preemption waiver permitted California to enact standards only to address "local and regional" pollution where the "causal factors are local to California"—which obviously did not include global climate change. 73 Fed. Reg. 12,156, 12,163 (Mar. 6, 2008).

The day after President Obama took office, California sought reconsideration of EPA's denial of its waiver application for greenhouse-gas standards. EPA granted reconsideration, reversed itself, and issued the waiver. 74 Fed. Reg. 32,755, 32,783 (July 8, 2009). A number of affected parties challenged EPA's decision, but the D.C. Circuit held that the case was moot because California had "deemed" compliance with federal standards to satisfy the State's standards as well. *Chamber of Commerce* v. *EPA*, 642 F.3d 192, 206 (D.C. Cir. 2011).

2. In 2012, California applied for a new waiver to allow it to impose even more standards aimed at curbing greenhouse-gas emissions. This case concerns that 2012 waiver application and the "Advanced Clean Car" standards that it imposes. Those standards govern all new passenger cars, light-duty trucks, and mediumduty vehicles sold in California for model years 2015 through 2025.

As relevant here, the Advanced Clean Car standards have two key features that require a preemption waiver. First, California's greenhouse-gas emission standards limit carbon-dioxide emissions and thus effectively force manufacturers to produce and sell fewer cars that run on liquid fuel. 78 Fed. Reg. 2,112, 2,137 (Jan. 9, 2013). Second, the California standards include a "zero-emission vehicle" mandate, which requires each car manufacturer to produce and deliver for sale an increasing percentage of electric or fuel-cell vehicles (or purchase regulatory "credits" instead). Cal. Regs. tit. 13, § 1962.2(b). By model year 2025, the required percentage of zero-emission vehicles will rise to around 22%. *Ibid.* California explained that the mandated zero-emission vehicles "can dramatically reduce

petroleum consumption . . . compared to conventional technologies." Advanced Clean Cars Waiver Request 7-9 (May 2012), https://www.regulations.gov/document/EPA-HQ-OAR-2021-0257-0006.

- 3. In 2013, EPA granted a waiver for California's Advanced Clean Cars program. EPA concluded that California's standards met Section 209(b)'s criteria including that they were "needed to meet compelling and extraordinary conditions"—because the threat of global climate change was itself "compelling." 78 Fed. Reg. at 2,130. Notably, EPA credited California's finding that the cost of its regulations would be "more than offset by consumer fuel savings over the life of the vehicles." 78 Fed. Reg. at 2,138. At the time, however, California had kept in place its "deemed to comply" provision, and so potential challengers were stuck with the unfavorable D.C. Circuit standing precedent. California maintained that provision until 2018, when EPA set less stringent standards during the Trump Administration.
- 4. After that change in Administration, EPA reverted to its original approach to Section 209. In a 2019 joint rulemaking with NHTSA, EPA rescinded the 2013 preemption waiver for California's greenhousegas standards and zero-emission-vehicle mandate, again reasoning that climate change is not the kind of "peculiar," California-specific condition covered by Section 209(b). 84 Fed. Reg. 51,310, 51,328, 51,339 (Sept. 27, 2019). In addition, EPA found that California did not "need" its standards to "meet" climatechange conditions because California's standards would likely result in "no change in temperatures or physical impacts resulting from anthropogenic climate change in California." *Id.* at 51,341 (emphasis added).

C. Proceedings Below

1. On his first day in office, President Biden signed Executive Order 13,990, directing EPA to consider "suspending, revising, or rescinding" the 2019 withdrawal of California's 2013 waiver. 86 Fed. Reg. 7,037 (Jan. 25, 2021). EPA dutifully followed the President's lead and reinstated California's waiver. App., *infra*, 57a

In granting the waiver, EPA flipped back to its 2013 understanding of Section 209(b). It concluded that the waiver was justified in part because California needs its emission program as a whole—not the particular Advanced Clean Cars program that was the subject of the waiver—to address its criteria-pollution problems. *Id.* at 158a-166a. EPA also found that California needed its emission standards and zero-emission vehicle mandate because the effects of global climate change in California are "extreme," and because there is a "logical link" between local air-pollution problems and greenhouse-gas emissions. *Id.* at 207a, 215a.

To date, 17 States and the District of Columbia have adopted California's greenhouse-gas standards, its zero-emission-vehicle mandate, or both. CARB, States that Have Adopted California's Vehicle Regulations (June 2024), https://ww2.arb.ca.gov/our-work/programs/advanced-clean-cars-program/states-have-adopted-californias-vehicle-regulations. Together with California, those jurisdictions account for more than 40% of the Nation's new vehicle market. 87 Fed. Reg. 14,332, 14,358 (Mar. 14, 2022).

2. Petitioners are entities that produce or sell liquid fuels and the raw materials used to produce them, along with associations whose members include such entities. They challenged EPA's waiver reinstatement in the D.C. Circuit. App., *infra*, 2a. Petitioners filed detailed standing declarations, explaining that the reinstatement of California's standards would depress demand for fuel, injuring them financially. *Id.* at 19a-20a. The State of Ohio, along with a coalition of 16 other States, separately challenged the reinstatement on the ground that California's unique exemption from nationwide preemption violates the constitutional equal-sovereignty principle.

In defending the waiver reinstatement, EPA did not contest petitioners' Article III standing. On the merits, it primarily relied on what it refers to as its "whole program" approach. Under this reading of the Clean Air Act, Section 209(b)(1)(B) is satisfied if "California needs its program as a whole to meet compelling and extraordinary conditions." C.A. EPA Br. 84 (emphasis added). In other words, EPA argued that California can tack on any emission standards it likes to its emissions "program," "so long as" the State's criteria-pollutant problems "persist." *Id.* at 66.

Unlike EPA, California and other state and local intervenors challenged petitioners' standing. They contended that automakers were independently increasing electric-vehicle production in response to consumer demand, and that petitioners had not "established any probability that manufacturers would change course if EPA's decision were vacated." C.A. California Br. 13-15. But their own intervention motion attached declarations asserting that "additional gasoline-fueled vehicles would be sold during these model years" if EPA's waiver were overturned. C.A. States' Int. Mot., Scheehle Decl. ¶¶ 17-18. In reply, petitioners further explained that they had standing because vacating the waiver would remove a "direct regulatory impediment"

to their products' use. C.A. Reply 4 (citing $Energy\ Future\ Coal.$, 793 F.3d at 144). They also noted that California itself had told EPA that the waiver was "critical for incentivizing production and deployment of zero-emission vehicles." Id. (quoting C.A. J.A. 237).

At oral argument, however, counsel for the state and local intervenors began making similar arguments under the rubric of mootness. California contended that automakers could no longer change their production and sales plans for vehicles covered by California's waiver—that is, through model year 2025. In response, petitioners moved to file a supplemental brief explaining why their petitions were not moot, along with supplemental declarations from individuals experienced in vehicle-emissions compliance, who explained that "automobile manufacturers could and likely would change their production, pricing, and/or distribution plans for Model Year 2025 as late as December 2025, but at a minimum well into 2025." C.A. Pet. Standing Addendum, Kreucher Decl. ¶ 5.

3. The court of appeals held that petitioners lack Article III standing to challenge EPA's waiver. App., infra, 30a. Although the court declined to "definitively decide" whether petitioners had established injury and causation, it did not question either showing. Id. at 21a. After all, petitioners had explained that "by requiring vehicle manufacturers to sell vehicles that use less or no liquid fuel, California's . . . requirements depress the demand for liquid fuel." Id. at 19a.

Instead, the court of appeals concluded that petitioners had failed to show that their injuries would be redressed if EPA's decision were set aside. App., *in-fra*, 30a. The court reasoned that petitioners' claims were not "mooted by the passage of time," but rather

that petitioners had lacked standing from the start. *Id.* at 25a. The court faulted petitioners for "fail[ing] to point to any evidence affirmatively demonstrating that vacatur of the waiver would be substantially likely to" prompt automakers to produce fewer electric vehicles or alter their prices so that fewer would be sold. *Id.* at 23a. According to the court, petitioners had thus failed to establish that—even if the waiver had been vacated at the moment it was reinstated *in 2022*—automakers would have changed *any* production or prices before the end of model year *2025*. *Id.* at 23a-24a.

The court of appeals also declined to consider petitioners' supplemental declarations. App., *infra*, 30a. The court reasoned that there was no "good cause" to supplement the record because petitioners could not "have reasonably believed" that "their standing was 'self-evident' from the record" when they filed their opening brief. *Id.* at 31a.

Finally, the court of appeals rejected the States' equal-sovereignty argument on the merits. App., *in-fra*, 49a. The court concluded that the "fundamental principle of equal sovereignty" does not operate "as a limit on the Commerce Clause or other Article I powers." *Id.* at 36a (citation omitted).

REASONS FOR GRANTING THE PETITION

When petitioners filed suit, EPA's waiver controlled for the next four years. As a matter of common sense, if that waiver were set aside and California were unable to require automakers to produce electric vehicles instead of liquid-fuel vehicles, automakers would make or sell at least one more liquid-fuel vehicle over the course of those four years. That is the whole point of this hard-fought litigation: that the waiver would do

something to reduce California vehicles' consumption of liquid fuel, which necessarily would shift at least a dollar of business away from liquid-fuel sellers. Indeed, the federal government—never one shy about raising standing objections—did not even contest petitioners' Article III standing. Yet the court of appeals blinded itself to the obvious, and faulted petitioners for failing to prove by affidavit how third-party automakers would naturally behave.

This Court should grant review to ensure that the court of appeals' contrived standing decision does not imperil future challenges to administrative action. The court of appeals' demand for record evidence of automaker behavior cannot be squared with this Court's standing decisions, and it conflicts with decisions from several courts of appeals. Especially concerning, it would chill many legitimate agency rule challenges that hinge on third-party action.

This Court should also resolve the merits of petitioners' challenge to EPA's statutory authority, together with the States' constitutional equal-sover-eignty challenge. For over a decade, the D.C. Circuit has avoided resolving whether EPA has authority to grant a preemption waiver to allow California to address global climate change. The court of appeals' dodge of the merits here is just the latest installment. EPA's (current) position that Congress granted California, alone among the States, the ability to set vehicle-emission standards to combat global climate change is patently wrong, and raises serious constitutional issues for the reasons discussed in the States' anticipated petition. Without this Court's immediate review, California's unlawful standards will continue to

dictate the composition of the Nation's automobile market.

I. THE D.C. CIRCUIT'S STANDING DECISION WARRANTS THIS COURT'S REVIEW

A. The Decision Below Is Wrong

In the decision below, the D.C. Circuit erected barriers to demonstrating redressability that have no basis in this Court's Article III jurisprudence. These barriers are especially problematic in cases like this one, in which the agency action concerns a question of national importance. Such major questions should not be artificially insulated from judicial review.

1. To demonstrate Article III standing, a plaintiff must show that he suffered a concrete injury, that the injury is fairly traceable to the challenged action, and that his "injury will be redressed by a favorable decision." Lujan v. Defenders of Wildlife, 504 U.S. 555, 561 (1992) (internal quotation marks omitted). A plaintiff seeking to demonstrate redressability does not need to establish that a favorable judicial decision would completely cure his injury. Rather, a judicial decision need only "take steps to slow or reduce" the injury. Massachusetts v. EPA, 549 U.S. 497, 525 (2007). Thus, when a plaintiff asserts an economic injury, he satisfies the redressability requirement if a favorable decision would put even one dollar back in his pocket. Uzuegbunam v. Preczewski, 592 U.S. 279, 292 (2021).

When a plaintiff is the direct object of the challenged regulation, there is usually little question that a favorable decision from the court would provide redress. *See Lujan*, 504 U.S. at 561-562. The same should be true when a plaintiff alleges an "injury pro-

duced by determinative or coercive effect" of the challenged regulation "upon the action of someone else." Bennett v. Spear, 520 U.S. 154, 169 (1997). Thus, as then-Judge Kavanaugh explained, if a plaintiff can show that a favorable decision "would remove a regulatory hurdle" to third-party conduct that would benefit the plaintiff, that is ordinarily "enough to demonstrate redressability." Energy Future Coalition, 793 F.3d at 141; see Corner Post, Inc. v. Board of Governors of the Fed. Rsrv. Sys., 603 U.S. __ (2024) (Kavanaugh, J., concurring) (slip op., at 8) ("[E]ntire classes of administrative litigation . . . have traditionally been brought by unregulated parties.").

Even when a plaintiff's injury arises from "the unfettered choices made by independent actors," rather than a rule's "coercive or determinative effect," a plaintiff still may establish standing, as long as he can "adduce facts showing that" the third party will behave in such a way as to "permit redressability of injury." Lujan, 504 U.S. at 562. This Court has explained that a plaintiff may rely on "the predictable effect of Government action on the decisions of third parties," and has not required affidavits detailing how third parties will behave. Department of Commerce, 588 U.S. at 768 (emphasis added). For example, it is predictable that government regulation of one business "may cause downstream or upstream economic injuries to others in the chain." FDA v. Alliance for Hippocratic Medicine, 602 U.S. (2024) (slip op., at 12). By contrast, standing may not rest on "speculation about the decisions of independent actors" when those decisions would be unlawful or irrational. Department of Commerce, 588 U.S. at 768 (internal quotation marks omitted). Thus, when a plaintiff's standing theory relies on

"counterintuitive" assumptions about third-party behavior, the plaintiff may need to support that theory with "stronger evidence." *California* v. *Texas*, 593 U.S. 659, 678 (2021).

Department of Commerce illustrates the difference between predictable effects (which do not require robust record evidence) and counterintuitive effects (which do). There, the plaintiffs challenged the inclusion of a census question about citizenship, which they contended would injure them by causing third-party noncitizen households to decline to respond to the census. 588 U.S. at 766-767. This Court concluded that those effects were sufficiently predictable to establish the causation element of standing—which, like the redressability element, turns on the effect of government action (or its removal). Id. at 768; see Alliance for Hippocratic Medicine, 602 U.S. at __ (slip op., at 8) ("[C]ausation and redressability . . . are often flip sides of the same coin.") (internal quotation marks omitted). The Court did not hold that affidavits from noncitizens attesting that they did not plan to participate in the census were required. The Court found it sufficient that noncitizen households had responded to the census at lower rates in the past, and credited the plaintiffs' common-sense prediction that noncitizens would respond at even lower rates if the census asked about their citizenship. See 588 U.S. at 768.

An effect of agency action is "predictable," rather than "counterintuitive," when the agency itself intends or presupposes that effect. In *Massachusetts*, for example, this Court relied on EPA's own statements and programs to find that ordering EPA to set emission standards would redress petitioners' injuries. 549 U.S.

- at 526. The Court observed that EPA promoted voluntary emissions-reductions programs, and "would presumably not bother with such efforts," unless it thought that emissions reductions would have some effect on the environment. *Ibid*. In other words, an agency's own assumptions about the design of its rule are strong evidence of the rule's predictable effects.
- 2. The decision below departs from this Court's sensible approach to redressability. First, petitioners' injury arises from the "determinative or coercive effect" of California's standards on third-party automakers. *Bennett*, 520 U.S. at 169. By requiring automakers to produce vehicles that consume less or no liquid fuel, California's standards and EPA's waiver pose a legal barrier to the use of petitioners' products that a favorable decision would redress. Under this Court's cases, no more is needed.

But even if petitioners had to show more, the predictable effect of EPA's waiver on automakers is obvious: automakers will make and sell more electric vehicles to comply with California's regulations. Otherwise, California and EPA "would presumably not bother with such efforts." *Massachusetts*, 549 U.S. at 526. If the regulations go away, then the government will no longer be forcing automakers to sell more electric vehicles than they would otherwise produce in response to market forces. So automakers will make more vehicles that run on liquid fuel, or they will adjust their prices in response to consumer demand. That is Economics 101, not a proposition that requires an affidavit for support.

The contrast with this Court's recent decision in *Murthy* v. *Missouri*, 603 U.S. __ (2024), further illus-

trates the point. Murthy involved several circumstances that may call redressability into question none of which is present here. First, Murthy did not involve direct government regulation. Here, whatever independent incentives automakers may have to increase electric-vehicle production, compliance with California's standards is mandatory. Second, in Murthy the government's alleged influence might not have mattered at all. Id., slip op., at 12. Here, the whole point of California's standards (and EPA's waiver) is to require electrification beyond what the market was demanding. Third, and what the Court in Murthy called the "key point," id. at 25 n.10, the government action there had concluded by the time suit was brought. There was no obvious ongoing harm for the courts to correct, and so this Court could only "speculat[e]" about redressability. Id. at 22, 24-25. Here, California's standards and mandate were reinstated just months before petitioners brought suit and are still in place today. Unlike Murthy, this is a case where redressability should be beyond question.

The court of appeals nevertheless demanded that the fuel-manufacturer petitioners supply additional evidence of automakers' plans. Although the court acknowledged that it was "possible" that automakers would change their plans and sell more liquid-fuel vehicles, it faulted petitioners for failing to supply "record evidence," such as "additional affidavits." App., infra, 24a (citation omitted). Petitioners' standing declarations already pointed out that California itself had projected that a waiver would "reduce emissions through reductions in fuel production." C.A. Pet. Standing Addendum, Swenton Decl. at 5-6 (citation

omitted). They further explained that petitioners' injuries would be ameliorated if the waiver were vacated. C.A. Pet. Standing Addendum, Swenton Decl. at 7. Another declaration from the California Air Resources Board projected that without the standards, "additional gasoline-fueled vehicles" would likely be "produced and sold during these model years." C.A. States' Int. Mot., Vanderspek Decl. ¶ 22. But that was not enough for the court of appeals. In context, it appears that the only kind of evidence the court would have found sufficient is an affidavit from an automaker itself, promising to change its production or pricing if the waiver were vacated.

That requirement is doubly wrong. First, the court of appeals did not even acknowledge *Bennett* or *Department of Commerce*, or recognize that this Court has allowed plaintiffs to draw logical inferences about rational economic behavior in assessing causation and redressability. The court below did not ask whether the withdrawal of a special regulatory license to California will have "predictable effects" on vehicle sales. *Department of Commerce*, 558 U.S. at 768. It simply assumed that every effect must match up to a line in an affidavit.

Second, making matters worse, the court of appeals appeared to require plaintiffs to obtain affidavits from the directly regulated parties—here, the automakers. But directly regulated parties may have good reasons for not wanting to participate in the litigation. Maybe they intend to pass on the costs, or the government has garnered their complicity with some carrot that makes up for the regulatory stick. Indeed, that is exactly what happened here, as several automakers entered into "California Framework Agreements" committing

themselves to acceding to California's standards in exchange for certain benefits, like additional lead time. C.A. Resp.-Int. Br. 4.

Petitioners' standing to challenge an agency action should not depend on automakers' current preferences to partner with them. The contrary rule adopted below creates an unworkable hurdle to establishing standing in agency rule challenges—one with no basis in precedent or logic.

3. The court's flawed approach could doom any challenge to a similar time-limited agency rule that requires some lead time to implement. Petitioners sued within 60 days of EPA's March 2022 order. Yet the court held that petitioners already could not obtain effective relief because of the waiver's "relatively short," four-year "duration." App., *infra*, 22a; *see Davis* v. *FEC*, 554 U.S. 724, 734 (2008) (standing is assessed as of the date "when the suit was filed"). Additionally, by grounding its determination in standing, rather than mootness, the court dismissed the case without even holding EPA to its burden of establishing mootness or considering exceptions.

B. The Decision Below Creates A Conflict Among The Courts Of Appeals

By imposing artificial hurdles on Article III's redressability requirement, the court of appeals split from several other courts of appeals. The Second, Fifth, Eighth, and Ninth Circuits—and even the D.C. Circuit in other cases—have all found standing based on a law's coercive or predictable effects on third parties, without requiring the sort of record evidence demanded below. Under the D.C. Circuit's approach in

this case, those cases would have been (wrongly) dismissed on standing grounds.

The Second Circuit has found standing based on a rule's predictable effects on third parties, without requiring third-party affidavits or similar evidence. In NRDC v. NHTSA, 894 F.3d 95 (2018), the court of appeals considered an environmental group's challenge to NHTSA's decision to delay a rule increasing civil penalties for violations of fuel-economy standards. The petitioners asserted an environmental injury based on the assumption that third-party automakers would be less compliant when civil penalties were lower. Id. at 104. In finding standing, the court relied on "common sense and basic economics," which "tell us that the increased cost of unlawful conduct will make that conduct less common." Id. at 105. Automaker intervenors there had posited that penalties "ha[d] the potential" to affect their decisions, ibid., but the court did not demand affidavits proving how automakers would behave.

The Fifth Circuit has followed the same approach. In *General Land Office* v. *Biden*, 71 F.4th 264 (2023), the court of appeals considered Texas's challenge to the Department of Homeland Security's decision to divert funds appropriated for the construction of a wall along the United States–Mexico border. Texas asserted injuries premised on the assumption that the diversion of funds would cause more unlawful immigration. Without requiring affidavits from undocumented immigrants, the Fifth Circuit adopted the commonsense point that "[i]n the absence of longer walls, at least some illegal aliens who otherwise would have been prevented from entering Texas will seek" benefits from the State. *Id.* at 273.

The Eighth Circuit, too, has relied on the predictable effects on third parties to find an injury redressable, without requiring a third-party affidavit. Wieland v. Department of Health & Human Services, 793 F.3d 949 (2015), involved a challenge to provisions of the Affordable Care Act that required certain insurers to cover contraceptive services. Plaintiffs, who wanted a contraception-free option, satisfied the redressability requirement, because although an order enjoining those laws would not require insurers to offer a contraception-free option, it was "likely" that insurers would respond that way. Id. at 957.

The Ninth Circuit has applied similar reasoning. It found standing for plaintiffs challenging a California directive that required certain insurers to offer abortion coverage. See Skyline Wesleyan Church v. California Dep't of Managed Health Care, 968 F.3d 738, 750 (2020). The court of appeals did not demand affidavits from the insurers, concluding both that the directive had a "determinative or coercive effect" on the insurers, and that "the predictable effect" of a favorable judicial decision included that "at least one insurer would be willing to sell" a plan without abortion coverage. Ibid.

The decision below is not even faithful to the D.C. Circuit's own standing precedent. In *Energy Future Coalition*, 793 F.3d 141, a case with strikingly parallel facts to this one, the D.C. Circuit held that fuel producers had standing to challenge an EPA rule effectively banning vehicle manufacturers from using certain fuel in emissions testing. *Id.* at 144. In an opinion by then-Judge Kavanaugh, the court held that the plaintiffs had established redressability, reasoning that "if EPA permitted vehicle manufacturers to use" the contested test

fuel, "there is substantial reason to think that at least some vehicle manufacturers would use it." *Ibid.* The court credited the common-sense, predictable effects of EPA's rule on third parties. Yet here, after criticizing petitioners for believing their standing was "self-evident" under existing precedent, App., *infra*, 31a, the court below failed to even *address* this precedent on which petitioners' standing argument was based.

Ultimately, the court of appeals' decision in this case cannot be squared with any of the above decisions. In each of them, the court relied on common-sense inferences about how third parties behave in response to *legal* barriers to certain behavior. None of them demanded the specific factual showing that the court required here. This Court should correct the D.C. Circuit's outlier position before it sows significant confusion in this area.

C. The Question Presented Is Important And Warrants Review In This Case

The question presented is exceptionally important, and this case presents an appropriate vehicle to consider it.

1. The decision below threatens to chill legitimate challenges to agency action. The D.C. Circuit has traditionally served as the primary home for litigation under the Administrative Procedure Act, and it is the exclusive venue for challenges to emission standards under the Clean Air Act. That makes its new barrier to agency challenges especially problematic. As the decisions in the circuit split illustrate, there are many circumstances in which third parties directly regulated by an agency rule have different incentives from an in-

jured petitioner. Here, the preferences of car manufacturers and fuel manufacturers may diverge, including because car manufacturers have been offered some regulatory flexibility that benefits them but not fuel producers. In other cases, a similar dynamic may arise between businesses and consumers, insurers and insured parties, and any number of other sets of potential petitioners. If every regulatory petitioner in the D.C. Circuit must secure the cooperation of a directly regulated party to establish standing, a significant swath of challenges may be doomed from the start.

The D.C. Circuit's unnecessary and unprecedented hurdle also creates bad incentives for agencies. Under the D.C. Circuit's rule, agencies could intentionally structure their actions to placate directly regulated parties and thereby foreclose future litigation. Those actions would still have sweeping effect, but the directly regulated parties—who would be the only potential plaintiffs—would have no incentive to sue. The result is an agency roadmap of particular concern because it insulates even major decisions like this one from judicial review.

At bottom, the entire purpose of California's standards and EPA's waiver determination is to reduce liquid-fuel usage and mandate electrification. That goal is existential to the liquid-fuel industry, even if it does not immediately threaten car manufacturers. Fuel manufacturers are the obvious parties to challenge the regulations. The fact that fuel producers are not directly *regulated*, when they are the direct *target* of the regulations, should not prevent them from establishing their standing to sue.

2. This case presents an appropriate vehicle for addressing this important issue. The redressability question determines petitioners' standing, as neither the court of appeals nor EPA has questioned petitioners' showing of injury or causation. See App., infra, 20a.

EPA has raised concerns about mootness, see id. at 25a, but the Court can decide this case before it becomes moot. The party asserting mootness "bears the burden to establish" that the case "has become moot." West Virginia, 597 U.S. at 719. To meet that burden, EPA would have to show that "it is *impossible*" for the Court "to grant any effectual relief." United States v. Washington, 596 U.S. 832, 837 (2022) (emphasis added). It is far from impossible here: if this Court grants certiorari in October Term 2024, it will presumably render a decision before model year 2025 ends. And even if EPA could somehow establish mootness, this case would fall within the capable-of-repetition exception because of the order's relatively short duration and the likelihood that petitioners will be "subjected to the same action again" when EPA issues future waivers. United States v. Sanchez-Gomez, 584 U.S. 381, 391 (2018).

II. THIS COURT SHOULD ALSO REACH THE MERITS AND VACATE THE WAIVER

This Court should also grant the second question presented and decide whether EPA has the authority to grant California a preemption waiver to address global climate change. Because California's waiver expires at the end of model year 2025, it is quite possible the D.C. Circuit will not decide the merits on remand in time for this Court's subsequent review. This EPA will no doubt take the same approach to its next waiver

determinations, and petitioners will then spend years suffering ongoing injury from California's next set of unlawful standards and mandates. Reaching the merits now would avoid that serious unfairness. EPA is wrong on the merits, and there is no reason to wait additional years to resolve an issue the D.C. Circuit has punted for over a decade.

A. EPA's Decision Is Wrong

The Clean Air Act does not authorize the preemption waiver that EPA granted here. Section 209(a) establishes federal control over motor-vehicle-emission standards. 42 U.S.C. § 7543(a). Because of California's unique "smog problem," Section 209(b) gives that State alone the right to set its own emission standards. *Motor Vehicle Mfrs. Ass'n of the U.S., Inc.* v. *New York State Dep't of Env't Conservation*, 17 F.3d 521, 526 (2d Cir. 1994). But Congress's baseline was nationwide preemption, and it tailored California's special exemption accordingly. As a result, EPA must find that California's standards are "need[ed]" to "meet compelling and extraordinary conditions" in California. 42 U.S.C. § 7543(b)(1)(B).

The California regulations here do not meet the statutory criteria. First, California's standards do not target conditions "extraordinary" to California because they were created to address global climate change, which is by definition not a phenomenon particular to California. Second, California does not "need" its separate standards to "meet" those conditions because, by EPA's own admission, the standards will not materially reduce the impacts of climate change in California or anywhere else. And EPA's pri-

mary defense—that California needs its "whole program," even if not these particular standards—lacks any basis in the text or common sense.

1. Global climate change is not an "extraordinary" California condition within the meaning of Section 209

The Clean Air Act's text, structure, and history demonstrate that the phrase "compelling and extraordinary conditions" refers to California's distinctive local pollution problems; it does not encompass conditions with global cause and effect, like climate change.

a. California's separate emission standards must be needed to meet "compelling and extraordinary conditions." 42 U.S.C. § 7543(b)(1)(B). The plain meaning of "extraordinary" is "most unusual." Webster's New International Dictionary 807 (3d ed. 1961). On its own, that definition could mean "most unusual" compared to ordinary pollution problems or "most unusual" compared to other States' problems. In context, it must mean the latter. The Clean Air Act pairs "extraordinary" with "compelling," and "compelling" already captures a sense of magnitude. To avoid rendering "extraordinary" redundant, it must mean "most unusual" as compared to other States.

Related statutory provisions support reading the phrase "compelling and extraordinary" to encompass severe local conditions. For example, Section 177 authorizes other States to adopt California's standards as part of approved plans for combatting the six criteria pollutants that cause local pollution problems. *See* 42 U.S.C. § 7507. Congress thus plainly contemplated that the standards California would adopt under Sec-

tion 209(b)—that other States might copy under Section 177—would help States attain local ambient airquality standards within their respective borders.

More generally, the structure of Section 209 makes clear that Section 209(b) is an *exception* from a uniform federal regulatory regime. See 42 U.S.C. § 7543(a). It would make little sense to permit California to function as a junior-varsity EPA and deviate from a national regulatory framework to address conditions that are broadly shared throughout the Nation.

Section 209(b)'s history and purpose confirm that it authorizes preemption waivers only for California standards aimed at local air-quality issues. In drafting Title II, Congress repeatedly identified California's "peculiar" circumstances: its "unique problems" resulting from local emissions and pollution concentrations interacting with the State's distinctive "climate and topography." H.R. Rep. No. 90-728, at 22 (1967). As the D.C. Circuit explained decades ago, "clearly the intent" of the waiver provision was to "focus on *local* air quality problems" that "may differ substantially from those in other parts of the nation." *Ford Motor Co.* v. *EPA*, 606 F.2d 1293, 1303 (1979) (emphasis added).

b. Global climate change is not a condition "extraordinary" to California. By definition, global climate change is neither unique to California nor uniquely felt by the State. As EPA found in 2019, when it comes to the effects of climate change, California is not "worse-positioned in relation to certain other areas." 84 Fed. Reg. at 51,348 n.278. In fact, "[m]any parts of the United States, especially western States, may have issues [caused by climate change] related to drinking water" and "wildfires, and effects on agriculture." *Id.* at

51,348. In other words, "effects related to climate change in California" are "not sufficiently different from the conditions in the nation as a whole to justify separate State standards." *Id.* at 51,344. EPA did not revisit that finding in 2022. The agency thus provided no basis to depart from its prior conclusion, let alone the heightened showing necessary for a reversal.

To be sure, EPA argued below that California's standards do address local conditions because they may have side effects on local criteria pollution. But that argument is wrong both legally and factually. As a legal matter, EPA cannot contrive a new goal for California's standards not presented in California's waiver application. No one disputes that the express purpose of California's standards was to regulate global climate change. Moreover, EPA's argument is not supported by the factual record. In the waiver reinstatement, EPA relied on the "logical link" between ozone pollution and greenhouse gases—namely, that ozone levels are "exacerbate[d]" by higher temperatures caused by global warming. App., infra, 207a-208a. But EPA previously found that the State's rules would produce "likely no change" to climate-change conditions—including rising temperatures—in California. 84 Fed. If California's standards will not Reg. at 51,341. change temperatures, then they cannot affect ozone levels under EPA's "logical link" theory either.

2. California does not "need" its own emission standards to "meet" climatechange conditions

Even if California faced "extraordinary" conditions within the meaning of Section 209(b) from global climate change, it does not "need" its greenhouse-gas

standards and zero-emission-vehicle mandate to "meet" those conditions. To the contrary, as EPA explained in 2019, California's standards "will not meaningfully address global air pollution problems of the sort associated with [greenhouse-gas] emissions." 84 Fed. Reg. at 51,347, 51,349.

a. A Section 209(b) waiver is authorized when "need[ed]" to "meet" conditions in California—that is, only if the proposed California emission standards would appreciably affect the conditions that warrant them. That accords with the ordinary meaning of the statutory terms "need" and "meet." The verb "need" means to "be necessary." Webster's New International Dictionary, supra, at 1512. And the term "necessary" typically means "essential; indispensable." American Heritage Dictionary 877 (1st ed. 1969). The verb "meet" is complementary. In this context, it means to "satisfy (a demand, need, obligation)." Id. at 816.

Putting the terms together, two things are clear. First, California must "need"—*i.e.*, require as essential or very important—specific standards that differ from federal standards. Second, California's standards must meaningfully address the conditions that give rise to California's need for separate standards. At a minimum, if the State's proposed standards have *no impact* on those conditions, then they cannot be said to be necessary, essential, or indispensable to "meet" the conditions the State faces.

b. Based on EPA's own undisturbed factual findings, California does not satisfy that standard for a preemption waiver. In vacating California's waiver, EPA previously found that California's greenhousegas standards would "lea[d] to little to no change" in "[greenhouse-gas] emissions at a national level," and

"would result in an *indistinguishable* change in global temperatures" and "likely *no change* in temperatures or physical impacts resulting from anthropogenic climate change in California." 84 Fed. Reg. at 51,341, 51,353 (emphases added). Critically, in reinstating the waiver, EPA did not disturb these findings about the futility of California's standards.

3. EPA's whole-program approach is wrong

In defending its reinstatement decision, EPA has primarily argued that Section 209(b)(1)(B)—which permits EPA to grant a preemption waiver only if "need[ed]" to "meet" "compelling and extraordinary conditions"—is effectively irrelevant. Under EPA's "whole program" approach, so long as California needs any separate standards at all—say, to combat smog—the State has satisfied the "need[s]...to meet" requirement. See 87 Fed. Reg. at 14,335. At that point, California can add on any other emission standards it wants and tackle global problems as it sees fit.

EPA's whole-program approach is atextual and defies common sense. EPA relies on language elsewhere in Section 209(b) requiring that California "determine[] that the State standards will be, in the aggregate, at least as protective of public health and welfare as applicable [federal] standards." 42 U.S.C. § 7543(b)(1) (emphasis added). But Section 209(b)'s "in the aggregate" language does not carry down to the rest of 209(b). For example, Subsection (b)(1)(C) requires EPA to ensure that manufacturers have sufficient lead time to meet California's standards. See 42 U.S.C. § 7521(a)(2). EPA does not assess whether manufacturers would have adequate lead time "in the aggregate"—that would make no sense. See 84 Fed.

Reg. at 51,332; see also C.A. Pet. Br. 46. Similarly, EPA has a separate duty in Section 209(b)(1)(B) to determine whether California "need[s] such State standards to meet compelling and extraordinary conditions." 42 U.S.C. § 7543(b)(1)(B). That determination, like the neighboring lead-time determination, is not done "in the aggregate."

EPA's reading would also make the Section 209(b)(1)(B) criteria meaningless. Congress already determined that California "need[s]" its own emissions program by creating the preemption exception in the first place. On EPA's view, however, subsection (b)(1)(B) serves no independent purpose so long as California has any air-quality issues. This assertion cannot withstand minimal scrutiny, as the D.C. Circuit panel appeared to recognize at oral argument. See C.A. Oral Arg. Tr. 45:30 (Wilkins, J., criticizing EPA's "whole program" approach).

4. Clear-statement rules favor petitioners' reading

Even if EPA's interpretation were possible, several clear-statement rules require petitioners' reading.

First, principles of constitutional avoidance require petitioners' reading. The State petitioners have argued that Section 209(b) deviates from the "fundamental principle of equality of the states under the Constitution." Bolln v. Nebraska, 176 U.S. 83, 89 (1900). This equal-sovereignty question is at least a serious one, and this Court should adopt a more modest interpretation of California's waiver authority to avoid it. Solid Waste Agency of N. Cook Cty. v. U.S. Army Corps of Eng'rs, 531 U.S. 159, 160 (2001). Even if Congress could grant California the unique ability to address a

local problem, it would be a far graver intrusion on equal-sovereignty principles to grant California alone the ability to address global climate change. Cf. South Carolina v. Katzenbach, 383 U.S. 301, 328-329 (1964) ("The doctrine of the equality of States" does not bar "remedies for local evils which have subsequently appeared.").

Second, the major-questions doctrine applies to California's efforts to tackle global climate change and force a transition to electric vehicles. This Court "expect[s] Congress to speak clearly if it wishes to assign to an agency decisions of 'vast economic and political significance." Utility Air Regul. Grp. v. EPA, 573 U.S. 302, 324 (2014) (citation omitted). On EPA's view, Section 209 authorizes the agency to permit California to adopt vehicle-emission standards to tackle climate change, and to force a transition to electric vehicles that would have enormous repercussions for the national economy, the States' electric grids, and national security. This Court should "greet" EPA's "assertions of 'extravagant statutory power over the national economy' with 'skepticism.'" West Virginia, 597 U.S. at 724 (citation omitted).

Third, the federalism canon points in the same direction. Congress must be "unmistakably clear in the language of the statute" if it "intends to alter the 'usual constitutional balance between the States and the Federal government." *Gregory* v. *Ashcroft*, 501 U.S. 452, 460-461 (1991) (citation omitted). Yet under EPA's view, California alone among the States can regulate the nation's automobile market in the service of addressing climate change and forcing a transition to electric vehicles.

B. The Question Presented Is Important, Is Implicated In The States' Petition, And Repeatedly Evades Review

The extent of EPA's authority to grant California a preemption waiver warrants this Court's immediate review. It is a question that shapes the direction of the entire country's automobile industry but that has repeatedly evaded judicial scrutiny. And it is a question antecedent to the constitutional issue that the court of appeals reached and that is the subject of the State petitioners' concurrent challenge.

1. Section 209(b), if construed to allow California to tackle nationwide issues like global climate change, becomes a huge source of power to regulate the country's economy. California is a significant market in its own right, and 17 States and the District of Columbia have opted into at least some of California's standards. See supra, p. 10. Additionally, both EPA and NHTSA have relied on California's standards in setting their respective (and unlawful) vehicle-emission standards aimed at forcing electrification. See 86 Fed. Reg. 74,434, 74,457-74,458 (Dec. 30, 2021); 87 Fed. Reg. 25,710, 25,762-25,765 (May 2, 2022). No wonder California's governor described its vehicle-emission rules as "one of the most significant steps to the elimination of the tailpipe as we know it." Coral Davenport et al., California to Ban the Sale of New Gasoline Cars, N.Y. Times (Aug. 24, 2022), https://www.nytimes.com/2022/ 08/24/climate/california-gas-cars-emissions.html.

A wait-and-see approach would embolden California to stretch Section 209(b) even further. Since petitioners brought this case, California has adopted its so-called Advanced Clean Cars II standards, along with Advanced Clean Fleets and Advanced Clean

Trucks rules, which collectively mandate 100% electrification of every class of new vehicles by 2036. See CARB, Public Hearing to Consider the Proposed Advanced Clean Cars II Regulations 12 (Apr. 12, 2022), http://ww2.arb.ca.gov/sites/default/files/barcu/regact/2022 /accii/isor.pdf. EPA's waiver thus enables California to force automakers to electrify their entire fleets in California and any State that adopts its standards.

- 2. This Court should also grant review of the second question because the States' anticipated petition presents a serious constitutional question that should be considered together with petitioners' statutory challenge. The State petitioners are challenging the D.C. Circuit's decision holding that Section 209 does not violate the principle of equal sovereignty. Unless this Court grants review of both questions presented here, the Court will not have complete briefing on a narrower statutory alternative to resolving the States' constitutional challenge.
- 3. Finally, this Court should hear the second question presented because it has for too long evaded this Court's review and may otherwise continue to do so. The challenged waiver is in effect only through model year 2025. That means that there will likely be time for a merits decision in this Court, or a remand and potential merits decision in the court of appeals, but not both without having to litigate mootness issues that may complicate this Court's review.

Petitioners have endured decades of regulatory whiplash only for the D.C. Circuit to repeatedly reject any legal challenges on threshold grounds. Absent this Court's review, petitioners will be back at square one, having to litigate additional challenges to EPA's next waiver, all while suffering ongoing injury. Critical

American industries deserve the certainty that can only be provided by this Court's finally defining the extent of EPA's authority under Section 209(b).

CONCLUSION

The petition for a writ of certiorari should be granted.

Respectfully submitted.

ERIC D. MCARTHUR SIDLEY AUSTIN LLP 1501 K Street NW Washington, DC 20005

Counsel for American Fuel & Petrochemical Manufacturers, Domestic Energy Producers Alliance, Energy Marketers of America, and National Association of Convenience Stores

MICHAEL BUSCHBACHER JARED M. KELSON BOYDEN GRAY PLLC 801 17th Street NW Suite 350 Washington, DC 20006

Counsel for Clean Fuels Development Coalition, ICM, Inc., Illinois Corn Growers Association, Kansas Corn Growers Association, Michigan Corn Growers Association, Missouri Corn Growers Association, and Valero Renewable Fuels Company, LLC

Jeffrey B. Wall
Counsel of Record
Morgan L. Ratner
Zoe A. Jacoby
Sullivan & Cromwell LLP
1700 New York Avenue NW
Suite 700
Washington, DC 20006
(202) 956-7660
wallj@sullcrom.com

LESLIE B. ARFFA SULLIVAN & CROMWELL LLP 125 Broad Street New York, NY 10004

Counsel for Valero Renewable Fuels Company, LLC RICHARD S. MOSKOWITZ AMERICAN FUEL & PETRO-CHEMICAL MANUFACTURERS 1800 M Street NW Suite 900 North Washington, DC 20036

Counsel for American Fuel & Petrochemical Manufacturers

BRITTANY M. PEMBERTON BRACEWELL LLP 2001 M Street NW Suite 900 Washington, DC 20036

Counsel for Diamond Alternative Energy, LLC and Valero Renewable Fuels Company, LLC

MATTHEW W. MORRISON SHELBY L. DYL PILLSBURY WINTHROP SHAW PITTMAN LLP 1200 Seventeenth Street NW Washington, DC 20036

Counsel for Diamond Alternative Energy, LLC, Iowa Soybean Association, The Minnesota Soybean Growers Association, and South Dakota Soybean Association

July 2, 2024



TABLE OF CONTENTS

Page
Appendix A — Court of appeals opinion (Apr. 9, 2024)
Appendix B — Decision of the Environmental Protection Agency (Mar. 14, 2022) 50a
Appendix C — Statutory provisions: 42 U.S.C. § 7507—New motor vehicle emission standards in nonattainment areas

APPENDIX A

UNITED STATES COURT OF APPEALS FOR THE DISTRICT OF COLUMBIA CIRCUIT

No. 22-1081, Consolidated with 22-1083, 22-1084, 22-1085

STATE OF OHIO, et al.,

Petitioners

v.

ENVIRONMENTAL PROTECTION AGENCY AND MICHAEL S. REGAN, IN HIS OFFICIAL CAPACITY AS ADMINISTRATOR OF THE U.S. ENVIRONMENTAL PROTECTION AGENCY,

Respondents,

ADVANCED ENERGY UNITED, et al.,

Intervenors

September 15, 2023, Argued; April 9, 2024, Decided

On Petitions for Review of a Final Action of the Environmental Protection Agency

OPINION OF THE COURT

Before: Wilkins, Childs, and Garcia, Circuit Judges.

Opinion for the Court filed PER CURIAM.

Per Curiam: These consolidated petitions for review concern a 2022 decision by the Environmental Protection Agency ("EPA") to reinstate the EPA's prior decision, in 2013, to waive federal preemption of two California regulations regarding automobile emissions under the Clean Air Act. The regulations in question are a standard limiting greenhouse gas emissions and a requirement that a certain percentage of new vehicles manufactured in the state each year be zero-emissions vehicles ("ZEV"), see 13 Cal. Code Regs. §§ 1961.3, 1962.2, respectively. Two sets of Petitioners challenge the EPA's decision. The first group of Petitioners comprises seventeen states ("State Petitioners"). The second group of Petitioners includes entities that produce or sell liquid fuels and the raw materials used to produce those fuels, along with associations whose members include such entities ("Fuel Petitioners"). Both State and Fuel Petitioners claim that

^{1.} State Petitioners are the State of Ohio, State of Alabama, State of Arkansas, State of Georgia, State of Indiana, State of Kansas, Commonwealth of Kentucky, State of Louisiana, State of Mississippi, State of Missouri, State of Montana, State of Nebraska, State of Oklahoma, State of South Carolina, State of Texas, State of Utah, and State of West Virginia.

^{2.} Fuel Petitioners are American Fuel & Petrochemical Manufacturers, Clean Fuels Development Coalition, Diamond Alternative Energy, LLC, Domestic Energy Producers Alliance,

the EPA was not authorized to grant California the waiver under the Clean Air Act. Fuel Petitioners argue that the EPA exceeded its statutory authority under the Clean Air Act. State Petitioners, meanwhile, contend that the EPA's waiver reinstatement decision was contrary to law because the relevant California regulations are preempted by a separate federal statute, the Energy Policy and Conservation Act of 1975 ("EPCA"), 49 U.S.C. § 32919(a). State Petitioners also claim that by granting a waiver to California alone, the EPA violated a constitutional requirement that the federal government treat states equally in terms of their sovereign authority. We hold that Fuel Petitioners lack standing to raise their statutory claim, and that State Petitioners lack standing to raise their preemption claim, because neither set of Petitioners has demonstrated that their claimed injuries would be redressed by a favorable decision by this Court. While we hold that State Petitioners have standing to raise their constitutional claim, we reject it on the merits.

I.

A.

While the Clean Air Act typically grants states broad discretion to meet federal air quality goals, emissions standards for new automobiles are promulgated at the

Energy Marketers of America, ICM, Inc., Illinois Corn Growers Association, Iowa Soybean Association, Kansas Corn Growers Association, Michigan Corn Growers Association, Minnesota Soybean Growers Association, Missouri Corn Growers Association, National Association of Convenience Stores, South Dakota Soybean Association, and Valero Renewable Fuels Company, LLC.

federal level. The Clean Air Act empowers the EPA to promulgate federal emissions standards for those vehicles, see 42 U.S.C. § 7521, and it preempts any corresponding state regulation, expressly preventing the adoption of emissions standards for new vehicles and/or engines as follows:

(a) Prohibition

No State or any political subdivision thereof shall adopt or attempt to enforce any standard relating to the control of emissions from new motor vehicles or new motor vehicle engines subject to this part. No State shall require certification, inspection, or any other approval relating to the control of emissions from any new motor vehicle or new motor vehicle engine as condition precedent to the initial retail sale, titling (if any), or registration of such motor vehicle, motor vehicle engine, or equipment.

Id. § 7543(a) ("Section 209(a)"). However, the Clean Air Act permits the EPA to waive application of Section 209(a) to any state under certain circumstances:

(b) Waiver

(1) The Administrator shall, after notice and opportunity for public hearing, waive application of this section to any State which has adopted standards (other than crankcase emission standards) for the control of emissions from new motor vehicles or new motor vehicle engines prior to March 30, 1966, if the State determines that the State standards will be, in the aggregate, at least as protective of public health and welfare as applicable Federal standards. No such waiver shall be granted if the Administrator finds that—

- (A) the determination of the State is arbitrary and capricious,
- (B) such State does not need such State standards to meet compelling and extraordinary conditions, or
- (C) such State standards and accompanying enforcement procedures are not consistent with section 7521(a) of this title.
- (2) If each State standard is at least as stringent as the comparable applicable Federal standard, such State standard shall be deemed to be at least as protective of health and welfare as such Federal standards for purposes of paragraph (1).
- (3) In the case of any new motor vehicle or new motor vehicle engine to which State standards apply pursuant to a waiver granted under paragraph (1), compliance with such State standards shall be treated as compliance with applicable Federal standards for purposes of this subchapter.

Id. § 7543(b) ("Section 209(b)"). California is the only state that had adopted standards (other than crankcase emission standards) for the control of emissions from new motor vehicles or new motor vehicle engines as of March 30, 1966. See Motor & Equip. Mfrs. Ass'n v. EPA, 627 F.2d 1095, 1100 n.1, 201 U.S. App. D.C. 109 (D.C. Cir. 1979).

If California applies to promulgate automobile emissions standards that it has determined are at least as protective of public health and welfare as the existing federal regulations, the Clean Air Act requires the EPA to waive preemption as to those regulations, unless certain criteria (the "waiver denial criteria") are met. 42 U.S.C. § 7543(b). The EPA may refuse to grant a waiver only if: (1) California's "determination . . . is arbitrary and capricious," (2) California "does not need such State standards to meet compelling and extraordinary conditions," or (3) the "standards and accompanying enforcement procedures are not consistent with [42 U.S.C. § 7521(a)]." *Id.* § 7543(b) (1)(A)–(C). In other words, the federal regulations continue to act as the floor for emissions regulations, but California can seek to enact its own more stringent regulatory program above those federal requirements.

Sections 209(a) and (b) of the Clean Air Act together make up a statutory compromise between several competing interests. When Congress enacted the Clean Air Act in 1967, California suffered from significant air quality and pollution problems caused by motor vehicle emissions, which federal emissions regulations were unlikely to adequately address. *See* S. Rep. No. 90-403, at 33–34 (1967); H.R. Rep. No. 90-728, at 21–23, 96–97

(1967). California was also the only state with its own motor vehicle emissions standards, and its leadership in automobile emissions regulation had been valuable to the federal government in crafting the Clean Air Act. See S. Rep. No. 90-403, at 33-34; H.R. Rep. No. 90-728, at 21–23, 96–97. At the same time, automobile manufacturers were growing concerned that other states might begin regulating automobile emissions, subjecting them to a patchwork of regulatory obligations and significantly increasing manufacturing costs. See H.R. Rep. No. 90-728, at 21; see also Engine Mfrs. Ass'n v. EPA, 88 F.3d 1075, 1079, 319 U.S. App. D.C. 12 (D.C. Cir. 1996). Congress enacted Sections 209(a) and (b) to balance the fears of automobile manufacturers, California's need for bespoke regulation, and the federal interest in allowing California to test new emissions regulations. Section 209(a) addresses the fears of automakers and ensures national uniformity in automobile emissions standards by preempting state regulation. See 42 U.S.C. § 7543(a). Meanwhile, Section 209(b) grandfathers in California's regulatory program and allows it to continue innovating new solutions to automobile pollution. See id. § 7543(b); see also Engine Mfrs. Ass'n, 88 F.3d at 1080.

В.

The D.C. Circuit is familiar with interpreting the Clean Air Act. Shortly following the enactment of Section 209(b), the D.C. Circuit addressed the question of how California should determine that its regulations are more protective than the federal regulations. See Motor & Equip. Mfrs. Ass'n, 627 F.2d at 1095. California sought to impose

regulations on oxides of nitrogen that were significantly more stringent than their federal counterparts. Id. at 1110 n.32. However, due to technological constraints, emissions control devices could not be constructed to meet both California's oxides of nitrogen standard and a carbon monoxide standard as stringent as the federal standard. Id. In an effort to impose its high oxides of nitrogen standard, California proposed a carbon monoxide standard that was less stringent than the federal carbon monoxide standard. Id. The EPA allowed California's stringent oxides of nitrogen standard to make up for its less stringent carbon monoxide standard, as long as its regulatory program as a whole was more protective than the federal regulations. Id. Dissatisfied with this decision, opponents of California's regulations argued that Section 209(b) required California to show that its carbon monoxide standard was individually more protective than the federal carbon monoxide standard. In 1977, Congress resolved this dispute by amending Section 209(b). The new language of Section 209(b) made explicit that California need only determine that its standards are, "in the aggregate, at least as protective of public health and welfare" as the federal standards. 42 U.S.C. § 7543(b) (1). So long as California has made that determination, the EPA must grant California a waiver unless the EPA finds that any of the waiver denial criteria are met. See id. § 7543(b)(1)(A)-(C). The amendment ensures that California is not required to determine that each new proposed regulation is more protective than its federal counterpart. Id. It was intended to give California the "broadest discretion in selecting the best means to protect

the health of its citizens and the public welfare." H.R. Rep. No. 95-294, at 301-02 (1977).

After Congress amended Section 209(b) to provide that California need only determine that its standards were, "in the aggregate," at least as protective as the federal standards, the EPA decided to apply a similar approach to its analysis of whether California's proposed standards met any of the waiver denial criteria. See 42 U.S.C. § 7543(b)(1)(A)–(C). Thus, in examining whether any of the waiver denial criteria applied, the EPA considered only whether California's proposed standards, in the aggregate, met any of the criteria—not whether each individual standard could be denied under any of the criteria. The EPA continued to evaluate California's waiver applications under Section 209(b) using this aggregate method of evaluation for decades. In the fiftyfive years since Section 209(b) was originally enacted, the EPA has granted California seventy-five waivers using the aggregate method of evaluation. See California State Motor Vehicle Pollution Control Standards; Advanced Clean Car Program; Reconsideration of a Previous Withdrawal of a Waiver of Preemption; Notice of Decision, 87 Fed. Reg. 14337 (Mar. 14, 2022) ("2022 Waiver Reinstatement Decision"); see also EPA, Vehicle

^{3.} The 1977 Clean Air Act amendments also empowered other states to choose between adopting the federal standards or the California standards. 42 U.S.C. § 7507. As of the date of this opinion, seventeen states have chosen to adopt some portion of the California regulations. See Cal. Air Res. Bd., States that Have Adopted California's Vehicle Regulations, https://perma.cc/HCS4-X7NP (last visited Mar. 27, 2024).

Emissions California Waivers and Authorizations, https://perma.cc/5T7U-L8GE (last visited Mar. 27, 2024).

In the 1960s and 1970s, California's emissions standards focused on ozone-generating pollutants, like nitrogen oxides, but over time, California expanded its regulatory program to restrict a variety of other emissions, such as methane and other greenhouse gases. See, e.g., California State Motor Vehicle Pollution Control Standards; Waiver of Federal Preemption, 43 Fed. Reg. 25729, 25735 (June 14, 1978); California State Motor Vehicle Pollution Control Standards; Waiver of Federal Preemption Notice of Decision, 49 Fed. Reg. 18887, 18890 (May 3, 1984). In 1993, the EPA approved a waiver of California's first ZEV standard, which required an annually increasing percentage of vehicles sold in California to produce zero tailpipe emissions. See California State Motor Vehicle Pollution Control Standards; Waiver of Federal Preemption; Decision, 58 Fed. Reg. 4166 (Jan. 13, 1993).

C.

In recent decades, California has continued to face significant pollution and climate challenges. It contains seven of the ten worst areas for ozone pollution in the country and six of the ten worst areas for small particulate matter. See 2022 Waiver Reinstatement Decision, 87 Fed. Reg. at 14377 n.469. It also faces "increasing risks from record-setting fires, heat waves, storm surges, sea-level rise, water supply shortages and extreme heat." Id. at 14363; see also id. at 14338–39 & nn.37, 43. And these conditions are exacerbated by climate change. Id. at

14350 & n.165. Moreover, pollution and climate change have particularly harmful impacts on California due to its large agriculture and ocean-based economies, dependence on an over-stressed water supply, long coastlines, and susceptibility to wildfires. See California State Motor Vehicle Pollution Control Standards; Notice of Decision Granting a Waiver of Clean Air Act Preemption for California's 2009 and Subsequent Model Year Greenhouse Gas Emission Standards for New Motor Vehicles, 74 Fed. Reg. 32744, 32746 (July 8, 2009).

To combat these challenges, in 2005, California applied for a waiver for a new set of regulations limiting greenhouse gas emissions. See Cal. Air Res. Bd., Low-Emission Vehicle Greenhouse Gas Program, https:// perma.cc/VC85-GQ2S (last visited Mar. 27, 2024). The request sparked disagreement among several subsequent presidential administrations. Under President George W. Bush's Administration, the EPA initially denied the waiver on the basis that the standards were not addressing "compelling and extraordinary conditions." California State Motor Vehicle Pollution Control Standards; Notice of Decision Denying a Waiver of Clean Air Act Preemption for California's 2009 and Subsequent Model Year Greenhouse Gas Emission Standards for New Motor Vehicles, 73 Fed. Reg. 12156, 12159-63 (Mar. 6, 2008). One year later, under the Obama Administration, the EPA determined that its initial decision to deny the waiver had been based on an incorrect interpretation of Section 209(b), and ultimately granted the waiver. California State Motor Vehicle Pollution Control Standards; Notice of Decision Granting a Waiver of Clean Air Act Preemption for California's 2009 and Subsequent Model Year Greenhouse Gas Emission Standards for New Motor Vehicles, 74 Fed. Reg. 32744, 32745–46 (July 8, 2009).

In 2012, California applied for the waiver at issue in this case, seeking to promulgate a new set of regulations called the Advanced Clean Car Program. See California State Motor Vehicle Pollution Control Standards; Notice of Decision Granting a Waiver of Clean Air Act Preemption for California's Advanced Clean Car Program and a Within the Scope Confirmation for California's Zero Emission Vehicle Amendments for 2017 and Earlier Model Years, 78 Fed. Reg. 2112 (Jan. 9, 2013). The new regulations included a Low Emission Vehicle ("LEV") Program, which set emissions requirements for new cars in Model Years 2017 to 2025 with the goal of reducing carbon dioxide emissions by thirty-four percent, and a ZEV Program, which required around fifteen percent of manufacturers' fleets to be electric cars by Model Year 2025. Id. The EPA initially granted the waiver in 2013. Id. In response, automobile manufacturers in California began making investments to meet both programs' requirements. See, e.g., Industry Resp.-Intervenor Br. 2-4.

In 2018, after car manufacturers had adjusted their fleets to comply with California's Advanced Clean Car Program, the EPA changed its course. It issued a notice of proposed rulemaking to withdraw the portions of the 2013 waiver covering California's LEV and ZEV standards. See The Safer Affordable Fuel-Efficient (SAFE) Vehicles Rule for Model Years 2021-2026 Passenger Cars and Light Trucks, 83 Fed. Reg. 42986 (Aug. 24, 2018).

The EPA withdrew the 2013 waiver on September 27, 2019. The Safer Affordable Fuel-Efficient (SAFE) Vehicles Rule Part One: One National Program, 84 Fed. Reg. 51310 (Sept. 27, 2019) ("2019 Withdrawal Decision"). The EPA offered three bases for the withdrawal. *Id.* at 51328–41. First, the 2013 waiver conflicted with a recent determination by the National Highway Traffic Safety Administration ("NHTSA") that state greenhouse gas regulations were preempted by a provision of the EPCA that prohibits states from enacting their own fuel economy standards. Id. at 51337–38; see also 49 U.S.C. § 32902(a), (f) (empowering the NHTSA to set federal fuel economy standards); id. § 32919(a) (preempting state fuel economy standards). Second, the EPA had decided that it would no longer follow a "whole program" interpretation of Section 209(b), and instead would evaluate whether each individual California standard met the requirement that it be necessary to "meet compelling and extraordinary conditions." 2019 Withdrawal Decision, 84 Fed. Reg. at 51341 (quoting 42 U.S.C. § 7543(b)(1)). Third, California could not show that its LEV and ZEV regulations were necessary to meet compelling and extraordinary conditions because California could not show a "particularized nexus" between greenhouse gas emissions and California's air pollution problems. Id. According to the EPA, because greenhouse gas pollution from global sources is blended in the atmosphere, the consequences of climate change from which California suffered were not "compelling and extraordinary." Id. at 51333-34.

Following recission of the 2013 waiver, automobile manufacturers such as Honda, Ford, Volvo, BMW,

and Volkswagen entered into independent agreements with California to continue reducing emissions. See Revised 2023 and Later Model Year Light-Duty Vehicle Greenhouse Gas Emissions Standards, 86 Fed. Reg. 74434, 74458 (Dec. 30, 2021). Under these agreements, the automakers would continue to meet the LEV and ZEV standards in the California regulations. Id. Automakers were motivated to sign these agreements by the investments they had already made in updating their fleets and growing consumer demand for electric vehicles. See J.A. 155–57.

In 2021, under the Biden Administration, the EPA revisited its 2019 withdrawal of the 2013 waiver. California State Motor Vehicle Pollution Control Standards; Advanced Clean Car Program; Reconsideration of a Previous Withdrawal of a Waiver of Preemption; Opportunity for Public Hearing and Public Comment, 86 Fed. Reg. 22421 (Apr. 28, 2021). On March 14, 2022, the EPA reinstated its 2013 waiver for California's Advanced Clean Car Program. 2022 Waiver Reinstatement Decision, 87 Fed. Reg. at 14332. As a result of that reinstatement, California's LEV and ZEV standards for Model Years 2017 through 2025 came back into force. Id. at 14333. The EPA provided three explanations for its 2022 Waiver Reinstatement Decision: the EPA exceeded its inherent authority to revisit its 2013 decision; it improperly rejected the "whole program" approach; and it improperly considered the NHTSA's view of the EPCA, which was beyond the scope of Section 209(b). *Id.* at 14333–35.

On May 12, 2022, State Petitioners filed a petition for review in this Court challenging the EPA's decision to reinstate the 2013 waiver (22-1081). That same day, three groups of Fuel Petitioners filed petitions for review of the same EPA action (22-1083, 22-1084, and 22-1085). The Court consolidated these cases (22-1081). California and several other states and cities (collectively, "California"), environmental organizations, and automobile manufacturers have intervened in support of respondents in the consolidated case.

^{4.} The state and city intervenors are the City of Los Angeles, the City of New York, Massachusetts, Pennsylvania, the District of Columbia, California, Colorado, Connecticut, Delaware, Hawaii, Illinois, Maine, Maryland, Minnesota, Nevada, New Jersey, New Mexico, New York, North Carolina, Oregon, Rhode Island, Vermont, and Washington.

^{5.} The environmental organization intervenors are the Center for Biological Diversity, the Clean Air Council, the Conservation Law Foundation, the Environmental Defense Fund, the Environmental Law and Policy Center, the National Parks Conservation Association, the Natural Resources Defense Council, the Public Citizen, the Sierra Club, and the Union of Concerned Scientists.

^{6.} The automobile manufacturer intervenors are Ford Motor Company, Volkswagen Group of America, Inc., BMW of North America, LLC, American Honda Motor Co., Inc., Volvo Car USA LLC, the National Coalition for Advanced Transportation, Advanced Energy Economy, Calpine Corporation, National Grid USA, the New York Power Authority, and the Power Companies Climate Coalition.

In their petition, Fuel Petitioners argue that the 2022 decision was arbitrary and capricious and exceeded the EPA's authority under Section 209(b) because climate change is not a "compelling and extraordinary condition," and California does not "need" its standards to "meet" its climate conditions. See, e.g., Fuel Pet. Br. 10–11. In challenging the EPA's determination of California's "need," Fuel Petitioners argue that the EPA's aggregate approach is wrong. Id. Meanwhile, State Petitioners claim that by granting a waiver to California, but not to any other state, the EPA has violated State Petitioners' constitutional right to equal sovereignty. See State Pet. Br. 28–33. State Petitioners also claim that the waiver is contrary to the preemption of state fuel economy standards set out in the EPCA. See State Pet. Br. 33–41.

We hold that neither Fuel Petitioners as to their statutory claims nor State Petitioners as to their EPCA claims establish standing to bring suit, and thus we do not reach the merits of their claims. We reject State Petitioners' constitutional claim on the merits.

II.

A.

We begin with the question whether either State or Fuel Petitioners have standing based on their assertions that the waiver will cause them economic injury. Fuel Petitioners premise their standing as to the entirety of their petition for review on their claimed economic injury. State Petitioners, meanwhile, premise their standing for their claim that the waiver is preempted by the EPCA on their alleged economic injury.

A "showing of standing is 'an essential and unchanging' predicate to any exercise of our jurisdiction." Fla. Audubon Soc. v. Bentsen, 94 F.3d 658, 663, 320 U.S. App. D.C. 324 (D.C. Cir. 1996) (en banc) (quoting Lujan v. Defs. of Wildlife, 504 U.S. 555, 560, 112 S. Ct. 2130, 119 L. Ed. 2d 351 (1992)). "[S]tanding is assessed as of the time a suit commences." Del Monte Fresh Produce Co. v. United States, 570 F.3d 316, 324, 386 U.S. App. D.C. 406 (D.C. Cir. 2009). The "irreducible constitutional minimum of standing contains three elements." Lujan, 504 U.S. at 560. First, the plaintiff must have suffered an injury-in-fact an "invasion of a judicially cognizable interest which is (a) concrete and particularized and (b) actual or imminent, not conjectural or hypothetical." Bennett v. Spear, 520 U.S. 154, 167, 117 S. Ct. 1154, 137 L. Ed. 2d 281 (1997). Second, there must be a "causal connection between the injury and the conduct complained of—the injury must be fairly traceable to the challenged action of the defendant, and not the result of the independent action of some third party not before the court." Id. Third, "it must be 'likely," as opposed to merely 'speculative,' that the injury will be 'redressed by a favorable decision." Lujan, 504 U.S. at 561 (quoting Simon v. E. Ky. Welfare Rts. Org., 426 U.S. 26, 38, 43, 96 S. Ct. 1917, 48 L. Ed. 2d 450 (1976)).

"A petitioner bears the burden of establishing each" of the elements of standing. *Chamber of Com. of U.S. v. EPA*, 642 F.3d 192, 200, 395 U.S. App. D.C. 193 (D.C. Cir. 2011). To meet that burden, a petitioner must "show

a 'substantial probability' that it has been injured, that the defendant caused its injury, and that the court could redress that injury." *Sierra Club v. EPA*, 292 F.3d 895, 899, 352 U.S. App. D.C. 191 (D.C. Cir. 2002) (quoting *API v. United States EPA*, 216 F.3d 50, 63, 342 U.S. App. D.C. 159 (D.C. Cir. 2000)). And a petitioner may not wait to attempt to meet its burden of demonstrating standing until after the respondent contests the issue. Rather, absent "good cause shown," a petitioner whose standing is not readily apparent must show that it has standing in "its opening brief." *Id.* at 900–01. A petitioner may carry this "burden of production by citing any record evidence relevant to its claim of standing and, if necessary, appending to its filing additional affidavits or other evidence sufficient to support its claim." *Id.*; *see also* D.C. Cir. R. 28(a)(7).

Whether a petitioner has standing to challenge a particular government action depends, in part, upon whether the petitioner is "an object of the action" at issue. Lujan, 504 U.S. at 561. When a petitioner is an object of the action it seeks to challenge, causation and redressability are usually easy to demonstrate. Id. But when, as here, the petitioner "is not [it]self the object of the government action or inaction he challenges, standing is not precluded, but is ordinarily 'substantially more difficult' to establish." Id. at 562 (quoting Allen v. Wright, 468 U.S. 737, 758, 104 S. Ct. 3315, 82 L. Ed. 2d 556 (1984)). Because any injury to petitioners "hinges on actions taken by manufacturers, the petitioners carry 'the burden of adduc[ing] facts showing that those [third-party] choices have been or will be made in such manner as to produce causation and permit redressability of injury." Chamber

of Com., 642 F.3d at 201 (alterations in original) (quoting Ctr. for Biological Diversity v. United States DOI, 563 F.3d 466, 477, 385 U.S. App. D.C. 257 (D.C. Cir. 2009)).

As we will explain, these principles compel the conclusion that both State and Fuel Petitioners lack standing premised on their claimed economic injuries because neither group of Petitioners has met their burden of demonstrating that those injuries are redressable.

В.

Fuel Petitioners argue that, by requiring vehicle manufacturers to sell vehicles that use less or no liquid fuel, California's LEV and ZEV requirements depress the demand for liquid fuels. Fuel Petitioners and their members, who produce and sell liquid fuels and the raw materials used to produce those fuels, are thereby financially injured by the reduction in demand for those products. Fuel Pet. Br. 16; Fuel Pet. Reply Br. 3–4. In support of Fuel Petitioners' contention that they are economically injured by the waiver, Fuel Petitioners offer over a dozen declarations by individuals who are affiliated with Fuel Petitioner entities and organizations; the individuals explain that the entity or organization is

^{7.} Fuel Petitioners include both associations and individual entities. Fuel Pet. Br. 16. Because, as we explain, we conclude that all Fuel Petitioners have failed to establish redressability, we need not address whether any of the Fuel Petitioner associations have established organizational standing. *Cf. Sierra Club*, 292 F.3d at 898 (laying out the requirements for establishing organizational standing).

involved with producing or selling fuel and that the waiver causes Fuel Petitioners economic injury by reducing the demand for fuel and related products.

State Petitioners, meanwhile, allege three financial injuries that they contend are caused by the waiver. First, the waiver causes manufacturers to increase the cost of conventional vehicles elsewhere in the country in order to account for the cost of meeting the requirements imposed on manufacturers by the waiver granted to California. State Petitioners explain that because they purchase conventional vehicles, the increase in the prices for those vehicles that results from the waiver causes State Petitioners financial harm. State Pet. Br. 14-15. Second, State Petitioners contend that the greater shift to electric vehicles that results from the waiver will cause State Petitioners to generate less fuel-tax revenue. Id. Finally, State Petitioners argue that the increase in electric vehicles caused by the waiver will affect the States' electrical grids. In support of their standing claims, State Petitioners offer a declaration from each individual State Petitioner and a declaration from an economist, Benjamin Zycher, Ph.D. Each State Petitioner's declaration states that the state purchases conventional (that is, gas-or diesel-powered) vehicles. State Pet. Add. 6–36. In his declaration, Dr. Zycher contends that California's ZEV requirement will have several economic impacts on State Petitioners, including an increase in the cost of conventional vehicles nationwide, a "decline in the quality of delivered state services," a reduction in "fuel tax revenues available for the provision of highway and

road services," and an "increase in the costs and prices of delivering electric power services." State Pet. Add. 38–39.

The EPA and California both dispute that State and Fuel Petitioners' allegations and evidence establish injury and causation sufficient to support standing. EPA Br. 23–28 (arguing State Petitioners fail to demonstrate standing); California Br. 9–15 (arguing both State and Fuel Petitioners fail to demonstrate standing). For example, as to causation, California argues that both groups of Petitioners fail to demonstrate that their alleged injuries are caused by the 2022 waiver reinstatement, rather than the original 2013 waiver or rising consumer demand for electric vehicles more generally. California Br. 11, 14. But this Court need not definitively decide whether either set of Petitioners has established injury or causation. However robust their claims of injury and causation are, State and Fuel Petitioners spend considerably less time explaining how those injuries are redressable. Indeed, even assuming that both sets of Petitioners have established injury and causation sufficient for standing, Petitioners' standing arguments fail for the same reason: Both groups of Petitioners fall far short of meeting their burden of demonstrating a "substantial probability" that their alleged injuries would be redressed by a favorable decision by this Court. Am. Petroleum, 216 F.3d at 63; see also Sierra Club, 292 F.3d at 899-900.

Fuel Petitioners assert in their opening brief—without explanation or citation—that this Court could redress their injuries "by setting aside the action." Fuel Pet. Br. 16. Fuel Petitioners' declarations offer little

more; to the extent that Fuel Petitioners' declarations discuss redressability at all, the declarations state that the injuries discussed therein "would be substantially ameliorated if EPA's decision were set aside." State Petitioners' opening brief is similarly conclusory regarding redressability. State Petitioners assert that their "injuries are redressable because a judgment setting aside the waiver would eliminate the source of their injuries." State Pet. Br. 16. However, none of the declarations submitted by State Petitioners with their opening brief addresses redressability at all.

The difficulty for Fuel and State Petitioners is that their claimed injuries "hinge[] on" the actions of third parties—the automobile manufacturers who are subject to the waiver. *Chamber of Com.*, 642 F.3d at 201. Redressability, too, "hinge[s] on the response of" those same automobile manufacturers. *Lujan*, 504 U.S. at 562. Both groups of Petitioners' injuries would be redressed only if automobile manufacturers responded to vacatur of the waiver by producing and selling fewer non-conventional vehicles or by altering the prices of their vehicles such that fewer non-conventional vehicles—and more conventional vehicles—were sold.

And, aside from turning on the actions of the automobile manufacturers subject to the waiver, redressability is further complicated by the relatively short duration of the waiver that Petitioners challenge. These petitions for review concern only the EPA's decision, in March 2022, to reinstate the waiver it had previously granted California as to Model Years 2017 through 2025. See 2022

Waiver Reinstatement Decision, 87 Fed. Reg. 14337. Thus, to meet their burden of demonstrating redressability, both sets of Petitioners must demonstrate a "substantial probability" not only that automobile manufacturers are likely to respond to a decision by this Court by changing their fleets in a way that alleviates their injuries in some way, but also that automobile manufacturers would do so relatively quickly—by Model Year 2025. *Am. Petroleum*, 216 F.3d at 63.

The record evidence provides no basis for us to conclude that manufacturers would, in fact, change course with respect to the relevant model years if this Court were to vacate the waiver. To begin, Petitioners fail to point to any evidence affirmatively demonstrating that vacatur of the waiver would be substantially likely to result in any change to automobile manufacturers' vehicle fleets by Model Year 2025. The only evidence points in the opposite direction, indicating that automobile manufacturers need years of lead time to make changes to their future model year fleets. In a comment submitted to the EPA during the rulemaking process regarding the EPA's 2019 recission of the 2013 waiver, for example, Ford Motor Company stated that its product cycle requires several years of lead time for planning, and that its "regulatory lead time (i.e., awareness of future regulatory requirements)" is seven years. J.A. 637. Ford explained that, as a result, if the regulatory landscape shifted in some way, "little or nothing could be done to re-optimize the company's product plans, which are largely fixed for the next few years." Id. Further, the record indicates that other automobile manufacturers would also require years

of lead time to alter their product plans. In comments submitted to the EPA during the EPA's rulemaking process regarding the 2022 waiver reinstatement, Tesla, Inc. and Toyota Motor North America, Inc., explained that their vehicle product cycles, too, can also begin years before a vehicle is launched. J.A. 371, 477; see also J.A. 370 n.5 (summarizing similar statements from Chrysler Group LLC, Hyundai America Technical Center, Inc., and Mitsubishi Motors North America). Thus, even if automobile manufacturers were inclined to change course so as to alleviate the Petitioners' injuries within the given model years, it is far from clear that they could do so within the model years covered by the waiver.8 To be sure, it is possible that manufacturers could change their prices without modifying their production cycles, which may redress Petitioners' injuries because pricing could affect the mix of conventional and electric vehicles purchased. But Petitioners point us to no evidence that manufacturers would change their prices by Model Year 2025 either.

Despite the paucity of evidence in the record regarding the redressability of their injuries, neither group of Petitioners attempts to explain in any detail how their injuries are redressable, let alone to "cit[e] any record evidence" or to file "additional affidavits or other

^{8.} We also note that several automobile manufacturers have intervened in support of the EPA in this case. Those manufacturers explain in their brief in support of the EPA that "both internal sustainability goals and external market forces" are prompting manufacturers to transition toward electric vehicles, irrespective of California's regulations. Industry Resp.-Intervenor Br. 6–7.

evidence sufficient to support" redressability. Sierra Club, 292 F.3d at 900–01. Nor, for that matter, does either set of Petitioners grapple with the relatively short nature of the waiver they challenge. Rather, all Petitioners seem to have treated redressability as a foregone conclusion. See Crete Carrier Corp. v. EPA, 363 F.3d 490, 494, 361 U.S. App. D.C. 54 (D.C. Cir. 2004) (petitioners lacked standing where they failed to produce "actual evidence" regarding how the regulated parties "would respond" to vacatur); Branton v. FCC, 993 F.2d 906, 912, 301 U.S. App. D.C. 244 (D.C. Cir. 1993) ("A court is rightly reluctant to enter a judgment which may have no real consequence, depending upon the putative cost-benefit analyses of third parties over whom it has no jurisdiction and about whom it has almost no information.").

When asked about redressability at oral argument, counsel for Fuel Petitioners emphasized that redressability—as with each prong of standing—is assessed when a lawsuit is first filed. Oral Argument Transcript 74; see also Del Monte, 570 F.3d at 325 ("[S] tanding is assessed as of the time a suit commences."). True enough. But that does not help Fuel Petitioners: Even "as of the time" this lawsuit commenced, Fuel Petitioners had failed to point to any evidence in the record showing that their alleged injuries were redressable. Del Monte, 570 F.3d at 325. Put differently, the flaw in Fuel Petitioners' standing arguments is not—as counsel for Fuel Petitioners contended at oral argument, Oral Argument Transcript 74-75—that their standing arguments were sufficient when originally filed, but that their claims have been mooted by the passage of time. Fuel Petitioners' standing arguments were deficient from the start.

State Petitioners, meanwhile, argue that, to the extent that there is any doubt that they have met their burden of demonstrating causation and redressability, this Court should resolve it in their favor given the "special solicitude" to which states are entitled when they seek to protect their "quasi-sovereign interests." Massachusetts v. EPA, 549 U.S. 497, 518-20, 127 S. Ct. 1438, 167 L. Ed. 2d 248 (2007); see also State Pet. Br. 16. We disagree. The "special solicitude" afforded to states can relax standing requirements only so far. Massachusetts, 549 U.S. at 520. Even the "greater leeway" afforded to states seeking to protect quasi-sovereign interests cannot save defective standing claims when, as here, the record is "almost completely silent" with respect to an element of a state's standing. Alaska v. U.S. Dep't of Agric., 17 F.4th 1224, 1230, 454 U.S. App. D.C. 493 (D.C. Cir. 2021).

State and Fuel Petitioners' sparse treatment of redressability is particularly surprising because, in a previous case, this Court noted that it could not presume redressability in essentially the same circumstances. In *Chamber of Commerce of the United States v. EPA*, the Chamber of Commerce and the National Automobile Dealers Association, on behalf of their automobile dealer members, petitioned for review in this Court of the EPA's decision to grant California a waiver, under Section 209(b), with respect to automobile Model Years 2009 through 2016. 642 F.3d at 196–97. There, the petitioners—automobile dealers who, like the Petitioners in this case, were not directly subject to the waiver—explained that automobile manufacturers' responses to the waiver injured them in two ways. First, automobile manufacturers would respond

to the waiver by altering the mix of vehicles they sold in California and other states; as a result, vehicle dealers would be injured because they would be unable to obtain specific vehicles that their customers wanted to buy. *Id.* at 201. And second, the California standards would increase automobile manufacturers' costs and, in turn, increase the prices of the automobiles they manufactured. *Id.* The automobile dealers believed they would be injured by those increased vehicle costs because they would have to choose whether to keep their prices the same, and accordingly lower their profit margins, or to increase their prices to account for the increased vehicle costs, at the risk of turning away customers. *Id.*

The Chamber of Commerce Court ultimately resolved petitioners' claims on mootness grounds, not standing. Id. at 204, 206. But before reaching that conclusion, the Court expressed serious doubts that the petitioners had met their burden of demonstrating redressability. Id. at 205. The record before the Court indicated that vacatur of the challenged waiver may not result in any change on the part of automobile manufacturers. And, the Court noted, "Petitioners ha[d] offered no evidence to the contrary, and no evidence that, if the waiver were vacated, [automobile manufacturers] would proceed on a different course more favorable to the petitioners." *Id*. at 205-06. So even if petitioners' claims were not moot, their failure to introduce redressability evidence made it—at a minimum—rather unclear whether their claims were redressable.

As the EPA and intervenors correctly recognize, State and Fuel Petitioners' standing submissions run into precisely the same problem here. In its response brief, the EPA explains in some detail how State Petitioners have failed to substantiate the redressability of their injuries. EPA Br. 26. California, meanwhile, argues that neither group of Petitioners has provided any evidence that vacatur would remedy their injuries. California Br. 13. Further underlining the point, California offers an expert declaration by Joshua M. Cunningham, the Chief of the Advanced Clean Cars Branch of the California Air Resources Board, who explains in specific terms why the Petitioners' claims are unlikely to be redressed by a favorable decision by this Court. California Add. 84–85; 96–99. Cunningham explains that automobile manufacturers have already made a number of public commitments regarding both vehicle pricing and availability with respect to the remaining model years covered by the challenged waiver; those public commitments would tend to suggest that neither group of Petitioners' claims are redressable. As Cunningham puts it, "manufacturers have likely already made pricing decisions for" the remaining model years. California Add. 96. Cunningham also states that "manufacturers are already selling *more* qualifying vehicles in California than the State's standards require," suggesting that vacatur of the zero-emission vehicle mandate would not redress Petitioners' injuries. California Add. 98. Indeed, record evidence supports the fact that manufacturers already exceed California's ZEV requirements. See J.A. 300-02. Yet despite these arguments against their theory of redressability, neither State nor Fuel Petitioners meaningfully addressed the redressability of their economic injuries in their reply briefs. State Pet. Reply Br. 3, Fuel Pet. Reply Br. 3–6.

Ultimately, the record evidence, coupled with the filings of the EPA and intervenors, provide this Court with no basis to conclude that Petitioners' claims are redressable—a necessary element of standing that Petitioners bear the burden of establishing. As in *Chamber* of Commerce, "Petitioners have offered no evidence to the contrary, and no evidence that, if the waiver were vacated, [automobile manufacturers] would proceed on a different course more favorable to the petitioners." 642 F.3d at 205. Rather, both State and Fuel Petitioners "offer only assertions, not facts, to support their claims about the likely response" of automobile manufacturers to a favorable decision by this Court. Crete Carrier Corp., 363 F.3d at 494. But "[s]peculative and unsupported assumptions regarding the future actions of third-party market participants are insufficient to establish Article III standing." Id. 10 Petitioners may not proclaim that their

^{9.} This Court denied a motion by State Petitioners to file with their reply brief new evidence regarding their standing. Per Curiam Order, Aug. 9, 2023. That proposed supplemental evidence, however, concerned only State Petitioners' allegations of economic injuries stemming from the waiver, not the redressability of those injuries. ECF 2019756. At any rate, the evidence was too late. See D.C. Cir. R. 28(a)(7).

^{10.} We conclude that Petitioners have failed to introduce sufficient evidence to raise a dispute of fact as to whether changes to the remaining model year fleets are substantially likely if vacatur were to occur, so we have no need to refer this to a district judge or special master as a disputed factual issue for resolution before making our ruling on redressability. See FCC v. ITT World

injuries are redressable and expect this Court to take them at their word. On this record, redressability poses a "fatal stumbling block" for both sets of Petitioners. *Cato Institute v. SEC*, 4 F.4th 91, 95, 453 U.S. App. D.C. 184 (D.C. Cir. 2021). We accordingly hold that both State and Fuel Petitioners lack standing premised on their economic injuries because they have failed to meet their burdens of demonstrating that their claims are redressable.

C.

After oral argument, Fuel Petitioners filed a motion to supplement the record and to file a supplemental brief regarding their standing. Fuel Pet. Mot. to Supp. 1. Fuel Petitioners contend that the EPA and California raised for the first time at oral argument the question whether Fuel Petitioners' claims could be redressed within the relevant model years—an issue Fuel Petitioners argue pertains to mootness, not the redressability of their claims. Fuel Pet. Mot. to Supp. 1–2. Fuel Petitioners argue that they should be allowed to file new evidence with this Court to "address that new argument." Fuel Pet. Mot. to Supp. 1.

We deny Fuel Petitioners' motion to supplement the record and to file a supplemental brief. As we have explained, a petitioner must generally demonstrate standing in its opening brief, either by "citing any record evidence relevant to its claim of standing" or, where necessary, by "appending to its filing additional affidavits

Commc'ns, Inc., 466 U.S. 463, 469, 104 S. Ct. 1936, 80 L. Ed. 2d 480 (1984) (citing 28 U.S.C. § 2347(b)(3)); Fed. R. App. R. 48(a).

or other evidence." Sierra Club, 292 F.3d at 900–01. This Court has, on rare occasion, accepted late affidavits or other evidence in support of standing for "good cause" shown. Am. Libr. Ass'n v. FCC, 401 F.3d 489, 495–96, 365 U.S. App. D.C. 207 (D.C. Cir. 2005) (quoting Sierra Club, 292 F.3d at 900); Nat'l Council for Adoption v. Blinken, 4 F.4th 106, 112, 453 U.S. App. D.C. 199 (D.C. Cir. 2021). We have found "good cause" when, for example, "'the parties reasonably, but mistakenly, believed' that they 'sufficiently demonstrated standing' or when they 'reasonably assumed that their standing was self-evident." Nat'l Council for Adoption, 4 F.4th at 111 (quoting Twin Rivers Paper Co. LLC v. SEC, 934 F.3d 607, 614, 443 U.S. App. D.C. 74 (D.C. Cir. 2019)).

No such good cause exists here. We do not think Fuel Petitioners could have reasonably believed that they had adequately demonstrated standing or that their standing was "self-evident" from the record when they filed their opening brief. Twin Rivers, 934 F.3d at 614. As this Court and the Supreme Court have repeatedly explained, redressability is "substantially more difficult' to establish" when, as here, Petitioners are not directly regulated by the government action they seek to challenge. Lujan, 504 U.S. at 562 (quoting *Allen*, 468 U.S. at 758). Indeed, as noted above, this Court has previously expressed doubt that petitioners seeking to challenge a Section 209 waiver had demonstrated redressability where they had failed to put any such evidence in the record. Chamber of Com., 642 F.3d at 205. And Fuel Petitioners should have been aware that redressability may pose a particularly challenging obstacle here, considering the relatively narrow timeframe of the particular waiver Petitioners challenge and the evidence in the record showing that automobile manufacturers generally require years of lead time to make changes to their future model year fleets. Yet Fuel Petitioners failed to meaningfully address redressability in their opening brief at all, either by "identify[ing] . . . record evidence" or by offering the Court evidence of their own. Sierra Club, 292 F.3d at 899.

Second, even if Fuel Petitioners reasonably believed that their standing was "self-evident" when they filed their opening brief, Petitioners offer no explanation for having failed to address redressability in their reply brief after California raised the issue in its opposition brief. Twin *Rivers*, 934 F.3d at 614. In this respect, Fuel Petitioners' motion relies on a false premise: Oral argument was plainly not the first time that California argued that Fuel Petitioners had failed to demonstrate redressability. Rather, as we have explained, California explicitly argued that Fuel Petitioners had offered no evidence regarding the redressability of their injuries, and California provided the Court with a declaration that addressed the point. Having failed even to attempt to respond to California's arguments regarding redressability at the reply stage, Fuel Petitioners provide this Court with no reason to allow them to do so now.

III.

State Petitioners also argue that the EPA's 2022 decision is "contrary to constitutional right" under 5 U.S.C. § 706(2)(B) because Section 209(b) of the Clean Air

Act is unconstitutional. They rely on the equal sovereignty principle, which the Supreme Court applied in Shelby County v. Holder, 570 U.S. 529, 133 S. Ct. 2612, 186 L. Ed. 2d 651 (2013), to hold that Fifteenth Amendment legislation that disparately impacts states' control over voting procedures must be "sufficiently related to the problem it targets." Id. at 542 (quoting Nw. Austin Mun. Util. Dist. No. One v. Holder, 557 U.S. 193, 203, 129 S. Ct. 2504, 174 L. Ed. 2d 140 (2009)). State Petitioners argue that this principle also categorically prohibits Congress from using its Commerce Clause power in a way that withdraws sovereign authority from some states but not others. And Section 209(b), they say, violates that principle by preempting the authority of every state but California to regulate motor vehicle emissions. We conclude that State Petitioners have standing to raise this constitutional claim, but we join the two other circuits to have considered the issue in rejecting State Petitioners' request to extend the equal sovereignty principle in this fashion. See NCAA v. Governor of New Jersey, 730 F.3d 208, 239 (3d Cir. 2013), abrogated on other grounds by Murphy v. NCAA, 584 U.S. 453, 138 S. Ct. 1461, 200 L. Ed. 2d 854 (2018); Mayhew v. Burwell, 772 F.3d 80, 95 (1st Cir. 2014).

A.

To assess State Petitioners' standing for this constitutional claim, we again "assume that on the merits" petitioners "would be successful." *City of Waukesha v. EPA*, 320 F.3d 228, 235, 355 U.S. App. D.C. 100 (D.C. Cir. 2003). Assuming State Petitioners' constitutional theory is correct, Section 209(b) and the EPA's 2022 decision

violate their constitutionally protected interest in equal sovereignty by leaving them with less regulatory authority over vehicle emissions than California. This claimed injury is akin to the type of dignitary injury recognized in equal protection cases. *Heckler v. Mathews*, 465 U.S. 728, 739–40, 104 S. Ct. 1387, 79 L. Ed. 2d 646 (1984). And, State Petitioners argue, invalidating the decision would redress that injury and restore their sovereign equality by removing California's greater authority.

Respondents resist that analysis on the ground that State Petitioners do not ask this court to increase their own sovereign authority over motor vehicle emissions. The States instead seek to reduce California's authority.

The Supreme Court has repeatedly held, however, that this type of "leveling down" remedy is sufficient to support standing when a party asserts a constitutional right to equality. As the Court has put it, "when the 'right invoked is that to equal treatment,' the appropriate remedy is a mandate of equal treatment, a result that can be accomplished by withdrawal of benefits from the favored class as well as by extension of benefits to the excluded class." Heckler, 465 U.S. at 740 (quoting Iowa-Des Moines Nat'l Bank v. Bennett, 284 U.S. 239, 247, 52 S. Ct. 133, 76 L. Ed. 265 (1931)); see also Sessions v. Morales-Santana, 582 U.S. 47, 72–73, 137 S. Ct. 1678, 198 L. Ed. 2d 150 (2017). That principle developed in cases applying the Fourteenth Amendment's Equal Protection Clause. But Respondents have not identified—and we do not perceive—any material reason to treat the right to equal sovereignty claimed here

any differently for standing purposes.¹¹ And under the logic of the Equal Protection cases, holding Section 209(b) unconstitutional and vacating the waiver would redress the claimed constitutional injury by leaving all states equally positioned, in that none could regulate vehicle emissions. Accordingly, unlike with their asserted economic injuries, State Petitioners' asserted constitutional injury can be redressed even absent evidence that manufacturers will change their plans before the waiver expires.¹²

^{11.} Although we find State Petitioners' claimed dignitary harm sufficiently analogous in kind for purposes of standing, we do not suggest that this harm is commensurate with equal protection injuries based on the perpetuation of "archaic and stereotypic notions" and stigmatization of members "of [a] disfavored group as "innately inferior." *Heckler*, 465 U.S. at 739 (quoting *Miss. Univ. for Women v. Hogan*, 458 U.S. 718, 725, 102 S. Ct. 3331, 73 L. Ed. 2d 1090 (1982)).

^{12.} State Petitioners' standing for their constitutional claim under 5 U.S.C. § 706(2)(B) does not revive their statutory preemption claim under § 706(2)(A). As the Supreme Court has repeatedly held, standing "is not dispensed in gross; rather, plaintiffs must demonstrate standing for each claim that they press and for each form of relief that they seek." *TransUnion LLC v. Ramirez*, 594 U.S. 413, 431, 141 S. Ct. 2190, 210 L. Ed. 2d 568 (2021); see also, e.g., *DaimlerChrysler Corp. v. Cuno*, 547 U.S. 332, 352, 126 S. Ct. 1854, 164 L. Ed. 2d 589 (2006); *Davis v. Fed. Election Comm'n*, 554 U.S. 724, 734, 128 S. Ct. 2759, 171 L. Ed. 2d 737 (2008). We are aware of no precedent or rationale that would allow parties to bring claims over which we otherwise would lack jurisdiction—such as State Petitioners' statutory preemption claim—by appending another claim subject to a different standing analysis.

В.

Turning to the merits, we reject State Petitioners' theory. The Supreme Court has held that the Constitution contains a "fundamental principle of equal sovereignty." *Shelby County*, 570 U.S. at 544. But neither the Supreme Court nor any other court has ever applied that principle as a limit on the Commerce Clause or other Article I powers. As explained below, the rationale of *Shelby County* and the cases on which it relied in fact suggests that the principle imposes no such limit. The parties' remaining arguments confirm that conclusion. We therefore hold that Section 209(b) is subject to traditional rational basis review for Commerce Clause legislation and—as no one disputes—that it is constitutional under that standard.

Shelby County addressed the constitutionality of the Voting Rights Act's ("VRA") coverage formula, which required some but not all states to obtain approval from federal authorities before enacting voting-related laws, a process known as preclearance. Id. at 536-37. The Court did not disturb South Carolina v. Katzenbach, 383 U.S. 301, 86 S. Ct. 803, 15 L. Ed. 2d 769 (1966), which held that the coverage formula in the Voting Rights Act of 1965 was constitutional because it was "rational in both practice and theory." Shelby County, 570 U.S. at 550 (quoting Katzenbach, 383 U.S. at 330); see also id. at 550-51 (discussing the exceptional circumstances that supported *Katzenbach's* conclusion). Instead, the core question in Shelby County was whether Congress had sufficient justification for continuing to subject those states to the preclearance requirement in its 2006 reauthorization of the Voting Rights Act. *Id.* The Supreme Court relied in part on the "fundamental principle of equal sovereignty," *id.* at 544, to hold that the VRA's coverage formula was unconstitutional because it was founded on "decades-old data and eradicated practices," *id.* at 551, instead of being tailored to "current conditions," *id.* at 557. The Court did not outright reject the coverage formula for treating states differently; instead, it held that the formula's "disparate geographic coverage" was not "sufficiently related to the problem that it targets." *Id.* at 550–51 (quoting *Nw. Austin*, 557 U.S. at 203).

State Petitioners do not ask us to apply Shelby County's test that a statute's "disparate geographic coverage" must be "sufficiently related to the problem that it targets." State Pet. Reply Br. 13 ("Shelby County never suggested this test applies in other contexts."). Indeed, they forfeited any argument that the waiver here fails Shelby County's "sufficiently related" test by raising that argument for the first time in their reply brief. State Pet. Reply Br. 14–15. Instead, State Petitioners rely on Shelby County to argue that the equal sovereignty principle operates as a categorical bar on Congress's Commerce Clause authority—that is, that the principle prohibits Congress from enacting Commerce Clause legislation that leaves some states with more sovereign authority than others, regardless of Congress's reasons for doing so. State Pet. Br. 24-25, 28-29 ("Section 209(b) violates the equal-sovereignty doctrine by allowing California to exercise sovereign authority that § 209(a) takes from every other State."); State Pet. Reply Br. 10, 12.13

^{13.} State Petitioners half-heartedly suggest that "Congress arguably complies with the equal-sovereignty doctrine when it

For several reasons, *Shelby County* does not support State Petitioners' request that we apply the equal sovereignty principle as a categorical limit on Congress's authority over interstate commerce. First, the central debate in *Shelby County* was the scope of Congress's power to enforce the Fifteenth Amendment "by appropriate legislation." 570 U.S. at 536 (quoting U.S. Const. amend. XV). The Court used equal sovereignty as a background principle in applying that phrase. *Id.* at 544–45. State Petitioners confirm that textual link in their brief to us, arguing that *Shelby County* means that "in deciding whether such legislation was 'appropriate,' courts must consult the background principle of equal sovereignty." State Pet. Br. 24.

But unlike the Fifteenth Amendment, Congress's Commerce Clause power is not limited to "appropriate legislation." The Commerce Clause instead declares unconditionally that Congress has the power "to regulate commerce with foreign nations, among states, and with the Indian tribes." U.S. Const. art. I § 8, cl. 3. As the Supreme Court has explained, the Commerce Clause is "a grant of plenary authority to Congress," *Hodel v. Va.*

empowers only a single State (or a single subset of States) to regulate a matter of unique concern to that State (or that subset of States)," such as if Congress were to allow just one state to regulate a mineral that exists only in that state. State Pet. Br. 26–27. Such a law would fit with State Petitioners' categorical theory because it would not "deny sovereign authority to any State capable of exercising it." State Pet. Reply Br. 14. As the example indicates, this suggestion is not substantively different from the theory that the equal sovereignty principle imposes a categorical limit on Congress's Commerce Clause authority, and we therefore do not address it separately.

Surface Mining & Reclamation Ass'n, 452 U.S. 264, 276, 101 S. Ct. 2352, 69 L. Ed. 2d 1 (1981), and "acknowledges no limitations" other than those "prescribed in the constitution" and "expressed in plain terms," Gibbons v. Ogden, 22 U.S. 1, 196, 6 L. Ed. 23 (1824).

Second, in requiring that the VRA's coverage formula be sufficiently related to the problem it targets, Shelby County repeatedly emphasized that the VRA was "extraordinary," 570 U.S. at 536, because it intruded on states' power to regulate elections, a "sensitive area of state and local policymaking," id. at 545 (quoting Lopez v. Monterey County, 525 U.S. 266, 282, 119 S. Ct. 693, 142 L. Ed. 2d 728 (1999)), which "the Framers of the Constitution intended the States to keep for themselves," id. at 543 (quoting Gregory v. Ashcroft, 501 U.S. 452, 461, 111 S. Ct. 2395, 115 L. Ed. 2d 410 (1991)). The VRA was therefore a "drastic departure from basic principles of federalism." *Id.* at 535. Because the VRA departed from the traditional balance of state and federal power over elections, the Court required a heightened showing that subjecting specific states to the preclearance requirement was still "appropriate" considering the nation's current conditions. Id. at 555.

Section 209(b) is not "extraordinary" in that way. The Constitution places regulation of all matters affecting interstate commerce—including vehicle emissions—squarely within Congress's domain, not that of the states. See Miss. Comm'n on Env't Quality v. EPA, 790 F.3d 138, 180–83, 416 U.S. App. D.C. 69 (D.C. Cir. 2015); see also Massachusetts, 549 U.S. at 528–29. Indeed, in discussing

the Constitution's assignment to Congress of control over interstate commerce the Court has stressed that "[n]o other federal power was so universally assumed to be necessary" and "no other state power was so readily relinquished." H. P. Hood & Sons, Inc. v. Du Mond, 336 U.S. 525, 534, 69 S. Ct. 657, 93 L. Ed. 865 (1949). Accordingly, no one questions that Congress could readily preempt all states from regulating motor vehicle emissions, or that Congress itself could set different vehicle emissions standards for different regions of the country. See Sec'y of Agric. v. Cent. Roig Refin. Co., 338 U.S. 604, 616, 70 S. Ct. 403, 94 L. Ed. 381 (1950) (recognizing Congress's authority to "devise ... a national policy with due regard for the varying and fluctuating interests of different regions"); *Hodel v.* Indiana, 452 U.S. 314, 332, 101 S. Ct. 2376, 69 L. Ed. 2d 40 (1981). Shelby County does not support requiring a heightened justification for disparate intrusions into areas over which the Constitution grants Congress such comprehensive control. As the Third Circuit put it, "there is nothing in *Shelby County* to indicate that the equal sovereignty principle is meant to apply with the same force outside the context of 'sensitive areas of state and local policymaking." NCAA, 730 F.3d at 239 (quoting Shelby County, 570 U.S. at 545).

Further, State Petitioners ask us not only to venture beyond the bounds *Shelby County* set for the equal sovereignty principle but also to dramatically increase its force. Recall that *Shelby County* did not establish a categorical bar against Congress leaving states with different levels of sovereign authority even in the traditionally state-dominated context of voting; it required only that Congress show the disparate treatment is

"sufficiently related to the problem that it targets." 570 U.S. at 550–51 (quoting *Nw. Austin*, 557 U.S. at 204). Indeed, the Court reaffirmed *Katzenbach*'s holding that Congress could do so with sufficient evidence. *Id.* Yet State Petitioners ask us to hold that the equal sovereignty principle operates as a categorical bar against treating states differently in the context of Commerce Clause legislation. State Pet. Br. 28–29; State Pet. Reply Br. 10–11. Given that the Constitution grants Congress primacy over interstate commerce, that would be a highly counterintuitive conclusion.

State Petitioners also rely on a series of cases known as the equal footing cases, which Shelby County cited as applying the equal sovereignty principle. See 570 U.S. at 544. Those cases involved congressional attempts to place limits on new states as a condition of admission to the Union and identified "equal sovereignty" as an "attribute...guaranteed to" each state "upon admission." United States v. Louisiana, 363 U.S. 1, 16, 80 S. Ct. 961, 4 L. Ed. 2d 1025 (1960), supplemented sub nom. United States v. Louisiana, 382 U.S. 288, 86 S. Ct. 419, 15 L. Ed. 2d 331 (1965). For example, Coyle v. Smith, 221 U.S. 559, 31 S. Ct. 688, 55 L. Ed. 853 (1911), held that Congress had no authority to prohibit Oklahoma from moving its state capital as a condition of admission into the United States. Id. at 567-68. The Court's opinion addressed whether Congress's power to admit new states into the Union allowed such a condition, which was concededly beyond any of Congress's other enumerated powers. Id. The Court explained that Congress could not use its admission power to require a new state to give up an aspect of sovereignty that the thirteen original states retained. *Id.* To do so, the Court concluded, would create a "union of states unequal in power, as including states whose powers were restricted only by the Constitution, with others whose powers had been further restricted by an act of Congress accepted as a condition of admission." *Id.* at 567. State Petitioners here seize on that and similar language to support their argument that the equal sovereignty principle must mean Congress generally has no power to legislate in ways that leave the states with unequal sovereign authority.

The equal footing cases, however, do not directly apply either outside of the admission context or to Article I powers like the Commerce Clause. *Shelby County* itself reaffirmed prior holdings that the doctrine is not a "bar on differential treatment outside th[e] context" of states' admission into the Union. 570 U.S. at 544. *Shelby County*, of course, drew on the equal footing cases and concluded that the principle of equal sovereignty they discuss remained "highly pertinent" in the context of that case. *Id.* But for all the reasons explained above, *Shelby County* does not extend the principle even further to any (let alone all) Article I legislation.

The equal footing cases themselves also support that conclusion. Those cases contemplated—though, to be sure, only in *dicta*—that even if Congress treated states differently at the time of admission, it would not violate the equal footing guarantee so long as it acted within the scope of its plenary powers over interstate commerce. The Court suggested in *Coyle* that Congress could treat states differently if—instead of using its admission

power—it enacted "legislation intended as a regulation of commerce," because to do that would be acting "within the sphere of the plain power of Congress." 221 U.S. at 574; see also id. at 572–74; Pollard v. Hagan, 44 U.S. 212, 229–30, 11 L. Ed. 565 (1845). Such a condition would not put states on an impermissibly unequal footing because it "would not operate to restrict the state's legislative power in respect of any matter which was not plainly within the regulating power of Congress." Coyle, 221 U.S. at 574. Accordingly, the equal footing cases fit neatly with the conclusion that the equal sovereignty principle is not a categorical bar on Congress deploying its plenary power over interstate commerce in ways that differentially affect states' legislative power.

The parties also debate whether State Petitioners' theory is supported by the Constitution's text, founding era history, and law of nations principles. We address each in turn and conclude these other indicators of constitutional meaning do not support State Petitioners' theory.

The Constitution does not contain any textual provision suggesting an equal sovereignty limit on Congress's Article I powers generally or on the Commerce Clause in particular. As already discussed, the Commerce Clause is a plenary grant of authority to regulate interstate commerce which the Supreme Court has held is subject only to those limitations "prescribed in the constitution" and "expressed in plain terms." *Gibbons*, 22 U.S. at 196.

To the extent the Constitution's text sheds light on the question, it appears to cut against State Petitioners, because the Constitution does impose certain equalitybased limitations on other Article I powers. For example, the text of Article I, Section 8 states that laws concerning bankruptcy, naturalization, and duties shall be "uniform." See U.S. Const. art. I, § 8, cl. 1, cl. 4. Similarly, Article I, Section 9 prohibits "[p]reference . . . given by any Regulation of Commerce or Revenue to the Ports of one State over those of another." U.S. Const. art. I, § 9, cl. 6. State Petitioners argue that these textual provisions do not suggest the absence of a general equal sovereignty limit on Article I. As they point out, these provisions speak only to whether Congress can treat states differently when Congress itself does the legislating, not whether Congress can allow some but not other states to exercise the sovereign authority to legislate on an issue. That is, the provisions guarantee the states equal treatment for only specific subjects rather than equal sovereignty for only those subjects. The key for present purposes, however, is that even though the Founders plainly knew how to include equality-based protections for states in Article I when they wished to, they did not include any mention of State Petitioners' broad equal sovereignty principle. The fact that some constitutional clauses explicitly contain an equality-based guarantee therefore supports a negative inference—though perhaps only a mild one—that the Commerce Clause is not so constrained.

There are, of course, "constitutional doctrines that are not spelled out in the Constitution but are nevertheless implicit in its structure and supported by historical practice," such as the doctrine of state sovereign immunity. *Franchise Tax Bd. v. Hyatt*, 139 S. Ct. 1485,

1498-99, 203 L. Ed. 2d 768 (2019). That category also includes limits on the Commerce Clause, such as the Tenth Amendment anticommandeering doctrine, which are supported by the historical context in which our federal structure was created. See New York v. United States, 505 U.S. 144, 163–66, 112 S. Ct. 2408, 120 L. Ed. 2d 120 (1992) (discussing founding era debates supporting the anticommandeering doctrine). The evidence the parties provide from the founding era, however, does not show that State Petitioners' version of the equal sovereignty principle has a comparable historical pedigree. It is true, as State Petitioners urge, that the general subject of state sovereignty and the states' relation to each other and the new federal government was a core focus at the founding. Despite that focus, however, State Petitioners have identified no evidence that the Founders contemplated the type of inviolable equal state sovereignty State Petitioners ask us to announce.

The equal sovereignty debate at the founding centered on how states would be represented in Congress, with the smaller states arguing for equal representation for each state and the larger states seeking "equality for each voter" in the form of proportional representation. See Wesberry v. Sanders, 376 U.S. 1, 11-14, 84 S. Ct. 526, 11 L. Ed. 2d 481 (1964) (summarizing the Great Compromise debates); Letter from James Madison to Thomas Jefferson (Oct. 24, 1787), in 12 The Papers of Thomas Jefferson 270, 279 (Julian P. Boyd ed., 1955) (discussing how the "little States insisted on retaining their equality in both branches" while the "large states . . . urged that as the new Government was to be drawn principally from the

people immediately"). For example, New Jersey delegate William Paterson used the concept of "equal sovereignty" to support his argument for a single legislative chamber with an equal vote for each state. Wesberry, 376 U.S. at 11 (citing 3 The Records of the Federal Convention of 1787, at 251 (Max Farrand ed., 1911)). Eventually, these debates led to the Great Compromise, which established two forms of equality central to Article I: "equal sovereignty" in the Senate in the form of equal representation for each state and equal representation for each voter in the House in the form of proportional representation. U.S. Const. art. I, §§ 2, 3; see also The Federalist No. 39, at 255 (James Madison) (Jacob E. Cooke ed., 1961). The Founders' preoccupation with the manner and extent of state equality under the Constitution also appears to have yielded the specific equality-based limits on Congress's legislative authority discussed above. If the Constitution also contained State Petitioners' fundamental yet unstated limit on Congress's authority to legislate, one would expect ample historical evidence of that limit at the founding. State Petitioners point us to none. Cf. New York, 505 U.S. at 163-66.

In fact, as State Petitioners admit, the Constitution includes one provision that expressly allows Congress to enhance the sovereign authority of some states without granting that authority equally to all states. State Pet. Reply Br. 10. Article I, Section 10 provides that "No State shall, without the Consent of Congress, lay any Duty of Tonnage, keep Troops, or Ships of War in time of Peace, enter into any Agreement or Compact with another State, or with a foreign Power." U.S. Const. art. I, § 10,

cl. 3. That is, Congress has the power to grant individual states greater authority to, for example, assess taxes and even enter compacts with foreign powers—indisputably elements of "sovereignty"—without extending the same authority to other states. See Wheeling, P. & C. Transp. Co. v. City of Wheeling, 99 U.S. 273, 283, 25 L. Ed. 412 (1878) ("Taxation, beyond all doubt, is the exercise of a sovereign power. . . ."); Cuyler v. Adams, 449 U.S. 433, 440, 101 S. Ct. 703, 66 L. Ed. 2d 641 (1981) (compacts necessarily "tend[] to the increase of political power" for the states (quoting U.S. Steel Corp. v. Multistate Tax Comm'n, 434 U.S. 452, 468, 98 S. Ct. 799, 54 L. Ed. 2d 682 (1978))). Early Congresses used Article I, Section 10 in this very manner, granting specific states legislative authority to impose tonnage duties. See, e.g., Act of Feb. 9, 1791, ch. 5, 1 Stat. 190 (consenting to Maryland statute imposing duty at Port of Baltimore); Act of Aug. 11, 1790, ch. 43, 1 Stat. 184 (consenting to Georgia, Maryland, and Rhode Island statutes imposing tonnage duties). If this provision were—as State Petitioners would have it—a limited exception to an otherwise generally applicable equal sovereignty guarantee implicit in the Constitution, we would expect to see some founding era discussion of how the provision interacts with that broader equal sovereignty principle. Yet State Petitioners identify no evidence of that either.

The only affirmative support State Petitioners identify for their theory comes from law of nations principles. They argue that the Founders expected international law of nations principles to govern the states and that those principles included a notion of equal sovereignty that would render federal legislation unconstitutional if it treated states differently. State. Pet. Br. 18; State Pet. Reply Br. 12 ("At the founding, the law of nations entitled all sovereigns to perfect equality." (internal quotation marks omitted)). That argument is unpersuasive. International law contained no analog for the relation in our constitutional system between the federal government and the states, and so it would be surprising if the law of nations dictated limits on Congress's authority in relation to the states. And, as it turns out, the Supreme Court has effectively explained that the Constitution's Supremacy Clause defeats State Petitioners' reasoning. In Gibbons, the Court held that while states may have equal sovereign authority to regulate commerce in the absence of federal action, that authority is "subjected . . . to the superior power of Congress" when Congress acts. 22 U.S. at 70. Indeed, the Gibbons Court cited the very same law of nations principles that State Petitioners rely on, but only to describe the relationship between the states when Congress has *not* acted pursuant to its commerce power. *Id.* at 69–70. No case since *Gibbons* has said otherwise.

The nature and extent of equality between the states has been a central debate throughout our country's history, from the founding to the admission of new states and beyond. But State Petitioners point us to no meaningful support for their novel request to apply the equal sovereignty principle as a categorical limit on Congress's power to regulate interstate commerce. The First and Third Circuits—the only appellate courts to have considered similar arguments—have found *Shelby County*'s discussion of the equal sovereignty principle inapplicable to Commerce Clause and Spending Clause legislation for similar reasons. *See NCAA*, 730 F.3d at

238–39 (Commerce Clause); *Mayhew*, 772 F.3d at 95 (Spending Clause).

Section 209(b) is subject to the rational basis review normally applied to Commerce Clause legislation. *See Hodel*, 452 U.S. at 276–77. And because State Petitioners present no argument that Section 209(b) or the waiver at issue cannot survive that review, we reject their constitutional challenge.

State Petitioners' request to set aside the Administrator's decision on these grounds is denied.

So ordered.

APPENDIX B

ENVIRONMENTAL PROTECTION AGENCY

[EPA-HQ-OAR-2021-0257; FRL-9325-01-OAR]

California State Motor Vehicle Pollution Control Standards; Advanced Clean Car Program; Reconsideration of a Previous Withdrawal of a Waiver of Preemption; Notice of Decision

AGENCY: Environmental Protection Agency.

ACTION: Notice of decision.

SUMMARY: The Environmental Protection Agency (EPA) has completed the reconsideration of its 2019 action withdrawing a 2013 Clean Air Act (CAA) waiver of preemption for California's greenhouse gas (GHG) emission standards and zero emission vehicle (ZEV) sale mandate, which are part of California's Advanced Clean Car (ACC) program. This decision rescinds EPA's 2019 waiver withdrawal, thus bringing back into force the 2013 ACC program waiver, including a waiver of preemption for California's ZEV sales mandate and GHG emissions standards. In addition, EPA is withdrawing the interpretive view of CAA section 177 included in its 2019 action, that States may not adopt California's GHG standards pursuant to section 177 even if EPA has granted California a waiver for such standards. Accordingly, other States may continue to adopt and enforce California's GHG standards under section 177 so long as they meet the requirements of that section.

DATES: Petitions for review must be filed by May 13, 2022.

ADDRESSES: EPA has established a docket for this action under Docket ID EPA-HQ-OAR-2021-0257. All documents relied upon in making this decision, including those submitted to EPA by CARB, are contained in the public docket. Publicly available docket materials are available electronically through www.regulations.gov. After opening the www.regulations.gov website, enter EPA-HQ-OAR-2021-0257 in the "Enter Keyword or ID" fill-in box to view documents in the record. Although a part of the official docket, the public docket does not include Confidential Business Information (CBI) or other information whose disclosure is restricted by statute. EPA's Office of Transportation and Air Quality (OTAQ) maintains a web page that contains general information on its review of California waiver and authorization requests. Included on that page are links to prior waiver Federal Register notices, some of which are cited in this notice; the page can be accessed at https://www.epa.gov/stateand-localtransportation/vehicle-emissionscaliforniawaivers-and-authorizations.

FOR FURTHER INFORMATION CONTACT:

David Dickinson, Office of Transportation and Air Quality, U.S. Environmental Protection Agency, 1200 Pennsylvania Ave. NW. Telephone: (202) 343–9256. Email: Dickinson.David@epa.gov or Kayla Steinberg, Office of Transportation and Air Quality, U.S. Environmental

Protection Agency, 1200 Pennsylvania Ave. NW. Telephone: (202) 564–7658. Email: *Steinberg.Kayla@epa.gov*.

SUPPLEMENTARY INFORMATION:

Table of Contents

- I. Executive Summary
- II. Background
 - A. California's Advanced Clean Car (ACC) Program and EPA's 2013 Waiver
 - B. Prior Waivers for GHG Standards
 - C. SAFE 1 Decision
 - D. Petitions for Reconsideration
- III. Principles Governing This Review
 - A. Scope of Preemption and Waiver Criteria Under the Clean Air Act
 - B. Deference to California
 - C. Standard and Burden of Proof
- IV. EPA did not Appropriately Exercise Its Limited Authority To Reconsider the ACC Program Waiver in SAFE 1
 - A. Comments Received
 - B. Analysis: EPA Inappropriately Exercised Its Limited Authority To Reconsider
 - C. Conclusion

- V. The SAFE 1 Interpretation of Section 209(b)(1)(B) was Inappropriate and, in any Event, California met Its Requirements
 - A. Historical Practice
 - B. Notice of Reconsideration of SAFE 1 and Request for Comment
 - C. Comments Received
 - D. Analysis: California Needs the ACC Program GHG Standards and ZEV Sales Mandate to Address Compelling and Extraordinary Conditions Under Section 209(b)(1)(B)
 - 1. EPA is Withdrawing the SAFE 1 Section 209(b)(1)(B) Interpretation
 - 2. California Needs the GHG Standards and ZEV Sales Mandate Even Under the SAFE 1 Interpretation
 - a. GHG Standards and ZEV Sales Mandates Have Criteria Emission Benefits
 - b. California Needs Its Standards To Address the Impacts of Climate Change in California
 - 3. California's ZEV Sales Mandate as Motor Vehicle Control Technology Development
 - E. Conclusion
- VI. EPA Inappropriately Considered Preemption Under the Energy and Policy Conservation Act (EPCA) in Its Waiver Decision
 - A. Historical Practice and Legislative History
 - B. Notice of Reconsideration of SAFE 1 and Request for Comment

- C. Comments Received
- D. Analysis: EPA is Rescinding its SAFE 1 Actions Related to Preemption Under EPCA
- 1. NHTSA Has Since Repealed Its Findings of Preemption Made in SAFE 1
- 2. EPA Improperly Deviated From its Historical Practice of Limiting its Review to Section 209(b) Criteria
- E. Conclusion
- VII. EPA Inappropriately set Forth an Interpretive View of Section 177 in SAFE 1
 - A. SAFE 1 Interpretation
 - B. Notice of Reconsideration of SAFE 1 and Request for Comment
 - C. Comments Received
 - D. Analysis: EPA Is Rescinding SAFE 1's Interpretive Views of Section 177
 - E. Conclusion
- VIII. Other Issues
 - A. Equal Sovereignty
 - B. CARB's Deemed-to-Comply Provision
- IX. Decision
- X. Statutory and Executive Order Reviews

I. Executive Summary

CAA section 209(a) generally preempts states from adopting emission control standards for new motor vehicles. But Congress created an important exception from preemption. Under CAA section 209(b), the State of California¹ may seek a waiver of preemption, and EPA must grant it unless the Agency makes one of three statutory findings. California's waiver of preemption for its motor vehicle emissions standards allows other States to adopt and enforce identical standards pursuant to CAA section 177. Since the CAA was enacted, EPA has granted California dozens of waivers of preemption, permitting California to enforce its own motor vehicle emission standards.

Of particular relevance to this action, in 2013, EPA granted California's waiver request for the state's Advanced Clean Car (ACC) program (ACC program waiver). California's ACC program includes both a Low Emission Vehicle (LEV) program, which regulates criteria pollutants and greenhouse gas (GHG) emissions, as well as a Zero Emission Vehicle (ZEV) sales mandate. These two requirements are designed to control smog- and soot-causing pollutants and GHG emissions in a single

^{1.} The CAA section 209(b) waiver is limited "to any State which has adopted standards . . . for the control of emissions from new motor vehicles or new motor vehicle engines prior to March 30, 1966," and California is the only State that had standards in place before that date. "California" and "California Air Resources Board" (CARB) are used interchangeably in certain instances in this notice when referring to the waiver process under section 209(b).

^{2. 78} FR 2111 (January 9, 2013).

coordinated package of requirements for passenger cars, light-duty trucks, and medium-duty passenger vehicles (as well as limited requirements related to heavy-duty vehicles). Between 2013 and 2019, twelve other States adopted one or both of California's standards as their own. But in 2019, EPA partially withdrew this waiver as part of a final action entitled "The Safer Affordable Fuel-Efficient (SAFE) Vehicles Rule Part One: One National Program" (SAFE 1), marking the first time the agency withdrew a previously granted waiver. In addition, in the context of SAFE 1, EPA provided an interpretive view of CAA section 177 asserting that other states were precluded from adopting California's GHG standards.

As Administrator of the Environmental Protection Agency (EPA), I am now rescinding EPA's 2019 actions in SAFE 1 that partially withdrew the ACC program waiver for California's ACC program. I am rescinding these actions because (1) EPA's reconsideration of the waiver under the particular facts and circumstances of this case was improper; (2) EPA's reconsideration was based on a flawed interpretation of CAA section 209(b); (3) even under that flawed interpretation, EPA misapplied the facts and inappropriately withdrew the waiver; (4) EPA erred in looking beyond the statutory factors in CAA 209(b) to action taken by another agency under another statute to justify withdrawing the waiver; (5) that agency has also since withdrawn the action EPA relied on in any event; and (6) EPA inappropriately provided an interpretive view of section 177.

^{3. 84} FR 51310 (September 27, 2019).

As a result of this action, EPA's 2013 waiver for the ACC program, specifically the waiver for California's GHG emission standards and ZEV sales mandate requirements for model years (MYs) 2017 through 2025, comes back into force. I am also rescinding the interpretive view set forth in SAFE 1 that States may not adopt California's GHG standards pursuant to CAA section 177 even if EPA has granted California a section 209 waiver for such standards. Accordingly, States may now adopt and enforce California's GHG standards so long as they meet the requirements of Section 177, and EPA will evaluate any State's request to include those provisions in a SIP through a separate notice and comment process.

Section II of this action contains a detailed history of EPA's waiver adjudications leading up to this action. In summary, in 2012, CARB submitted the ACC waiver request to EPA, which included ample evidence of the criteria pollution benefits of the GHG standards and the ZEV sales mandate. As it had in all prior waiver decisions with two exceptions (including SAFE 1), in considering the request EPA relied on its "traditional" interpretation of section 209(b)(1)(B), which examines whether California

^{4.} In SAFE 1, EPA did not withdraw the entire 2013 waiver, but instead only withdrew the waiver as it related to California's GHG emission standards and the ZEV sales mandate. The waiver for the low-emission vehicle (LEV III) criteria pollutant standards in the ACC program remained in place. EPA's reconsideration of SAFE 1 and the impact on the ACC waiver therefore relates only to the GHG emission standards and the ZEV sales mandate, although "ACC program waiver" is used in this document. This action rescinds the waiver withdrawal in SAFE 1. In this decision, the Agency takes no position on any impacts this decision may have on state law matters regarding implementation.

needs a separate motor vehicle program as a whole—not specific standards—to address the state's compelling and extraordinary conditions. In 2013, EPA granted California's waiver request for its ACC program in full. In 2018, however, EPA proposed to withdraw portions of its waiver granted in 2013 based on a new interpretation of section 209(b)(1)(B) that looked at whether the specific standards (the GHG standards and ZEV sales mandate), as opposed to the program as a whole, continued to meet the second and third waiver prongs (found in sections 209(b) (1)(B) and (C)).⁵ In addition, EPA proposed to look beyond the section 209(b) criteria to consider the promulgation of a NHTSA regulation and pronouncements in SAFE 1 that declared state GHG emission standards and ZEV sales mandates preempted under EPCA. In 2019, after granting CARB a waiver for its ACC program in 2013 and after 12 states had adopted all or part of the California standards under section 177, EPA withdrew portions of the waiver for CARB's GHG emission standards and ZEV sales mandates. In SAFE 1, EPA cited changed circumstances and was based on a new interpretation of the CAA and the agency's reliance on an action by NHTSA that has now been repealed.6

^{5.} EPA's 2018 proposal was jointly issued with the National Highway Traffic Safety Administration (NHTSA). 83 FR 42986 (August 24, 2018) (the "SAFE proposal"). In addition to partially withdrawing the waiver, that proposal proposed to set less stringent greenhouse gas and CAFE standards for model years 2021-2026. NHTSA also proposed to make findings related to preemption under the Energy Policy and Conservation Act (EPCA) and its relationship to state and local GHG emission standards and ZEV sales mandates.

^{6. 84} FR 51310. In SAFE 1, NHTSA also finalized its action related to preemption under EPCA. NHTSA's action included both

On January 20, 2021, President Biden issued Executive Order 13990, directing the Federal Agencies to "immediately review" SAFE 1 and to consider action "suspending, revising, or rescinding" that action by April 2021. On April 28, 2021, EPA announced its Notice of Reconsideration, including a public hearing and an opportunity for public comment. The Agency stated its belief that there were significant issues regarding whether SAFE 1 was a valid and appropriate exercise of Agency authority, including the amount of time that had passed since EPA's ACC program waiver decision, the approach and legal interpretations used in SAFE 1, whether EPA took proper account of the environmental conditions (e.g., local climate and topography, number of motor vehicles, and local and regional air quality) in California, and the environmental consequences from the waiver withdrawal in SAFE 1. Further, EPA stated it would be addressing issues raised in the related petitions for reconsideration of EPA's SAFE 1 action. In the meantime, having reconsidered its own action, and also in response to Executive Order 13990, NHTSA repealed its conclusion that state and local laws related

regulatory text and well as pronouncements within the preamble of SAFE 1. In 2020, EPA finalized its amended and less stringent carbon dioxide standards for the 2021-2026 model years in an action titled "The Safer Affordable Fuel-Efficient (SAFE) Vehicles Rule for Model Years 2021-2026 Passenger Cars and Light Trucks" (SAFE 2). 85 FR 24174 (April 30, 2020).

^{7. &}quot;California State Motor Vehicle Pollution Control Standards; Advanced Clean Car Program; Reconsideration of a Previous Withdrawal of a Waiver of Preemption; Opportunity for Public Hearing and Public Comment." 86 FR 22421 (April 28, 2021).

to fuel economy standards, including GHG standards and ZEV sales mandates, were preempted under EPCA,⁸ and EPA revised and made more stringent the Federal GHG emission standards for light-duty vehicles for 2023 and later model years, under section 202(a).⁹

Section III of this action outlines the principles that govern waiver reconsiderations. It sets forth the statutory background and context for the CAA preemption of new motor vehicle emission standards, the criteria for granting a waiver of preemption, and the ability of other States to adopt and enforce California's new motor vehicle emission standards where a waiver has been issued if certain CAA criteria are met. In brief, CAA section 209(a) generally preempts all States or political subdivisions from adopting and enforcing any standard relating to the control of emissions from new motor vehicles or new motor vehicle engines. But section 209(b) contains an important exception that allows only California to submit a request to waive preemption for its standards. Importantly, EPA must grant the waiver unless the Administrator makes at least one of three findings: (1) That California's determination that its standards will be, in the aggregate, at least as protective of public health and welfare as applicable Federal standards, is arbitrary and capricious (the "first waiver prong," under section 209(b)(1)(A)); (2) that California does not need such State standards to meet compelling and extraordinary conditions (the "second waiver prong," under section 209(b)(1)(B)); or (3)

^{8. 86} FR 74236 (December 29, 2021).

^{9. 86} FR 74434 (December 30, 2021).

that California standards are not consistent with section 202(a), which contains EPA's authority to regulate motor vehicles (the "third waiver prong," under section 209(b)(1) (C)). In the 1977 amendments to the CAA, section 177 was added to allow other States that may be facing their own air quality concerns to adopt and enforce the California new motor vehicle emission standards for which California has been granted a waiver under section 209(b) if certain criteria are met.

Section III also provides more context to indicate that Congress intended that, when reviewing a request for a waiver, EPA treat with deference the policy judgments on which California's vehicle emission standards are based. It discusses the history of Congress allowing states to adopt more stringent standards. Ultimately, Congress built a structure in section 209(b) that grants California authority to address its air quality problems, and also acknowledges the needs of other states to address their air quality problems through section 177. Lastly, Section III describes the burden and standard of proof for waiver decisions.

Section IV of this action then discusses EPA's first basis for rescinding the SAFE 1 waiver withdrawal: That EPA did not appropriately exercise its limited authority to withdraw a waiver once granted. Section 209 does not provide EPA with express authority to reconsider and withdraw a waiver previously granted to California. EPA's authority thus stems from its inherent reconsideration authority. In the context of reconsidering a waiver grant, that authority may only be exercised sparingly. EPA

believes its inherent authority to reconsider a waiver decision is constrained by the three waiver criteria that must be considered before granting or denying a waiver request under section 209(b). EPA's reconsideration may not be broader than the limits Congress placed on its ability to deny a waiver in the first place. EPA notes further support for limiting its exercise of reconsideration authority, relevant in the context of a waiver withdrawal, is evidenced by Congress's creation of a state and federal regulatory framework to drive motor vehicle emissions reduction and technology innovation that depends for its success on the stable market signal of the waiver grant—automobile manufacturers must be able to depend reliably on the continuing validity of the waiver grant in order to justify the necessary investments in cleaner vehicle technology. Accordingly, EPA now believes it may only reconsider a previously granted waiver to address a clerical or factual error or mistake, or where information shows that factual circumstances or conditions related to the waiver criteria evaluated when the waiver was granted have changed so significantly that the propriety of the waiver grant is called into doubt. Even then, as with other adjudicatory actions, when choosing to undertake such a reconsideration EPA believes it should exercise its limited authority within a reasonable timeframe and be mindful of reliance interests. EPA expects such occurrences will be rare. The Agency's waiver withdrawal in SAFE 1 was not an appropriate exercise of EPA's limited authority; there was no clerical error or factual error in the ACC program waiver, and SAFE 1 did not point to any factual circumstances or conditions related to the three waiver prongs that have changed so significantly that the

propriety of the waiver grant is called into doubt. Rather, the 2019 waiver withdrawal was based on a change in EPA's statutory interpretation, an incomplete assessment of the record, and another agency's action beyond the confines of section 209(b). EPA erred in reconsidering a previously granted waiver on these bases. Accordingly, EPA is rescinding its 2019 withdrawal of its 2013 ACC program waiver.

Sections V and VI further explain why, even if SAFE 1 were an appropriate exercise of EPA's limited authority to reconsider its previously-granted waiver, the Agency would still now rescind its waiver withdrawal.

As discussed in Section V, the Agency's reinterpretation of the second waiver prong in SAFE 1 was flawed. While EPA has traditionally interpreted the second waiver prong, section 209(b)(1)(B), to require a waiver unless the Agency demonstrates that California does not need its own motor vehicle emissions program, to meet compelling and extraordinary conditions, the SAFE 1 waiver withdrawal decision was based on a statutory interpretation that calls for an examination of the need for the specific standard at issue. Section V explains why EPA believes that its traditional interpretation is, at least, the better interpretation of the second waiver prong because it is most consistent with the statutory language and supported by the legislative history. Accordingly, we reaffirm the traditional interpretation—in which EPA reviews the need for California's motor vehicle program in this action.

Additionally, Section V explains why even if the focus is on the specific standards, when looking at the record before it, EPA erred in SAFE 1 in concluding that California does not have a compelling need for the specific standards at issue—the GHG emission standards and ZEV sales mandate. In particular, in SAFE 1, the Agency failed to take proper account of the nature and magnitude of California's serious air quality problems, including the interrelationship between criteria and GHG pollution.¹⁰ Section V further discusses EPA's improper substitution in SAFE 1 of its own policy preferences for California's, and discusses the importance of deferring to California's judgment on "ambiguous and controversial matters of public policy" that relate to the health and welfare of its citizens. 11 Based on a complete review of the record in this action, EPA now believes that, even under the SAFE 1 interpretation, California needs the ZEV sales mandate and GHG standards at issue to address compelling and extraordinary air quality conditions in

^{10.} As explained herein, the requirements in the ACC program were designed to work together in terms of the technologies that would be used to both lower criteria emissions and GHG emissions. The standards, including the ZEV sales mandate and the GHG emission standards, were designed to address the short- and long-term air quality goals in California in terms of the criteria emission reductions (including upstream reductions) along GHG emission reductions. The air quality issues and pollutants addressed in the ACC program are interconnected in terms of the impacts of climate change on such local air quality concerns such as ozone exacerbation and climate effects on wildfires that affect local air quality.

^{11. 40} FR 23102, 23104 (May 28, 1975); 58 FR 4166 (January 13, 1993).

the state. EPA's findings in SAFE 1, which were based on the Agency's inaccurate belief that these standards were either not intended to or did not result in criteria emission reductions to address California's National Ambient Air Quality Standard (NAAQS) obligations, are withdrawn.

Section VI discusses SAFE 1's other basis for withdrawing the ACC program waiver, EPCA. In SAFE 1, EPA reached beyond the waiver criteria in section 209(b)(1) and considered NHTSA's regulations in SAFE 1 that state or local regulation of carbon dioxide emission from new motor vehicles (including California's ZEV sales mandate and GHG standards) are related to fuel economy and as such are preempted under EPCA. NHTSA has since issued a final rule that repeals all regulatory text and additional pronouncements regarding preemption under EPCA set forth in SAFE 1.12 This action by NHTSA effectively removes the underpinning and any possible reasoned basis for EPA's withdrawal decision based on preemption under EPCA in SAFE 1. Additionally, the Agency has historically refrained from consideration of factors beyond the scope of the waiver criteria in section 209(b)(1) and the 2013 ACC program waiver decision was undertaken consistent with this practice. EPA believes that the consideration of EPCA preemption in SAFE 1 led the Agency to improperly withdraw the ACC program waiver on this non-CAA basis. EPA's explanation that withdrawal on this basis was justified because SAFE 1 was a joint action, and its announcement that this would be a single occurrence, does not justify the ACC waiver

^{12 86} FR 74236.

withdrawal. Thus, EPA is rescinding the withdrawal of those aspects of the ACC program waiver that were based on NHTSA's actions in SAFE 1.

Section VII addresses SAFE 1's interpretive view of section 177 that States adopting California's new motor vehicle emission standards could not adopt California's GHG standards. EPA believes it was both unnecessary and inappropriate in a waiver proceeding to provide an interpretive view of the authority of states to adopt California standards when section 177 does not assign EPA any approval role in states' adoption of the standards. Therefore, as more fully explained in Section VII, the Agency is rescinding the interpretive view on section 177 set out in SAFE 1. Section VIII discusses certain other considerations, including the equal sovereignty doctrine and California's deemed-to-comply provision, and concludes that they do not disturb EPA's decision to rescind the 2019 waiver withdrawal action.

Section IX contains the final decision to rescind the withdrawal of the 2013 ACC program waiver. In summary, I find that although EPA has inherent authority to reconsider its prior waiver decisions, that authority to reconsider is limited and may be exercised only when EPA has made a clerical or factual error or mistake, or where information shows that factual circumstances or conditions related to the waiver criteria evaluated when the waiver was granted have changed so significantly that the propriety of the waiver grant is called into doubt. Further,

^{13. 84} FR at 51310, 51350.

EPA's reconsideration may not be broader than the limits Congress placed on its ability to deny a waiver in the first place. Even where those conditions are met, I believe that any waiver withdrawal decision should consider other factors such as the length of time since the initial decision and California and others' reliance on the initial decision. Because there were no factual or clerical errors or such significantly changed factual circumstances or conditions necessary to trigger EPA's authority to reconsider its previously granted waiver during the SAFE 1 proceeding, I believe SAFE 1 was not an appropriate exercise of EPA's authority to reconsider. In addition, even if it were an appropriate exercise, EPA should not have departed from its traditional interpretation of the second waiver prong (section 209(b)(1)(B)), which is properly focused on California's need for a separate motor vehicle emission program—not specific standards—to meet compelling and extraordinary conditions. And even under EPA's SAFE 1 interpretation of the second waiver prong, a complete review of the factual record demonstrates that California does need the GHG emission standards and ZEV sales mandate to meet compelling and extraordinary conditions in the State. Therefore, EPA should not have withdrawn the ACC program waiver based upon the second waiver prong in SAFE 1 and recission of the withdrawal is warranted. Additionally, I find that EPA inappropriately relied on NHTSA's finding of preemption, now withdrawn, to support its waiver withdrawal, and rescind the waiver withdrawal on that basis as well. Finally, independently in this action, I am rescinding the interpretive views of section 177 that were set forth in SAFE 1, because it was inappropriate to include those views as part of this waiver proceeding.

For these reasons, I am rescinding EPA's part of SAFE 1 related to the CAA preemption of California's standards. This recission has the effect of bringing the ACC program waiver back into force.

II. Background

This section provides background information needed to understand EPA's decision process in SAFE 1, and this decision. This context includes: A summary of California's ACC program including the record on the criteria pollutant benefits of its ZEV sales mandate and GHG emission standards; a review of the prior GHG emission standards waivers in order to explain EPA's historical evaluation of the second waiver prong; an overview of the SAFE 1 decision; a review of the petitions for reconsideration filed subsequent to SAFE 1; and a description of the bases and scope of EPA's reconsideration of SAFE 1. EPA's sole purpose in soliciting public comment on its reconsideration was to determine whether SAFE 1 was a valid and appropriate exercise of the Agency's authority. In the Notice of Reconsideration, EPA therefore noted that reconsideration was limited to SAFE 1 and that the Agency was not reopening the ACC program waiver decision.

A. California's Advanced Clean Car (ACC) Program and EPA's 2013 Waiver

On June 27, 2012, CARB notified EPA of its adoption of the ACC program regulatory package that contained amendments to its LEV III and ZEV sales mandate, and requested a waiver of preemption under section 209(b) to

enforce regulations pertaining to this program.¹⁴ The ACC program combined the control of smog- and soot-causing pollutants and GHG emissions into a single coordinated package of requirements for passenger cars, light-duty trucks, and medium-duty passenger vehicles (as well as limited requirements related to heavy-duty vehicles for certain model years).¹⁵

In its 2012 waiver request, CARB noted that the 2012 ZEV amendments would also result in additional criteria pollutant benefits in California in comparison to the

^{14. 2012} Waiver Request, EPA-HQ-OAR-2012-0562-0004 (2012 Waiver Request) at 1, 3-6. CARB's LEV III standards include both its criteria emission standards and its GHG emission standards. SAFE 1 did not address the LEV III criteria emission standards and as such the ACC program waiver remained in place. SAFE 1 did address CARB's GHG emission standards and ZEV sales mandate and this action addresses these two standards as well. As noted in CARB's 2012 Waiver Request, these three standards are interrelated and comprehensive in order to address the State's serious air quality problems including its criteria pollutants and climate change challenges.

^{15.} As noted in CARB's waiver request, "[a]t the December 2009 hearing, the Board adopted Resolution 09-66, reaffirming its commitment to meeting California's long term air quality and climate change reduction goals through commercialization of ZEV technologies. The Board further directed staff to consider shifting the focus of the ZEV regulation to both GHG and criteria pollutant emission reductions, commercializing ZEVs and PHEVs in order to meet the 2050 goals, and to take into consideration the new LEV fleet standards and propose revisions to the ZEV regulation accordingly." 2012 Waiver Request at 2 (emphasis added). EPA stated in SAFE 1 that California's ZEV standard initially targeted only criteria pollutants. 84 FR at 51329. See also 78 FR at 2118.

earlier ZEV regulations and would likely provide benefits beyond those achieved by complying with the LEV III criteria pollutant standard for conventional vehicles only. CARB attributed these benefits not to vehicle emissions reductions specifically, but to increased electricity and hydrogen use that would be more than offset by decreased gasoline production and refinery emissions. 16 CARB's waiver request attributed the criteria emissions benefits to its LEV III criteria pollutant fleet standard and did not include similar benefits from its ZEV sales mandate. According to the request, the fleet would become cleaner regardless of the ZEV sales mandate because the ZEV sales mandate is a way to comply with the LEV III standards and, regardless of the ZEV sales mandate, manufacturers might adjust their compliance response to the standard by making less polluting conventional vehicles. CARB further explained that because upstream criteria and PM emissions are not captured in the LEV III criteria pollutant standard, net upstream emissions are reduced through the increased use of electricity and concomitant reductions in fuel production.¹⁷

On August 31, 2012, EPA issued a notice of opportunity for public hearing and written comment on CARB's request and solicited comment on all aspects of a full waiver analysis for such request under the criteria of section 209(b). Commenters opposing the waiver

^{16. 2012} Waiver Request at 6.

^{17.} *Id.* at 15-16.

^{18. 77} FR 53119 (August 31, 2012).

asked EPA to deny the waiver under the second waiver prong, section 209(b)(1)(B), as it applied to the GHG provisions in the ACC Program, calling on EPA to adopt an alternative interpretation of that provision focusing on California's need for the specific standards. Following public notice and comment and based on its traditional interpretation of section 209(b), on January 9, 2013, EPA granted California's request for a waiver of preemption to enforce the ACC program regulations. The traditional interpretation, which EPA stated is the better interpretation of section 209(b)(1)(B), calls for evaluating California's need for a separate motor vehicle emission program to meet compelling and extraordinary conditions. As explained, EPA must grant a waiver to

^{19.} Set forth in the ACC program waiver decision is a summary discussion of EPA's earlier decision to depart from its traditional interpretation of section 209(b)(1)(B) (the second waiver prong) in the 2008 waiver denial for CARB's initial GHG standards for certain earlier model years along with EPA's return to the traditional interpretation of the second prong in the waiver issued in 2009. 78 FR at 2125-31. These interpretations are discussed more fully in Section III.

^{20.} *Id.* at 2128 ("The better interpretation of the text and legislative history of this provision is that Congress did not intend this criterion to limit California's discretion to a certain category of air pollution problems, to the exclusion of others. In this context it is important to note that air pollution problems, including local or regional air pollution problems, do not occur in isolation. Ozone and PM air pollution, traditionally seen as local or regional air pollution problems, occur in a context that to some extent can involve long range transport of this air pollution or its precursors. This long range or global aspect of ozone and PM can have an impact on local or regional levels, as part of the background in which the local or regional air pollution problem occurs.").

California unless the Administrator makes at least one of the three statutorily-prescribed findings in section 209(b) (1). Concluding that opponents of the waiver did not meet their burden of proof to demonstrate that California does not have such need, EPA found that it could not deny the waiver under the second waiver prong.²¹

Without adopting the alternative interpretation, EPA noted that, to the extent that it was appropriate to examine the need for CARB's specific GHG standards to meet compelling and extraordinary conditions, EPA had explained at length in its earlier 2009 GHG waiver decision that California does have compelling and extraordinary conditions directly related to regulation of GHGs. This conclusion was supported by additional evidence submitted by CARB in the ACC program waiver proceeding, including reports that demonstrate record-setting wildfires, deadly heat waves, destructive storm surges, and loss of winter snowpack. Many of these extreme weather events and other conditions have the potential to dramatically affect human health and well-

^{21.} Because EPA received comment on this issue during the ACC program waiver proceeding, as it pertained to both CARB's GHG emission standards and ZEV sales mandate, the Agency recounted the interpretive history associated with standards for both GHG emissions and criteria air pollutants to explain EPA's belief that section 209(b)(1)(B) should be interpreted the same way for all air pollutants. *Id.* at 2125-31 ("As discussed above, EPA believes that the better interpretation of the section 209(b)(1)(B) criterion is the traditional approach of evaluating California's need for a separate motor vehicle emission program to meet compelling and extraordinary conditions. Applying this approach with the reasoning noted above, with due deference to California, I cannot deny the waiver.").

being.²² Similarly, to the extent that it was appropriate to examine the need for CARB's ZEV sales mandate, EPA noted that the ZEV sales mandate in the ACC program enables California to meet both its air quality and climate goals into the future. EPA recognized that CARB's coordinated strategies reflected in the ACC program for addressing both criteria pollutants and GHGs and the magnitude of the technology and energy transformation needed to meet such goals.²³ Therefore, EPA determined that, to the extent the second waiver prong should be interpreted to mean a need for the specific standards at issue, CARB's GHG emission standards and ZEV sales mandate satisfy such a finding.

In the context of assessing the need for the specific ZEV sales mandate in the ACC program waiver, EPA noted CARB's intent in the redesign of the ZEV regulation

^{22.} *Id.* at 2126-29. Within the 2009 GHG waiver, and again in the 2013 ACC program waiver, EPA explained that the traditional approach does not make section 209(b)(1)(B) a nullity, as EPA must still determine whether California does not need its motor vehicle program to meet compelling and extraordinary conditions as discussed in the legislative history. Conditions in California may one day improve such that it may no longer have a need for its motor vehicle program.

^{23.} *Id.* at 2131 ("Whether or not the ZEV standards achieve additional reductions by themselves above and beyond the LEV III GHG and criteria pollutant standards, the LEV III program overall does achieve such reductions, and EPA defers to California's policy choice of the appropriate technology path to pursue to achieve these emissions reductions. The ZEV standards are a reasonable pathway to reach the LEV III goals, in the context of California's longer-term goals.").

of addressing both criteria pollutants and GHG emissions, and CARB's demonstration of "the magnitude of the technology and energy transformation needed from the transportation sector and associated energy production to meet... the goals set forth by California's climate change requirements" and found that the ZEV standards would help California achieve those "long term emission benefits as well as . . . some [short-term] reduction in criteria pollutant emissions."²⁴

B. Prior Waivers for GHG Standards

For over fifty years, EPA has evaluated California's requests for waivers of preemption under section 209(b), primarily considering CARB's motor vehicle emission program for criteria pollutants.²⁵ More recently, the Agency has worked to determine how section 209(b)(1)(B) should be interpreted and applied to GHG standards, including

^{24.} *Id.* at 2130-31. *See also* 2012 Waiver Request at 15-16); CARB Supplemental Comments, EPA-HQ-OAR-2012-0562-0373 at 4 (submitted November 14, 2012).

^{25.} EPA notes that the 1990 amendments to the CAA added subsection (e) to section 209. Subsection (e) addresses the preemption of State or political subdivision regulation of emissions from nonroad engines or vehicles. Section 209(e)(2)(A) sets forth language similar to section 209(b) in terms of the criteria associated with EPA waiving preemption, in this instance for California nonroad vehicle and engine emission standards. Congress directed EPA to implement subsection (e). See 40 CFR part 1074. EPA review of CARB requests submitted under section 209(e)(2)(A)(ii) includes consideration of whether CARB needs its nonroad vehicle and engine program to meet compelling and extraordinary conditions. See 78 FR 58090 (September 20, 2013).

consideration of the relationship of GHG standards to California's historical air quality problems, the public health impacts of GHG emissions on NAAQS pollutants, and the direct impacts of GHG emissions and climate change on California and its inhabitants. While the SAFE 1 withdrawal and revocation of the waiver for CARB's ACC program represents a singular snapshot of this task, it is important to examine EPA's long-standing and consistent waiver practice in general, including EPA's interpretations in prior waiver decisions pertaining to CARB's GHG emission standards, in order to determine whether EPA properly applied the waiver criterion in section 209(b)(1)(B) in SAFE 1.²⁶

Historically, EPA has consistently interpreted and applied the second waiver prong by considering whether California needed a separate motor vehicle emission program as compared to the specific standards at issue to meet compelling and extraordinary conditions. ²⁷ At the same time, in response to commenters that have argued that EPA is required to examine the specific standards at issue in the waiver request, EPA's practice has been to nevertheless review the specific standards to determine whether California needs those individual standards to meet compelling and extraordinary conditions. ²⁸ This

^{26.} EPA notes that, in the history of EPA waiver decisions, it has only denied a waiver once (in 2008) and withdrawn a waiver once (in 2019). Each instance was under this second waiver prong in section 209(b)(1)(B).

^{27. 49} FR 18887, 18890 (May 3, 1984).

^{28.} For example, in EPA's 2009 GHG waiver that reconsidered the 2008 GHG waiver denial, the Agency noted that "Given the

does not mean that EPA has adopted an "alternative approach" and required a demonstration for the need for specific standards; rather, this additional Agency review has been afforded to address commenters' concerns and this secondary analysis has been done to support the Agency's primary assessment. For example, EPA granted an authorization for CARB's In-use Off-road Diesel Standards (Fleet Requirements) that included an analysis under both approaches.²⁹ The only two departures from this traditional approach occurred first in 2008 when EPA

comments submitted, however, EPA has also considered an alternative interpretation, which would evaluate whether the program or standards has a rational relationship to contributing to amelioration of the air pollution problems in California. Even under this approach, EPA's inquiry would end there. California's policy judgment that an incremental, directional improvement will occur and is worth pursuing is entitled, in EPA's judgment, to great deference. EPA's consistent view is that it should give deference to California's policy judgments, as it has in past waiver decisions, on California's choice of mechanism used to address air pollution problems. EPA does not second-guess the wisdom or efficacy of California's standards. EPA has also considered this approach with respect to the specific GHG standards themselves, as well as California's motor vehicle emissions program." 74 FR at 32766 (citing to Motor & Equip. Mfrs. Ass'n, Inc. v. EPA, 627 F.2d 1095, 1110-11 (D.C. Cir. 1979)).

29. 78 FR at 58090. The United States Court of Appeals for the Ninth Circuit reviewed EPA's grant of a waiver of preemption under the traditional approach, and because of comments seeking an alternative interpretation, an assessment of the need for the standards contained in California's request. *Dalton Trucking v. EPA*, No. 13-74019 (9th Cir. 2021) (finding that EPA was not arbitrary in granting the waiver of preemption under either approach). The court opinion noted that "[t]his disposition is not appropriate for publication and is not precedent except as provided by Ninth Circuit Rule 36-3."

adopted an "alternative approach" to the second waiver prong and second in 2019 when EPA adopted the "SAFE 1 interpretation" of the second waiver criterion.

EPA's task of interpreting and applying section 209(b) (1)(B) to California's GHG standards and consideration of the State's historical air quality problems that now include the public health and welfare challenge of climate change began in 2005, with CARB's waiver request for 2009 and subsequent model years' GHG emission standards. On March 6, 2008, EPA denied the waiver request based on a new interpretive finding that section 209(b) was intended for California to enforce new motor vehicle emission standards that address local or regional air pollution problems, and an Agency belief that California could not demonstrate a "need" under section 209(b) (1)(B) for standards intended to address global climate change problems. EPA also employed this new alternative interpretation to state a belief that the effects of climate change in California are not compelling and extraordinary in comparison with the rest of the country. Therefore, in the 2008 waiver denial, EPA did not evaluate whether California had a need for its motor vehicle emission program to meet compelling and extraordinary conditions (the traditional interpretation) but rather focused on the specific GHG emission standard in isolation and not in conjunction with the other motor vehicle emission standards for criteria pollutants.

In 2009, EPA initiated a reconsideration of the 2008 waiver denial. The reconsideration resulted in granting CARB a waiver for its GHG emission standards

commencing in the 2009 model year.³⁰ In granting the waiver, EPA rejected the Agency's alternative interpretation of the second waiver prong announced in the 2008 waiver denial. Instead, EPA returned to its traditional approach of evaluating California's need for a separate motor vehicle emission program to meet compelling and extraordinary conditions because the Agency viewed it as the better interpretation of the second waiver prong. Under the traditional interpretation, EPA found that the opponents of the waiver had not met their burden of proof to demonstrate that California did not need its motor vehicle emission program to meet compelling and extraordinary conditions. In responding to comments on this issue, EPA also determined that, even if the alternative interpretation were to be applied, the opponents of the waiver had not demonstrated that California did not need its GHG emissions standards to meet compelling and extraordinary conditions.³¹

Since EPA's 2009 GHG waiver decision and before SAFE 1 the Agency applied the traditional interpretation of the second waiver prong in its GHG-related waiver

^{30. 74} FR 32743, 32745 (July 8, 2009).

^{31. 74} FR at 32759-67. For example, EPA noted that the analysis of the need for CARB's GHG standards in the 2008 waiver denial failed to consider that although the factors that cause ozone are primarily local in nature and that ozone is a local or regional air pollution problem, the impacts of global climate change can nevertheless exacerbate this local air pollution problem. EPA noted that California had made a case that its greenhouse gas standards are linked to amelioration of its smog problems. See also 76 FR 34693 (June 14, 2011).

proceedings, including the on-going review of California's GHG emission standards for vehicles. In the first instance, in 2009, CARB adopted amendments to its certification requirements that would accept demonstration to the Federal GHG standards as compliance with CARB's GHG program. This provision is known as a "deemed-to-comply" provision.³² In 2011, EPA determined that this deemed-to-comply provision was within-the-scope of the waiver issued in July 2009, relying on the traditional interpretation of the second waiver prong.³³ As such,

^{32.} California Code of Regulations, Title 13 1961(a)(1)(B). Under this provision, automakers could comply with the California GHG standards for model years 2017-2025 by meeting Federal GHG standards for the same model years.

^{33. 76} FR 34693. EPA's "within-the-scope" decisions are generally performed when CARB has amended its regulations that were previously waived by EPA under section 209(b)(1) and include an analysis of whether EPA's prior evaluation of the waiver criteria has been undermined by CARB's amendments. EPA received comment during the reconsideration of SAFE 1 that questioned whether CARB needed its GHG standards if it was otherwise accepting compliance with the Federal GHG standards. EPA addressed the issue in its final decision (76 FR at 34696-98) and continues to believe EPA's analysis applies. The existence of federal emission standards that CARB may choose to harmonize with or deem as compliance with its own State standards (or that CARB may choose to set more stringent standards) does not on its own render California's as not needed. CARB continues to administer an integrated and comprehensive motor vehicle emission program (including its ZEV sales mandate and GHG emission standards and other applicable emission standards for light-duty vehicles) and this program continues to evolve to address California's serious air quality issues. CARB's decision to select some federal emission standards as

in the June 14, 2011 within-the-scope decision EPA determined that CARB's 2009 amendments did not affect or undermine the Agency's prior determination made in the 2009 GHG waiver decision, including the technological feasibility findings in section 209(b)(1)(C).³⁴ EPA also acted on two requests for waivers of preemption for CARB's heavy-duty (HD) tractor-trailer GHG emission standards.³⁵ Once again, EPA relied upon its traditional approach of evaluating California's need for a separate motor vehicle emission program to meet compelling and extraordinary conditions and found that no evidence had been submitted to demonstrate that California no

sufficient to comply with its own State emission standards does not negate the overall design and purpose of section 209 of the CAA. In the within-the-scope decision issued in 2011, EPA agreed with Global Automakers comment that the deemed-to-comply provision renders emission benefits equally protective as between California and Federal programs. *Id.* at 34696.

34. Id. at 34696-97.

35. The first HD GHG emissions standard waiver related to certain new 2011 and subsequent model year tractor-trailers. 79 FR 46256 (August 7, 2014). In this waiver decision EPA responded to comments regarding whether CARB had quantified how the GHG regulations would contribute to attainment of ozone or particulate matter standards by noting that nothing in section 209(b)(1)(B) calls for California to quantify specifically how its regulations would affect attainment of the NAAQS in the State. Rather, EPA noted, the relevant question is whether California needs its own motor vehicle emission program and not whether there is a need for specific standards. The second HD GHG emissions standard waiver related to CARB's "Phase I" regulation for 2014 and subsequent model year tractor-trailers. 81 FR 95982 (December 29, 2016).

longer needed its motor vehicle emission program to meet compelling and extraordinary conditions. ³⁶ EPA's second waiver for the HD GHG emission standards made a similar finding that California's compelling and extraordinary conditions continue to exist under the traditional approach for the interpretation of the second waiver criterion. ³⁷

^{36.} Relatedly, California explained the need for these standards based on projected "reductions in NO X emissions of 3.1 tons per day in 2014 and one ton per day in 2020 due to the HD GHG Regulations. California state[d] that these emissions reductions will help California in its efforts to attain applicable air quality standards. California further projects that the HD GHG Regulations will reduce GHG emissions in California by approximately 0.7 million metric tons (MMT) of carbon dioxide equivalent emissions (CO $_2$ e) by 2020." 79 FR at 46261. See also 81 FR at 95982.

^{37. 81} FR at 95987. At the time of CARB's Board adoption of the HD Phase I GHG regulation, CARB determined in Resolution 13-50 that California continues to need its own motor vehicle program to meet serious ongoing air pollution problems. CARB asserted that "[t] he geographical and climatic conditions and the tremendous growth in vehicle population and use that moved Congress to authorize California to establish vehicle standards in 1967 still exist today. EPA has long confirmed CARB's judgment, on behalf of the State of California, on this matter." See EPA Air Docket at regulations. gov at EPA-HQ-OAR-2016-0179-0012. In enacting the California Global Warming Solutions Act of 2006, the Legislature found and declared that "Global warming poses a serious threat to the economic well-being, public health, natural resources, and the environment of California. The potential adverse impacts of global warming include the exacerbation of air quality problems, a reduction in the quality and supply of water to the state from the Sierra snowpack, a rise in sea levels resulting in the displacement of thousands of coastal businesses and residences, damage to the marine ecosystems and the natural environment, and an increase in the incidences of infectious diseases, asthma, and other health-related problems."

C. SAFE 1 Decision

In 2018, NHTSA issued a proposal for new Corporate Average Fuel Economy (CAFE) standards that must be achieved by each manufacturer for its car and light-duty truck fleet while EPA revisited its light-duty vehicle GHG emissions standards for certain model years in the SAFE Proposal.³⁸ EPA also proposed to withdraw the waiver for the ACC program GHG emission standards and ZEV sales mandate, referencing both sections 209(b)(1)(B) and (C). EPA posited that since the grant of the initial waiver a reassessment of California's need for its GHG standards and ZEV sales mandate under the second waiver prong, section 209(b)(1)(B), was appropriate. EPA further posited that its own Federal GHG rulemaking in the SAFE proposal raised questions about the feasibility of CARB's standards under the third waiver prong, section 209(b)(1)(C).³⁹ In addition, EPA reasoned that the SAFE proposal presented a unique situation that required EPA to consider the implications of NHTSA's proposed conclusion that California's GHG emission standards and ZEV sales mandate were preempted by EPCA.⁴⁰ EPA

^{38.} The Safer Affordable Fuel-Efficient (SAFE) Vehicles Rule for Model Years 2021-2026 Passenger Cars and Light Trucks, 83 FR at 42986.

^{39.} As explained below, EPA did not make a determination regarding section 209(b)(1)(C) in SAFE 1.

^{40. &}quot;To the extent that NHTSA has determined that these standards are void *ab initio* because EPCA preempts standards that relate to fuel economy, that determination presents an independent basis for EPA to consider the validity of the initial grant of a waiver

thus also posited that state standards preempted under EPCA cannot be afforded a valid section 209(b) waiver and then proposed that it would be necessary to withdraw the waiver separate and apart from section 209(b)(1)(B) and (C) if NHTSA finalized its interpretation regarding preemption under EPCA.

During the SAFE 1 proceeding, EPA received additional information demonstrating that the ZEV sales mandate plays a role in reducing criteria pollution, including CARB's comments that EPA's prior findings in the ACC program waiver were correct. As noted by a number of States and Cities, "[f]or example, CARB modeled the consequences of the actions proposed in SAFE, which included withdrawing California's waiver for its GHG and ZEV standards and freezing the federal GHG standards at MY 2020 levels. CARB concluded these actions, which would eliminate California's ZEV and GHG standards and leave in place only federal GHG standards at MY 2020 levels, would increase NOx emissions in the South Coast air basin alone by 1.24 tons per day."41 The SAFE 1 record also includes information that demonstrates that California is "one of the most climate challenged" regions of North America, and that it is home to some of the country's hottest and driest areas,

for these standards, separate and apart from EPA's analysis under the criteria that invalidate a waiver request." 84 FR at 51338.

^{41.} States and Cities in Support of EPA Reversing Its SAFE 1 Actions (States and Cities), Docket No. EPA-HQ-OAR-2021-0257-0132 at 10 (citing CARB, Docket No. NHTSA-2018-0067-11873 at 287-88, 290-91 (upstream emission impacts), 308).

which are particularly threatened by record-breaking heatwaves, sustained droughts, and wildfire, as a result of GHG emissions. ⁴² This record also includes information from the United States Fourth National Climate Assessment that documents the impact of climate change in exacerbating California's record-breaking fires seasons, multi-year drought, heat waves, and flood risk, and notes that California faces a particular threat from sea-level rise and ocean acidification and that the State has "the most valuable ocean-based economy in the country." ⁴³ EPA received information during the SAFE 1 public comment period regarding the criteria emission benefits of CARB's ZEV sales mandate and GHG emission standards. ⁴⁴

^{42.} States and Cities at 43-47 (citing EPA-HQ-OAR-2018-0283-5481, EPA-HQ-OAR-2018-0283-5683, and EPA-HQ-OAR-2018-0283-5054).

^{43.} *Id.* at 45 (EPA-HQ-OAR-2018-0283-7447—U.S. Global Research Program, *Impacts, Risks, and Adaptation in the United States: Fourth National Climate Assessment, Volume II*, Chapter 25., 2018). (*E.g.*, "The California coast extends 3,400 miles (5,500 km), 8 with 200,000 people living 3 feet (0.9 m) or less above sea level.9 The seaports of Long Beach and Oakland, several international airports, many homes, and high-value infrastructure lie along the coast. In addition, much of the Sacramento-San Joaquin River Delta is near sea level. California has the most valuable ocean-based economy in the country, employing over half a million people and generating \$20 billion in wages and \$42 billion in economic production in 2014.10 Coastal wetlands buffer against storms, protect water quality, provide habitat for plants and wildlife, and supply nutrients to fisheries. Sea level rise, storm surges, ocean warming, and ocean acidification are altering the coastal shoreline and ecosystems."

^{44.} During the current reconsideration proceeding, EPA received additional comment regarding the criteria pollution

On September 27, 2019, EPA and NHTSA published the final SAFE 1 action that promulgated preemption regulations which supported NHTSA's conclusion that EPCA preempted California's GHG standards and ZEV sales mandate. In the same action, EPA withdrew the waiver of preemption for California to enforce the ACC program GHG and ZEV sales mandate on two grounds. 45

First, in SAFE 1 the Agency posited that standards preempted under EPCA could not be afforded a valid waiver of preemption under section 209(b). EPA explained that Agency pronouncements in the ACC program waiver decision on the historical practice of disregarding the preemptive effect of EPCA in the context of evaluating California's waiver applications were "inappropriately broad, to the extent it suggested that EPA is categorically

benefits of California's GHG and ZEV standards. The States and Cities at 10-11. Likewise, CARB notes this connection in comments on the SAFE proposal. Multi-State SAFE Comments, EPA-HQ-OAR-2018-0283-5481 at 24. The States and Cities provided supplemental information in response to the Notice of Reconsideration by submitting California's latest analyses of the criteria pollutant benefits of its GHG standards. For example, CARB estimated those benefits for calendar years by which the South Coast air basin must meet increasingly stringent NAAQS for ozone: 2023, 2031, and 2037. States and Cities app. A at 2-4, app. C at 8-9.

^{45. 84} FR at 51328-29. Parties subsequently brought litigation against EPA on its SAFE 1 decision. See generally Union of Concerned Scientists, et al. v. NHTSA, et al., No. 19-1230 (D.C. Cir. filed Oct. 28, 2019) (on February 8, 2021, the D.C. Circuit granted the Agencies' motion to hold the case in abeyance in light of the reconsideration of the SAFE 1 action). EPA also received three petitions for reconsideration of this waiver withdrawal.

forbidden from ever determining that a waiver is inappropriate due to consideration of anything other than the 'criteria' or 'prongs' at section 209(b)(1)(B)(A)-(C)."46 EPA further explained that those pronouncements were made in waiver proceedings where the Agency was acting solely on its own in contrast to a joint action with NHTSA such as SAFE 1. Additionally, EPA expressed its intention not to consider factors other than statutory criteria set out in section 209(b)(1)(A)-(C) in future waiver proceedings, explaining that addressing the preemptive effect of EPCA and its implications for EPA's waiver for California's GHG standards and ZEV sales mandate was uniquely called for in SAFE 1 because EPA and NHTSA were coordinating regulatory actions in a single notice.⁴⁷

Second, EPA withdrew the waiver for the GHG standards and ZEV sales mandate under the second waiver prong, section 209(b)(1)(B), on two alternative grounds. Specifically, EPA determined first that California does not need the GHG standards "to meet compelling and extraordinary conditions," under section 209(b)(1)(B), and second, even if California does have compelling and extraordinary conditions in the context of global climate change, California does not "need" the specific GHG standards under section 209(b)(1)(B) because they will not meaningfully address global air pollution problems of the type associated with GHG emissions.⁴⁸ EPA also reasoned

^{46. 84} FR at 51338.

^{47.} *Id*.

^{48.} Id. at 51341-42.

that because CARB had characterized the ZEV sales mandate as a compliance mechanism for GHG standards, both were "closely interrelated" given the overlapping compliance regimes for the ACC program, and as a result the ZEV sales mandate was inextricably interconnected with CARB's GHG standards.⁴⁹ In support of its overall determination that the ZEV sales mandate was not needed to meet compelling and extraordinary conditions, EPA relied on a single statement in the ACC program waiver support document where CARB did not attribute criteria emission reductions to the ZEV sales mandate, but rather noted its LEV III criteria pollutant fleet standard was responsible for those emission reductions.⁵⁰ Relying on this reasoning, EPA also withdrew the waiver for the ZEV sales mandate under the second waiver prong finding that California had no "need" for its own ZEV sales mandate.

In withdrawing the waiver, EPA relied on an alternative view of the scope of the Agency's analysis of California waiver requests and posited that reading "such State standards" as requiring EPA to only and always consider California's entire motor vehicle program would limit the application of this waiver prong in a way that EPA did not believe Congress intended.⁵¹ EPA further

^{49.} Id. at 51337.

^{50.} Id. at 51330.

^{51.} In other words, EPA asserted that once it determines that California needed its very first set of submitted standards to meet extraordinary and compelling conditions, EPA would never have the discretion to determine that California did not need any subsequent standards for which it sought a successive waiver. EPA based its

noted that the Supreme Court had found that CAA provisions may apply differently to GHGs than they do to traditional pollutants in UARG v. EPA, 134 S. Ct. 2427 (2014) (partially reversing the GHG "Tailoring" Rule on grounds that the CAA section 202(a) endangerment finding for GHG emissions from motor vehicles did not compel regulation of all sources of GHG emissions under the Prevention of Significant Deterioration and Title V permit programs). EPA then interpreted section 209(b)(1)(B) as requiring a particularized, local nexus between (1) pollutant emissions from sources, (2) air pollution, and (3) resulting impact on health and welfare. 52 Interpreting section 209(b)(1)(C) to be limited to the specific standards under the waiver, EPA stated that "such State standards" in sections 209(b)(1)(B) and (C) should be read consistently with each other, which EPA asserted was a departure from the traditional approach where this phrase in section 209(b)(1)(B) is read as referring back to "in the aggregate" in section 209(b)(1).53

reading also on an assertion of ambiguity in the meaning of "such State standards" in section 209(b)(1)(B).

^{52.} *Id.* at 51339-40.

^{53.} *Id.* at 51344-45.EPA notes that this SAFE 1 position was taken despite the Agency previously stating in the ACC program waiver that "Similarly, although the Dealers might suggest that EPA only be obligated to determine whether each of CARB's ACC regulatory components, in isolation, is consistent with section 202(a) we believe the better approach is to determine the technological feasibility of each standard in the context of the entire regulatory program for the particular industry category. In this case, we believe CARB has in fact recognized the interrelated, integrated approach the industry must take in order to address the regulatory

In the SAFE proposal, as an additional basis for the waiver withdrawal, EPA proposed to find that CARB's ZEV sales mandate and GHG standards are not consistent with section 202(a) of the CAA under the third waiver prong, section 209(b)(1)(C).⁵⁴ However, in the final SAFE 1 action, EPA and NHTSA explained they were not finalizing the proposed assessment regarding the technological feasibility of the Federal GHG and CAFE standards for MY 2021 through 2025 in SAFE 1, and thus EPA did not finalize any determination with respect to section 209(b)(1)(C).⁵⁵

In justifying the withdrawal action in SAFE 1, EPA opined that the text, structure, and context of section 209(b) supported EPA's authority to reconsider prior waiver grants. Specifically, EPA asserted that the Agency's authority to reconsider the grant of ACC

components of the ACC program. As noted above, the House Committee Report explained as part of the 1977 amendments to the Clean Air Act that California was to be afforded flexibility to adopt a complete program of motor vehicle emission controls (emphasis added). As such, EPA believes that Congress intended EPA to afford California the broadest possible discretion in selecting the best means to protect the health of its citizens and the public welfare.32 EPA believes this intent extends to CARB's flexibility in designing its motor vehicle emission program and evaluating the aggregate effect of regulations within the program." 78 FR at 2217.

^{54. 83} FR at 43240.

^{55. 84} FR at 51350. EPA explained that it may make a determination in connection with a future final action with regard to Federal standards. EPA's subsequent regulation to issue Federal standards did not address this issue. 85 FR 24174.

program waiver was implicit in section 209(b) given that revocation of a waiver is implied in the authority to grant a waiver. The Agency noted that further support for the authority to reconsider could be found in a single sentence in the 1967 legislative history of provisions now codified in sections 209(a) and (b) and the judicial principle that agencies possess inherent authority to reconsider their decisions. According to the Senate report from the 1967 CAA amendments, the Administrator has "the right . . . to withdraw the waiver at any time [if] after notice and an opportunity for public hearing he finds that the State of California no longer complies with the conditions of the waiver."⁵⁶ EPA also noted that, subject to certain limitations, administrative agencies possess inherent authority to reconsider their decisions in response to changed circumstances: "It is well settled that EPA has inherent authority to reconsider, revise, or repeal past decisions to the extent permitted by law so long as the Agency provides a reasoned explanation." This authority exists in part because EPA's interpretations of the statutes it administers "are not carved in stone."58

Finally, in SAFE 1, EPA provided an interpretive view of section 177 as not authorizing other states to adopt California's GHG standards for which EPA had granted a waiver of preemption under section 209(b). Although section 177 does not require states that adopt California's

^{56. 84} FR at 51332 (citing S. Rep. No. 90-403, at 34 (1967)).

^{57.} Id. at 51333.

^{58.} Chevron U.S.A. Inc. v. NRDC, Inc., 467 U.S. 837, 863 (1984).

emission standards to submit such regulations for EPA review and provides no statutory role for EPA in states' decision to adopt California's standards, EPA chose to nevertheless provide an interpretation that this provision is available only to states with approved nonattainment plans. EPA stated that nonattainment designations exist only as to criteria pollutants and GHGs are not criteria pollutants; therefore, states could not adopt GHG standards under section 177. Notably, California in previous waiver requests addressed the criteria pollutant benefits of GHG emissions reductions, specifically related to ground level ozone.

D. Petitions for Reconsideration

After issuing SAFE 1, EPA received three petitions for reconsideration urging the Agency to reconsider the waiver withdrawal of the ACC program's GHG standards and ZEV sales mandate and to rescind part or all of the SAFE 1 action.⁵⁹ The first Petition for Clarification/Reconsideration was submitted by the State of California and a number of States and Cities on October 9, 2019 (California Petition for Clarification).⁶⁰ These Petitioners sought both clarification and reconsideration of the scope of SAFE 1. Citing somewhat contradictory statements in the action, they claimed that SAFE 1 created confusion regarding which model years of the ACC program were

^{59.} The California Petition for Clarification only sought reconsideration of SAFE 1 to the extent it withdrew the ACC program waiver for model years outside those proposed. The other two petitions sought reconsideration of the full SAFE 1 action.

^{60.} EPA-OAR-2021-0257-0015.

affected by the waiver withdrawal.⁶¹ They based their request for reconsideration of the withdrawal on the grounds that the SAFE 1 action relied on analyses and justifications not presented at proposal and, thus, was beyond the scope of the proposal.

A second Petition for Reconsideration was submitted by several non-governmental organizations on November 25, 2019 (NGOs' Petition). ⁶² These Petitioners claimed that EPA's reconsideration of the ACC program waiver was not a proper exercise of agency authority because the Agency failed to consider comments submitted after the formal comment period—which they charged as inadequate and because the EPA's rationale was a pretextual cover

^{61.} The California Petition for Clarification notes that, "[i]n the Final Actions, EPA makes statements that are creating confusion, and, indeed, appear contradictory, concerning the temporal scope of its action(s)—specifically, which model years are covered by the purported withdrawal of California's waiver for its GHG and ZEV standards. In some places, EPA's statements indicate that it has limited its action(s) to the model years for which it proposed to withdraw and for which it now claims to have authority to withdraw—namely model years 2021 through 2025. In other places, however, EPA's statements suggest action(s) with a broader scope—one that would include earlier model years." *Id.* at 2. In SAFE 1, EPA withdrew the waiver for California's GHG and ZEV standards for model years 2017-2025 on the basis of EPCA preemption and for model years 2021-2025 on the basis of the second waiver prong.

^{62.} EPA-HQ-OAR-2021-0257-0014. This Petition was joined by The Center for Biological Diversity, Chesapeake Bay Foundation, Environment America, Environmental Defense Fund, Environmental Law & Policy Center, Natural Resources Defense Council, Public Citizen, Inc., Sierra Club, and the Union of Concerned Scientists.

for the Administration's political animosity towards California and the oil industry's influence. The late comments summarized in the Petition address SAFE 1's EPCA preemption and second waiver prong arguments. On EPCA preemption, the summarized comments asserted that EPCA does not preempt GHG standards because GHG emission standards are not the "functional equivalent" of fuel economy standards, as SAFE 1 claimed. On the second waiver prong, the summarized comments asserted both that GHG and ZEV standards do have criteria pollutant benefits, and that the threat of climate change is compelling and extraordinary and will have California-specific impacts. In addition to objections to SAFE 1's EPCA preemption and second waiver prong arguments, the summarized comments asserted that ZEV standards play a key role in SIPs, which were disrupted by SAFE 1. This disruption, Petitioners claimed, violated "conformity" rules prohibiting federal actions from undermining state's air quality plans. 63

A third Petition for Reconsideration was submitted by several states and cities on November 26, 2019 (States and Cities' Petition). These Petitioners sought reconsideration

^{63.} These "late comments" can be found in the "Appendix of Exhibits" attached to the Petition for Reconsideration. These comments are considered part of EPA's record for purposes of the reconsideration of SAFE 1.

^{64.} See EPA-HQ-OAR-2021-0257-0029. This Petition was joined by the States of California, Colorado, Connecticut, Delaware, Hawaii, Illinois, Maine, Maryland, Minnesota, Nevada, New Jersey, New Mexico, New York, North Carolina, Oregon, Rhode Island, Vermont, Washington, Wisconsin, Michigan, the Commonwealths of Massachusetts, Pennsylvania, and Virginia, the District of Columbia,

of the withdrawal on the grounds that EPA failed to provide an opportunity to comment on various rationales and determinations, in particular on its authority to revoke argument, flawed re-interpretation and application of the second waiver prong, its flawed new rationale for considering factors outside section 209(b) (namely, EPCA preemption), and its determination that states cannot adopt California's GHG standards under section 177. For example, these Petitioners claimed they did not have an adequate opportunity to comment on EPA's use of equal sovereignty or the endangerment finding as rationales for its new "particularized nexus" interpretation of the second waiver prong. These Petitioners also claimed that EPA's statements concerning the burden of proof applicable to a waiver revocation were either unclear or inaccurate, particularly whether the Agency bears the burden of proof in withdrawing a previously granted waiver and, if not, how and why this burden of proof is different from the burden of proof for denying a waiver request. 65 Finally, these Petitioners asserted that the Agency failed to consider comments, submitted after the formal comment period, that challenged EPA's interpretation of the second waiver prong, including new evidence of California's need for its GHG emission standards and ZEV sales mandate. and alleged that EPA's rationale was pretextual and based on the Administration's political animosity towards California and on the oil industry's influence.

and the Cities of Los Angeles, New York, San Francisco, and San Jose.

^{65.} The applicable burden of proof for a waiver withdrawal is discussed in Section III of this decision.

EPA notified the petitioners in the above-noted Petitions for Reconsideration that the Agency would be considering issues raised in their petitions as part of the proceeding to reconsider SAFE 1. This action addresses these petitions in the broader context of EPA's adjudicatory reconsideration of SAFE 1 commenced in response to a number of significant issues with SAFE 1.

III. Principles Governing This Review

The CAA has been a paradigmatic example of cooperative federalism, under which "States and the Federal Government [are] partners in the struggle against air pollution." In Title II, Congress authorized EPA to promulgate emission standards for mobile sources and generally preempted states from adopting their own standards. At the same time, Congress created an important exception for the State of California.

 $^{66.\;}$ General Motors Corp. v. United States, 496 U.S. 530, 532 (1990).

^{67. &}quot;The regulatory difference [between Titles I and II] is explained in part by the difficulty of subjecting motor vehicles, which readily move across state boundaries, to control by individual states." Engine Mfrs. Ass'n v. EPA, 88 F.3d 1075, 1079 (D.C. Cir. 1996). Congress also asserted federal control in this area to avoid "the specter of an anarchic patchwork of federal and state regulatory programs" nationwide. See Motor & Equip. Mfrs. Ass'n, Inc. v. EPA, 627 F.2d 1095, 1109 (D.C. Cir. 1979) (MEMA I).

A. Scope of Preemption and Waiver Criteria Under the Clean Air Act

The legal framework for this decision stems from the waiver provision first adopted by Congress in 1967, and subsequent amendments. In Title II of the CAA, Congress established only two programs for control of emissions from new motor vehicles—EPA emission standards adopted under the CAA and California emission standards adopted under its state law. Congress accomplished this by preempting all state and local governments from adopting or enforcing emission standards for new motor vehicles, while at the same time providing that California could receive a waiver of preemption for its emission standards and enforcement procedures in keeping with its prior experience regulating motor vehicles and its serious air quality problems. Accordingly, section 209(a) preempts states or political subdivisions from adopting or attempting to enforce any standard relating to the control of emissions from new motor vehicles. 68 Under the terms of section 209(b)(1), after notice and opportunity for public hearing, EPA must waive the application of section 209(a) to California unless the Administrator finds

^{68. 42} U.S.C. 7543(a)-(a) Prohibition No State or any political subdivision thereof shall adopt or attempt to enforce any standard relating to the control of emissions from new motor vehicles or new motor vehicle engines subject to this part. No State shall require certification, inspection, or any other approval relating to the control of emissions from any new motor vehicle or new motor vehicle engine as condition precedent to the initial retail sale, titling (if any), or registration of such motor vehicle, motor vehicle engine, or equipment.

at least one of three criteria to deny a waiver in section 209(b)(1)(A)-(C) has been met.⁶⁹ EPA may thus deny a waiver only if it makes at least one of these three findings based on evidence in the record, including arguments that opponents of the waiver have provided. This framework struck an important balance that protected manufacturers from multiple and different state emission standards and preserved a pivotal role for California in the control of emissions from new motor vehicles. Congress intentionally structured this waiver provision to restrict and limit EPA's ability to deny a waiver and did this to ensure that California had broad discretion in selecting the means it determined best to protect the health and welfare of

69. 42 U.S.C. 7543(b)(1):

- (1) The Administrator shall, after notice and opportunity for public hearing, waive application of this section to any State which has adopted standards (other than crankcase emission standards) for the control of emissions from new motor vehicles or new motor vehicle engines prior to March 30, 1966, if the State determines that the State standards will be, in the aggregate, at least as protective of public health and welfare as applicable Federal standards. No such waiver shall be granted if the Administrator finds that—
- (A) the determination of the State is arbitrary and capricious,
- (B) such State does not need such State standards to meet compelling and extraordinary conditions, or
- (C) such State standards and accompanying enforcement procedures are not consistent with section 7521(a) of this title.

its citizens in recognition of both the harsh reality of California's air pollution and to allow California to serve as a pioneer and a laboratory for the nation in setting new motor vehicle emission standards and developing control technology. Accordingly, section 209(b) specifies that EPA must grant California a waiver if California determines that its standards are, in the aggregate, at least as protective of the public health and welfare as applicable Federal standards.

EPA has consistently interpreted the waiver provision as placing the burden on the opponents of a waiver and EPA to demonstrate that one of the criteria for a denial has been met. In this context, since 1970, EPA has recognized its limited discretion in reviewing California waiver requests. For over fifty years, therefore, EPA's role upon receiving a request for waiver of preemption from California has been limited and remains only to determine whether it is appropriate to make any of the three findings specified by the CAA. If the Agency cannot make at least one of the three findings, then the waiver must be granted. The three waiver criteria are also properly seen as criteria for a denial. This reversal of the normal statutory structure embodies and is consistent with the congressional intent of providing deference to California to maintain its own new motor vehicle emission program.

^{70.} See, e.g., S. Rep. No. 403, 90th Cong., 1st Sess. 33 (1967) (The waiver of preemption is for California's "unique problems and pioneering efforts."); 113 Cong. Rec. 30950, 32478 ("[T]he State will act as a testing agent for various types of controls and the country as a whole will be the beneficiary of this research.") (Statement of Sen. Murphy); *MEMA I*, 627 F.2d 1095, 1111 (D.C. Cir. 1979).

The 1970 CAA Amendments strengthened EPA's authority to regulate vehicular "emission[s] of any air pollutant," while reaffirming the corresponding breadth of California's entitlement to regulate those emissions (amending CAA section 202 and recodifying the waiver provision as section 209(b), respectively). Congress also established the NAAQS program, under which EPA issues air quality criteria and sets standards for so-called "criteria" pollutants, and states with regions that have not "attained" those federal standards must submit SIPs indicating how they plan to attain the NAAQS (which is often a multi-year, comprehensive plan). With the CAA Amendments of 1977, Congress allowed California to consider the protectiveness of its standards "in the aggregate," rather than requiring that each standard proposed by the State be as or more stringent than its federal counterpart.⁷¹ Congress also approved EPA's interpretation of the waiver provision as providing appropriate deference to California's policy goals and consistent with Congress's intent "to permit California to proceed with its own regulatory program" for new motor vehicle emissions.⁷²

In previous waiver decisions, EPA has noted that the statute specifies particular and limited grounds for rejecting a waiver and has therefore limited its review to those grounds. EPA has also noted that the structure Congress established for reviewing California's decisionmaking is deliberately narrow, which further supports this approach. This has led EPA to reject arguments that

^{71. 42} U.S.C. 7543(b)(1).

^{72.} H.R. Rep. No. 95-294, at 301 (1977).

are not specified in the statute as grounds for denying a waiver:

The law makes it clear that the waiver requests cannot be denied unless the specific findings designated in the statute can properly be made. The issue of whether a proposed California requirement is likely to result in only marginal improvement in air quality not commensurate with its cost or is otherwise an arguably unwise exercise of regulatory power is not legally pertinent to my decision under section 209, so long as the California requirement is consistent with section 202(a) and is more stringent than applicable Federal requirements in the sense that it may result in some further reduction in air pollution in California. Thus, my consideration of all the evidence submitted concerning a waiver decision is circumscribed by its relevance to those questions that I may consider under section 209(b).73

Given the text, legislative history, and judicial precedent, EPA has consistently interpreted section 209(b) as requiring it to grant a waiver unless opponents of a waiver can demonstrate that one of the criteria for a denial has been met.⁷⁴

^{73. 78} FR at 2115 (footnote omitted).

^{74.} *MEMA I*, 627 F.2d at 1120-21 ("The language of the statute and its legislative history indicate that California's regulations, and California's determination that they comply with the statute, when

The 1977 CAA Amendments additionally demonstrated the significance of California's standards to the Nation as a whole with Congress' adoption of a new section 177. Section 177 permits other states addressing their own air pollution problems to adopt and enforce California new motor vehicle standards "for which a waiver has been granted if certain criteria are met." Also known as the

presented to the Administrator are presumed to satisfy the waiver requirements and that the burden of proving otherwise is on whoever attacks them."); *Motor & Equip. Mfrs. Ass'n, Inc.* v. *Nichols,* 142 F.3d 449, 462 (D.C. Cir. 1998) (*MEMA II*) ("[S]ection 209(b) sets forth the only waiver standards with which California must comply If EPA concludes that California's standards pass this test, it is obligated to approve California's waiver application.").

75. This provision was intended to continue the balance, carefully drawn in 1967, between states' need to meet increasingly stringent federal air pollution limits and the burden of compliance on auto-manufacturers. See, e.g., H.R. Rep. No. 294, 95th Cong., 1st Sess. 309-10 (1977) ("[S]ection 221 of the bill broadens State authority, so that a State other than California . . . is authorized to adopt and enforce new motor vehicle emission standards which are identical to California's standards. Here again, however, strict limits are applied This new State authority should not place an undue burden on vehicle manufacturers "); Motor Vehicle Mfrs. Ass'n v. NYS Dep't of Env't Conservation, 17 F.3d 521, 527 (2d Cir. 1994) ("Many states, including New York, are in danger of not meeting increasingly stringent federal air pollution limits It was in an effort to assist those states struggling to meet federal pollution standards that Congress, as noted earlier, directed in 1977 that other states could promulgate regulations requiring vehicles sold in their state to be in compliance with California's emission standards or to "piggyback" onto California's preemption exemption. This opt-in authority, set forth in §177 of the Act, 42 U.S.C. 7507, is carefully circumscribed to avoid placing an undue burden on the automobile manufacturing industry.").

"opt-in" provision, section 177 of the Act, 42 U.S.C. 7507, provides:

Notwithstanding section 7543(a) of this title, any State which has plan provisions approved under this part may adopt and enforce for any model year standards relating to control of emissions from new motor vehicles or new motor vehicle engines and take such other actions as are referred to in section 7543(a) of this title respecting such vehicles if—

- (1) such standards are identical to the California standards for which a waiver has been granted for such model year, and
- (2) California and such State adopt such standards at least two years before commencement of such model year (as determined by regulations of the Administrator).

Nothing in this section or in Subchapter II of this chapter shall be construed as authorizing any such State to prohibit or limit, directly or indirectly, the manufacture or sale of a new motor vehicle or motor vehicle engine that is certified in California as meeting California standards, or to take any action of any kind to create, or have the effect of creating, a motor vehicle or motor vehicle engine different that a motor vehicle or engine certified in California under California standards (a "third vehicle") or otherwise create such a "third vehicle."

Any state with qualifying SIP provisions may exercise this option and become a "Section 177 State," without first seeking the approval from EPA.⁷⁶ Thus, over time, Congress has recognized the important state role, for example, by making it easier (by allowing California to consider its standards "in the aggregate") and by expanding the opportunity (via section 177) for states to adopt standards different from EPA's standards.⁷⁷

^{76.} In 1990 Congress amended the CAA by adding section 209(e) to section 209. Section 209(e) sets forth the terms of CAA preemption for nonroad engines and vehicles and the ability of States to adopt California emissions standards for such vehicles and engines if certain criteria are met. 42 U.S.C. 7543(e)(2)(B) ("Any State other than California which has plan provisions approved under part D of subchapter I may adopt and enforce, after notice to the Administrator, for any period, standards relating to control of emissions from nonroad vehicles or engines . . . if (i) such standards and implementation and enforcement are identical, for the period concerned, to the California standards "). Courts have interpreted these amendments as reinforcing the important role Congress assigned to California. See Engine Mfrs. Ass'n v. EPA, 88 F.3d 1075, 1090 ("Given the indications before Congress that California's regulatory proposals for nonroad sources were ahead of the EPA's development of its own proposals and the Congressional history of permitting California to enjoy coordinated regulatory authority over mobile sources with the EPA, the decision to identify California as the lead state is comprehensible. California has served for almost 30 years as a 'laboratory' for motor vehicle regulation."); MEMA I, 627 F.2d 1095, 1110 (D.C. Cir. 1979) ("Its severe air pollution problems, diverse industrial and agricultural base, and variety of climatic and geographical conditions suit it well for a similar role with respect to nonroad sources.").

^{77. 40} FR at 23104; see also LEV I waiver at 58 FR 4166, Decision Document at 64.

B. Deference to California

EPA has consistently noted that the text, structure, and history of the California waiver provision clearly indicate both congressional intent and appropriate EPA practice of leaving the decision on "ambiguous and controversial matters of public policy" to California's judgment. In waiver decisions, EPA has thus recognized that congressional intent in creating a limited review of California waiver requests based on the section 209(b)(1) criteria was to ensure that the federal government did not second-guess the wisdom of state policy. In an early waiver decision EPA highlighted this deference:

It is worth noting * * * I would feel constrained to approve a California approach to the problem which I might also feel unable to adopt at the federal level in my own capacity as a regulator. The whole approach of the Clean Air Act is to force the development of new types of emission control technology where that is needed by compelling the industry to "catch up" to some degree with newly promulgated standards. Such an approach * * * may be attended with costs, in the shape of reduced product offering, or price or fuel economy penalties, and by risks that a wider number of vehicle classes may not be able to complete their development work in time. Since a balancing of these risks and costs against the potential benefits from reduced emissions is a central policy decision for any regulatory agency under the statutory scheme outlined above, I believe I am required to give very substantial deference to California's judgments on this score.⁷⁸

As noted above, Congress amended the CAA in 1977. Within these amendments, Congress had the opportunity to reexamine the waiver provision and elected to expand California's flexibility to adopt a complete program of motor vehicle emission controls. The House Committee Report explained that "[t]he amendment is intended to ratify and strengthen the California waiver provision and to affirm the underlying intent of that provision, *i.e.*, to afford California the broadest possible discretion in selecting the best means to protect the health of its citizens and the public welfare."⁷⁹

SAFE 1 was a departure from congressional intent and EPA's typical practice of deference to California on matters of state public policy regarding how best to address its serious air quality problems. In SAFE 1, EPA adopted a new interpretation of section 209(b)(1)(B) more than five years after the initial grant of the ACC program waiver and applied it to CARB's GHG standards and ZEV sales mandate. Specifically, EPA premised its finding on a consideration of California's "need" for the

^{78. 40} FR at 23104.

^{79.} *MEMA I*, 627 F.2d at 1110 (citing H.R. Rep. No. 294, 95th Cong., 1st Sess. 301-02 (1977)). Congress amended section 209(b) (1)(A) regarding California's determination that its standards are as at least as protective as applicable Federal standards so that such determination may be done "in the aggregate" looking at the summation of the standards within the vehicle program.

specific standards, instead of the "need" for a separate motor vehicle emission program to meet compelling and extraordinary conditions, stating that "such State standards" in section 209(b)(1)(B) was ambiguous with respect to the scope of the Agency's analysis. EPA further determined that California did not need the ZEV sales mandate to meet compelling and extraordinary conditions by relying on a single statement in the ACC program waiver support document taken out of context, where it noted that the ZEV sales mandate had no criteria emissions benefits in terms of vehicle emissions and its LEV III criteria pollutant fleet standard was responsible for those emission reductions. In response to the SAFE 1 proposal, California had provided further context and additional data on net upstream emissions benefits of the ZEV sales mandate, but EPA did not consider them in arriving at the findings and conclusions in SAFE 1. The final decision in SAFE 1 was not based on the third waiver prong.⁸⁰ EPA also explained in SAFE 1 that the task of interpreting section 209(b)(1)(B) required no deference to California.81

C. Standard and Burden of Proof

In Motor and Equipment Manufacturers' Association v. EPA, 627 F.2d 1095 (D.C. Cir. 1979) ($MEMA\ I$), the

^{80. 84} FR at 51322-33. EPA notes that when reviewing California's standards under the third waiver prong, the Agency may grant a waiver to California for standards that EPA may choose not to adopt at the federal level due to different considerations. *See* 78 FR at 2133.

^{81. 84} FR at 51339-40.

U.S. Court of Appeals for the District of Columbia stated, with regard to the standard and burden of proof, that the Administrator's role in a section 209 proceeding is to "consider all evidence that passes the threshold test of materiality and . . . thereafter assess such material evidence against a standard of proof to determine whether the parties favoring a denial of the waiver have shown that the factual circumstances exist in which Congress intended a denial of the waiver."82 The court in MEMA I considered the standards of proof under section 209 for the two findings necessary to grant a waiver for an "accompanying enforcement procedure" (as opposed to the standards themselves): (1) Protectiveness in the aggregate and (2) consistency with CAA section 202(a) findings. The court instructed that "the standard of proof must take account of the nature of the risk of error involved in any given decision, and it therefore varies with the finding involved. We need not decide how this standard operates in every waiver decision."83 The court upheld the Administrator's position that to deny a waiver, there must be clear and compelling evidence to show that the proposed procedures undermine the protectiveness of California's standards. The court noted that this standard of proof also accords with the congressional intent to provide California with the broadest possible discretion in setting regulations it finds protective of the public health and welfare.⁸⁴ With respect to the consistency finding,

^{82.} MEMA I, 627 F.2d at 1122.

^{83.} *Id*.

^{84.} Id.

the court did not articulate a standard of proof applicable to all proceedings but found that the opponents of the waiver were unable to meet their burden of proof even if the standard were a mere preponderance of the evidence.

Although MEMA I did not explicitly consider the standards of proof under section 209 concerning a waiver request for "standards," as compared to accompanying enforcement procedures, there is nothing in the opinion to suggest that the court's analysis would not apply with equal force to such determinations. EPA's past waiver decisions have consistently made clear that: "[E]ven in the two areas concededly reserved for Federal judgment by this legislation—the existence of compelling and extraordinary conditions and whether the standards are technologically feasible—Congress intended that the standard of EPA review of the State decision to be a narrow one."85 Although EPA evaluates whether there are compelling and extraordinary conditions in California, the Agency nevertheless accords deference to California on its choices for how best to address such conditions in light of the legislative history of section 209(b).

^{85.} See, e.g., 40 FR at 23102-03. See also MEMA I, 627 F.2d at 1109 ("Congress had an opportunity to restrict the waiver provision in making the 1977 amendments, and it instead elected to expand California's flexibility to adopt a complete program of motor vehicle emissions control. Under the 1977 amendments, California need only determine that its standards will be 'in the aggregate, at least as protective of public health and welfare than applicable Federal standards,' rather than the "more stringent" standard contained in the 1967 Act.") (citing H.R. Rep. No. 294, 95th Cong., 1st Sess. 301-02 (1977), U.S. Code Cong. & Admin. News 1977, p. 1380).

As noted earlier, the burden of proof in a waiver proceeding is on EPA and the opponents of the waiver. This is clear from the statutory language stating that EPA "shall . . . waive" preemption unless one of three statutory factors is met. This reading was upheld by the D.C. Circuit in *MEMA I*, which concluded that this obligation rests firmly with opponents of the waiver in a section 209 proceeding, holding that: "[t]he language of the statute and its legislative history indicate that California's regulations, and California's determinations that they must comply with the statute, when presented to the Administrator are presumed to satisfy the waiver requirements and that the burden of proving otherwise is on whoever attacks them. California must present its regulations and findings at the hearing and thereafter the parties opposing the waiver request bear the burden of persuading the Administrator that the waiver request should be denied."86

The Administrator's burden, on the other hand, is to make a reasonable evaluation of the information in the record in coming to the waiver decision. As the court in *MEMA I* stated, "Here, too, if the Administrator ignores evidence demonstrating that the waiver should not be granted, or if he seeks to overcome that evidence with unsupported assumptions of his own, he runs the risk of having his waiver decision set aside as 'arbitrary and capricious." Therefore, the Administrator's burden is to act "reasonably." 88

^{86.} MEMA I, 627 F.2d at 1121.

^{87.} Id. at 1126.

^{88.} Id.

In this instance, EPA has withdrawn a previously granted waiver and is now reconsidering whether that withdrawal was an appropriate exercise of authority, whether the reinterpretation of the second waiver prong was appropriate, and whether EPA's evaluation and findings of fact under the second waiver prong meet the applicable burden of proof in the context of deference to California's policy choices. EPA believes that the same burden that is applicable to those opposed to an initial waiver request from CARB (this applies to any party including the Administrator as explained in MEMAI) is also applicable to EPA's actions in SAFE 1 (e.g., the burden of proof of whether California does not need its standards to meet compelling and extraordinary conditions rests on those opposing a waiver for California).⁸⁹

^{89.} In EPA's 2009 evaluation of the 2008 GHG waiver denial the Agency applied a similar test. See 74 FR at 32745 ("After a thorough evaluation of the record, I am withdrawing EPA's March 6, 2008 Denial and have determined that the most appropriate action in response to California's greenhouse gas waiver request is to grant that request. I have determined that the waiver opponents have not met their burden of proof in order for me to deny the waiver under any of the three criteria in section 209(b)(1)."). In the context of 2009 GHG waiver that reconsidered the Agency's 2008 GHG waiver denial, EPA determined it was appropriate to apply the same burden of proof during the reconsideration as would apply at the time of the initial waiver evaluation. EPA received comment suggesting that the entire burden of proof shifts to California in order for the prior 2008 denial to be reversed. EPA, in response, stated that "... regardless of the previous waiver denial, once California makes its protectiveness determination the burden of proof falls on the opponents of the waiver This is consistent with the legislative history, which indicates that Congress intended a narrow review by EPA and to preserve the broadest possible discretion for California." Id. at 32749. EPA acknowledges that in SAFE 1 the Agency not only

IV. EPA Did Not Appropriately Exercise Its Limited Authority To Reconsider the ACC Program Waiver in SAFE 1

The first question this final action tackles is whether the agency properly exercised its reconsideration authority to withdraw its previously-granted waiver in SAFE 1. EPA concludes that it did not, and on that independent basis rescinds SAFE 1's waiver withdrawal.

Section 209 does not provide EPA with express authority to reconsider and withdraw a waiver previously granted to California. EPA's authority thus stems from its inherent reconsideration authority. For several reasons, in the context of reconsidering a waiver grant, that authority may only be exercised sparingly. First, EPA believes its inherent authority to reconsider a waiver

adopted an interpretation of the second waiver prong which was similar to the previously rejected interpretation, but that in doing so also questioned its previous position that the burden of proof in evaluating the need for standards at issue resides with those that oppose the waiver, including EPA. See 84 FR at 51344 n.268. In this action, however, EPA now finds that the historical deference provided to California regarding its policy choices on how best to address its serious air quality conditions also requires that the burden of proof should reside in those seeking to demonstrate that standards are not needed under the second waiver prong regardless of whether the rationale is characterized as a new interpretation or not. The language of section 209(b)(1) requires California to make a protectiveness finding under the first waiver prong. Moreover, nothing in section 209(b) could be read as support for drawing a distinction between the burden of proof when the Agency considers an initial waiver request and one where the Agency reconsiders a waiver decision based on a new interpretation of the statutory criteria. That burden properly resides with opponents of the waiver.

decision is constrained by the three waiver criteria that must be considered before granting or denying a waiver request under section 209(b). A contrary approach, which treats reconsiderations as more broadly appropriate, would undermine Congress' intent that California be able to exercise its policy judgments and develop motor vehicle controls programs to address California's air pollution problems, and make advances which could be built on by EPA or adopted by other states. Second, EPA believes it may only reconsider a previously granted waiver to address a clerical or factual error or mistake, or where information shows that factual circumstances or conditions related to the waiver criteria evaluated when the waiver was granted have changed so significantly that the propriety of the waiver grant is called into doubt. Even when EPA is acting within the appropriate bounds of its authority to reconsider, during that reconsideration EPA should exercise its limited authority within a reasonable timeframe and be mindful of reliance interests.

The Agency's reconsideration in SAFE 1 was not an appropriate exercise of authority; there was no clerical error or factual error in the ACC program waiver, and SAFE 1 did not point to any factual circumstances or conditions related to the three waiver prongs that had changed so significantly that the propriety of the waiver grant is called into doubt. Rather, the 2019 waiver withdrawal was based on a change in EPA's statutory interpretation, an incomplete and inaccurate assessment of the record, and another agency's action beyond the confines of section 209(b). EPA erred in reconsidering a previously granted waiver on these bases. Moreover, in considering the passage of time between the initial

waiver and the SAFE 1 action, and the development of reliance interests based on the waiver, EPA finds those factors do not support the reconsideration of the ACC program waiver that occurred in SAFE 1. Accordingly, as explained in detail below, EPA is rescinding SAFE 1's withdrawal of its 2013 ACC program waiver because it was an inappropriate exercise of reconsideration authority.

A. Comments Received

EPA received several comments in the reconsideration proceeding on the Agency's authority to reconsider waivers. Comments on explicit authority focused on whether any language in section 209(b)(1), on its face, permits EPA to reconsider a previously granted waiver. Some of these commenters also distinguished between the denial of the 2008 waiver and the reconsideration and grant of the GHG waiver in 2009, and EPA's grant of the ACC program waiver in 2013 and the reconsideration and withdrawal of the ACC program waiver in 2019.

EPA received comments in support of and against the view that EPA has inherent authority to reconsider waivers. As support for EPA's implied authority to reconsider, one commenter cited relevant language from the Senate Committee Report from 1967 that stated, "implicit in [§209] is the right of [EPA] to withdraw the waiver [if] at any time after notice and an opportunity for public hearing he finds that the State of California no longer complies with the conditions of that waiver."

^{90.} Urban Air Initiative (Urban Air), Docket No. EPA-HQ-OAR-2021-0257-0223 at 22 (quoting S. Rep. 90-403, at 34 (1967)).

According to the commenter because "the waiver authorizes *future* regulation, which always remains open to change," EPA must have the authority to reconsider a waiver. Otherwise, EPA would be unable to monitor CARB's continued compliance with the waiver conditions in light of updated information.⁹¹ The same commenter also argued that an agency generally retains the authority to reconsider and correct any earlier decision unless Congress acts to displace the authority with a process to rectify the Agency's mistakes and that explicit statutory authority to withdraw a waiver is therefore not necessary, because "the power to reconsider is inherent in the power to decide." The commenter claimed that, under *Chevron*, "[a]n agency has a 'continuing' statutory obligation to consider the 'wisdom of its policy." ⁹³

In contrast, several commenters maintained that section 209(b) strongly indicates that EPA's authority to

^{91.} *Id.* at 21 ("A determination that California's state standards are technologically feasible and appropriate requires complex technical projections at the frontiers of science, which must be continually updated 'if the actual future course of technology diverges from expectation." (quoting *NRDC Inc.* v. *EPA*, 655 F.2d 318, 329 (D.C. Cir. 1981))).

^{92.} Urban Air at 20 (citing *Ivy Sports Med.*, *LLC* v. *Burwell*, 767 F.3d 81, 86, 93 (D.C. Cir. 2014)). This commenter also notes that, in EPA's 2009 action to reconsider its prior denial of a GHG waiver in 2008, CARB submitted a letter to EPA stating that "California believes EPA has inherent authority to reconsider the denial and should do so in order to restore the interpretations and applications of the Clean Air Act to continue California's longstanding leadership role in setting emission standards." *Id*.

^{93.} Id. at 21.

withdraw a previously issued waiver is, at most, limited. Several commenters argued that, absent language in a statute, administrative agencies lack inherent authority to reconsider adjudicatory decisions. He has commenters noted that courts highly scrutinize administrative revocations and are "unwilling[] to wrest a standardless and open-ended revocation authority from a silent statute." Instead, these commenters argued, EPA may act only with the authorities conferred upon it by Congress, and thus the Agency may only act if the CAA explicitly or implicitly grants it power to do so. According to these

^{94.} Institute for Policy Integrity Amicus Brief at 4 ("Lacking textual support, EPA invokes so-called 'inherent authority'—'more accurate[ly] label[ed] . . . 'statutorily implicit' authority,' $HTH\ Corp.$ v. NLRB, 823 F.3d 668, 679 (D.C. Cir. 2016)—to justify its action. 84 FR at 51,331. But this Court is 'unwilling[] to wrest a standardless and open-ended revocation authority from a silent statute,' $Am.\ Methyl.$, 749 F.2d 826, 837 (D.C. Cir. 1984), and EPA fails to justify the implicit authority it claims."); Twelve Public Interest Organizations app 1 at 32 (citing $Am.\ Methyl$ for "rejecting 'implied power' as 'contrary to the intention of Congress and the design of' the Act and quoting $HTH\ Corp.$'s statement that agencies, as creatures of statute, lack inherent authority); States and Cities at 16 (also citing $Am.\ Methyl$).

^{95.} Institute for Policy Integrity at 1 (citing Am. Methyl).

^{96.} States and Cities at 15 (citing HTH Corp. v. NLRB, 823 F.3d 668, 679 (D.C. Cir. 2016)); Twelve Public Interest Organizations, Docket No. EPA-HQ-OAR-2021-0257-0277 app. 1 at 28 ("The Clean Air Act preserves state authority to regulate emissions unless expressly 'provided' otherwise. 42 U.S.C. 7416. In statutes like this where preemption is the exception, only Congress's 'precise terms' can produce preemption. CTS Corp. v. Waldburger, 573 U.S. 1, 12-13 (2014)."); National Coalition for Advanced Transportation (NCAT), Docket No. EPA-HQ-OAR-2021-0257-0131 at 7-8; Institute for Policy Integrity at New York University School of Law (Institute

commenters, section 209(b) is silent on waiver withdrawal, its text indicates that EPA may only consider 209(b)'s three factors before either granting or denying a waiver, and its purpose and structure affords broad deference to California's standards. "Taken together, these factors indicate that EPA may not withdraw a previously-issued waiver based solely upon a reconsideration of its initial judgment." Commenters suggested that Congress, by listing the three waiver criteria and directing that EPA evaluate such criteria prior to granting the waiver, only authorized EPA to perform the evaluation once and that it "cannot later second-guess the wisdom of legal and policy judgments made as part of that evaluation." ⁹⁸

for Policy Integrity), Docket No. EPA-HQ-OAR-2021-0257-0115 at 2, citing its Final Brief of the Institute for Policy Integrity at New York University School of Law as Amicus Curiae in Support of Petitioners (Institute for Policy Integrity Amicus Brief) at 4, *Union of Concerned Scientists*, et al. v. NHTSA, et al., No. 19-1230 (D.C. Cir. filed Oct. 28, 2019), reprinted in the Institute's comments on the 2021 Notice of Reconsideration.

^{97.} Institute for Policy Integrity at 2, citing its Amicus Brief at 6-11.

^{98.} *Id.* at 7. *See also* Twelve Public Interest Organizations app. 1 at 28-29 ("Section 209(b)(1)'s precise terms mandate that EPA "shall" grant California a waiver unless EPA finds one of the three specified bases for denial. This language charges EPA "with undertaking a single review in which [the Administrator] applies the deferential standards set forth in Section 209(b) to California and either grants or denies a waiver." *Ford Motor Co.* v. *EPA*, 606 F.2d 1293, 1302 (D.C. Cir. 1979). It evinces no intent to provide EPA with the different and greater authority to withdraw a previously granted waiver, thereby arresting the State's ongoing implementation of its own laws.")

Similarly, commenters noted that section 209 does not textually "provide" EPA any authority nor specify any process by which EPA might revoke the rights given by an earlier-granted waiver. 99 In response to SAFE 1's claim of inherent reconsideration authority and the other commenters' reliance on the relevant excerpt from the 1967 Senate Report, these commenters argued that this "single sentence . . . does not establish any withdrawal authority," either generally or for the SAFE 1 withdrawal specifically. 100 That statement, commenters argued, "predate[s] the creation of the NAAQS program and Congress's invitations to development of numerous state reliance interests."101 Moreover, according to these commenters, the statement only discusses authority in the case that "California no longer complies with the conditions of the waiver," which commenters believe means California's "compliance with waiver conditions and, specifically, its cooperation with EPA concerning

^{99.} See South Coast Air Quality Management District (SCAQMD), Docket No. EPA-HQ-OAR-2021-0257-0228 at 3. This commenter argued that section 116 of the CAA (which explicitly references section 209) provides that there needs to be a textual basis for any exercise of authority to deny California the right (which it achieved via the 2013 waiver) to enforce its emission standards. Thus, the commenter continued, because there is no language in section 209 that gives any authority nor specifies any process for EPA to revoke the rights/waiver previously granted then EPA may not do so by the terms of section 116.

^{100.} States and Cities at 16. See also Twelve Public Interest Organizations app. 1 at 33-34.

^{101.} States and Cities at 16; See also Twelve Public Interest Organizations app. 1 at 33-34.

enforcement and certification procedures," not "redefined waiver criteria." ¹⁰²

In response to the argument made by EPA in SAFE 1 that, given the "considerable degree of future prediction" required by the third waiver prong, "where circumstances arise that suggest that such predictions may have been inaccurate, it necessarily follows that EPA has authority to revisit those predictions," some commenters claimed that California's standards do not become inconsistent with federal standards simply because they become more stringent than federal standards (in other words, a weakening of the federal standards does not necessarily create an inconsistency). The commenters noted also that EPA did not in fact revise its section 202(a) standards between issuing and withdrawing the waiver at issue, nor did EPA in fact make any final findings under the third waiver prong. 104

Many commenters stated that in order to exercise any implied or inherent authority, an agency must

^{102.} Twelve Public Interest Organizations app. 1 at 34. See also States and Cities at 16 (arguing that, although EPA proposed to withdraw the waiver on multiple grounds, such as the third waiver prong, "EPA's final action was based entirely on its own changed policy positions, namely its interpretation of Section 209(b) (1) to create a categorical bar against state regulation of vehicular GHG emissions and its decision to rely on another agency's newly articulated views of a different statute [EPCA].").

^{103. 84} FR at 51332.

^{104.} Institute for Policy Integrity at 2.

provide a "detailed justification" when departing from a policy that has "engendered serious reliance interests" and should not "rest on mere 'policy changes" alone. 105 Thus, supporters and opponents of SAFE 1 also provided comments on whether, assuming EPA did have authority to reconsider the ACC program waiver—either because of language in the CAA or because of its inherent authority to reevaluate decisions because of changed conditions—it was appropriate to exercise that authority in SAFE 1. Some commenters summarized precedent as requiring that the Agency consider reliance interests that have attached to its original decision, that reversals of informal adjudications occur within a reasonable time after the original decision, and that the reversal is not for the sole purpose of applying some change in administrative policy. 106 Opponents and supporters of SAFE 1 did, however, disagree on the significance of each of these factors.107

 $^{105.\,}$ States and Cities at 21-22 (quoting FCC v. Fox, 556 U.S. 502, 515 (2009)).

^{106.} Id. at 17 (citing Am. Methyl, 749 F.2d at 835; Chapman v. El Paso Nat. Gas Co., 204 F.2d 46, 53-54 (D.C. Cir. 1953); DHS v. Regents of the Univ. of California, 140 S. Ct. 1891, 1914 (2020); United States v. Seatrain Lines Inc., 329 U.S. 424, 429 (1947)).

^{107.} Urban Air at 21 (arguing that agencies need only provide a "detailed justification" to overcome reliance interests); Competitive Enterprise Institute (CEI), Docket No. EPA-HQ-OAR-2021-0257-0398 (correction to an earlier comment by the same commenter, which can be found at Docket No. EPA-HQ-OAR-2021-0257-0140) at 9 ("As for reliance interests, all costly wasteful, or otherwise defective government programs create reliance interests. Usurpations of power do as well. If the creation of reliance interests is enough to legitimize bad or unlawful policies,

Commenters who argued that reliance interests were relevant to EPA's authority to reconsider also offered evidence of reliance interests that had accrued over the five years the ACC program waiver had been in effect, with several commenters providing specific details regarding their reliance on the GHG and ZEV standards. As commenters noted. California's standards are incorporated into plans and regulations aimed at achieving state and federal air pollution goals. These plans can be complex and cannot "change on a dime." 108 According to one commenter "[w]ithout the full Waiver, past decisionmaking was blighted and planned-for reductions to meet Air District goals need to be reassessed. The emission reductions are key to combatting climate change, curbing ozone formation, preventing additional wildlife impacts, and attaining California [air quality goals] and [NAAQS]."109 Revoking a waiver and disrupting existing air quality plans, they argue, also has "far-reaching ripple effects" because "businesses operating in California base their own long-term plans on the State's policies" and, if California cannot reduce emissions from the

anything goes."). Compare to States and Cities at 17-18 (citing their comments on SAFE 1 at 130-31 and citing Ctr. for Sustainable Econ. v. Jewell, 779 F.3d 588, 595 (D.C. Cir. 2015)) (describing reliance interests as "weighty," stating that "[t]he Clean Air Act and long-standing Executive branch policy both place substantial importance on States' interests in implementing the plans and laws they have determined best meet the needs of their States"—plans and laws such as SIPs, which can and do include California standards).

^{108.} Twelve Public Interest Organizations app. 1 at 29.

^{109.} Bay Area Air Quality Management District (BAAQMD), Docket No. EPA-HQ-OAR-2021-0257-0278 at 2.

automobile sector, it will have to "consider requiring further reductions from other sectors of the economy." Additionally, they said that by the time of the SAFE proposal, twelve states had already adopted at least one or both of the California standards under section 177. Several of these states submitted comments attesting to their need for these standards to achieve both greenhouse gas and criteria emission reductions. 112 Like the reliance

^{110.} Twelve Public Interest Organizations app. 1 at 29.

^{111.} States and Cities at 17. With these state adoptions, automanufacturers would then need to meet program requirements in these states.

^{112.} See, e.g., Delaware Department of Natural Resources and Environmental Control (Delaware), Docket No. EPA-HQ-OAR-2021-0257-0109 at 1 ("The GHG program allowed by the waiver is vitally important, as it enables long-term plans and yields critical emission reductions that will contribute significantly to Delaware's ability to attain and maintain the health-based National Ambient Air Quality Standards (NAAQS) for criteria pollutants."); Connecticut Department of Transportation and Connecticut Department of Energy and Environmental Protection (Connecticut), Docket No. EPA-HQ-OAR-2021-0257-0104 at 2 ("These programs enable long-term planning and yield critical emission reductions that are critical to meeting Connecticut's climate goals as well as our statutory obligations to reach attainment with the ozone NAAQS."); Minnesota Pollution Control Agency and Minnesota Department of Transportation (Minnesota), Docket No. EPA-HQ-OAR-2021-0257-0113 at 2 ("The MPCA is in the process of adopting the LEV and ZEV standards in Minnesota as allowed under section 177 of the CAA. These rules are vitally important in helping our state achieve our GHG emission reduction goals and reduce other harmful air pollutants. . . . "); Maine Department of Environmental Protection (Maine), Docket No. EPA-HQ-OAR-2021-0257-0130 at 1, 3 ("While the LEV program was initially created to help attain and maintain the health-based [NAAQS] for criteria pollutants, the

interests of Californian air districts, several of these section 177 states and other opponents of SAFE 1 claim that "reliance interests in State Implementation Plans are particularly acute" because "they set expectations for extended periods of time and for many sectors of the economy, making it challenging (if not impossible) to change them quickly." These commenters note that "planning failures can carry significant consequences, including the imposition of federal plans that limit local flexibility and control, as well as penalties such as loss of highway funds." Some automakers and industry groups

California GHG and ZEV standards will contribute significantly to states' abilities to meet their emission reduction goals. . . . [T]he transportation sector is the largest source of ozone forming pollution in Maine . . . and California's ability to set ZEV standards under the [CAA] is an essential tool for addressing both criteria pollutants and GHGs."); Virginia Department of Environmental Quality (Virginia), Docket No. EPA-HQ-OAR-2021-0257-0112 at 2 ("These standards provide important and necessary reductions in both GHG and criteria pollutant emissions needed to meet state and local air quality goals and address federal CAA requirements.")

^{113.} Twelve Public Interest Organizations app. 1 at 30; Delaware at 3 (explaining that, without the California standards, adopted into Delaware's SIP, the State will not be able to meet air quality goals). These reliance interests, one commenter argued, are another reason to doubt the implicit authority of EPA to reconsider an already granted waiver: "It would be quite surprising, then, for EPA to have implicit authority to upend this multi-actor, multi-step scheme by pulling the rug out from under it after the fact." States and Cities at 16 (citing *Am. Methyl*, 749 F.2d at 840).

^{114.} Twelve Public Interest Organizations app. 1 at 30-31 (citing 42 U.S.C. 7410(c)(1) (establishing triggers for imposition of federal plan), 7509 (outlining sanctions for state planning failures)).

also discussed their reliance interests. ¹¹⁵ For example, the National Coalition for Advanced Transportation,

^{115.} See Ford Motor Company (Ford), Docket No. EPA-HQ-OAR-2021-0257-0028 at 1 ("Ford supports EPA's rescission of its SAFE I action, which withdrew California's waiver for zero emission vehicle (ZEV) mandate and greenhouse gas (GHG) emission standards within California's Advanced Clean Car (ACC) program. Ford does not believe this previous action was appropriate. Ford firmly supports recognition of California's authority to implement ZEV and GHG standards in support of its air quality targets pursuant to its 2012 waiver application. We have relied on California's actions pursuant to the waiver and California's related pronouncements in negotiating and agreeing to the California Framework Agreement, and in the development of our own product and compliance plans. Ultimately, Ford considered EPA's and NHTSA's rationales and California's statements regarding SAFE I and took action in the best interests of the company and of the environment."). See also Tesla, Docket No. EPA-HQ-OAR-2021-0257-0136 at 4 ("Because of the sizeable investments required to develop alternative fuel and advanced technology vehicles, regulatory stability is vital for ensuring the level of manufacturer and investor confidence necessary to facilitate innovation.") and at n.5 (quoting comments from several automakers and auto industry groups about reliance interests on the waiver from the MTE). See also Toyota, Docket No. EPA-HQ-OAR-2021-0381 ("Should EPA reinstate California's waiver, we request it be reinstated as it was originally granted, including the "deemed-to-comply" provision that was so important in establishing One National Program (ONP) over a decade ago. . . . Reinstatement of California's waiver for model years 2021 and 2022 poses significant lead time challenges considering that 2021 model year is well underway, and 2022 model year vehicles are generally already designed, sourced, certified to various regulatory requirements, and ready to begin production. Some manufacturers may have already begun production of 2022 model year vehicles. As a result, a reinstatement of California's waiver by EPA should apply prospectively to model years 2023 and later.").

an industry coalition group, stated "NCAT members have invested billions of dollars with the well-founded expectation that increased demand for electric vehicles would be propelled by California and the section 177 States' continued ability to drive technology innovation and emission reductions."¹¹⁶ EPA also received comment from CARB, by and through the comments of the States and Cities, that provided data on manufacturer compliance.¹¹⁷

According to commenters, these reliance interests were compounded by the considerable passage of time between the granting of the ACC program waiver in 2013 and SAFE 1's withdrawal in 2019. Commenters also remarked that the more than five years that had passed was too long a delay and well beyond the "weeks, not years" sometimes referenced as guidance for reasonableness. SAFE 1, they noted "comes years after the waiver was granted, years after multiple sovereign States adopted California's standards, and years into long-term plans States developed in reliance on anticipated emission reductions from those standards—including, but not

^{116.} NCAT at 13; Rivian as a member of NCAT (Rivian), Docket No. EPA-HQ-OAR-2021-0135.

^{117.} States and Cities at 55-57, including app. D and app. E.

^{118.} *Id.* at 17 (citing Mazaleski v. Treusdell, 562 F.2d 701, 720 (D.C. Cir. 1977)). Twelve Public Interest Organizations app. 1 at 73. In addition, this commenter notes that the time period for seeking judicial review of the ACC program waiver had run long ago and that no one had sought that review (citing Am. Methyl Corp., 749 F.2d at 835); NCAT at 14-15.

limited to, multiple EPA approved State Implementation Plans."¹¹⁹

Other commenters argued that SAFE 1 did not upend reliance interests and was not untimely. They agreed with the SAFE 1 decision that the 2018 Mid-Term Evaluation (MTE), which was agreed to in 2013, prevented any reliance interests from accruing. Although this MTE was for the federal GHG standards for MYs 2022-2025, not the California GHG standards approved under the ACC program waiver, these commenters argued that the two were linked through the "deemed to comply" provision approved in the ACC program waiver, which allowed manufacturers to comply with the California standards by meeting the federal standards. They

^{119.} Twelve Public Interest Organizations app. 1 at 58.

^{120.} America Fuel & Petrochemical Manufacturers, EPA-HQ-OAR-2021-0257-0139 (AFPM) at 26 ("And no reliance interests derive from this decision because one could not reasonably expect that the standards approved in that waiver would remain untouched. As part of the 2013 waiver decision, EPA and CARB committed to a 2018 mid-term evaluation of the federal standards for MYs 2022-2025."); Urban Air at 22; NADA at 6 ("as discussed at length repeatedly in EPA's 2013 CAA preemption waiver rule, a coordinated mid-term evaluation (MTE) involving EPA and NHTSA's MY 2022-2025 rules was expected to be conducted.").

^{121.} AFPM at 26 ("Because California's deemed-to-comply provision linked those standards to compliance with its own state program, any change in federal standards from the mid-term review would have required an equal overhaul of California's emissions program for those future MYs."); Urban Air at 22-23 ("The 2018-reevaluation is relevant because California's deemed-to-comply provision allowed a manufacturer to satisfy state GHG standards

also noted that California separately agreed to a 2016 mid-term evaluation of its own state standards for the same model years. Therefore, they argued, because the initial grant of the waiver was contingent on two subsequent mid-term evaluations, no one could have reasonably believed the ACC program waiver was "set in stone." Additionally, at least one commenter argued that California and other states' purported reliance interests were further undermined because they "have known for years that NHTSA's longstanding position is that state carbon dioxide regulations and zero-emissions vehicle mandates are related to average fuel economy standards and therefore preempted by CAFE" and "could not have reasonably believed that EPA would continue to ignore NHTSA's view of the law in perpetuity. 123

Some commenters also argued that even if reliance interests are relevant, automakers and industry groups have reliance interests of their own affected by CARB's 2018 deemed to comply amendments and the SAFE 1 action itself. One commenter wrote that "CARB tossed automakers' reliance interests out the window when it refused to be bound by the results of the EPA and

simply by complying with federal standards."); NADA at 6 ("[A]s noted above, CA's GHG mandates included both a "deem-to-comply" rule enabling vehicle manufacturers to meet those mandates by complying with applicable federal rules, and a commitment on the part of the state to conduct a mid-term evaluation of its own GHG standards.").

^{122.} AFPM at 26-27; Urban Air at 22; NADA at 6.

^{123.} Urban Air at 23.

NHTSA's Mid-Term Evaluation (MTE)... and refused to honor its 'deemed to comply' pledge to automakers unless they complied with the standards set by the EPA in 2012 and 2017."¹²⁴ Another commenter noted that "[w]hatever 'reliance interests' are disturbed when EPA reverses a waiver grant are no more real, and no more serious for the parties involved, than the reliance interests upended by reversal of a waiver denial."¹²⁵

Some commenters also argued that SAFE 1 was timely, disputing opponents' claims that a "reasonable" amount of time is measured in "weeks, not years." Commenters noted that "courts have not reached consensus on the amount of time that is reasonable." Moreover, one commenter argued that "timeliness depends on reliance interests" and, because those could not have accrued prior to the MTE, the time period at issue is only four months (between the conclusion of the MTE and the reconsideration of the ACC program waiver, starting in 2018). This "short time," the commenter claimed, "lies

^{124.} CEI at 9.

^{125.} AFPM at 27. See also Urban Air at 20-21 ("And under the presumption that 'an agency retains authority to reconsider and correct an earlier decision," the grant of a waiver is as liable to change as the denial of a waiver. No greater reliance interests attach to the grant of a waiver authorizing regulation than to the denial of a waiver preventing regulation, so reliance interests provide no support for California's ratchet argument.").

^{126.} Urban Air at 23-24.

^{127.} Id. at 24. Another commenter disagreed with this accounting of time, stating that "timeliness for reconsidering an

in the acceptable range given the intervening events."¹²⁸ Another commenter argued that, if "time elapsed" is a factor to be considered in the appropriateness of an action, it cuts in favor of SAFE 1, as thirty years passed between EPCA's enactment in 1975 and California's first request for a "waiver implicitly authorizing the State to regulate fuel economy."¹²⁹ Even if the time period at issue was nearly six years between the grant of the ACC program and the final SAFE 1 action, that commenter wrote, such a length of time is not unreasonable, since "[i]f six years locks a policy in place and puts it beyond revision or repeal by the next administration, elections no longer matter."¹³⁰

In addition to reliance interests and timeliness, some commenters claimed that EPA's authority to revoke, if it existed, requires the Agency to have a purpose other than "applying some . . . change in administrative policy." SAFE 1, they argued, did not meet this requirement. Instead, in SAFE 1, EPA "chose to *sua sponte* reconsider its 2013 Waiver Grant for the sole purpose of applying new policy determinations," specifically "NHTSA's views of

adjudication is measured from the date of the agency's decision, not from the date of activity resulting from that decision. *E.g.*, *Am. Methyl*, 749 F.2d at 835 (tethering timeliness to period for appeal of agency decision)." Twelve Public Interest Organizations app. 1 at 38.

^{128.} Urban Air at 23-24.

^{129.} CEI at 8 (calling "time elapsed" a "frivolous objection.").

^{130.} Id.

^{131.} States and Cities at 17 (quoting Chapman v. El Paso Nat. Gas Co., 204 F.2d 46, 53-54 (D.C. Cir. 1953)).

EPCA preemption" and "new interpretations [of section 209(b)(1)(B)] that served only to categorically bar state standards that reduce vehicular GHG emissions." Still, another commenter disagreed, arguing that EPA's reconsideration was an appropriate reevaluation of the legal interpretation and facts upon which the initial waiver determination was based because—"reconsideration determinations do not become 'policy' decisions simply because they address substantive errors." ¹³³

EPA also received comment on whether EPA's actions were inappropriate because the Agency failed to satisfy the "requirements of reasoned decision-making." Some commenters noted that EPA had taken the position in SAFE 1 that "reducing criteria pollution is of overriding importance" yet failed to "consider[] the criteria-pollution and SIP consequences of its Waiver Withdrawal and Section 177 Determination." Similarly, EPA received comments claiming that the decision to apply a new

^{132.} *Id.* at 8, 19 ("No statute compelled EPA to reconsider the 2013 waiver at all, let alone to apply new policies to that long-settled decision rather than to new waiver requests."); Twelve Public Interest Organizations app. 1 at 35 ("EPA relied exclusively on its purported discretion to reinterpret Section 209(b)(1)(B) of the Clean Air Act... and its purported discretion to consider factors not enumerated in Section 209(b)(1)."). *See also* SCAQMD at 3 ("Because the 2013 waiver decision was not pending judicial review in 2019 and was a long-closed matter, the EPA could not rightfully reopen its adjudication.").

^{133.} Urban Air at 24 (citing *Civil Aeronautics Bd.* v. *Delta Air Lines*, 367 US 316, 321 (1961)).

^{134.} States and Cities at 8-9, 12.

approach to the ACC program waiver section 209(b) (1)(B) was both unnecessary and unjustified because, as EPA acknowledged in SAFE 1, the Agency has consistently posited that section 209(b)(1)(B) calls for determining whether the State needs its own regulatory program, separate from that of the federal government, not whether the State needs each specific standard or package of standards for which it seeks a waiver. ¹³⁵ One of these commenters pointed out that EPA also acknowledged that the phrase "such State standards" could reasonably remain the program-level interpretation (EPA's traditional interpretation) yet the Agency chose to adopt a new interpretation and apply it to the more than five-year old ACC program waiver, impacting expectations and reliance interests.

The Agency also received comments on whether NHTSA's finding of preemption under EPCA in the joint action granted EPA authority to reconsider the ACC program waiver. Commenters argued that NHTSA is charged with interpreting and implementing EPCA and that its finding "that Congress prohibited California's standards" in the same action cannot be ignored. ¹³⁶ Still other commenters pointed to the language of section 209(b)(1) itself, where only three criteria are provided by which EPA can deny a waiver. As such, they argued, EPA cannot have broad, implicit authority to revoke a waiver on entirely different grounds than by which it may deny

^{135.} Id. at 22.

^{136.} See, e.g., CEI at 11.

a waiver.¹³⁷ The commenters also argued that the joint context of the action did not grant the Agency special authority to reconsider, explaining that "[w]hat Congress directed EPA to consider when it wrote Section 209(b)(1) does not change depending on whether EPA acts alone or with another agency."¹³⁸ Some commenters also pointedly noted that SAFE 1's distinction between single-agency and joint actions is arbitrary and capricious and therefore not a valid basis for reconsideration because EPA stated it "does not intend in future waiver proceedings concerning submissions of California programs in other subject areas to consider factors outside the statutory criteria in section 209(b)(1)(A)-(C),"¹³⁹ and because NHTSA and EPA now consider SAFE 1 as "two severable actions."¹⁴⁰

B. Analysis: EPA Inappropriately Exercised Its Limited Authority To Reconsider

EPA finds it does have authority to reconsider waivers, although its reconsideration of previously-granted waivers is limited and circumscribed. In the context of adjudicatory decisions (as contrasted to rulemakings), administrative

^{137.} States and Cities at 16-17.

^{138.} $\it Id.$ at 20. $\it See~also$ Twelve Public Interest Organizations app. 164-65.

^{139.} Northeast States for Coordinated Air Use Management (NESCAUM), Docket No. EPA-HQ-OAR-2021-0257-0126 at 3; Twelve Public Interest Organizations app. 164-65; States and Cities at 20.

^{140.} SCAQMD at 7 (citing 86 FR at 22439 n.40).

law principles and case law support limited reconsideration authority for waiver proceedings. For example, in Ivy Sports Med., LLC v. Burwell, 767 F.3d 81, 86, 93 (D.C. Cir. 2014), the D.C. Circuit noted that where a statute "does not contain an express provision granting [the agency] authority to reconsider," "administrative agencies are assumed to possess at least some inherent authority to revisit prior decisions, at least if done in a timely fashion," noting the baseline limitations of such inherent authority. And in Chapman v. El Paso Nat. Gas Co., 204 F.2d 46, 53-54 (D.C. Cir. 1953), the D.C. Circuit made clear that once concluded, an adjudicatory decision granting a right "may not be repudiated for the sole purpose of applying some quirk or change in administrative policy."141 These precedents suggest that, while agencies do generally possess some inherent authority to reconsider previous adjudicatory decisions, that authority is limited in scope.

Section 209 does not provide EPA with express authority to reconsider and withdraw a waiver previously granted to California. EPA's authority thus stems from its

^{141.} See also Am. Methyl, 749 F.2d 826, 835 (D.C. Cir. 1984) ("We have held that agencies have an inherent power to correct their mistakes by reconsidering their decisions within the period available for taking an appeal."); Mazaleski v. Treusdell, 562 F.2d 701, 720 (D.C. Cir. 1977) ("We have many times held that an agency has the inherent power to reconsider and change a decision if it does so within a reasonable period of time.") (quoting Gratehouse v. United States, 512 F.2d 1104, 1109 (Ct. Cl. 1975)); Albertson v. FCC, 182 F.2d 397, 399 (D.C. Cir. 1950) ("in the absence of any specific limitation," reconsideration available "within the period for taking an appeal"). See generally Daniel Bress, Note, Administrative Reconsideration, 91 VA. L. REV. 1737 (2005).

inherent reconsideration authority. The 1967 legislative history provides some indication of congressional intent to preserve some implied authority for EPA to reconsider previous waiver decisions, but also to place limitations on it. This legislative history explains: "[i]mplicit in this provision is the right of the [Administrator] to withdraw the waiver at any time [if] after notice and an opportunity for public hearing he finds that the State of California no longer complies with the conditions of the waiver."142 Thus, from the earliest days of the program it has been understood that any withdrawal of a waiver should be tied to the statutory criteria and California's compliance with them. This legislative history must be taken into account along with Congress's intent expressed in the 1977 legislative history, which, as discussed previously, sought to ensure deference to California and to strengthen that state's role in driving emissions-reducing technological innovation. Congress was also mindful to ensure the ability of other states to adopt California's standards. 143 Ultimately, EPA concludes it has authority to reconsider previously-granted waivers, but that this authority may only be exercised sparingly. As discussed below, there are several considerations that support narrow authority to reconsider waiver grants.

First and most important, EPA believes its inherent authority to reconsider a waiver decision is constrained by the three waiver criteria that must be considered before granting or denying a waiver request under section

^{142.} S. Rep. No. 90-403, at 34 (1967).

^{143.} See supra Section III.B.

209(b). It would be inappropriate and inconsistent with congressional intent for EPA to reconsider and withdraw a waiver on a ground outside the limited scope of those which Congress specified for EPA to consider when reviewing a waiver in the first place. In the few instances where the Agency reconsidered prior waiver decisions prior to SAFE 1, EPA focused its review on the section 209(b) statutory waiver criteria.

A circumscribed approach to reconsideration of waivers is consistent with the deference to California's policy judgment that Congress built into the waiver process. 146 Congress explicitly required that EPA "shall" grant the waiver unless one of three limited criteria are met. The use of the word "shall" (versus "may") was heavily debated by the enacting Congress, with the successful proponents of "shall" explaining that such language would "guarantee" that California could regulate with the burden placed on EPA to demonstrate

^{144.} See MEMA I, 627 F.2d at 1115 (noting that section 209(b) creates "a narrowly circumscribed proceeding requiring no broad policy judgments").

^{145.} EPA initiated reconsideration of certain motor-cycle standards, under the third waiver prong, section 209(b)(1)(C), in order to "vacate that portion of the waiver previously granted under section 209(b)." 47 FR 7306, 7309 (February 18, 1982). EPA affirmed the grant of the waiver in the absence of "findings necessary to revoke California's waiver of Federal preemption for its motorcycle fill-pipe and fuel tank opening regulations." *Id.* at 7310.

^{146.} *See MEMA I*, 627 F.2d at 1124-25 (describing Congress's intent to defer to California's judgments regarding its motor vehicle program).

why California should not be allowed to go beyond federal limitations. ¹⁴⁷ Congress's legislative enactments since its creation of the waiver program—including adding section 177 to allow other states to adopt California's standards in 1977 and section 209(e)(2)(A) to create parallel deference for nonroad engines and vehicles in 1990—reinforce the important role it envisioned for, and deference it afforded to, California. ¹⁴⁸

In SAFE 1, EPA argued instead that deference to California was not merited where the Agency was interpreting its "own statute." But in Title II of the Clean Air Act, Congress envisioned two standards—California and Federal. Congress recognized California's early attempts to address motor vehicle emissions intended to address its extraordinary environmental conditions as well as being a laboratory for motor vehicle emissions control. Congress called for EPA deference to California

^{147.} H.R. Rep. No 90-728 ("Are we now to tell California that we don't quite trust her to run her own program, that big government should do it instead?").

^{148. 40} FR 23104; 58 FR 4166.

^{149. 84} FR at 51344 n.268.

^{150.} Motor vehicles are "either 'federal cars' designed to meet the EPA's standards or 'California cars' designed to meet California's standards." *Engine Mfrs. Ass'n* v. *EPA*, 88 F.3d 1075, 1079-80, 1088 (D.C. Cir. 1996) ("Rather than being faced with 51 different standards, as they had feared, or with only one, as they had sought, manufacturers must cope with two regulatory standards.").

^{151.} See, e.g., S. Rep. No. 403, 90th Cong., 1st Sess. 33 (1967) (The waiver of preemption is for California's "unique problems and

in implementing section 209(b) by not only limiting EPA review of California waiver requests to three specific criteria but also instructing that EPA is "to afford California the broadest possible discretion in selecting the best means to protect the health of its citizens and the public welfare." 152 Similarly, "[t]he Administrator, . . . is not to overturn California's judgment lightly. Nor is he to substitute his judgment for that of the State."153 Additionally, the D.C. Circuit has explained that "Congress consciously chose to permit California to blaze its own trail with a minimum of federal oversight" and "[t]he statute does not provide for any probing substantive review of the California standards by federal officials."154 Further, "[t]here is no indication in either the statute or the legislative history that . . . the Administrator is supposed to determine whether California's standards are in fact sagacious and beneficial."155 Thus, early in the waiver program's history, EPA explained the deference that Congress intended for the Agency's review of waiver requests by noting that it would feel constrained to

pioneering efforts."); 113 Cong. Rec. 30950, 32478 ("[T]he State will act as a testing agent for various types of controls and the country as a whole will be the beneficiary of this research.") (Statement of Sen. Murphy).

^{152.} H.R. Rep. No. 95-294, at 301-02 (1977).

^{153.} H.R. Rep. No. 95-294, at 302 (1977), reprinted in 1977 U.S.C.C.A.N. at 1381)).

^{154.} Ford Motor Co. v. EPA, 606 F.3d 1293, 1297, 1300 (D.C. Cir. 1979).

^{155.} Id. at 1302.

approve a California approach to a problem that the EPA Administrator might not feel able to adopt at the federal level as a regulator. EPA explained that the balancing of risks and costs against potential benefits from reduced emissions is a central policy decision for any regulatory agency and substantial deference should be provided to California's judgement on such matters.¹⁵⁶

In addition, limiting reconsideration of waivers undergirds Congress' intent that California would be a laboratory for the country driving emissions-reducing technological innovation when it created the program in the first place. As the D.C. Circuit explained in MEMA I: "The history of congressional consideration of the California waiver provision, from its original enactment up through 1977, indicates that Congress intended the State to continue and expand its pioneering efforts at adopting and enforcing motor vehicle emission standards different from and in large measure more advanced than the corresponding federal program; in short, to act as a kind of laboratory for innovation." Indeed, broad authority to reconsider waiver grants could undermine the very structure that Congress built in Title II. Specifically, while EPA does not consider section 177 when reviewing waiver requests under section 209, Congress built a structure wherein EPA must grant California a waiver under section 209 unless one of the three statutory criteria are met, and then other states may adopt California's standards under section 177 as part of their overall air quality programs.

^{156. 40} FR at 23104.

^{157.} MEMA I, 627 F.2d at 110-11.

Limited inherent authority to reconsider previouslygranted waivers as described in this action is important to the success of Congress's structure.

Finally, even the sentence in the legislative history that suggests EPA has inherent reconsideration authority in the first place, and which SAFE 1 relied on for its assertion of inherent reconsideration authority, lends weight to the view that this authority is limited. According to the Senate report from the 1967 CAA amendments, the Administrator has "the right . . . to withdraw the waiver at any time [if] after notice and an opportunity for public hearing he finds that the State of California no longer complies with the conditions of the waiver." That specific circumstance—where California does not comply with the conditions of a waiver—should not be expanded to include a gaping hole for discretionary administrative policy changes.

Given all of the above considerations, several principles emerge. EPA's authority to reconsider a grant of a waiver, which is an adjudicatory action by the Administrator, is not open-ended. Any reconsideration is constrained to the criteria that Congress set out in section 209(b). Even within those statutory criteria, considering all of the factors that weigh in favor of a narrow interpretation of the Agency's authority and the importance of not disrupting Congress's scheme, EPA believes reconsideration is limited to situations where the Agency has made a clerical or factual error or mistake, or where information shows

^{158.} S. Rep. No. 90-403, at 34 (1967).

that factual circumstances or conditions related to the waiver criteria evaluated when the waiver was granted have changed so significantly that the propriety of the waiver grant is called into doubt.

Even if the bases for EPA's reconsideration did satisfy one of the foregoing conditions such that reconsideration may be appropriate, during that reconsideration EPA believes it should consider the passage of time and reliance interests. In the context of CAA waiver grants in general, and the 2013 ACC program waiver grant in particular, California is relying on its standards to meet short- and long-term emission reduction goals. ¹⁵⁹ In addition, by the time the SAFE proposal was published, twelve states had already adopted at least one or both of the GHG and ZEV standards. ¹⁶⁰ Several of these states incorporated these adopted standards into their SIPs. ¹⁶¹ Several automakers

^{159.} States and Cities at 17-18.

^{160.} Id. at 17.

^{161.} *Id.* at 10; Wisconsin Department of Natural Resources (Wisconsin), Docket No. EPA-HQ-OAR-2021-0257-0095 at 1 ("These standards provide important and necessary reductions in both GHG and criteria pollutant emissions needed to meet state and local air quality goals and address federal CAA requirements."); Connecticut at 2 ("These programs enable long-term planning and yield critical emission reductions that are critical to meeting Connecticut's climate goals as well as our statutory obligations to reach attainment with the ozone NAAQs."); Delaware 2 ("Delaware adopted the California LEV regulation and incorporated the LEV and GHG standards into the State Implementation Plan.... Delaware will not meet air quality goals without more protective vehicle emission standards."); Maine at 1 ("[T]he LEV program was initially created to help attain and

and industry groups have also indicated reliance on these standards. 162

Reconsideration thus must carefully consider the factors noted and should not be undertaken where immense degrees of uncertainty are introduced in settled expectations of California, other states, and regulated industry or to allow for the continual questioning of EPA's decisions, thus impairing needed finality. Such reconsideration could frustrate congressional intent in designing the waiver program and ultimately discourage reliance by the recipient of EPA's waiver decision (CARB), states that may have adopted CARB's regulations under the terms of section 177 (and are permitted to enforce the regulations once EPA grants a waiver to California) as well as the regulated industry.

We now turn to whether the reconsideration in SAFE 1 was a proper exercise of EPA's inherent reconsideration authority. As an initial matter, SAFE 1 did not assert that any clerical or factual error or mistake was made in the 2013 ACC program waiver. Nor did SAFE 1 point to any evidence showing that factual circumstances or conditions related to the waiver criteria evaluated when the waiver was granted have changed so significantly that

maintain the health-based National Ambient Air Quality Standards (NAAQS)... The California ZEV and GHG programs enable long-term planning for both the states and the regulated community and have been drivers of technological change across the industry.").

 $^{162.\} E.g.$, Ford at 1; Tesla at n.5, 4; Rivian (as a member of NCAT) at 13-14.

the propriety of the waiver grant is called into doubt. For example, SAFE 1 did not assert that California was not complying with the terms of the waiver. Instead, SAFE 1's reconsideration was premised on retroactive application of discretionary policy changes. Therefore, EPA believes it did not appropriately exercise its inherent authority in SAFE 1 to reconsider the prior ACC program waiver. Upon reconsideration, and as further shown in Sections V and VI, EPA now believes that SAFE 1 amounted to an improper exercise of the Agency's limited inherent authority to reconsider. ¹⁶³

SAFE 1 gave two primary reasons for withdrawing the $2013\,\mathrm{ACC}$ program waiver. Neither was an appropriate

^{163.} EPA acknowledges that, in the SAFE 1 proceedings, it had noted that at the time of proposal that CARB had given notice that it was considering amending its "deemed to comply" provision and that by the time of SAFE 1, California had entered into agreements with several automobile manufacturers to accept less stringent standards than the California program or the Federal standards as promulgated in 2012. As noted in SAFE 1, EPA believed that neither of these matters were necessary for EPA's action in SAFE 1, but that they provided further support for the action. 84 FR at 51334 n.230. By this action, EPA finds that neither of these matters amounted to a change in circumstances or conditions associated with the three waiver criteria and EPA's evaluation of the criteria in the ACC program waiver. EPA did not predicate its ACC program waiver on CARB's deemed-to-comply provision or any changes to the deemed-to-comply provision. (EPA does not take a position as to whether that provision has changed in its purpose as a result of CARB's 2018 amendment). Further, to the extent CARB utilized a deemed-to-comply provision or uses non-regulatory mechanisms to achieve its air quality objectives, this had no bearing on EPA's assessment of whether CARB has a need for its standards under the second waiver prong at the time of SAFE 1 or now.

basis for reconsideration. First, SAFE 1 premised the revocation on its interpretation of the second waiver prong, section 209(b)(1)(B), that called for the Agency's scrutiny of specific standards under the waiver rather than California's program as a whole. As explained in detail in Section V of this final action, that statutory interpretation is flawed, and EPA does not believe a new statutory interpretation should be the basis of reconsidering the grant of a waiver.

SAFE 1 premised the withdrawal of the ACC program waiver under section 209(b)(1)(B) on the perceived lack of record support on the causal link between GHG emission standards and air quality conditions in California. He the underlying record from the ACC program waiver, and the record of SAFE 1, have shown that CARB's ZEV sales mandate and GHG emission standards are designed to address California's serious air quality problems, including both its NAAQS pollutants and a variety of climate impacts from GHG emissions. As discussed in greater detail in Section V, EPA has since at least 2009 recognized that greenhouse gas pollution exacerbates criteria pollution, and climate change impacts on California's air quality conditions (e.g., heat

^{164. &}quot;California's *approach* in its ACC program waiver request differed from the state's approach in its waiver request for MY 2011 and subsequent heavy-duty tractor-trailer GHG standards, where California quantified NO X emissions reductions attributed to GHG standards and explained that they would contribute to PM and ozone NAAQS attainment." 84 FR at 51337 n.252 (citing 79 FR at 46256, 46257 n.15, 46261, 46262 n.75).

exacerbation of ozone). The ACC program was especially designed to address both criteria and GHG pollution,

165. The first HD GHG emissions standard waiver related to certain new 2011 and subsequent model year tractor-trailers. 79 FR 46256 (August 7, 2014). CARB projected, for example, "reductions in NO X emissions of 3.1 tons per day in 2014 and one ton per day in 2020" in California. *Id.* at 46261. The second HD GHG emissions standard waiver related to CARB's "Phase I" regulation for 2014 and subsequent model year tractor-trailers. 81 FR 95982 (December 29, 2016).

CARB also noted the scientific findings since EPA's 2009 GHG waiver including the report titled "Our Changing Climate 2012 Vulnerability & Adaptation to the Increasing Risks from Climate Change in California." The summary report highlights new insights for the energy, water, agriculture, public health, coastal, transportation, and ecological resource sectors that are vital to California residents and businesses. The study also predicts that peak concentrations of dangerous airborne particles will increase in the San Joaquin Valley because of climate change on wind patterns. This study provides further evidence of what is known as the "climate penalty," where rising temperatures increase groundlevel ozone and health-damaging particles, despite the reductions achieved by successful programs targeting smog-forming emissions from cars, trucks, and industrial sources. Id. at 8-9. See also "The Impacts of Climate Change on Human Health in the United States: A Scientific Assessment" Chapter 3 Air Quality Impacts—Key Finding ("Climate change will make it harder for any given regulatory approach to reduce ground-level ozone pollution in the future as meteorological conditions become increasingly conducive to forming ozone over most of the United States. Unless offset by additional emissions reductions, these climate-driven increases in ozone will cause premature deaths, hospital visits, lost school days, and acute respiratory symptoms.") at https://health2016.globalchange.gov/airquality-impacts; Chapter 13: Air Quality, Fourth National Climate Assessment at https://nca2018.globalchange.gov/chapter/13/.

including the effects of GHG pollution on criteria pollution in California. 166 As also further discussed in Section V, in SAFE 1 the Agency dismissed the criteria pollutant benefits of California's ZEV sales mandate requirements based on a snippet from the 2012 waiver request, taken out of context. 167 This was also remarkable considering EPA's prior waivers for ZEV sales mandate requirements that demonstrated criteria pollutant emissions reduction benefits. 168 The record also includes information that demonstrates that a withdrawal of the waiver for the GHG emission standards and ZEV sales mandate (and leaving the Federal GHG standards at the 2020 levels as proposed in SAFE) would increase NOx emissions in the South Coast air basin alone by 1.24 tons per day. 169 In sum, EPA opted to elide the available ample technical support from the ACC program waiver proceedings.

^{166. 2012} Waiver Request at 1, 9-11, 15-17 ("[A]s detailed below, the ACC program will result in reductions of both criteria pollutants and GHG emissions that, in the aggregate, are more protective than the federal standards that exist."). 78 FR at 2122 ([T]he ACC program will result in reductions of both criteria pollutants and GHG emissions.").

^{167. 84} FR at 51337 (quoting CARB's statement that "[t]here is no criteria emissions benefit from including the ZEV proposal in terms of vehicle (tank-to-wheel or TTW) emissions."). As explained in more detail below, this statement merely reflected how CARB attributed pollution reductions between its different standards and compliance mandates, not the reality of how those standards and mandates actually drive pollution reductions.

^{168. 58} FR 4156. 71 FR 78190 (December 28, 2006); 75 FR 11878 (March 12, 2010) and 76 FR 61095 (October 3, 2011).

^{169.} States and Cities at 10.

EPA's factual predicates in SAFE 1—that there was no criteria pollutant benefit of the GHG standards and ZEV sales mandate—for reconsideration based on the second waiver prong were simply inaccurate and inappropriate. Reconsideration was thus improper on this basis because there were no factual errors in the ACC program waiver and EPA should not be exercising authority to reconsider prior valid waivers that present no factual errors based on different statutory interpretations.

Second, SAFE 1 premised its revocation on NHTSA's finding of preemption under EPCA. This, too, was an inappropriate ground for reconsideration. As earlier noted, EPA believes its inherent authority to reconsider a waiver decision is constrained by the three waiver criteria that must be considered before granting or denying a waiver request under section 209(b). Preemption under EPCA is not one of these criteria and was not considered in CARB's ACC program waiver request or in EPA's granting of that waiver. In fact, in its waiver grant, the Agency expressly found that consideration of preemption under EPCA would be inappropriate and unnecessary. In SAFE 1, the Agency did not premise its consideration of preemption under EPCA on any of the three statutory criteria. Therefore, EPA believes that SAFE 1 was not a proper exercise of the authority to reconsider on this basis, and any subsequent action in SAFE 1 to withdraw the ACC program waiver was inappropriate.

Although SAFE 1 was an inappropriate exercise of inherent authority given that the Agency did not correct a factual error and there was no change in factual

circumstances so significant that the propriety of the waiver would be called into doubt, it is nevertheless relevant to note that SAFE 1 did not give appropriate consideration to the passage of time and the reliance interests that had developed between the granting and the revocation of the ACC program waiver. Several automakers and industry groups have also indicated reliance on these standards, as previously discussed. California and section 177 states were, by the time of the reconsideration, into the long-term plans they had developed relying on the ACC program waiver standards. California and other states rely on waivers

^{170.} E.g., Ford at 1; Tesla at n.5, 4; Rivian (as a member of NCAT) at 13-14. EPA notes that it received limited comment on whether reliance interests had formed since the issuance of SAFE 1 but nothing to demonstrate error in the findings regarding section 209(b)(1)(C) made within the ACC program waiver. See Toyota, Docket No. EPA-HQ-OAR-2021-0381 ("Reinstatement of California's waiver for model years 2021 and 2022 poses significant lead time challenges considering that 2021 model year is well underway, and 2022 model year vehicles are generally already designed, sourced, certified to various regulatory requirements, and ready to begin production."). Further, as discussed elsewhere, the short passage of time since the promulgation of SAFE 1 and ongoing litigation over that action has, as automakers have noted in that briefing, prevented automakers from relying on the waiver revocation. See also Twelve Public Interest Organizations at 11 (noting filings by automakers suggesting lack of reliance on the waiver withdrawal).

^{171.} E.g., States and Cities at 17 (the length between the waiver grant and reconsideration was too long "by any measure."); Twelve Public Interest Organizations at app. 36. EPA acknowledges the commenter who argued that "timeliness depends on reliance interests" and, because the standards were not final before the MTE, the time period at issue is the four months between the MTE and

that EPA has approved to meet short- and long-term emission reduction goals.¹⁷² In addition, by the time

the SAFE 1 proposal. Urban Air at 24. EPA also received comment that disagreed with this accounting of time stating that timeliness for reconsidering an adjudication is measured from the date of the agency's decision, not from the date of activity resulting from that decision. E.g., Am. Methyl, 749 F.2d at 835 (tethering timeliness to period for appeal of agency decision)." Twelve Public Interest Organizations app. 1 at 38. EPA believes it is not necessary to resolve the permissible amount of time, or the existence or lack of a bright line, that may pass before reconsideration of its prior adjudication is no longer appropriate. However, EPA did not "condition" its ACC program waiver on any subsequent actions, including the MTE, which explicitly applied to the federal standards. See 78 FR at 2137. EPA expects its waiver adjudications to be final and that appropriate reliance may flow to affected parties. Moreover, in this instance EPA did not make any final determination regarding the third waiver prong at section 209(b)(1)(C). EPA notes that it has administered the California waiver program for a number of decades and acknowledges that emission standards continue to evolve at the California and the federal levels. This evolution in the standards has rested on regulatory certainty and the enforceability of CARB's emission standards once a waiver has been issued by EPA under section 209(b) of the CAA. As for the inclusion of the deemed-tocomply provision in the California standards, California provided documentation demonstrating that the deemed-to-comply provision was reliant upon the federal standards having a certain level of stringency, a fact that EPA had recognized. See States and Cities at 18-19 n. 14, 57-60. EPA found that the California standards were feasible even without the deemed-to-comply provision, 78 FR at 2138, making it irrelevant to the waiver grant. California's own actions with respect to its standards, such as its independent review of the ACC program, cannot disturb California's or other state's reliance on the federal waiver.

172. States and Cities at 17-18.

the SAFE proposal was published, twelve states had already adopted at least one or both of the GHG and ZEV standards.¹⁷³ Several of these states incorporated these adopted standards into their SIPs.¹⁷⁴

SAFE 1 barely mentioned these reliance interests, explaining only that the Agency "will consider whether and how to address SIP implications of this action, to the extent that they exist, in separate actions; EPA believes that it is not necessary to resolve those implications in the course of this action." EPA now believes that, when exercising its inherent authority to reconsider the 2013 waiver decision, it was inappropriate to ignore these possible reliance interests and to "resolve" any potential

^{173.} Id. at 17.

^{174.} Id. at 10; Wisconsin Department of Natural Resources (Wisconsin), Docket No. EPA-HQ-OAR-2021-0257-0095 at 1 ("These standards provide important and necessary reductions in both GHG and criteria pollutant emissions needed to meet state and local air quality goals and address federal CAA requirements."); Connecticut at 2 ("These programs enable long-term planning and yield critical emission reductions that are critical to meeting Connecticut's climate goals as well as our statutory obligations to reach attainment with the ozone NAAQs."); Delaware 2 ("Delaware adopted the California LEV regulation and incorporated the LEV and GHG standards into the State Implementation Plan.... Delaware will not meet air quality goals without more protective vehicle emission standards."); Maine at 1 ("[T]he LEV program was initially created to help attain and maintain the health-based National Ambient Air Quality Standards (NAAQS)... The California ZEV and GHG programs enable longterm planning for both the states and the regulated community and have been drivers of technological change across the industry.").

^{175.} Id. at 51324 n.167.

implications at a later time. In the SAFE 1 context, while it was not necessary to resolve the status of every SIP, it was inappropriate to not even consider the reliance interests raised by the adoption of California standards by section 177 states (including, but not limited to, their adoption into SIPs). EPA has consistently recognized the importance of long-term planning in the attainment and maintenance of NAAQS. ¹⁷⁶ Given the long-term nature of these plans, it is "challenging (if not impossible) to change them quickly," and any changes in one part of a SIP can affect multiple sectors of the economy. ¹⁷⁷

^{176.} EPA is responsible for approving SIPs and SIP amendments, which span years. *See, e.g.,* 82 FR 42233 (September 7, 2017) (approval of Maine's SIP revision including updates to be consistent with California's updated LEV program); 80 FR 13768 (March 17, 2015) (approval of Connecticut's SIP revision, including the adoption of elements of California's LEV program). For example, states with areas that achieve attainment for any air pollutant must submit for EPA approval a revised SIP that sets out the State's plan for maintaining attainment for at least ten years after the redesignation. At the end of that ten-year period, the State must submit another ten-year maintenance plan to EPA for approval. 42 U.S.C. 7505a.

^{177.} Twelve Public Interest Organizations app. 1 at 29, 30. Several states also commented, during this reconsideration, that they rely on the California GHG standards and ZEV sales mandate to reach their own state emission reduction goals. *E.g.*, Connecticut at 2 ("Reducing GHG emissions from the transportation sector is required to achieve Connecticut's economy-wide targets of at least 45 percent below 2001 levels by 2030 and 80 percent below 2001 levels by 2050, as required by the 2008 Global Warming Solutions Act (GWSA) and the 2018 Act Concerning Climate Change Planning and Resiliency."); Minnesota at 2 ("[California's standards] are vitally important in helping our state achieve our GHG emission reduction goals and

As noted above, EPA also received other comments regarding reliance interests, including those noting that the midterm evaluation (MTE) was an indication that the technological feasibility of the GHG emission standards was not a settled matter and hence no certainty or reliance could accrue. EPA, however, did not "condition" its ACC program waiver on any subsequent actions, including the MTE.¹⁷⁸ EPA expects its waiver adjudications to be final and that appropriate reliance may flow to affected parties. Moreover, in this instance EPA did not make any final determination regarding the third waiver prong at section 209(b)(1)(C). EPA notes that it has administered the California waiver program for a number of decades and acknowledges that emission standards continue to evolve at the California and the federal levels. This evolution in the standards has rested on regulatory certainty and the enforceability of CARB's emission standards once a waiver has been issued by EPA under section 209(b) of the CAA.

EPA's historic practice of properly affording broad discretion to California has meant that in almost fifty

reduce other harmful air pollutants, especially in communities of color and lower-income communities, which are disproportionately impacted by vehicle pollution. The MPCA found that these rules are needed to address GHG emissions in our state and take steps towards achieving Minnesota's statutory Next Generation Energy Act GHG reduction goals. On May 7, 2021, an independent Administrative Law Judge affirmed the MPCA findings."); Maine at 1 n.3 ("Maine statute at 38 M.R.S 576-A establishes tiered GHG emission reduction requirements culminating in gross annual reductions of at least 80% from 1990 baseline levels.").

^{178.} See 78 FR at 2137.

years of administering the California waiver program the Agency had never withdrawn any waiver prior to SAFE 1. And while SAFE 1 cited prior reconsideration actions as support for the Agency's authority to reconsider prior waiver decisions, as previously noted, EPA has historically limited reconsideration of prior waived standards to statutory criteria and most important, none of these prior reconsideration actions resulted in a revocation. ¹⁷⁹ As further shown in Sections V and VI, SAFE 1 was the result of a "probing substantive review of the California standards," with the Agency substituting its own judgment for California's contrary to both congressional exhortation of deference to California and the Agency's review practice.

This present reconsideration is an appropriate exercise of the Agency's reconsideration authority. It is not at all clear that the reasons for limiting reconsideration of waiver grants apply to the same degree to reconsideration of waiver denials and withdrawals. However, EPA need not resolve the question in this action, because this action falls well within the bounds of even the limited authority this action concludes the Agency possesses for reconsideration of waiver grants. First, this action corrects factual errors made in the SAFE 1 waiver withdrawal. Specifically, even under SAFE 1's flawed interpretation of section 209(b)(1)(B), SAFE 1 ignored facts demonstrating that California does need the specific standards at issue to meet

^{179.} See, e.g., 43 FR at 7310 (affirming the grant of the waiver in the absence of "findings necessary to revoke California's waiver of Federal preemption for its motorcycle fill-pipe and fuel tank opening regulations.").

compelling and extraordinary conditions. Second, in this reconsideration EPA properly constrains its analysis to whether SAFE 1 made one of the three statutory findings necessary to deny a waiver. Third, this reconsideration is timely with respect to the finalization of SAFE 1 and limited, if any, reliance interests have developed as a result of SAFE 1 (which has been subject to judicial review since its promulgation).

C. Conclusion

In SAFE 1, EPA inappropriately exercised its limited inherent authority to reconsider the ACC program waiver for several reasons. EPA believes its exercise of reconsideration authority to reinterpret the language of section 209(b)(1)(B) was not taken to correct any factual or clerical error or based upon factual circumstances or conditions related to the waiver criteria evaluated when the waiver was granted that have changed so significantly that the propriety of the waiver grant is called into doubt. Rather, as discussed in detail in Section V, it was based upon a flawed statutory interpretation and a misapplication of the facts under that interpretation. Likewise, EPA's decision to reconsider the ACC program waiver based on NHTSA's rulemaking within SAFE 1, which raised issues beyond the statutory waiver criteria, was inappropriate. For these reasons EPA now believes it is appropriate to rescind its actions within SAFE 1.

V. The SAFE 1 Interpretation of Section 209(b)(1)(B) was Inappropriate and, in any Event, California met its Requirements

Even if SAFE 1's reconsideration of the 2013 program waiver grant was appropriate, EPA concludes for two independent reasons that its waiver withdrawal in SAFE 1 based upon its new statutory interpretation was flawed. First, EPA concludes that the SAFE 1 interpretation of the second waiver prong was not an appropriate reading of that second waiver prong, section 209(b)(1)(B). It bears noting that the traditional interpretation is, at least, the better interpretation. Informed by but separate from the factual analysis discussed next, the Agency finds that the new interpretation set out in SAFE 1 was inconsistent with congressional intent and contrary to the purpose of section 209(b). Under the traditional interpretation of the second waiver prong, California's need for its own motor vehicle program, including its GHG emission standards and ZEV sales mandate, to meet compelling and extraordinary conditions is clear and the waiver should not have been withdrawn.

Second, even if the interpretation in SAFE 1 were appropriate, EPA concludes that SAFE 1 incorrectly found that California did not have a need for its specific standards. EPA has evaluated California's need for both requirements by applying both the traditional and the SAFE 1 interpretations of section 209(b)(1)(B). In doing so, EPA reviewed the record from the ACC program waiver proceedings, including CARB's ACC program waiver request and supporting documents, as well as the comments received as part of the SAFE 1 proceeding and

the comments received under the present reconsideration of SAFE 1.180 The record review focused on salient pronouncements and findings in the ACC program waiver decision, such as the relationship of both criteria and GHG pollutants and the impacts of climate change on California's serious air quality conditions. For example, the effects of climate change and the heat exacerbation of tropospheric ozone is well established. California's ACC program is established, in part, to address this. California's program, including its GHG emission standards, is also designed to address upstream criteria emission pollutants. The review did so primarily because SAFE 1 premised the withdrawal of the GHG standards at issue on the lack of a causal link between GHG standards and air quality conditions in California. The review included EPA's prior findings regarding heat exacerbation of ozone, a serious air quality issue recognized by EPA as presenting compelling and extraordinary conditions under the second waiver prong.

^{180.} EPA notes that it reviewed the factual record within the ACC program waiver proceeding and finds there was no factual error in its evaluation of whether CARB's standards satisfied the second waiver prong. EPA also notes, merely as confirming the finding it made at the time of the ACC program waiver but not for purposes of making a new factual finding from that made at the time of the ACC program waiver decision, that the record and information contained in the SAFE 1 proceeding as well as the record and information contained in the Agency's reconsideration of SAFE 1 (including late comments submitted during the SAFE 1 proceeding and, in some cases, resubmitted during the Agency's reconsideration of SAFE 1) at each point in time clearly demonstrates the need of California's standards (whether evaluated as a program or as specific standards) to meet compelling and extraordinary conditions within California.

On completion of this review, EPA finds no basis for discounting the ample record support on California's need for both the GHG standards and the ZEV sales mandate to address compelling and extraordinary conditions in California when using both the traditional and SAFE 1 interpretation to the second waiver prong. Additionally, because of the way CARB's motor vehicle emission standards operate in tandem and are designed to reduce both criteria and GHG pollution and the ways in which GHG pollution exacerbates California's serious air quality problems, including the heat exacerbation of ozone, the Agency in SAFE 1 should not have evaluated California's specific "need" for GHG standards. In sum, in reconsidering SAFE 1, and after having now reviewed and evaluated the complete factual record, EPA reaffirms that California needs the GHG standards and ZEV sales mandate at issue to "meet compelling and extraordinary conditions."

A. Historical Practice

Under section 209(b)(1)(B), EPA shall not grant a waiver if California "does not need such State standards to meet compelling and extraordinary conditions." For nearly the entire history of the waiver program, EPA has read the phrase "such State standards" in section 209(b) (1)(B) as referring back to standards "in the aggregate," in the root paragraph of section 209(b)(1), which calls for California to make a protectiveness finding for its standards. EPA has interpreted the phrase "in the aggregate" as referring to California's program as a whole, rather than each State standard, and as such not

calling for the Agency's standard-by-standard analysis of California's waiver request. EPA has thus reasoned that both statutory provisions must be read together so that the Agency reviews the same standards that California considers in making its protectiveness determination and to afford California discretion. The D.C. Circuit has also stated that "[t]he expansive statutory language gives California (and in turn EPA) a good deal of flexibility in assessing California's regulatory needs. We therefore find no basis to disturb EPA's reasonable interpretation of the second criterion." 183

^{181. &}quot;The interpretation that my inquiry under (b)(1)(B) goes to California's need for its own mobile source program is borne out not only by the legislative history, but by the plain meaning of the statue as well." 49 FR at 18890.

^{182. 74} FR at 32751 n. 44; 32761 n.104. EPA cited Entergy Corp. v. Riverkeeper, Inc., 129 S. Ct. 1498 (2009) ("That view governs if it is a reasonable interpretation of the statute—not necessarily the only possible interpretation, nor even the interpretation deemed most reasonable by the courts"), and Chevron U.S.A. Inc. v. Natural Resources Defense Council, Inc., 467 U.S. 837, 843-844 (1984).) ("It seems to us, therefore, that the phrase "best available," even with the added specification "for minimizing adverse environmental impact," does not unambiguously preclude cost-benefit analysis."). See also 78 FR at 2126-2127 n. 78.

^{183.} Am. Trucking Ass'n v. EPA, 600 F.3d 624, 627 (D.C. Cir. 2010) (ATA v. EPA). See also Dalton Trucking v. EPA, No. 13-74019 (9th Cir. 2021) ("The EPA was not arbitrary and capricious in declining to find that 'California does not need such California standards to meet compelling and extraordinary conditions,' §7543(e) (2)(A)(ii), under the alternative version of the needs test, which requires 'a review of whether the Fleet Requirements are per se needed to meet compelling and extraordinary conditions,' 78 FR

In addressing the Agency's reading of section 209(b) (1)(B), for example, in the 1983 LEV waiver request EPA explained that:

This approach to the "need" criterion is also consistent with the fact that because California standards must be as protective as Federal standards in the aggregate, it is permissible for a particular California standard or standards to be less protective than the corresponding Federal standard. For example, for many years, California chose to allow a carbon monoxide standard for passenger cars that was less stringent than the corresponding Federal standard as a "trade-off" for California's stringent nitrogen oxide standard. Under a standard of review like that proposed by MVMA/AIAM, EPA could not approve a waiver request for only a less stringent California standard because such a standard, in isolation,

at 58,103. The EPA considered 'the relevant factors,' *Motor Vehicle Mfrs. Ass'n of U.S.* v. *State Farm Mut. Auto. Ins. Co., Inc.*, 463 U.S. 29, 42-43 (1983), including statewide air quality, 78 FR 58,104, the state's compliance with federal National Ambient Air Quality standards for ozone and PM 2.5 on a statewide basis, *id.* at 58,103-04, the statewide public health benefits, *id.* at 58,104, and the utility of the Fleet Requirements in assisting California to meet its goals, *id.* at 58,110. Contrary to Dalton's argument, the EPA did not limit its review to two of California's fourteen air quality regions. The EPA examined the relevant data provided by CARB, and it articulated a 'satisfactory explanation for its action including a rational connection between the facts found and the choice made.' *See Motor Vehicle Mfrs. Ass'n of U.S., Inc.*, 463 U.S. at 43, 103 S.Ct. 2856 (cleaned up).").

necessarily could be found to be contributing to rather than helping, California's air pollution problems.¹⁸⁴

In 1994, EPA again had cause to explain the Agency's reading of section 209(b)(1)(B) in the context of California's particulate matter standards waiver request:

[T]o find that the 'compelling and extraordinary conditions' test should apply to each pollutant would conflict with the amendment to section 209 in 1977 allowing California to select standards 'in the aggregate' at least as protective as federal standards. In enacting that change, Congress explicitly recognized that California's mix of standards could 'include some less stringent than the corresponding federal standards.' See H.R. Rep. No. 294, 95th Cong., 1st Sess. 302 (1977). Congress could not have given this flexibility to California and simultaneously assigned to the state the seemingly impossible task of establishing that 'extraordinary and compelling conditions' exist for each standard. 185

Congress has also not disturbed this reading of section 209(b)(1)(B) as calling for EPA review of California's whole program. With two noted exceptions described below, EPA has consistently interpreted this provision

^{184. 58} FR 4166, LEV Waiver Decision Document at 50-51.

^{185. 49} FR at 18887, 18890.

as requiring the Agency to consider whether California needs a separate motor vehicle emission program as compared to the specific standards in the waiver request at issue to meet compelling and extraordinary conditions.

Congress intended to allow California to address its extraordinary environmental conditions and foster its role as a laboratory for motor vehicle emissions control. The Agency's long-standing practice therefore has been to evaluate CARB's waiver requests with the broadest possible discretion to allow California to select the means it determines best to protect the health and welfare of its citizens in recognition of both the harsh reality of California's air pollution and to serve as a pioneer and a laboratory for the nation in setting new motor vehicle emission standards and developing control technology. EPA notes that "the statute does not provide for any probing substantive review of the California standards by federal officials." 187

As a general matter, EPA has applied the traditional interpretation in the same way for all air pollutants,

^{186.} See, e.g., S. Rep. No. 403, 90th Cong., 1st Sess. 33 (1967) (The waiver of preemption is for California's "unique problems and pioneering efforts."); 113 Cong. Rec. 30950, 32478 ("[T]he State will act as a testing agent for various types of controls and the country as a whole will be the beneficiary of this research.") (Statement of Sen. Murphy).

^{187.} $Ford\ Motor\ Co.,\ v.\ EPA,\ 606\ F.2d\ 1293,\ 1300\ (D.C.\ Cir.\ 1979).$

criteria and GHG pollutants alike. 188 As discussed in Section II, there have only been two exceptions to this practice: one in 2008 and one in 2019. In 2008, EPA for the first time analyzed California's waiver request under an alternative approach and denied CARB's waiver request. EPA concluded that section 209(b) was intended to allow California to promulgate state standards applicable to emissions from new motor vehicles to address air pollution problems that are local or regional, but that section 209(b)(1)(B) was not intended to allow California to promulgate state standards for emissions from new motor vehicles designed to address global climate change problems. Or, in the alternative, EPA concluded that effects of climate change in California were not compelling and extraordinary compared to the effects in the rest of the country. 189 EPA rejected this view a little over a year later in 2009 by applying the traditional interpretation in granting California's waiver request for the same GHG standard, finding no support in the statute or congressional intent for the alternative application of the statute.¹⁹⁰

^{188. 74} FR at 32763-65; 76 FR 34693; 79 FR 46256; 81 FR 95982.

^{189. 73} FR at 12160-64.

^{190. 74} FR at 32744, 32746, 32763 ("The text of section 209(b) and the legislative history, when viewed together, lead me to reject the interpretation adopted in the March 6, 2008 Denial, and to apply the traditional interpretation to the evaluation of California's greenhouse gas standards for motor vehicles. If California needs a separate motor vehicle program to address the kinds of compelling and extraordinary conditions discussed in the traditional interpretation, then Congress intended that California could have such a program. Congress also intentionally provided California the

In evaluating the ACC program waiver in 2013, EPA applied the traditional interpretation to the ACC program waiver request and found that the Agency could not deny the waiver request under the second waiver prong. ¹⁹¹ Further, without adopting the alternative interpretation that had been applied in the 2008 GHG waiver denial, EPA assessed California's need for the GHG standards at issue and found that the Agency could not deny the ACC program waiver request, even applying the alternative interpretation. EPA noted that to the extent that it was appropriate to examine the CARB's need for the GHG standards at issue to meet compelling and extraordinary conditions, the Agency had discussed at length in the 2009 GHG waiver decision that California has compelling and

broadest possible discretion in adopting the kind of standards in its motor vehicle program that California determines are appropriate to address air pollution problems and protect the health and welfare of its citizens. The better interpretation of the text and legislative history of this provision is that Congress did not use this criterion to limit California's discretion to a certain category of air pollution problems, to the exclusion of others. EPA concluded that even under this alternative approach California GHG standards were intended at least in part to address a local or regional problem because of the 'logical link between the local air pollution problem of ozone and GHG.").

191. 78 FR at 2129 ("CARB has repeatedly demonstrated the need for its motor vehicle program to address compelling and extraordinary conditions in California. As discussed above, the term compelling and extraordinary conditions 'does not refer to the levels of pollution directly.' Instead, the term refers primarily to the factors that tend to produce higher levels of pollution—geographical and climatic conditions (like thermal inversions) that, when combined with large numbers and high concentrations of automobiles, create serious air pollution problems. California still faces such conditions.").

extraordinary conditions directly related to regulations of GHGs. ¹⁹² Similarly, EPA explained that to the extent it was appropriate to examine California's need for the ZEV sales mandate, these requirements would enable California to meet both air quality and climate goals into the future. ¹⁹³ Additionally, EPA recognized CARB's coordinated strategies reflected in the technologies envisioned to meet the ACC program requirements and in turn addressing both criteria pollutants and GHGs and the magnitude of the technology and energy transformation needed to meet such goals. ¹⁹⁴

192. Id. at 2129-30.

193. Id. at 2129 ("[A]s EPA discussed at length in its 2009 GHG waiver decision, California does have compelling and extraordinary conditions directly related to regulations of GHG. EPA's prior GHG waiver contained extensive discussion regarding the impacts of climate change in California. In addition, CARB has submitted additional evidence in comment on the ACC waiver request that evidences sufficiently different circumstances in California. CARB notes that "Record-setting fires, deadly heat waves, destructive storm surges, loss of winter snowpack—California has experienced all of these in the past decade and will experience more in the coming decades. California's climate—much of what makes the state so unique and prosperous—is already changing, and those changes will only accelerate and intensify in the future. Extreme weather will be increasingly common as a result of climate change. In California, extreme events such as floods, heat waves, droughts and severe storms will increase in frequency and intensity. Many of these extreme events have the potential to dramatically affect human health and well-being, critical infrastructure and natural systems." (footnotes omitted)).

194. *Id.* at 2130-31 ("As CARB notes in its waiver request, the goal of the CARB Board in directing CARB staff to redesign the ZEV regulation was to focus primarily on zero emission drive—that

The only other exception to the application of the traditional interpretation was in SAFE 1, when EPA again used a standard-specific level of review and focused on California's need for GHG standards at issue under the waiver. There, EPA posited that section 209(b)(1)

is BEV, FCV, and PHEVs in order to move advanced, low GHG vehicles from demonstration phase to commercialization. CARB also analyzed pathways to meeting California's long term 2050 GHG reduction targets in the light-duty vehicle sector and determined that ZEVs would need to reach nearly 100 percent of new vehicle sales between 2040 and 2050. CARB also notes that the "critical nature of the LEV III regulation is also highlighted in the recent effort to take a coordinated look at strategies to meet California's multiple air quality and climate goals well into the future. This coordinated planning effort, Vision for Clean Air: A Framework for Air Quality and Climate Planning (Vision for Clean Air) demonstrates the magnitude of the technology and energy transformation needed from the transportation sector and associated energy production to meet federal standards and the goals set forth by California's climate change requirements. . . . The Vision for Clean Air effort illustrates that in addition to the cleanup of passenger vehicles (at issue here) as soon as possible as required in the LEV III regulation, transition to zero- and near-zero emission technologies in all on- and off-road engine categories is necessary to achieve the coordinated goals. Therefore, EPA believes that CARB's 2018 and later MY ZEV standards represent a reasonable pathway to reach these longer term goals. Under EPA's traditional practice of affording CARB the broadest discretion possible, and deferring to CARB on its policy choices, we believe there is a rational connection between California ZEV standards and its attainment of long term air quality goals. Whether or not the ZEV standards achieve additional reductions by themselves above and beyond the LEV III GHG and criteria pollutant standards, the LEV III program overall does achieve such reductions, and EPA defers to California's policy choice of the appropriate technology path to pursue to achieve these emissions reductions." (footnote omitted)).

(B) called for a "particularized nexus" for California's motor vehicle standards: "Congress enacted the waiver authority for California under section 209(b) against the backdrop of traditional, criteria pollutant environmental problems, under which all three links in this chain bear a particularized nexus to specific local California features: (1) Criteria pollutants are emitted from the tailpipes of the California motor vehicle fleet; (2) those emissions of criteria pollutants contribute to air pollution by concentrating locally in elevated ambient levels, which concentration, in turn; (3) results in health and welfare effects (e.g., from ozone) that are extraordinarily aggravated in California as compared to other parts of the country, with this extraordinary situation being attributable to a confluence of California's peculiar characteristics, e.g., population density, transportation patterns, wind and ocean currents, temperature inversions, and topography."195 As support for the nexus test, EPA, for the first time in waiver decisions, relied on section 202(a) and its own terms of authority to inform interpretation of the second waiver prong. 196 In addition, EPA relied on legislative history to interpret "compelling and extraordinary" conditions as a reference to "peculiar local conditions" and "unique problems" in California. 197 Accordingly, EPA reasoned that California

^{195. 84} FR at 51339.

^{196.} Id. at 51339-40.

^{197.} *Id.* at 51342 (quoting S. Rep. No. 403, 90th Cong. 1st Sess., at 32 (1967)) ("Congress discussed 'the unique problems faced in California as a result of its climate and topography.' H.R. Rep. No. 728, 90th Cong. 1st Sess., at 21 (1967). *See also* Statement of Cong. Holifield (CA), 113 Cong. Rec. 30942-43 (1967). Congress also noted the large effect of local vehicle pollution on such local

must demonstrate "compelling and extraordinary circumstances sufficiently different from the nation as a whole to justify standards on automobile emissions which may, from time to time, need to be more stringent than national standards." ¹⁹⁸

In SAFE 1, EPA then posited that the nexus test should be applied to California's GHG standards specifically, rather than California's program "as a whole" under the traditional "aggregate" approach, "to ensure that such standard is linked to local conditions that giv[e] rise to the air pollution problem, that the air pollution problem is serious and of a local nature, and that the State standards at issue will meaningfully redress that local problem." As support for the GHG-specific scrutiny, EPA reasoned that "[t]he Supreme Court's opinion in *UARG* v. *EPA*, 134 S. Ct. 2427 (2014), instructs that Clean Air Act provisions cannot necessarily rationally be applied identically to GHG as they are to traditional pollutants." ²⁰⁰

problems. See, e.g., Statement of Cong. Bell (CA) 113 Cong. Rec. 30946. As explained at proposal, Congress focus was on California's ozone problem, which is especially affected by local conditions and local pollution. See Statement of Cong. Smith (CA) 113 Cong. Rec. 30940-41 (1967); Statement of Cong. Holifield (CA), id., at 30942. See also, MEMA I, 627 F.2d 1095, 1109 (D.C. Cir. 1979) (noting the discussion of California's 'peculiar local conditions' in the legislative history). In sum and as explained at proposal, conditions that are similar on a global scale are not 'extraordinary,' especially where 'extraordinary' conditions are a predicate for a local deviation from national standards, under section 209(b). 83 FR 43247.").

^{198.} Id.

^{199.} Id. at 51345.

^{200.} Id. at 51340.

Applying the nexus test, EPA concluded that California did not need its GHG standards to meet "compelling and extraordinary conditions" because they were missing a particularized nexus to specific local features. EPA in the alternative posited that "even if California does have compelling and extraordinary conditions in the context of global climate change. California does not 'need' these standards under section 209(b)(1)(B) because they will not meaningfully address global air pollution problem of the sort associated with GHG emissions."201 EPA also dismissed the 2009 GHG waiver conclusion on deleterious effects of GHG emissions on ozone (e.g., how increases in ambient temperature are conducive to ground-level ozone formation), stating that such a relationship "does not satisfy this requirement for a particularized nexus, because to allow such attenuated effects to fill in the gaps would eliminate the function of requiring such a nexus in the first place."202

B. Notice of Reconsideration of SAFE 1 and Request for Comment

In the Notice of Reconsideration of SAFE 1, EPA noted its interest in any new or additional information or comments regarding whether it appropriately interpreted and applied section 209(b)(1)(B) in SAFE 1. The Agency noted that EPA's finding in SAFE 1, that such standards were only designed to address climate change and a global air pollution problem, led EPA to a new interpretation of section 209(b)(1)(B). EPA solicited views on whether it was

^{201.} Id. at 51349.

^{202.} Id.

permissible to construe section 209(b)(1)(B) as calling for a consideration of California's need for a separate motor vehicle program where criteria pollutants are at issue as well as California's specific standards where GHG standards are at issue.

The Notice of Reconsideration also set forth that EPA's decision to withdraw the ACC program waiver as it relates to California's ZEV sales mandate was based on the same new interpretation and application of the second waiver prong and rested heavily on the conclusion that California only adopted the ZEV sales mandate requirement for purposes of achieving GHG emission reductions. EPA recognized that this conclusion in turn rested solely on a specific reading of a single sentence in CARB's ACC program waiver request. EPA requested comment on these specific conclusions and readings as well as whether the withdrawal of the ACC program waiver, within the context of California's environmental conditions and as applied to the GHG standards and ZEV sales mandate requirement, was permissible and appropriate.

C. Comments Received

EPA received multiple comments on its decision to evaluate California's need for its GHG standards separate from its need for a separate motor vehicle

^{203.} *Id.* at 51330 ("Regarding the ACC program ZEV mandate requirements, CARB's waiver request noted that there was no criteria emissions benefit in terms of vehicle (tank-to-wheel—TTW) emissions because its LEV III criteria pollutant fleet standard was responsible for those emission reductions.").

emission program as a whole. Some commenters agreed that EPA could evaluate waiver requests for the specific GHG standards under the waiver along the lines of the Agency's pronouncements in SAFE 1. Additionally, commenters pointed to the method of EPA's review in SAFE 1—evaluating the standards individually, as they are received, rather than in the aggregate—as evidence of the flaw in the traditional interpretation. Some commenters also echoed SAFE 1's concern that "once EPA had determined that California needed its very first set of submitted standards to meet extraordinary and compelling conditions, EPA would never have the discretion to determine that California did not need any subsequent standards."

Under this analysis of the specific standards at issue under the waiver, these commenters continued, California could not demonstrate that its GHG and ZEV standards were, on their own, compelling and extraordinary. These commenters agreed with SAFE 1's "particularized nexus" interpretation of "compelling and extraordinary," arguing that the words required unique consequences in order to give adequate meaning to the words themselves and in order to overcome equal sovereignty implications. ²⁰⁶ Using this interpretation, these commenters concluded that, because "GHG concentrations are essentially uniform throughout the globe, and are not affected by

^{204.} CEI at 13-14.

^{205. 84} FR at 51341. See, e.g., NADA at 5; Urban Air at 25, 29-33; AFPM at 22-23.

^{206.} AFPM at 12; Urban Air at 4.

California's topography and meteorology," and because the entire nation would be affected by climate change, neither the effects of the regulations on climate change, nor the impacts of climate change on California could be considered "compelling and extraordinary." Some commenters also argued that these standards were unnecessary given California's "deemed to comply" provision, which would theoretically allow all automobile manufacturers to comply with California's standards by meeting the less stringent Federal GHG standards. ²⁰⁸

In contrast, other commenters asked that EPA reverse its SAFE 1 section 209(b)(1)(B) determination by reverting to EPA's long-standing "program-level" approach to the "need" inquiry, where "EPA considers California's need for its own mobile-source-emissions program as a whole, not whether California needs a particular standard for which

^{207.} CEI at 14-16 ("The resulting "global pool" of GHG emissions is not any more concentrated in California than anywhere else . . . [E]ven if one assumes "compelling and extraordinary conditions" can refer to climate change impacts, such as heat waves, drought, and coastal flooding, California's vulnerability is not "sufficiently different" from the rest of the nation to merit waiving federal preemption of state emission standards. Thus, California is not "extraordinary" in regard to either the "causes" of the "effects" of global climate change."); NADA at 5 ("while vehicle GHG emissions also were, by definition, local, their impact on serious local air quality concerns could not be shown."); AFPM at 11-14 ("Neither the causes nor effects of GHG emissions are compelling and extraordinary conditions, as they are global rather than local conditions, and California's GHG standards and ZEV mandate will not meaningfully address the causes or effects of these GHG emissions.").

^{208.} NADA at 4-5; Urban Air at 33.

it has requested a waiver."²⁰⁹ These commenters noted the long tradition of interpreting California's need in the aggregate, an interpretation that SAFE 1 acknowledged was reasonable.²¹⁰ This interpretation, they argued, best aligned with the text, legislative history, and purpose of the waiver program.²¹¹ For example, some commenters argued that, because feasibility was evaluated under an aggregate approach, it would be unreasonable for California's need for the program to be evaluated under a more restrictive approach.²¹² These commenters also argued that Congress had expressed approval of this

^{209.} States and Cities at 22 n.16.

^{210.} Twelve Public Interest Organizations at 7 ("The Trump EPA in turn acknowledged that this longstanding interpretation of Section 209(b)(1)(B) was a reasonable one, 84 FR at 51,341").

^{211.} States and Cities at 22 (citing 84 FR at 51341); Tesla at 11 ("The plural reference to 'such State standards' requires that the standards be considered in the aggregate as a group. This language stands in stark contrast to alternate phrasing that was available to Congress and that would have permitted a non-aggregate determination, such as: 'such State does not need a State standard to meet compelling and extraordinary conditions.' Indeed, alternative language referencing individual standards is present in subsection (b)(2), which references 'each State standard.").

^{212.} States and Cities at 25-26; Twelve Public Interest Organizations at 8 ("An aggregate approach to the consistency inquiry also makes sense under Section 209(b)(1)(C) because technological feasibility is effectively evaluated on a program basis. The feasibility of a new standard cannot be evaluated on its own if there are interactions with pre-existing standards. Such interactions between standards are what prompted Congress to add the "in the aggregate" phrase to section 209 in the first place.").

aggregate approach, citing legislative history from 1977 and 1990. 213 This approach, they continued, aligns with the Waiver Program's broad deference to California to create an entire regulatory program, which is comprised of regulations that interact with and affect each other.²¹⁴ One commenter also responded directly to the question EPA posed in its Notice of Reconsideration, whether it was "permissible for EPA to construe section 209(b)(1)(B) as calling for consideration of California's need for a separate motor vehicle program where criteria pollutants are at issue and consideration of California's individual standards where GHG standards are at issue."215 According to the commenter, "The Supreme Court has rejected this 'novel interpretive approach' of assigning different meanings to the same statutory text in the same provision, depending on the application, because it 'would render every statute a chameleon."216

^{213.} States and Cities at 26-27; Ozone Transport Commission (OTC), Docket No. EPA-HQ-OAR-2021-0257-0283 at 4.

^{214.} States and Cities at 27-28.

^{215. 86} FR at 22429.

^{216.} States and Cities at 24 (quoting Clark v. Martinez, 543 U.S. 371, 382 (2005) and citing U.S. v. Santos, 553 U.S. 507, 522 (2008); U.S. Dep't of the Treasury v. FLRA, 739 F.3d 13,21 (D.C. Cir. 2014)). The commenter notes that in the SAFE 1 brief, EPA claimed that its new approach to section 209(b)(1)(B) would apply "for all types of air pollutants" but EPA could point to nowhere in SAFE 1 decision where this was said. Id. at 25. And "only two sentences later," EPA acknowledged that its review under this second prong would change "depending upon which 'air quality concerns' were implicated." Id.

These commenters also asked EPA to revert to the traditional interpretation of "compelling and extraordinary" instead of SAFE 1's "particularized nexus" formulation. Commenters noted the SAFE 1 requirement appears nowhere in the text of the statute.²¹⁷ Because of this absence, they continued, EPA's references to the legislative history from 1967 have no "tether" to the statutory text and cannot justify the nexus requirement.²¹⁸ Further, commenters argued that EPA's reliance on the equal sovereignty doctrine improperly informed how EPA should interpret the phrase "compelling and extraordinary conditions" in the second waiver prong, and therefore requiring such conditions to be sufficiently different or unique among states, was inappropriate.²¹⁹ Commenters argued that the equal sovereignty doctrine was inapplicable to the second waiver prong. They explained that the Supreme Court has only applied the "rarely invoked" doctrine of equal sovereignty in the "rare instance where Congress undertook 'a drastic departure from basic principles of federalism' by authorizing 'federal intrusion into sensitive areas of state and local policymaking."220 Congress's exercise of its Commerce Clause power in regulating air pollution

^{217.} States and Cities at 34 (noting the lack of the words "nexus," "particularized," "peculiar," and "local" anywhere in sections 209(b) or 202(a)(1)).

^{218.} Id. at 35.

^{219.} *Id.* at 41-43; Twelve Public Interest Organizations at 4-6.

^{220.} States and Cities at 42 (quoting *Shelby Cnty.* v. *Holder*, 570 U.S. 529, 535, 545 (2013)).

from new motor vehicles, commenters continued, is not such an "intrusion." Moreover, they wrote, applying the equal sovereignty doctrine in this instance would actually "diminish most States' sovereignty" because it would "reduce the regulatory options available to California and to other [section 177] States." This diminished sovereignty, they argued, would not "enhance[e] the sovereignty of any State" or "alleviate" any unjustified burden because "Section 209(b)(1) imposes no such burden."²²¹

Similarly, commenters rebutted SAFE 1's use of words like "peculiar" and "unique" to further define "compelling and extraordinary." These words, they noted, appear nowhere in the text of section 209(b)(1)(B) and do not align with the plain meaning of the word "extraordinary." Further, they argued, this narrow interpretation "would render the waiver provision unworkable" as, "for any given air pollutant, it is possible to identify other areas of the

^{221.} *Id.* at 43; Twelve Public Interest Organizations at 5 ("Clean Air Act Section 209(b) places no extraordinary burden or disadvantage on one or more States. Rather, the statute benefits California by allowing the exercise of its police power authority to address its particular pollution control needs").

^{222.} States and Cities at 38-39 (explaining that the existence of those words in the legislative history "simply highlight that Congress did *not* codify [them] in Section 209(b)(1)(B)" and that plain meaning of "extraordinary" is "out of the ordinary"); Twelve Public Interest Organizations app. 1 at 49 ("Congress understood, even in 1967, that '[o]ther regions of the Nation may develop air pollution situations related to automobile emissions which will require standards different from those applicable nationally.' S. Rep. No. 90-403, at 33.").

country that suffer from a similar pollution problem."²²³ In fact, they continued, this argument was rejected in the 1967 legislative history and in 1984, "when EPA thoroughly rebutted the assertion that California could not receive a waiver if individual pollutant levels were 'no worse than some other areas of the country."²²⁴ Moreover, they argued, the existence of section 177 necessarily acknowledges that other states may have the same or similar air pollution problems as California.²²⁵

Other commenters argued that California needed GHG standards to address "compelling and extraordinary" conditions in California even under the SAFE 1 interpretation of the second waiver prong. These commenters argued that GHG and ZEV standards produce both GHG and criteria pollution benefits, pointing to language in the ACC program waiver that acknowledged these dual benefits and to subsequent SIP approvals that incorporated the California standards in order to achieve criteria emission reductions. ²²⁶ In particular,

^{223.} Tesla at 9.

^{224.} *Id.* (quoting 49 FR at 18887, 18891) (stating that EPA explained that "there is no indication in the language of section 209 or the legislative history that California's pollution problem must be the worst in the country, for a waiver to be granted.")).

^{225.} Twelve Public Interest Organizations app. 1 at 49; States and Cities at 38-39.

^{226.} States and Cities at 9-14, 30-31; Center for Biological Diversity, Docket No. EPA-HQ-OAR-2021-0257-0358 at 2 ("The Trump EPA improperly separated California's need for greenhouse gas regulations from its need for criteria pollutant standards. In

commenters explained that the 2012 California waiver request established that the ZEV standard would reduce criteria pollution both "by reducing emissions associated with the production, transportation, and distribution of gasoline" and "by driving the commercialization of zero-emission-vehicle technologies necessary to reduce future emissions and achieve California's long-term air

reality, these two goals are tightly linked, and both are critical to the Clean Air Act's goals of safeguarding public health and welfare."); San Joaquin Valley Air Pollution Control District (SJVAPCD), Docket No. EPA-HQ-OAR-2021-0257-0105 at 3 ("The District's 2016 Plan for the 2009 9-Hour Ozone Standard adopted June 16, 2016, and 2018 Plan for the 1997, 2006, and 2012 PM 2.5 Standards, adopted November 15, 2018, both rely on emission reductions from California's Advanced Clean Cars regulation and other mobile source measures to support the Valley's attainment of the federal health-based NAAQS."); NCAT at 11 ("In addition, California's ZEV standards are intended to and do achieve significant incremental reductions of NOx and other non-GHG emissions."); Tesla at 10-11 ("In comments submitted to the EPA in 2009 regarding a preemption waiver, [California] explained that it 'specifically designed its GHG standards for criteria pollutants.' It also emphasized that it has 'frequently referenced the science to support GHG standards as a necessary method for controlling ozone and particulate matter pollution' and has 'consistently recognized that the State's ability to reduce nonattainment days for ozone and wildfire-caused particulate matter depends on its ability to reduce GHG emissions. . . . EPA also has repeatedly expressed its own understanding that GHG standards should be viewed as a strategy to help control criteria pollutants to address National Ambient Air Quality Standards nonattainment."); Twelve Public Interest Organizations at 5 ("For example, atmospheric heating due to global warming can increase the production of ground-level ozone in California, which suffers from extraordinary amounts of locally reacting nitrogen oxides and volatile organic compounds.").

quality goals."²²⁷ As for the GHG standards, commenters noted that, as acknowledged in the ACC program waiver, "global warming exacerbates criteria pollution and makes it harder to meet air pollution standards."²²⁸ Thus, they argue, "EPA expressly and improperly limited its Determination to consideration of the 'application of section 209(b)(1)(B) to California's need *for a GHG climate program.*"²²⁹ Given EPA's consistent acceptance that "California's criteria pollution 'conditions' are 'extraordinary and compelling' and that the record demonstrates that California's GHG and ZEV standards reduce criteria emissions in California," EPA should

^{227.} Center for Biological Diversity at 2-3. In contrast, some commenters, echoing SAFE 1, argued that these upstream emission benefits should not be considered in determining the criteria pollutant benefits of these standards. CEI at 16 ("Although NHTSA and EPA are required to consider all relevant factors when determining CAFE and tailpipe CO2 standards, it is inappropriate to elevate stationary source criteria pollutant emissions into a make-or-break factor in waivers for mobile source programs. The Clean Air Act already provides the EPA with ample authorities to regulate stationary sources, including the NAAQS program, New Source Performance Standards program, Prevention of Significant Deterioration of Air Quality program, Acid Rain program, and Regional Haze program. If Congress wanted NHTSA's CAFE program and EPA's mobile source program to prioritize reductions of indirect stationary source emissions, it could easily have said so. The indirect effects on stationary source emissions are not even mentioned.").

^{228.} Center for Biological Diversity at 3.

^{229.} States and Cities at 28 (citing 84 FR at 51339 (emphasis added)) (limiting section 209(b)(1)(B) consideration to "the case of GHG emissions.").

"reverse its SAFE 1 section 209(b)(1)(B) determination and the waiver withdrawal that rested on it—regardless of whether EPA reverts to its traditional, program-level approach." ²³⁰

Regardless of the emissions benefits of the standards, some commenters argued that California's plan to address both long-term and short-term climate and criteria pollutant reduction goals is entitled to deference. Thus, even if "the mandate truly added nothing to the emission benefits of California's standards for vehicular emissions of criteria and greenhouse gas pollutants," commenters claimed, "the mandate would simply constitute the State's choice of means for automakers to comply with its standards." These commenters further argued that section 209(b)(1)(B) "does not authorize EPA to inquire into whether the means to comply with California emission standards, as opposed to the actual

^{230.} States and Cities at 29. The commenter notes that EPA never considered whether California needed those criteria emission reductions from its ZEV and GHG standards because it refused to consider those criteria reductions at all: "EPA attempted to justify disregarding record evidence and its own prior findings concerning the criteria emission benefits of these California standards by mischaracterizing CARB's 2012 waiver request.... But, having chosen to sua sponte reopen the question whether California continues to need standards it has been implementing for six years,, EPA could not limit its consideration to what the standards were intended to achieve when they were originally designed or presented. CARB (and others) asserted clearly in SAFE 1 comments that both the GHG and ZEV standards produce criteria pollution benefits upon which California and other States rely to improve air quality." Id. at 29-30.

^{231.} Twelve Public Interest Organizations at 9-10.

standards themselves, are needed to meet compelling and extraordinary conditions."²³² Commenters also claimed that EPA's argument, that California cannot need the GHG and ZEV standards because those standards alone would not "meaningfully address global air pollution problems" posed by climate change, "lacks merit" and "is illogical."²³³ Such an approach, they explained "amounts to a conclusion that California is forbidden from acting precisely because climate change is a global threat—when in fact the global aspect of this problem demonstrates the need for California to take action," a conclusion, they noted, that was rejected by the Supreme Court in *Massachusetts* v. *EPA*. ²³⁴ Even if there was some merit to

^{232.} Id. (citing $MEMA\ I,\ 627\ F.2d\ 1095,\ 1111-14$ (D.C. Cir. 1979)).

^{233.} States and Cities at 40, 49-50; NCAT at 11 ("EPA's argument that California does not 'need' vehicle standards that reduce GHG emissions because such standards alone cannot meaningfully reduce the impacts of climate change in California lacks merit. 84 FR at 51,346-47. EPA's approach in SAFE 1 read requirements into the statute that Congress did not choose to impose: That a single standard be sufficient to resolve an environmental problem caused by multiple and diverse sources. Instead, need should be defined by reference to the underlying problem, and California's standards are one important element of the broader response."); Tesla at 8-9 (citing *Massachusetts* v. *EPA*, 549 U.S. 497, 525-26 (2007)) ("Nor is it dispositive that developing countries such as China and India are poised to increase greenhouse gas emissions substantially over the next century: A reduction in domestic emissions would slow the pace of global emissions increases, no matter what happens elsewhere.").

^{234.} Tesla at 8-9 ("Indeed, the Supreme Court rejected this logic in *Massachusetts* v. *EPA*, 549 U.S. 497 (2007), explaining: "Because of the enormity of the potential consequences associated

the argument, one commenter argued, SAFE 1's assertion that the regulations "would have only a *de minimis* effect on climate change understates the impact that collective action by California and the Section 177 states can have on GHG emissions." The commenter noted that "[w] ith a total population of over 140 million people, these 19 jurisdictions collectively account for more than 42 percent of the U.S. population . . . and more than 40 percent of the U.S. new car market." ²³⁶

Finally, these commenters also argued that climate change and its impacts are, themselves, "extraordinary and compelling" conditions. They provided evidence of increased weather events, agricultural effects, and wildfires, amongst other impacts of climate change, which have already begun to severely affect California.²³⁷

with man-made climate change, the fact that the effectiveness of a remedy might be delayed during the (relatively short) time it takes for a new motor-vehicle fleet to replace an older one is essentially irrelevant."); States and Cities at 41.

235. NESCAUM at 7.

236. Id.

237. States and Cities at 43-48; Twelve Public Interest Organizations at 5; Center for Biological Diversity at 3; Tesla at 8-9. States and Cities at 43-48; Twelve Public Interest Organizations at 5-6; Center for Biological Diversity at 3 ("California also experiences uniquely dangerous effects from increases in greenhouse gases. For example, the California legislature has found that global warming will cause adverse health impacts from increased air pollution and a projected doubling of catastrophic wildfires. Many of the state's most extreme weather events have occurred in the last decade, including a severe drought from 2012-2016, an almost non-existent

D. Analysis: California Needs the ACC Program GHG Standards and ZEV Sales Mandate To Address Compelling and Extraordinary Conditions Under Section 209(b)(1)(B)

In this action, EPA first finds that the Agency should not have reinterpreted section 209(b)(1)(B) in evaluating California's "need" for GHG standards and ZEV sales mandate requirements at issue. The analysis below walks through the statutory language and history associated with this provision. As part of this discussion, the relationship of this provision and California's authority and deference is highlighted. The two interpretations of the waiver prong are then reviewed, presenting the Agency's rationale for its findings of the inappropriate SAFE 1 interpretation and support for its conclusion about the better interpretation. Second, as shown below, the factual record before the Agency at the time of SAFE 1 supports the GHG standards and ZEV sales mandate requirements at issue under either the traditional or SAFE 1 interpretation of section 209(b)(1)(B).

1. EPA Is Withdrawing the SAFE 1 Section 209(b)(1)(B) Interpretation

Except for two short-lived exceptions in the context of the 2008 waiver denial and SAFE 1, EPA has consistently

Sierra Nevada winter snowpack in 2014-2015, three of the five deadliest wildfires in state history, and back-to-back years of the warmest average temperatures on record. These ongoing disasters demonstrate California's status as 'one of the most 'climate-challenged' regions of North America.").

recognized that reading the "needs" test of the second waiver prong as calling for a standard-specific evaluation would be inconsistent with congressional intent given the text of section 209(b)(1) legislative history, as well as the way the different standards in the ACC program work together to reduce criteria and GHG pollution and spur innovation. As further explained below, all of these aspects lend support to the Agency practice of not subjecting California's waiver requests to review of the specific standards under the second waiver prong, and we agree that the traditional interpretation of section 209(b) is, at least, the better interpretation.

Under section 209(b)(1)(B), EPA must grant a waiver request unless the Agency finds that California "does not need such State standards to meet compelling and extraordinary conditions." EPA has historically read the phrase "such State standards" in section 209(b)(1)(B) as referring back to standards "in the aggregate" in section 209(b)(1), which addresses the protectiveness finding that California must make for its waiver requests. In addition, as EPA has explained in the past, reading the provision otherwise would conflict with Congress's 1977 amendment to the waiver provision to allow California's standards to be "at least as protective" as the federal standards "in the aggregate." This amendment must mean that some of California's standards may be weaker than federal standards counterbalanced by others that are stronger. If, however, a waiver can only be granted if each standard on its own meets a compelling need, then California could never have a standard that is weaker than the federal standard, rendering Congress's 1977

amendment inoperative. Congress would not have created the option for California's individual standards to be at least as protective "in the aggregate" and then taken that option away in the second waiver prong's "compelling need" inquiry.

In addition, EPA has reasoned that giving effect to section 209(b)(1) means that both subparagraph (b)(1) (B) and paragraph (b)(1) must be read together such that the Agency reviews the same standards that California considers in making its protectiveness determination. "\$209 (formerly \$208) was amended to require the U.S. Environmental Protection Agency (EPA) to consider California's standards as a package, so that California could seek a waiver of preemption if its standards 'in the aggregate' protected public health at least as well as federal standards."²³⁸

EPA has thus explained the reasoning for the reading of "such State standards" for instance, as follows:

[I]f Congress had intended a review of the need for each individual standard under (b)(1)(B), it is unlikely that it would have used the phrase "... does not need such state standards," which apparently refers back to the phrase "State standards... in the aggregate," as used in the first sentence of section 209(b)(1), rather than to the particular standard being considered.

^{238.} Motor Vehicle Mfrs. Ass'n v. NYS Dep't of Env't Conservation, 17 F.3d 521, 525 (2d Cir. 1994).

The use of the plural, *i.e.*, "standards," further confirms that Congress did not intend EPA to review the need for each individual standard in isolation.²³⁹

EPA has also explained that "to find that the 'compelling and extraordinary conditions' test should apply to each pollutant would conflict with the amendment to section 209 made in 1977 allowing California to select standards 'in the aggregate' at least as protective as federal standards. In enacting that change, Congress explicitly recognized that California's mix of standards could include some less stringent than the corresponding federal standards."²⁴⁰ This is in accord with *MEMA I*, where the D.C. Circuit explained that:

The intent of the 1977 amendment was to accommodate California's particular concern with oxides of nitrogen, which the State regards as a more serious threat to public health and welfare than carbon monoxide. California was eager to establish oxides of nitrogen standards considerably higher than applicable federal standards, but technological developments posed the possibility that emission control devices could not be constructed to meet both the high California oxides of nitrogen standard and the high federal carbon monoxide standard.²⁴¹

^{239. 49} FR at 18890.

^{240.} Id. at 18890 n.24.

^{241.} MEMA I, 627 F.2d 1095, 1110 n.32 (D.C. Cir. 1979).

EPA has further explained that the crucial consequence of the 1977 Amendment was to require waiver grants for California's specific standards that are part of the State's overall approach to reducing vehicle emissions to address air pollution even if those specific standards might not be needed to address compelling and extraordinary conditions. 242 For instance, EPA has previously granted a waiver for what was then described as "harmless emissions constituents such as methane" while reminding objectors of "EPA's practice to leave the decisions on controversial matters of public policy, such as whether to regulate methane emissions, to California."243 Similarly, in the 1984 p.m. standards waiver decision, EPA also discussed California's "need" for its own standards at length in response to comments that California must have worse air quality problems than the rest of the country to qualify for a waiver.²⁴⁴ There, EPA explained that California need not "have a 'unique' particulate problem, i.e., one that is demonstrably worse than in the rest of the country [because], there is no indication in the language of section 209 or the legislative history that California's pollution

^{242. 74} FR at 32761 ("Congress decided in 1977 to allow California to promulgate individual standards that are not as stringent as comparable federal standards, as long as the standards are 'in the aggregate, at least as protective of public health and welfare as applicable federal standards."); "[T]he 1977 amendments significantly altered the California waiver provision." *Ford Motor Co.*, 606 F.2d 1293, 1302 (D.C. Cir. 1979).

^{243. 43} FR at 25735.

^{244.} It bears note that these are the same kinds of comments that EPA received in the context of the ACC program waiver proceedings on California's need for GHG standards.

problem must be the worst in the country, for a waiver to be granted."²⁴⁵ Indeed, the word "unique" is not contained in the statutory provision. EPA further explained that "even if it were true that California's total suspended particulate problem is, as certain manufacturers argue, no worse than some other areas of the country, this does not mean that diesel particulates do not pose a special problem in California."²⁴⁶

As explained at length earlier, EPA believes Congress intended the Agency to grant substantial deference to California on its choice of standards that are appropriate to meet its needs. EPA has explained that "Congress has made it abundantly clear that the manufacturers would face a heavy burden in attempting to show 'compelling and extraordinary conditions' no longer exist: The Administrator, thus, is not to overturn California's judgment lightly. Nor is he to substitute his judgment for that of the State. There must be "clear and compelling" evidence that the State acted unreasonably in evaluating the relative risks of various pollutants in light of the air quality, topography, photochemistry, and climate in that State, before EPA may deny a waiver."247 Likewise, the House Committee Report explained for instance that "[t] he [1977] amendment is intended to ratify and strengthen the California waiver provision and to affirm the underlying intent of that provision, i.e., to afford California

^{245. 49} FR at 18891.

^{246.} Id.

^{247.~}Id. at 18890 n.25 (citing H.R. Rep. No. 95-294, 95th Cong., 1st Sess. 302 (1977)).

the *broadest possible discretion* in selecting the best means to protect the health of its citizens and the public welfare."²⁴⁸ EPA's past practice prior to SAFE 1, except for one instance, was consistent with this deferential stance.

In enacting section 209(b)(1), Congress struck a deliberate balance first in 1967 when it acknowledged California's serious air quality problems as well as its role as a laboratory for emissions control technology for the country, ²⁴⁹ and again, in the 1977 Amendments that allowed for California to seek and obtain waivers for standards that are less stringent than the federal standards (by amending section 209(b)(1)(A)) and also added section 177 to acknowledge that states may have air quality problems similar to California's by allowing states, subject to certain conditions, to adopt California's

^{248.} *MEMA I*, 627 F.2d at 1110 (citing H.R. Rep. No. 294, 95th Cong., 1st Sess. 301-02 (1977)) (emphasis added). Congress amended section 209(b)(1)(A) so that California's determination that its standards are as at least as protective as applicable Federal standards so that such determination may be done "in the aggregate" looking at the summation of the standards within the vehicle program.

^{249.} The CAA has been a paradigmatic example of cooperative federalism, under which "States and the Federal Government [are] partners in the struggle against air pollution." *General Motors Corp.* v. *United States*, 496 U.S. 530, 532 (1990). Motor vehicles "must be either 'federal cars' designed to meet the EPA's standards or 'California cars' designed to meet California's standards." *Engine Mfrs.*, 88 F.3d at 1079-80, 1088 ("Rather than being faced with 51 different standards, as they had feared, or with only one, as they had sought, manufacturers must cope with two regulatory standards."). *See also MEMA II*, 142 F.3d at 463.

new motor vehicle standards once waived by EPA.²⁵⁰ These provisions struck a balance between having only one national standard and having 51 different state standards by settling on two standards—a federal one and a California one that other states may also adopt. Since 1967, in various amendments to section 209, Congress has also not disturbed this reading of section 209(b)(1)(B) as calling for the review of the standards as a whole program. Likewise, Congress has also not placed any additional constraints on California's ability to obtain waivers beyond those now contained in section 209(b)(1). The Agency has thus viewed the text, legislative history, and structure of section 209(b)(1) as support for the program-level review of waiver requests as well for the conclusion that California's air quality need not be worse than the rest of the country for EPA to grant a waiver of preemption. In addition, to the extent that SAFE 1 was intended to preclude California's regulation of all greenhouse gases from light-duty vehicles, the SAFE 1 interpretation creates a structural conflict within the relevant CAA provisions and could also create an inability for California to address GHG emissions and its contribution to the serious air quality problems within the State. There is a fundamental relationship between sections 209(a) and 209(b). Section 209(a) preempts states from adopting or enforcing new motor vehicle emission standards, and section 209(b) calls for EPA to waive that preemption for California vehicular

^{250. &}quot;\$177 . . . permitted other states to 'piggyback' onto California 's standards, if the state's standards 'are identical to the California standards for which a waiver has been granted for such model year." *Motor Vehicle Mfrs. Ass'n* v. *New York State Dep't of Envtl. Conservation*, 17 F.3d 521, 525 (2d Cir. 1994).

emission standards unless EPA finds that one or more of the waiver criteria set out therein are not met. Nothing on the face of the CAA or applicable legislative history indicates that the scope of section 209(b)—the pollutants for which California may obtain a waiver—is more limited than the scope of section 209(a).²⁵¹ The D.C. Circuit has already held as much as to section 209(a): "whatever is preempted [by section 209(a)] is subject to waiver under subsection (b)."252 As demonstrated by EPA's review of the record in this decision, California's GHG emission standards at issue meet the SAFE 1 interpretation of the second waiver prong. Nevertheless, to the extent that SAFE 1 was intended to preclude all California regulation of greenhouse gases, EPA believes it improper to exclude entirely a pollutant from a waiver under section 209(b) that is otherwise preempted by section 209(a).

In addition, Congress has cited California's GHG standards and ZEV sales mandate in subsequent legislation. Federal procurement regulations direct the EPA to issue guidance identifying the makes and models

^{251.} EPA believes that, to the extent the SAFE 1 interpretation has the practical effect of defining or implementing the scope of section 209(b) differently depending on the pollutants involved, the interpretation is contrary to legislative intent and the Agency's historic practice given the criteria emission benefits of CARB's GHG emission standards and ZEV sales requirements as well as the impacts of climate change on California's local and regional air quality.

^{252.} MEMA I, 627 F.2d 1095, 1106-08 (D.C. Cir. 1979).

numbers of vehicles that are low GHG emitting vehicles.²⁵³ In a clear reference to California's motor vehicle GHG standards, Congress has required EPA when identifying those vehicles to "take into account the most stringent standards for vehicle greenhouse gas emissions applicable to and enforceable against motor vehicle manufacturers for vehicles sold anywhere in the United States."254 And in its State Implementation Plan provision regarding fleet programs required for certain non-attainment areas relating to issuing credits for cleaner vehicles, Congress stated that the "standards established by the Administrator under this paragraph . . . shall conform as closely as possible to standards which are established for the State of California for ULEV and ZEV vehicles in the same class. 255 Congress would not likely have adopted California's standards into its own legislation if it believed those standards to be preempted.

EPA also disagrees with SAFE 1's related argument that the statutory criteria must be interpreted in the context of the constitutional doctrine of "equal sovereignty." As explained in detail in Section VIII, waiver requests should be reviewed based solely on the criteria in section 209(b)(1) and the Agency should not consider constitutional issues in evaluating waiver requests. ²⁵⁶ The constitutionality of section 209 is not one of the three statutory criteria for

^{253. 42} U.S.C. 13212(f)(3).

^{254.} Id.

^{255. 42} U.S.C. 7586(f)(4).

^{256. 78} FR at 2145.

reviewing waiver requests. However, because the Agency asserted in SAFE 1 that the equal sovereignty doctrine formed a gloss on its statutory interpretation of the three criteria, EPA addresses that argument here briefly. In short, in SAFE 1, EPA stated that because section 209(b)(1) provides "extraordinary treatment" to California, the second waiver prong should be interpreted to require a "state-specific" and "particularized" pollution problem. ²⁵⁷ But section 177's grant of authority to other states to adopt California's standards undermines the notion that the regulatory scheme treats California in an extraordinary manner. Indeed, if section 209(b) is interpreted to limit the types of air pollution that California may regulate, it would diminish the sovereignty of California and the states that adopt California's standards pursuant to section 177 without enhancing any other state's sovereignty. Nor does section 209(b) impose any burden on any state. For these reasons, EPA agrees with commenters who argued that the Supreme Court's decision in Shelby County is inapposite. In section 209(b), Congress did not authorize "federal intrusion into sensitive areas of state and local policymaking."258 Rather, it underscored a foundational principle of federalism—allowing California to be a laboratory for innovation. Nor is section 209(b) an "extraordinary departure from the traditional course of relations between the States and the Federal Government."²⁵⁹ To the contrary, it is just one of many

^{257. 84} FR 51340, 51347.

^{258.} Shelby County v. Holder, 570 U.S. 529, 535, 545 (2013).

^{259.} Id.

laws Congress passes that treat States differently, and where, as discussed more fully below, Congress struck a reasonable balance between authorizing one standard and authorizing 51 standards in deciding to authorize two. SAFE 1's invocation of the rarely used equal sovereignty principle as an aid in interpreting the second waiver prong simply does not fit section 209.

SAFE 1 dismissed the Agency's traditional interpretation of the second waiver prong under which EPA reviews the same standards that California considers in making its protectiveness determination, asserting that the practical implications of reviewing standards in the "aggregate" compared to specific standards presented in a waiver request meant that the Agency would never have the discretion to determine that California did not need any subsequent standards. But nothing in section 209(b)(1)(B) can be read as calling for scrutinizing the specific California standards under the waiver.²⁶⁰ Under section 209(b)(1)(B), EPA is to grant a waiver unless California does not need "such State standards" (plural). EPA interprets section 209(b)(1)(B) to refer back to the phrase "in the aggregate" in section 209(b)(1), which was added in the 1977 CAA Amendments when Congress

^{260.} In the 2009 GHG waiver, and again in the 2013 ACC program waiver, EPA explained that the traditional approach does not make section 209(b)(1)(B) a nullity, as EPA must still determine whether California does not need its motor vehicle program to meet compelling and extraordinary conditions as discussed in the legislative history. Conditions in California may one day improve such that it may no longer have a need for its motor vehicle program, or a program designed for a particular type of air pollution problem, if the underlying specific air pollutant is no longer at issue.

removed the stringency requirements for waiver of California standards allowing instead for standards that are not as stringent as comparable federal standards, so long as the standards were "in the aggregate, at least as protective of public health and welfare as applicable Federal standards." EPA believes that referring back to section 209(b)(1) is appropriate given that it precedes the language prior to section 209(b)(1)(B) and is in accord with the deference Congress intended by the 1977 Amendments.²⁶¹ Conversely, EPA believes that under the SAFE 1 interpretation California would, of necessity, be required to make a protectiveness finding for each of the specific standards, and the Agency believes this would be an inappropriate outcome from SAFE 1. Under the 1977 Amendments, California can "include some less stringent [standards] than the corresponding federal standards."262 As previously explained, "Congress could not have given this flexibility to California and simultaneously assigned to the state the seemingly impossible task of establishing that 'extraordinary and compelling conditions' exist for each standard."263

^{261.} EPA had applied the traditional interpretation of the second waiver prong prior to the 1977 Amendments.

^{262.} See H.R. Rep. No. 294, 95th Cong., 1st Sess. 302 (1977); "In further amendments to the Act in 1977, \$209 (formerly \$208) was amended to require the U.S. Environmental Protection Agency (EPA) to consider California's standards as a package, so that California could seek a waiver from preemption if its standards 'in the aggregate' protected public health at least as well as federal standards." Motor Vehicle Mfrs. Ass'n v. NYS Dep't of Env't Conservation, 17 F.3d at 525.

^{263. 49} FR at 18890 n.24.

SAFE 1 further argued that its interpretation read the use of "such standards" consistently between the second and third waiver prongs, sections 209(b)(1)(B) and (C).²⁶⁴ It is true that section 209(b)(1)(C) employs the same phrase "such State standards" as employed in section 209(b)(1)(B), and it similarly uses that phrase to refer to standards in the aggregate. Indeed, section 209(b)(1)(C) involves an analysis of feasibility that can take more than the feasibility and impacts of the new standards into account. The feasibility assessment conducted for a new waiver request focuses on the standards in that request but builds on the previous feasibility assessments made for the standards already in the program and assesses any new feasibility risks created by the interaction between the standards in the petition and the existing standards.²⁶⁵

In sum, EPA now views as inconsistent with congressional intent the SAFE 1 interpretation, which was a flawed interpretation and also a significant departure from the traditional interpretation under which the

^{264.} Section 209(b)(1)(C) provides that no such waiver shall be granted if the Administrator finds that "such State standards and accompanying enforcement procedures are not consistent with section 7521(a) [202(a)] of this title."

^{265.} For example, in the 2013 ACC waiver that contains CARB's LEV III criteria pollutant standards and GHG emission standards, as well as the ZEV sales mandate, EPA assessed information submitted by CARB regarding the technological feasibility, lead time available to meet the requirements, and the cost of compliance and the technical and resource challenges manufacturers face in complying with the requirements to simultaneously reduce criteria and GHG emissions. 78 FR at 2131.

Agency reviews California's need for the same standards as those that the State determines are "in the aggregate" as protective of public health and welfare, under section 209(b)(1).²⁶⁶ EPA believes the traditional interpretation is, at least, the better reading of the statute.

As previously explained, in reviewing waiver requests EPA has applied the traditional interpretation in the same way for all air pollutants, criteria and GHG pollutants

^{266. 84} FR at 51345. EPA notes that in SAFE 1 the following rationale was used to interpret both 209(b)(1)(C) and then connect it with 209(b)(1)(B): "[B]ecause both sections 209(b)(1)(B) and (C) employ the term 'such state standards,' it is appropriate for EPA to read the term consistently between prongs (B) and (C). Under section 209(b)(1)(C), EPA conducts review of standards California has submitted to EPA for the grant of a waiver to determine if they are consistent with section 202(a). It follows then that EPA must read 'such state standards' in section 209(b)(1)(B) as a reference to the same standards in subsection (C)." Although the Agency has not pointed to 209(b)(1)(C) as a basis of statutory construction to support the traditional interpretation of 209(b)(1)(B), EPA nevertheless believes it is supportive. EPA notes that the term "such state standards" in 209(b)(1)(C) allows the Agency, in appropriate circumstances, to review the consistency of CARB's suite of standards, for a particular vehicle category, with section 202(a). For example, EPA evaluated all of the standards (LEV III criteria pollutant, ZEV sales mandate, and GHG standards) of the ACC program in recognition of the aggregate costs and lead time associated with CARB's standards as well as technologies that may be employed to meet more than one standard. 78 FR 2131-45. EPA's assessment under 209(b)(1)(C) is not in practice a standard-by-standard review. EPA believes it appropriate to read the entirety of 209 together, along with its purposes, in order to properly interpret its components such as 209(b)(1)(B).

alike.²⁶⁷ In SAFE 1, however, EPA reinterpreted section 209(b)(1)(B) and further set out a particularized nexus test and applied this test separately to GHG standards at issue. SAFE 1 then concluded that no nexus exists for GHG emissions in California.²⁶⁸ SAFE 1 further posited that California must demonstrate "compelling and extraordinary circumstances sufficiently different from the nation as a whole to justify standards on automobile emissions which may, from time to time, need to be more stringent than national standards."²⁶⁹ This has resulted in potentially different practical results depending on whether GHG standards or criteria emission pollutants are at issue, a distinction neither found in nor supported by the text of section 209(b)(1)(B) and legislative history.

^{267. 74} FR at 32763-65; 76 FR at 34693; 79 FR at 46256; 81 FR at 95982.

^{268.} SAFE 1 also relied on UARG v. EPA, 134 S. Ct. 2427 (2014), where the Supreme Court disagreed with the Agency's decision to regulate all sources of GHG under Titles I and V as the consequence of the Agency's section 202(a) endangerment finding for motor vehicle GHG emissions. In EPA's view upon reconsideration of SAFE 1, UARG is distinguishable because here the Agency is acting under a specific exemption to section 202(a) that allows for California to set its own standards for motor vehicle GHG standards under California state law, and thus, regulate major sources of GHG emissions within the State. California's authority to promulgate standards is neither contingent nor dependent on the Agency's section 202(a) endangerment finding for GHG. See 74 FR at 32778-80; 79 FR at 46262. Moreover, as discussed above, EPA's waiver authority under section 209(b) is coextensive with preemption under section 209(a). See MEMA I, 627 F.2d at 1107. UARG is inapplicable to the scope of preemption under section 209(a).

^{269. 84} FR at 51341.

Specifically, SAFE 1 would have the ACC program MYs 2017-2025 criteria pollutants standards subject to review under the traditional interpretation while GHG standards at issue would be subject to review under the SAFE 1 particularized nexus test or individualized scrutiny.²⁷⁰ This uneven application is even more irreconcilable given that California's motor vehicle emission program includes two GHG standards for highway heavy-duty vehicles that EPA previously reviewed under the traditional approach.²⁷¹ EPA acknowledges that ascribing different meanings to the same statutory text in the same provision here, depending on its application, "would render every statute a chameleon."272 Nothing in either section 209 or the relevant legislative history can be read as calling for a distinction between criteria pollutants and GHG standards and thus, the individualized scrutiny under

^{270.} Id. at 51337.

^{271.} The first HD GHG emissions standard waiver related to certain new 2011 and subsequent model year tractor-trailers. 79 FR 46256 (August 7, 2014). The second HD GHG emissions standard waiver related to CARB's "Phase I" regulation for 2014 and subsequent model year tractor-trailers. 81 FR 95982 (December 29, 2016).

^{272.} See States and Cities at 24 (quoting Clark v. Martinez, 543 U.S. 371, 382 (2005) and citing U.S. v. Santos, 553 U.S. 507, 522 (2008); U.S. Dep't of the Treasury v. FLRA, 739 F.3d 13, 21 (D.C. Cir. 2014)). The commenter notes that in the SAFE 1 brief, EPA claimed that its new approach to section 209(b)(1)(B) would apply "for all types of air pollutants" but EPA could point to nowhere in SAFE 1 decision where this was said. Id. at 25. And "only two sentences later," EPA acknowledged that its review under this second prong would change "depending upon which 'air quality concerns' were implicated." Id.

the SAFE 1 particularized nexus test. 273 Nothing in section 209(b) can be read as calling for EPA to waive preemption only if California seeks to enforce criteria pollutant standards. The Administrator is required to waive the preemption in section 209(a) unless California "does not need such State standards to meet compelling and extraordinary conditions."274 This is in stark contrast to, for example, section 211(c)(4)(C), which calls for a waiver of preemption only if a state demonstrates that a fuel program is "necessary" to achieve the NAAQS.²⁷⁵ Moreover, as previously noted, "[I]f Congress had intended a review of the need for each individual standard under (b)(1)(B), it is unlikely that it would have used the phrase "... does not need such state standards" (emphasis in original), which apparently refers back to the phrase "State standards . . . in the aggregate as used in the first sentence of section 209(b)(1), rather than the particular standard being considered."276 EPA has also explained that an individualized review of standards would mean that Congress "g[ave] flexibility to California and simultaneously assigned to the state the seemingly impossible tasks of establishing that 'extraordinary

^{273.} H.R. Rep. No. 294, 95th Cong., 1st Sess. 302 (1977); 49 FR at 18890 n.24.

^{274. &}lt;sup>274</sup>CAA section 209(b)(1)(B) (emphasis added).

^{275.} Section 211(c)(4)(C) allows EPA to waive preemption of a state fuel program respecting a fuel characteristic or component that EPA regulates through a demonstration that the state fuel program is necessary to achieve a NAAQS.

^{276. 49} FR at 18890.

and compelling conditions' exist for each less stringent standard."277

Similarly, nothing in either section 209 or legislative history can be read as requiring EPA to grant GHG standards waiver requests only if California's GHG pollution problem is the worst in the country. 278 "There is no indication in either the statute or the legislative history that . . . the Administrator is supposed to determine whether California's standards are in fact sagacious and beneficial."279 And most certainly, nothing in either section 209 or the legislative history can be read as calling for EPA to draw a comparison between California's GHG pollution problem and the rest of the country (or world) when reviewing California's need for GHG standards. Instead, the crucial consequence of the 1977 Amendment was to require waiver grants for California's specific standards that are part of the State's overall approach to reducing vehicle emissions to address air pollution even if those specific standards might not be needed to address compelling and extraordinary conditions.²⁸⁰ Thus,

^{277.} Id. at 18890 n.24.

^{278.} Id. at 18891.

^{279.} Ford Motor Co., v. EPA, 606 F.2d 1293, 1302 (D.C. Cir. 1979).

^{280. 74} FR at 32761 ("Congress decided in 1977 to allow California to promulgate individual standards that are not as stringent as comparable federal standards, as long as the standards are 'in the aggregate, at least as protective of public health and welfare as applicable federal standards."); "[T]he 1977 amendments significantly altered the California waiver provision." *Ford Motor Co.*, 606 F.2d 1293, 1302 (D.C. Cir. 1979).

"even if it were true that California's [GHG] problem is, . . . no worse than some other areas of the country, this does not mean that [GHG] do not pose a special problem in California."²⁸¹ Rather, "EPA's practice [is] to leave the decisions on controversial matters of public policy, such as whether to regulate [GHG] emissions, to California."²⁸²

In addition, in Title II, Congress established only two programs for control of emissions from new motor vehicles: EPA emission standards adopted under the Clean Air Act and California emission standards adopted under its state law. And states other than California may not "tak[e] any action that has the effect of creating a car different from those produced to meet either federal or California emission standards, a so-called 'third vehicle."²⁸³

As previously explained, and noted in the Notice of Reconsideration, since the grant of the initial GHG waiver request in 2009, the Agency has applied the traditional interpretation in granting two additional waivers for CARB's Heavy-Duty Vehicle GHG emission standards and these GHG standards are now part of California's motor vehicle program, but EPA did not address these waivers in SAFE 1.²⁸⁴ It also bears note that, given the limited analysis and application of the SAFE 1 interpretation

^{281. 49} FR at 18891.

^{282. 43} FR at 25735.

^{283.} Motor Vehicle Mfrs. Ass'n v. NYS Dep't of Env't Conservation, 17 F.3d 521, 526, 528 (2d Cir. 1994).

 $^{284.\;\;79\;\}mathrm{FR}\;46256$ (August 7, 2014); 81 FR 95982 (December 29, 2016).

of the second waiver prong, it is uncertain whether the traditional interpretation remains otherwise applicable to earlier model year GHG standards under prior waivers. Ambiguity also applies to SAFE 1's interpretation of this prong in respect to all criteria pollutant standards in the ACC program. While SAFE 1 stated it was only applicable to the GHG standards at issue, in at least one instance the Agency indicated that the SAFE 1 interpretation could also be applicable to future evaluation of waiver requests for criteria pollutant standards.²⁸⁵ This uncertainty between these statements in SAFE 1 further highlights the inappropriateness of the new interpretation of the second prong.

^{285. 84} FR at 51341 n.263. "EPA determines in this document that GHG emissions, with regard to the lack of a nexus between their State-specific sources and their State specific impacts, and California's GHG standard program, are sufficiently distinct from criteria pollutants and traditional, criteria pollutant standards, that it is appropriate for EPA to consider whether California needs its own GHG vehicle emissions program. EPA does not determine in this document and does not need to determine today how this determination may affect subsequent reviews of waiver applications with regard to criteria pollutant control programs." (Emphasis added). See also id. at 51344 n.268 ("EPA is adopting an interpretation of CAA section 209(b)(1)(B), specifically its provision that no waiver is appropriate if California does not need standards "to meet compelling and extraordinary conditions," similar to the interpretation that it adopted in the 2008 waiver denial but abandoned in the 2009 and 2013 waiver grants, and applying that interpretation to determine to withdraw the January 2013 waiver for California's GHG and ZEV program for model years 2021 through 2025"), and at 51346 ("EPA therefore views this interpretation and application of CAA section 209(b)(1)(B) set forth here as, at minimum, a reasonable one that gives appropriate meaning and effect to this provision.").

In sum, for the reasons noted above, EPA is withdrawing the SAFE 1 interpretation and reinstating certain aspects of the ACC program waiver that were earlier granted under the traditional interpretation and approach. EPA concludes it erred by not properly evaluating the statutory interpretation of section 209, the associated legislative history including the policy deference that should be afforded to California to address its serious air quality problems and to serve as a laboratory for the country, and because the "need" for a motor vehicle emission program and related standards within the program are necessarily better viewed as a comprehensive and interrelated effort to address the range of air quality problems facing California.²⁸⁶ At the

^{286.} As noted previously, in the context of evaluating the "need" for California's motor vehicle emission standards the Agency is informed by the legislative history from 1967 and 1977, whereby California is properly viewed as a laboratory for the country and that its policy decisions on how best to address its serious air quality issues, and that deference on the question of "need" is in order. Therefore, EPA believes it misapplied the concept of deference in the context of the second waiver prong application in SAFE 1. See e.g., 84 FR at 51344 n.268. While EPA believes it appropriate to not defer when it is interpreting its own statute, the Agency nevertheless determines that California's policy choices in term of its "need" in how best to address compelling and extraordinary conditions in California requires deference by the Agency. This is consistent with EPA's longstanding waiver practice and its integration of the legislative history behind section 209. In any event, EPA would reach the same conclusions regarding the second waiver prong even if it did not defer to California regarding the nature of its air quality problems. 86 FR at 74489 ("The 2009 Endangerment Finding further explained that compared with a future without climate change, climate change is expected to increase tropospheric ozone pollution over broad areas of the U.S., including in the largest metropolitan

same time, EPA notes that the traditional interpretation is reasonable and consistent with the text, structure and congressional intent and purpose of section 209(b) and EPA is thus confirming that the traditional interpretation of section 209(b)(1)(B) was appropriate and is, at least, the better interpretation.²⁸⁷

2. California Needs the GHG Standards and ZEV Sales Mandate Even Under the SAFE 1 Interpretation

Even if the SAFE 1 interpretation of section 209 (b)(1)(B) was appropriate, the record of both the ACC program waiver and SAFE 1 proceeding demonstrate that California has a need for the GHG standards and ZEV sales mandate at issue under the SAFE 1 interpretation as well. The opponents of the waiver (including EPA in SAFE 1) did not met their burden of proof to demonstrate that California does not need its GHG emission standards and ZEV sales mandate, whether individually or as part of California's motor vehicle emission program, to meet compelling and extraordinary conditions.²⁸⁸

areas with the worst tropospheric ozone problems, and thereby increase the risk of adverse effects on public health (74 FR 66525)."). See also 86 FR at 74492.

^{287. &}quot;The interpretation that my inquiry under (b)(1)(B) goes to California's need for its own mobile source program is borne out not only by the legislative history, but by the plain meaning of the statue as well." 49 FR at 18890.

^{288.} EPA notes that by this action it is rescinding the interpretation of section 209(b)(1)(B) as set forth in SAFE 1. Nevertheless, EPA believes it appropriate to address comments received that suggest the SAFE 1 interpretation was not only correct,

As previously explained, the 1977 CAA Amendments allow California to promulgate standards that might not be considered needed to meet compelling and extraordinary circumstances but would nevertheless be part of California's overall approach of reducing vehicle emissions to address air pollution in California. Thus, CARB may now design motor vehicle emission standards, individually, that might sometimes not be as stringent as federal standards but collectively with other standards would be best suited for California air quality problems because under the 1977 Amendments, California can "include some less stringent [standards] than the corresponding federal standards." And EPA is "required to give very substantial deference to California's judgments on this score." 291

Indeed, as EPA noted in the ACC program waiver, Congress intentionally provided California the broadest possible discretion in adopting the kind of standards in its motor vehicle program that California determines are appropriate to address air pollution problems that exist in California, whether or not those problems are only

but that the factual record supported the SAFE 1 withdrawal of the ACC waiver based on this interpretation.

^{289.} See Ford Motor Co., v. EPA, 606 F.2d 1293, 1296-97 (D.C. Cir. 1979); See H.R. Rep. No. 294, 95th Cong., 1st Sess. 302 (1977).

 $^{290.\;\;43\;\}mathrm{FR}\;25729,25735$ (June 14, 1978). See Ford Motor Co., 606 F.2d at 1296-97.

 $^{291.\;\;40\;\}mathrm{FR}$ at 23104. See also LEV I (58 FR 4166 (January 13, 1993)) Decision Document at 64.

local or regional in nature, and to protect the health and welfare of its citizens:

Congress did not intend this criterion to limit California's discretion to a certain category of air pollution problems, to the exclusion of others. In this context it is important to note that air pollution problems, including local or regional air pollution problems, do not occur in isolation. Ozone and PM air pollution, traditionally seen as local or regional air pollution problems, occur in a context that to some extent can involve long range transport of this air pollution or its precursors. This long range or global aspect of ozone and PM can have an impact on local or regional levels, as part of the background in which the local or regional air pollution problem occurs. ²⁹²

In the context of implementing section 209(b)(1)(B) and assessing the "need" for California's standards even

^{292. 78} FR at 2128-29. See "Our Changing Climate 2012 Vulnerability & Adaptation to the Increasing Risks from Climate Change in California." Publication # CEC-500-2012- 007. Posted: July 31, 2012; available at https://ucanr.edu/sites/Jackson_Lab/files/155618.pdf at 4 ("Higher temperatures also increase ground-level ozone levels. Furthermore, wildfires can increase particulate air pollution in the major air basins of California. Together, these consequences of climate change could offset air quality improvements that have successfully reduced dangerous ozone concentrations. Given this "climate penalty," as it is commonly called, air quality improvement efforts in many of California's air basins will need to be strengthened as temperatures increase in order to reach existing air quality goals.").

under the SAFE 1 interpretation, EPA sees no reason to distinguish between "local or regional" air pollutants versus other pollutants that may be more globally mixed. Rather, it is appropriate to acknowledge that all pollutants and their effects may play a role in creating air pollution problems in California and that EPA should provide deference to California in its comprehensive policy choices for addressing them. Again, even if a new interpretation of section 209(b)(1)(B) were appropriate in SAFE 1, and EPA believes it is not, it is important to note that historically, criteria pollutant reductions have been relevant to section 209(b)(1)(B). As previously noted, nothing in section 209(b) can be read as calling for EPA to waive preemption only if California seeks to enforce criteria pollutant standards. The Administrator is required to waive the preemption in section 209(a) unless California "does not need such State standards to meet compelling and extraordinary conditions."293 As also previously noted this is in stark contrast to, for example, section 211(c)(4)(C), which calls for a waiver of preemption only if a state demonstrates that a fuel program will result in criteria pollutant reductions that will enable achievement of applicable NAAQS.

The first section below focuses on criteria pollution reduction, which has long been relevant to section 209 (b)(1)(B). EPA has never put in doubt that California's serious criteria air pollution problems (such as NAAQS nonattainment and the factors that give rise to those conditions, including the geographic and climate conditions in the State, the number of motor vehicles in California,

^{293.} CAA section 209(b)(1)(B) (emphasis added).

and local and regional air quality) are "compelling and extraordinary," or that California "needs" regulations that address such emissions in order to achieve every fraction of criteria pollutant emissions it can achieve.²⁹⁴ The factual record before the Agency in 2013 and again in 2019 includes ample documentation of criteria emission reductions from California's GHG standards and ZEV sales mandate.²⁹⁵ Nothing in the record is sufficient to demonstrate that California does not need the ACC program (or the motor vehicle emission program) or, in the context of the SAFE 1 interpretation, the specific GHG emission standards and the ZEV sales mandate to meet compelling needs related to criteria pollution. These benefits have a clear connection to California's "need" for its specific GHG standards and ZEV sales mandate, at issue under the waiver. The second section below focuses on the GHG reduction benefits of California's GHG standards and ZEV sales mandate. EPA acknowledges that California is particularly impacted by climate change, including increasing risks from record-setting fires,

^{294.} In SAFE 1, EPA found that California's criteria pollution conditions remain "compelling and extraordinary and that California needs standards to produce any and all reductions in criteria pollutant emissions." 84 FR at 51344, 51346.

^{295.} When California originally adopted a ZEV sales mandate into its regulations, a significant factor in support of its action was addressing criteria pollutant emissions. In SAFE 1 EPA acknowledged that California's ZEV mandate initially targeted only criteria pollution. 84 FR at 51329. EPA's 2013 waiver grant recognized that with California's ACC program California had shifted to relying on the ZEV requirements to reduce both criteria and GHG pollution. 78 FR at 2114.

heat waves, storm surges, sea-level rise, water supply shortages and extreme heat, and that climate-change impacts in California are therefore "compelling and extraordinary conditions" for which California needs the GHG standards and ZEV sales mandate.

a. GHG Standards and ZEV Sales Mandates Have Criteria Emission Benefits

As shown below, criteria pollutant reductions are demonstrably connected to California's "need" for its GHG standards and ZEV sales mandate at issue under the waiver. ²⁹⁶ EPA first concluded that there is a "logical link between the local air pollution problem of ozone and GHGs" in the 2009 California GHG waiver by explaining,

^{296.} In response to comments arguing that upstream emission benefits should not be considered in determining the criteria pollutant benefits of CARB' standards or that it is inappropriate to elevate stationary source criteria pollutant emissions into a makeor-break factor in waivers for motor vehicle emission programs, EPA believes it appropriate to reiterate the air quality problems facing California, as evidenced by NAAQS attainment challenges. Waiver practice and applicable case law, as previously noted, afford California wide deference in its policy and regulatory approaches in addressing these challenges. Therefore, EPA believes that to the degree a nexus between CARB's standards and addressing its serious air quality problems is required, that it is reasonable to base the need on related criteria emission impacts. EPA notes that, in setting its federal light-duty vehicle GHG standards, it is afforded discretion under the CAA to consider upstream emission impacts and does include such consideration in its own rulemakings. 77 FR 62624, 62819 (October 15, 2012) (taking fuel related upstream GHG emissions into account in setting compliance values for vehicle GHG emissions standards).

for instance, that "the impacts of global climate change can nevertheless exacerbate this local air pollution problem."²⁹⁷ Moreover, as previously explained, in two additional GHG waiver requests and associated EPA waiver decisions since the 2009 GHG waiver, EPA acknowledged that CARB had demonstrated the need for GHG standards to address criteria pollutant concentrations in California. In the 2014 HD GHG waiver request, CARB projected, for example, "reductions in NO X emissions of 3.1 tons per day in 2014 and one ton per day in 2020" in California. ²⁹⁸

^{297. 74} FR at 32763. According to California, "California's high ozone levels-clearly a condition Congress considered-will be exacerbated by higher temperatures from global warming...[T]here is general consensus that temperature increases from climate change will exacerbate the historic climate, topography, and population factors conducive to smog formation in California, which were the driving forces behind Congress's inclusion of the waiver provision in the Clean Air Act." Id. (quoting comments submitted by CARB during the 2009 reconsideration). CARB also explained that "the factors that cause ozone are primarily local in nature and [] ozone is a local or regional air pollution problem, the impacts of global climate change can nevertheless exacerbate this local air pollution problem. Whether or not local conditions are the primary cause of elevated concentrations of greenhouse gases and climate change, California has made a case that its greenhouse gas standards are linked to amelioration of California's smog problems There is a logical link between the local air pollution problem of ozone and California's desire to reduce GHGs as one way to address the adverse impact that climate change may have on local ozone conditions." Id.

^{298. 79} FR at 46261. See *also* 81 FR at 95985-86 n.27 (referencing Resolution 13-50's statements supporting California's continued need for its own motor vehicle program in order to meet serious ongoing pollution problems).

In SAFE 1, EPA distinguished prior GHG waivers from the ACC program GHG waiver solely on grounds of how CARB attributed the pollution benefits in its waiver request. EPA explained that CARB had linked those prior waived GHG standards to criteria pollutant benefits but had not done so in the ACC program waiver request: "California's approach in its ACC program waiver request differed from the state's approach in its waiver request for MY 2011 and subsequent heavy-duty tractor-trailer GHG standards, where California quantified NO X emissions reductions attributed to GHG standards and explained that they would contribute to PM and ozone NAAQS attainment."²⁹⁹ Moreover, how CARB attributes the pollution reductions for accounting purposes from its various standards does not reflect the reality of how the standards deliver emissions reductions and should not drive whether or not a waiver can be withdrawn. EPA believes, based on its historical deference to CARB in waiver proceedings, that CARB is entitled to this discretion.

EPA also believes that prior waiver decisions indicate that the "approach" taken by California in its waiver requests needs to be carefully assessed and understood by the Agency before discounting the benefits of its mobile source emission standards. The characterization of CARB's "approach," as not calling out criteria emissions benefits (such as upstream criteria emission benefits) of GHG standards, was incorrect and should not have

 $^{299.\;\;84\;}FR$ at 51337 n.252 (citing 79 FR at 46256, 46257 n.15, 46261, 46262 n.75).

undermined EPA's findings and grant of the initial ACC program waiver request for the following reasons: (1) As previously noted, the ACC program standards are interrelated and all serve to reduce both criteria and GHG pollution; (2) CARB conducted a combined emissions analysis of the elements of the ACC program because the program was designed to work as an integrated whole: and (3) EPA has always considered California's standards as a whole or "in the aggregate" under the traditional interpretation of section 209(b)(1)(B).³⁰⁰ EPA noted the associated criteria pollutant and GHG emissions benefits for the whole ACC program: "the ACC program will result in reductions of both criteria pollutants and GHG emissions that, in the aggregate, are more protective than the pre-existing federal standards."301 EPA also made the requisite finding that California's protectiveness finding for the ACC program was not arbitrary and capricious, under section 209(b)(1)(A), by explaining that "California's ZEV and GHG emission standards are an addition to its LEV program."302

In SAFE 1, EPA further asserted that "California's responses to the SAFE proposal do not rebut the Agency's views that the ZEV standards for MY 2021-2025 are inextricably interconnected with the design and purpose

^{300.} ZEV ISOR, EPA-HQ-OAR-2012-0562-0008 at 72; CARB Supplemental Comments, EPA-HQ-OAR-2012-0562-0373 at 3.

^{301. 74} FR at 2122.

^{302.} Id. at 2125.

of California's overall GHG reduction strategy."303 For the following reasons, however, EPA was also incorrect in the assessment of criteria emission benefits of CARB's ZEV sales mandate. EPA focused on only the following snippet from one salient paragraph in CARB's 2012 waiver request as support for the lack of criteria emissions benefits: "There is no criteria emissions benefit from including the ZEV proposal in terms of vehicle (tank-to-wheel or TTW) emissions. The LEV III criteria pollutant fleet standard is responsible for those emission reductions in the fleet; the fleet would become cleaner regardless of the ZEV regulation because manufacturers would adjust their compliance response to the standard by making less polluting conventional vehicles."304 But, as discussed above, that was merely an attribution of benefits and did not reflect the practical reality of how California's standards work. Moreover, the paragraph in its entirety goes on to explain that CARB's ZEV sales mandate would achieve criteria emission reductions: "However, since upstream criteria and PM emissions are not captured in the LEV III criteria pollutant standard, net upstream emissions are reduced through the increased use of electricity and concomitant reductions in fuel production."305

^{303. 84} FR at 51337.

^{304.} *Id.* at 51337, 51330, 51337, 51353-54, 51356, 51358.

^{305. 2012} Waiver Request at 15-16. CARB also noted that criteria and PM emission benefits will vary by region throughout the State depending on the location of emission sources. Refinery emission reductions will occur primarily in the east Bay Area and South Coast region where existing refinery facilities operate. As refinery operations reduce production and emissions, the input

It bears note that this attribution of criteria pollutant reductions was similar to the one that CARB made almost a decade ago for the 2009 GHG waiver request. 306 For example, CARB provided "extensive evidence of its current and serious air quality problems and the increasingly stringent health-based air quality standards and federally required state planning efforts to meet those standards firmly."307 The States and Cities also commented that "the attribution CARB made as part of its waiver request was never intended to, and did not, establish the absence of any vehicular emission benefits from the ZEV standard." EPA believes that CARB's statement was merely a "simplification that distinguished the standards based on the *primary* objectives of the two, complementary standards."308 EPA agrees that the

and output activities, such as truck and ship deliveries, will also decline. This includes crude oil imported through the Los Angeles and Oakland ports, as well as pipeline and local gasoline truck distribution statewide. EPA again notes that in its light-duty vehicle GHG rulemaking in 2012 it also noted the upstream emission impacts. 77 FR at 62819.

306. "The establishment of greenhouse gas emission standards will result in a reduction in upstream emissions (emission due to the production and transportation of the fuel used by the vehicle) of greenhouse gas, criteria and toxic pollutants due to reduced fuel usage." EPA-HQ-OAR-2006-0173-0010.107 at 8.

307. CARB, EPA-HQ-OAR-2012-0562-0371. CARB estimated benefits of the ZEV and GHG standards for calendar years by which the South Coast air basin must meeting increasingly stringent NAAQS for ozone: 2023, 2031, and 2037. States and Cities app. A at 2-4, app. C at 8-9.

308. States and Cities at 31 (original emphasis).

record from 2013, and 2019, demonstrates that CARB's attribution of short-term emissions benefits did not undercut the long-term vehicular emission benefits of the ZEV standards. Thus, regardless of how the emissions reductions are attributed, the GHG standards and ZEV sales mandate drive reductions in criteria pollution.

EPA has also consistently explained that "consideration of all the evidence submitted concerning a waiver decision is circumscribed by its relevance to those questions . . . consider[ed] under section 209(b)."³⁰⁹ And so, as earlier noted, any reconsideration of a prior waiver decision must comport with criteria in section 209(b)(1) as well as have record support. Moreover, in prior waiver requests for ZEV sales mandate requirements, CARB has discussed criteria pollutant emissions reductions because of the mandate for sale of vehicles that have zero emissions. ³¹⁰ CARB's 2012 waiver request also indicated the clear intent regarding the evolution of the ZEV program and California's decision to focus both on criteria pollutant and GHG reductions. ³¹¹ EPA's reading of and reliance on

^{309. 74} FR at 32748. See also 78 FR at 2115.

^{310. 71} FR 78190 (December 28, 2006); 75 FR 11878 (March 12, 2010) and 76 FR 61095 (October 3, 2011).

^{311.} See 2012 Waiver Request at 2. At the December 2009 hearing, the Board adopted Resolution 09-66, reaffirming its commitment to meeting California's long term air quality and climate change reduction goals through commercialization of ZEV technologies. The Board further directed staff to consider shifting the focus of the ZEV regulation to both GHG and criteria pollutant emission reductions, commercializing ZEVs and PHEVs in order to

the snippet from CARB's waiver request describing the ZEV sales mandate requirements in the ACC program was both incorrect and improper, as well as contrary to congressional intent and EPA's historic practice of affording broad discretion to California in selecting the best means for addressing the health and welfare of its citizens.

b. California Needs Its Standards To Address the Impacts of Climate Change in California

Under section 209(b)(1)(B), EPA is to grant a waiver request unless California does not need the standards at issue to address "compelling and extraordinary conditions." In applying the traditional approach, EPA has consistently reasoned that "compelling and extraordinary conditions" refers primarily to the factors that tend to produce higher levels of pollution in California geographical and climatic conditions (like thermal inversions) that, when combined with large numbers and high concentrations of automobiles, create serious air pollution problems. 312 These conditions continue to exist in California and CARB, since the initial 2009 GHG waiver, has consistently drawn attention to the existential crisis that California faces from climate change and maintained that air quality issues associated with GHG emissions have exacerbated this crisis and have yet to attenuate.³¹³

meet the 2050 goals, and to take into consideration the new LEV fleet standards and propose revisions to the ZEV regulation accordingly.

^{312. 49} FR at 18890 (citing legislative history).

^{313. 2012} Waiver Request at 1.

EPA now recognizes that CARB, as part of its original waiver request and in comments in response to SAFE 1, submitted ample evidence of multiple ways California is particularly impacted by climate change, including increasing risks from record-setting fires, heat waves, storm surges, sea-level rise, water supply shortages and extreme heat; in other words that GHG emissions contribute to local air pollution, and that climate-change impacts in California are "compelling and extraordinary conditions." For example, CARB noted that "[r]ecordsetting fires, deadly heat waves, destructive storm surges, loss of winter snowpack—California has experienced all of these in the past decade and will experience more in the coming decades. California's climate—much of what makes the State so unique and prosperous—is already changing, and those changes will only accelerate and intensify in the future. Extreme weather will be increasingly common as a result of climate change. In California, extreme events such as floods, heat waves, droughts and severe storms will increase in frequency and intensity. Many of these extreme events have the potential to dramatically affect human health and well-being, critical infrastructure and natural systems."314 Within the

^{314.} CARB supplemental comment at EPA-HQ-OAR-2012-0562-0371. CARB notes that EPA's reasoning that the "compelling and extraordinary conditions" criteria should be viewed as a "program as a whole" was upheld as "eminently reasonable" in $ATA \times EPA$, 600 F.3d 624, 627-29 (D.C. Cir. 2010), and that the ACC program appropriately integrates the passenger vehicle program to address multiple pollutant types, which also reflects the intent of Congress in 1977 to broaden California's discretion to adjust its program as needed (Ford Motor Co. v. EPA, 606 F.2d at 1294). This comment extensively lays out the compelling and extraordinary

ACC waiver request, CARB provided a summary report on the third assessment from the California Climate Change Center (2012), which described dramatic sea level rises and increases in temperatures in California and associated impacts on local air quality and other conditions in California.³¹⁵

conditions associated with California's air quality challenges and the need to reduce criteria emissions and greenhouse gas emissions associated with CARB's ZEV sale mandate and GHG standards. *Id.* at 5 ("The critical nature of the LEV III regulation is also highlighted in the recent effort to take a coordinated look at strategies to meet California's multiple air quality and climate goals well into the future. This coordinated planning effort, Vision for Clean Air: A Framework for Air Quality and Climate Planning (Vision for Clean Air) demonstrates the magnitude of the technology and energy transformation needed from the transportation sector and associated energy production to meet federal standards and the goals set forth by California's climate change requirements.").

315. 78 FR at 2129 ("To the extent that it is appropriate to examine the need for CARB's GHG standards to meet compelling and extraordinary conditions, as EPA discussed at length in its 2009 GHG waiver decision, California does have compelling and extraordinary conditions directly related to regulations of GHG. EPA's prior GHG waiver contained extensive discussion regarding the impacts of climate change in California. In addition, CARB has submitted additional evidence in comment on the ACC waiver request that evidences sufficiently different circumstances in California. CARB notes that 'Record-setting fires, deadly heat waves, destructive storm surges, loss of winter snowpack—California has experienced all of these in the past decade and will experience more in the coming decades. California's climate—much of what makes the state so unique and prosperous—is already changing, and those changes will only accelerate and intensify in the future. Extreme weather will be increasingly common as a result of climate change. In California, extreme events such as floods, heat waves, droughts

To the extent that SAFE 1 relied on the premise that GHG emissions from motor vehicles located in California become globally-mixed as part of global climate change, and therefore do not pose a local air quality issue (placing aside the impacts of heat on ozone as well as air quality impacts from the dramatic increase in wildfires), EPA notes that in addition to the record from the ACC waiver proceeding noted above, the SAFE 1 record contains sufficient and unrefuted evidence that there can be locally elevated carbon dioxide concentrations resulting from nearby carbon dioxide emissions. This can have local

and severe storms will increase in frequency and intensity. Many of these extreme events have the potential to dramatically affect human health and well-being, critical infrastructure and natural systems.") ("Our Changing Climate 2012 Vulnerability & Adaptation to the Increasing Risks from Climate Change in California. Publication # CEC-500-2012-007. Posted: July 31, 2012; available at http://www.climatechange.ca.gov/adaptation/third-assessment "). EPA also noted that "the better interpretation of the text and legislative history of this provision is that Congress did not intend this criterion to limit California's discretion to a certain category of air pollution problems, to the exclusion of others. In this context it is important to note that air pollution problems, including local or regional air pollution problems, do not occur in isolation. Ozone and PM air pollution, traditionally seen as local or regional air pollution problems, occur in a context that to some extent can involve long range transport of this air pollution or its precursors. This longrange or global aspect of ozone and PM can have an impact on local or regional levels, as part of the background in which the local or regional air pollution problem occurs." 78 FR at 2128.

316. CARB comment at EPA-HQ-OAR-2018-0283-5054 at 305-06 (California's Fourth Climate Assessment; https://www.energy.ca.gov/sites/default/files/2019-12/Governance_External_Ekstrom_ada.pdf).

impacts on, for instance, the extent of ocean acidification.³¹⁷ Thus, like criteria pollution, emissions of GHGs can lead to locally elevated concentrations with local impacts, in

317. See, for example, reports from California's Fourth Climate Change Assessment, "California Mussels as Bio-indicators of Ocean Acidification," available at https://www.energy.ca.gov/sites/default/ files/2019-12/Oceans CCCA4-CNRA-2018-003 ada.pdf ("Because of the coupling between natural (upwelling-driven) and anthropogenic (CO2 emission-driven) processes, California waters are already experiencing declines in pH that are not expected in other areas of the world's oceans for decades (Feely et al. 2008; Chan et al. 2017). These perturbations to seawater chemistry join others associated with changing seawater temperatures (García-Reyes and Largier 2010) and reductions in ocean oxygenation (Bograd et al. 2008; Chan et al. 2008). Therefore, marine communities along the coast of California are increasingly subjected to a suite of concurrent environmental stressors. Substantial impetus exists to understand, quantify, and project biological and ecological consequences of these stressors, which current work suggests may be pervasive and diverse (Kroeker et al. 2010, 2013; Gaylord et al. 2015)."). Further, evidence in the record from a 2019 study demonstrated that locally enhanced carbon dioxide concentrations above Monterey Bay, California, fluctuate by time of day likely because of the magnitude of nearby urban carbon dioxide pollution and the effects of topography on offshore winds, and that this fluctuation increases the expected rate of acidification of the Bay. See Northcott, et al., Impacts of urban carbon dioxide emissions on sea-air flux and ocean acidification in nearshore waters, PLoS ONE (2019). For decades, the monthly average carbon dioxide concentrations off California's coast have been consistently higher and more variable than those at Mauna Loa (which are commonly used as the global measurements). In fact, another more recent study shows that the waters of the California Current Ecosystem, off the coast of Southern California, have already acidified more than twice as much as the global average. E.g., Cal. Office of Environmental Health Hazard Assessment, Atmospheric Greenhouse Gas Concentrations (Feb. 11, 2019).

addition to the longer-term global impacts resulting from global increases in GHG concentrations.

Finally, in demonstrating the need for GHG standards at issue, CARB attributed GHG emissions reductions to vehicles in California. For instance, "CARB project[ed] that the standards will reduce car CO 2 emissions by approximately 4.9%/year, reduce truck CO 2 emissions by approximately 4.1%/year (the truck CO 2 standard target curves move downward at approximately 3.5%/ year through the 2016-2021 period and about 5%/year from 2021-2025), and reduce combined light-duty CO 2 emissions by approximately 4.5%/year from 2016 through 2025."318 CARB also projected that its GHG emissions standards for MYs 2017-2025 will reduce fleet average CO 2 levels by about 34 percent from MY 2016 levels of 251 g/ mile down to about 166 g/mile, based on the projected mix of vehicles sold in California."319 CARB further noted that there might be a GHG emission deficit if only the Federal GHG standards were implemented in California. 320 The GHG emissions from California cars, therefore, are particularly relevant to both California's air pollution problems and GHG standards at issue.

In SAFE 1, EPA dismissed California's "need" for the GHG standards at issue because their impact on GHG emissions would be too small to "meaningfully address global air pollution problems of the sort associated with GHG

^{318. 78} FR at 2139.

^{319.} Id. at 2135.

^{320.} Id. at 2122.

emissions": "[T]he most stringent regulatory alternative considered in the 2012 final rule and [Final Regulatory Impact Analysis]..., which would have required a seven percent average annual fleetwide increase in fuel economy for MYs 2017-2025 compared to MY 2016 standards, was forecast to decrease global temperatures by only 0.02 °C in 2100."321 EPA also received similar comments in response to the Notice of Reconsideration. But since the inception of the waiver program, EPA has never applied a test to determine whether a California waiver request under 209(b)(1) would independently solve a pollution problem. EPA has never applied a de minimis exemption authority to California waiver request under section 209(b)(1).322 EPA believes there is no basis for exercise of such a test under section 209(b), considering that CARB continues to maintain that emissions reductions in California are essential for meeting the NAAQS. 323 EPA has reiterated that "California's policy judgment that an incremental, directional improvement will occur and is worth pursuing is entitled, in EPA's judgment, to great deference."324

^{321. 84} FR at 51349.

^{322.} See, e.g., 74 FR at 32766 ("As noted by the Supreme Court in *Massachusetts* v. *EPA*, while it is true that regulating motor vehicle GHG emissions will not by itself reverse global warming, a reduction in domestic automobile emissions would slow the pace of global emissions increase no matter what happens with regard to other emissions.").

^{323.} See Alabama Power Co. v. Costle, 636 F.2d 323, 360-66, n.89 (D.C. Cir. 1979).

^{324. 74} FR at 32766 ("Under this approach, there is no need to delve into the extent to which the GHG standards at issue here would address climate change or ozone problems. That is an issue

As the Supreme Court has recognized, "[a]gencies, like legislatures, do not generally resolve massive problems in one fell regulatory swoop. . . They instead whittle away at them over time, refining their preferred approach as circumstances change and as they develop a more nuanced understanding of how best to proceed."³²⁵ And so, in the ACC program waiver decision, EPA also explained that "[t]he issue of whether a proposed California requirement is likely to result in only marginal improvement in air quality not commensurate with its cost or is otherwise an arguably unwise exercise of regulatory power is not legally pertinent to my decision under section 209."³²⁶

Further, nothing in either section 209 or the legislative history could be read as requiring EPA to grant GHG standards waiver requests only if California's GHG pollution problem is the worst in the country.³²⁷ CARB further demonstrated a "need" for its GHG standards by projecting GHG emissions reductions deficits from implementation of only the Federal GHG program in

appropriately left to California's judgment.... Given the comments submitted, however, EPA has also considered an alternative interpretation, which would evaluate whether the program or standards has a rational relationship to contributing to amelioration of the air pollution problems in California. Even under this approach, EPA's inquiry would end there. California's policy judgment that an incremental, directional improvement will occur and is worth pursuing is entitled, in EPA's judgment, to great deference.").

^{325.} Massachusetts v. EPA, 549 U.S. 497, 524 (2007).

^{326. 78} FR at 2134.

^{327. 49} FR at 18891.

California. "[I]f a National Program standard was theoretically applied only to California new vehicle sales alone, it might create a GHG deficit of roughly two million tons compared to the California standards." 328

3. California's ZEV Sales Mandate as Motor Vehicle Control Technology Development

Congress also envisioned that California's other role under section 209(b) would be an innovative laboratory for motor vehicle emission standards and control technology. California is to serve as "a kind of laboratory for innovation" and to "blaze its own trail with a minimum of federal oversight. California's "unique [air pollution] problems and [its] pioneering efforts justif[ied] a waiver of the preemption section." Congress stressed that California should serve the Nation as a "testing area" for more protective standards." In the 2009 GHG waiver,

^{328. 78} FR at 2122 (citing EPA-HQ-OAR-2012-0562-0374 at 3). CARB also noted that "to the extent a manufacturer chooses not to exercise their National Program compliance option in California this would actually provide additional GHG benefits in California, so compliance in California can never yield fewer cumulative greenhouse gas reductions from the industry wide fleet certified in California." *Id.* at 2122 n.61.

^{329.} MEMA I, 627 F.2d 1095, 1111 (D.C. Cir. 1979).

^{330.} Ford Motor Co., v. EPA, 606 F.2d 1293, 1297 (D.C. Cir. 1979).

^{331.} S. Rep. No. 90-403, at 33 (1967).

^{332.} Id.

for example, EPA explained that "the basic nature of the compromise established by Congress [is that] California could act as the laboratory for the nation with respect to motor vehicle emission control, and manufacturers would continue to face just two sets of emissions standards— California's and EPA's."333 California's ZEV sales mandates have so far supported development of technologies such as battery electric and fuel cell vehicles that embody the pioneering efforts Congress envisaged. EPA acknowledged this important role in the ACC program waiver by explaining that California needs the ZEV sales mandate requirement to ensure the development and commercialization of technology required for the future, deeper vehicular emission reductions California will have to attain to meet its NAAQS obligations as well as achieve other long-term emission goals of new vehicle sales between 2040 and 2050. 334 In SAFE 1, however, EPA did not consider this additional role carved out in section 209(b)(1) for California as a proven ground for motor vehicle control emissions technology. 335

In sum, while nothing in section 209 or the legislative history limits EPA's waiver authority to standards that

^{333. 74} FR at 32763.

^{334. 78} FR at 2123, 2130-31.

^{335. 84} FR at 51343 ("[I]n a statute designed to address public health and welfare, it certainly cannot mean standards that allow a state to be "a laboratory for innovation" in the abstract, without any connection to a need to address pollution problems.").

reduce criteria pollution, 336 analyses in this section again recognize the way the different requirements in the ACC program work together to reduce criteria and GHG pollution and spur technological innovation. These analyses conclude that GHG pollution exacerbates tropospheric ozone pollution, worsening California's air quality problems, and the manner in which GHG and criteria pollutant standards work together to reduce both forms of pollution. Ample record support exists on California's need for both GHG standards and ZEV sales mandate at issue to address compelling and extraordinary conditions in California. As noted above, in SAFE 1 EPA, however, relied on an excerpt of the ACC program waiver record to determine the lack of criteria emission benefits of GHG emission standards and ZEV sales mandate at issue. In doing so, EPA did not evaluate the complete record from the ACC waiver proceeding and the nature of California's air quality problem, including the relationship of climate change to California's ability to achieve the ozone NAAQS in the assessment of California's need for these requirements.³³⁷

^{336.} The Agency again notes that, unlike provisions of the CAA such as section 211(c)(4)(C) which allows EPA to waive preemption of a state fuel program respecting a fuel characteristic or component that EPA regulates through a demonstration that the state fuel program is necessary to achieve a NAAQS, section 209(b) makes no mention of NAAQS pollutants or otherwise indicates that air pollutants should be treated differently.

^{337.} For example, CARB's ISOR for its ZEV standards identifies at Table 6.2 the well to wheel emission benefits of the ZEV program compared to the LEV III program. ZEV ISOR, EPA-HQ-OAR-2012-0562-0008 at 78. See also 2012 Waiver Request

As noted above, in SAFE 1, EPA established a new test under section 209, requiring a particularized, local nexus between (1) pollutant emissions from sources, (2) air pollution, and (3) resulting impact on health and welfare, a test that would exclude GHG pollution from the scope of the waiver.³³⁸ But this test is found nowhere in the text of section 209— the statute does not contain this requirement, or even use these terms.

EPA's review of the complete record confirms the Agency's conclusions in the ACC program waiver that California needs the GHG standards at issue to meet a compelling and extraordinary conditions regardless of whether the Agency focuses on criteria or greenhouse gas pollution reduction. This review also indicates that opponents of the waiver (including EPA in SAFE 1) did not meet the burden of proof necessary to demonstrate

at 16. CARB noted in its comments on the SAFE proposal that "Rising temperatures exacerbate California's ozone problem by increasing ground-level ozone concentrations." CARB, EPA-HQ-OAR-2018-0283-5054 at 371-72 (citing the 2012 Waiver Request). In addition, "Several studies indicate that a warming climate is expected to exacerbate surface ozone in California's two major air basins: South Coast Air Basin and San Joaquin Valley. *Id.* at 372 (citing Jacob & Winner. *Effect of Climate Change on Air Quality*, 43:1 ATMOS. ENVIRON. 51 (Jan. 2009); Wu, et al., *Effects of 2000-2050 Global Change on Ozone Air Quality in the United States*, 113, D06302, J. GEOPHYS. RES.-ATMOS. (Mar. 19, 2008), available at https://doi.org/10.1029/2007JD008917; Rasmussen, et al., The Ozone-climate Penalty: Past, Present, and Future, 47:24 ENVTL. SCI. & TECH. 14258 (Dec. 17, 2013), available at https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3990462/).

^{338. 84} FR at 51339-40.

that California did not have a need for the GHG standards, including under the nexus test applied in SAFE 1. It also bears note that EPA's longstanding practice, based on the statutory text, legislative history, and precedent calls for deference to California in its approach to addressing the interconnected nature of air pollution within the state and is not limited to criteria pollutant problems. Critically, EPA is not to engage in "probing substantive review" of waiver requests, 339 but rather "afford California the broadest possible discretion in selecting the best means to protect the health of its citizens and the public welfare." 340

E. Conclusion

Considering the text, legislative history, and precedent that support the Agency's historical practice of interpreting section 209(b)(1)(B) as calling for a program-level evaluation of waiver requests, as well as the uncertainty in settled expectations created by the SAFE 1 interpretation, EPA rescinds its actions in SAFE 1 regarding both the interpretation of section 209(b)(1)(B) and the findings regarding California's need for the GHG standards and ZEV sales mandate. EPA believes that the burden of proof had not been met in SAFE 1, based on the complete factual record, to demonstrate that California did not have a need for the GHG standards and ZEV sales mandate under the SAFE 1 interpretation of the second waiver prong nor had the burden been met to support a

^{339.} Ford Motor Co. v. EPA, 606 F.2d 1293, 1300 (D.C. Cir. 1979).

^{340.} *MEMA II*, 142 F.3d 449, 453 (D.C. Cir. 1998).

finding that the ample evidence in the record at the time of the ACC waiver decision did not demonstrate that California had a need for its standards to meet compelling and extraordinary conditions. As noted above, the result of the recission of the SAFE 1 action is the reinstatement of the ACC program waiver. EPA confirms the traditional interpretation of section 209(b)(1)(B) was appropriate and continues to be, at least, a better interpretation regardless of the recission of the SAFE 1 interpretation of this criterion.³⁴¹

VI. EPA Inappropriately Considered Preemption Under the Energy Policy and Conservation Act (EPCA) in Its Waiver Decision

SAFE 1's other justification for withdrawing the ACC program waiver was that California's GHG standards and ZEV sales mandate were preempted under EPCA. As explained in detail in Section IV, EPA believes this basis for reconsideration was outside the appropriate bounds of EPA's authority to reconsider previously granted waivers. In particular, if EPA could reconsider and withdraw a waiver based on a factor not contained in the specified criteria for denial in section 209(b)(1), EPA could circumvent the specified criteria for denial via reconsideration of previously granted waiver.

Even if it were appropriate for EPA to reconsider a previously granted waiver based on non-statutory factors, in this action, EPA concludes that it was inappropriate to

^{341.} See 84 FR at 51344 n.269.

rely on preemption under EPCA as a basis for withdrawing certain aspects of the ACC program waiver. In SAFE 1, a joint action between NHTSA and EPA, NHTSA concluded that state or local regulations of tailpipe carbon dioxide emissions are "related to fuel economy standards" and are therefore preempted under EPCA.³⁴² As a direct result of NHTSA's codified text and pronouncements on preemption set forth in SAFE 1, EPA withdrew the ACC program waiver for California's GHG standards and ZEV sales mandate on grounds that they were preempted under EPCA. In SAFE 1, EPA believed it was appropriate to consider the effect of NHTSA's actions, including the view that California cannot enforce standards that are void ab initio, and thus EPA stated that "to the extent that administrative action is necessary on EPA's part to reflect that state of affairs, EPA hereby withdraws that prior grant of a waiver on this basis."343 NHTSA has since issued a new final rule that formally repeals the

^{342. 49} U.S.C. 32919(a) ("When an average fuel economy standard prescribed under this chapter is in effect, a State or a political subdivision of a State may not adopt or enforce a law or regulation related to fuel economy standards or average fuel economy standards for automobiles covered by an average fuel economy standard under this chapter."). NHTSA noted that a law or regulation having the direct or substantial effect of regulating or prohibiting tailpipe carbon dioxide emissions from automobiles or automobile fuel economy is a law or regulation related to fuel economy standards and expressly preempted under 49 U.S.C. 32919(a). 84 FR at 51317-18. NHTSA's rule was codified at 49 CFR 531.7 ("Preemption") and 533.7 ("Preemption"), as well as each Appendix B in 49 CFR part 531 ("APPENDIX B TO PART 531—PREEMPTION") and Part 533 ("APPENDIX B TO PART 533—PREEMPTION").

^{343. 84} FR at 51338.

codified text and pronouncements regarding preemption under EPCA found in SAFE 1. Upon reconsideration, EPA now believes that, given NHTSA's repeal of its regulation and pronouncements in SAFE 1, preemption under EPCA cannot serve as a basis for the withdrawal of the ACC program waiver as it did in SAFE 1—if it could ever legitimately serve as such basis. EPA thus believes it is appropriate to rescind the portion of the waiver withdrawal that was based on preemption under EPCA.

In addition, given the unique consideration of preemption under EPCA in SAFE 1 and its effect on an otherwise validly issued waiver under the CAA, EPA believes it is helpful to provide additional information regarding the Agency's historical practice and views to demonstrate why consideration of preemption under EPCA was inappropriate. Consideration of preemption under EPCA is beyond the statutorily prescribed criteria for EPA in section 209(b)(1). Preemption under EPCA was not a factor that California addressed under the applicable waiver criteria in its initial request nor was it a factor that EPA considered in granting the ACC program waiver. Until SAFE 1, the Agency consistently refrained from reviewing waiver requests against factors beyond the statutorily listed criteria under section 209(b)(1). Thus, EPA also believes that in the reconsideration of a waiver where EPA had previously declined to consider preemption under EPCA, SAFE 1 was contrary to congressional intent and the Agency's historic practice of hewing to section 209(b)(1) statutory criteria in reviewing waiver requests. Given this backdrop, EPA believes that the joint rulemaking context of SAFE 1 was an improper

basis to deviate from EPA's long held belief to not consider factors outside the scope of section 209(b)(1), especially given that the Agency indicated it would only be a singular occurrence. EPA continues to view the text and congressional intent of the statute, as well as subsequent case law, as best supporting a limited scope of review for waiver requests under section 209(b)(1)—irrespective of whether a waiver proceeding is undertaken either solely by EPA or in unison with another agency. Therefore, based on EPA's historical practice of not considering factors outside of the section 209(b)(1) criteria and because EPA believes the "joint-action" premise was improper, the Agency is rescinding its withdrawal of the ACC program waiver based on preemption under EPCA.

A. Historical Practice and Legislative History

Historically, in reviewing California's waiver requests, EPA has refrained from the consideration of factors beyond those criteria set out in section 209(b)(1).³⁴⁴

^{344.} See, e.g., 43 FR at 32184 (rejecting objections to the procedures at state level, objections that section 207(c)(3)(A) establishes field protection, and constitutional objections all as beyond the "narrow" scope of the Administrator's review); 74 FR at 32783 (rejecting comments asking for the consideration of EPCA because it is not one of the three statutorily prescribed criteria); 78 FR at 2145 (again rejecting comments asking for the consideration of EPCA because it is outside the statutory criteria); 79 FR at 46265 (rejecting the argument that the HD GHG Regulations "impermissibly regulate fuel economy" because, like the commerce clause and Federal Aviation Administration Authorization Act of 1994 (FAAAA) issues, this issue is "outside the proper scope of review since it is not among the criteria listed under section 209(b).").

EPA has generally explained that the text, structure, and purpose of the California waiver provision indicate congressional intent for EPA to provide significant deference to California's judgment, especially on "ambiguous and controversial matters of public policy." 345 In section 209(a), Congress generally preempted state standards relating to the control of emissions from new motor vehicles and engines, but, in section 209(b), Congress carved out an exception for California, directing EPA to grant California a waiver of section 209(a) unless the Agency can make a finding under section 209(b). Congress recognized that California's "compelling and extraordinary circumstances," and its historical practice of regulating in the area, were sufficient "to justify standards on automobile emissions which may, from time to time, need be more stringent than national standards."³⁴⁶ In creating the waiver program, Congress intended not only for California to be able to meet its own emission reduction needs, but also for California to act as "a kind of laboratory for innovation" for motor vehicle standards and control technology."347 Thus "Congress

^{345. 78} FR at 2112, 2115; 40 FR at 23103-04; 58 FR 4166.

^{346.} H.R. Rep. No. 90-728, 90th Cong., 1st Sess. 21 (1967); S. Rep. No. 403, 90th Cong., 1st Sess. 33 (1967) ("The waiver of preemption is for California's 'unique problems and pioneering efforts.").

^{347.} *MEMA I*, 627 F.2d 1095, 1111 (D.C. Cir. 1979); 113 Cong. Rec. 30950, 32478 (Statement of Sen. Murphy) ("The United States as a whole will benefit by allowing California to continue setting its own more advanced standards for control of motor vehicle emissions. . . [The] State will act as a testing agent for various

consciously chose to permit California to blaze its own trail with a minimum of federal oversight."³⁴⁸

Legislative history makes clear that the Administrator must "presume" that the California standards "satisfy the waiver requirements" and that the burden of proving otherwise rests on the Administrator or other parties favoring denial of the waiver. 349 Further, according to the House Committee Report for the 1977 amendments that strengthened California's waiver provisions, EPA is "to afford California the broadest possible discretion in selecting the best means to protect the health of its citizens and the public welfare."350 According to the House Report, "The Administrator, thus, is not to overturn California's judgment lightly. Nor is he to substitute his judgment for that of the State. There must be "clear and compelling evidence that the State acted unreasonably in evaluating the relative risks of various pollutants in light of the air quality, topography, photochemistry, and climate in that State, before EPA may deny a waiver."351

types of controls and the country as a whole will be the beneficiary of this research.").

^{348.} Ford Motor Co. v. EPA, 606 F.2d 1293, 1297 (D.C. Cir. 1979).

^{349.} *MEMA I*, 627 F.2d at 1121-22 (citing, for example, S. Rep. No. 403, 90th Cong., 1st Sess. 33 (1967)).

^{350.} MEMA~II, 142 F.3d 449, 453 (D.C. Cir. 1998) (quoting H.R. Rep. No. 95-294, at 301-02 (1977)).

^{351.} H.R. Rep. No. 95-294, at 302 (1977), reprinted in 1977 U.S.C.C.A.N. at 1381.

EPA's historic practice of considering only listed criteria is thus in keeping with the highly deferential review of waiver requests that Congress intended in carving out the exception from preemption of new motor vehicle and engine standards in section 209(a).³⁵²

Courts have generally agreed with the Agency's consideration of only listed CAA criteria in reviewing waiver requests, also pointing to the statute's lack of any indication of the ability to consider non-statutory criteria as well as the waiver program's significant deference to California. The D.C. Circuit has stated that, under the text of the statute, the section 209(b) criteria are "the only waiver standards with which California must comply" and that, therefore, "[i]f EPA concludes that California's standards [meet section 209(b)], it is obligated to approve California's waiver application."353 The D.C. Circuit has repeatedly described EPA's waiver approval role as "limited" and "narrow." In MEMA I, for example, the court explained that "the Administrator has consistently held since first vested with the waiver authority, [that] his inquiry under section 209 is modest in scope. He has no 'broad and impressive' authority to modify California regulations."354 The court further noted that "there is no such thing as a 'general duty' on an administrative agency to make decisions based on factors other than those Congress expressly or impliedly

^{352.} See, e.g., 74 FR at 32783; 78 FR at 2145.

^{353.} MEMA II, 142 F.3d at 463.

^{354.} MEMA 1, 627 F.2d at 1119 (internal citations omitted).

intended the agency to consider."355 Similarly, the court has stated that "[t]he statute does not provide for any probing substantive review of the California standards by federal officials" and that "EPA's only role is to review California's proposed rules under a narrowly defined set of statutory criteria."356 Thus, the court has consistently rejected arguments requiring EPA to consider factors outside of the statutory criteria. In MEMA I, the court rejected a constitutional objection to a waiver, explaining that, because "the Administrator operates in a narrowly circumscribed proceeding requiring no broad policy judgments on constitutionally sensitive matters," "[n]othing in section 209 requires him to consider the constitutional ramifications of the regulations for which California requests a waiver . . . although nothing in section 209 categorically forbids" it. ³⁵⁷ In the same case, the court also rejected an antitrust objection as outside the scope of the Administrator's review.³⁵⁸ The court again upheld EPA's decision to not consider constitutional objections in American Trucking Association (ATA) v. EPA, stating, "We agree with EPA that ATA is seeking improperly to engraft a type of constitutional Commerce Clause analysis

^{355.} *Id.* at 1116-17.

^{356.} Ford Motor Co. v. EPA, 606 F.2d 1293, 1300 (D.C. Cir. 1979), and ATA v. EPA, 600 F.3d 624, 628 (2010), respectively.

^{357.} *MEMA I*, 627 F.2d at 1115 (declining to consider whether California standards are constitutional).

^{358.} *Id.* at 1117 ("[N]othing in section 209 or elsewhere in the Clean Air Act can fairly be read to imply a duty on the Administrator to deny a waiver on the basis of the antitrust implications of California regulations.").

onto EPA's [s]ection 7543(e) waiver decisions that is neither present in nor authorized by the statute."³⁵⁹

It is against this backdrop that EPA has reviewed waiver requests by evaluating them solely under the criteria of section 209(b). For instance, prior to SAFE 1, EPA had solicited comment, in the context of the 2008 and 2009 GHG notices for comment on CARB's first waiver request for GHG emission standards, as to whether the EPCA fuel economy preemption provisions were relevant to EPA's consideration of CARB's authority to implement its motor vehicle GHG regulations.³⁶⁰ In both instances, EPA declined to consider preemption under EPCA.³⁶¹ In the 2009 waiver, EPA explained that "section 209(b) of the Clean Air Act limits our authority to deny California's requests for waivers to the three criteria therein."362 EPA further pointed to its historic practice of "refrain[ing] from denying California's requests for waivers based on any other criteria," which had been reviewed and upheld by the Court of Appeals for the District of Columbia Circuit. 363 In the 2013 review of the ACC program waiver request, the Agency again declined to consider factors outside the statutory criteria, explaining that "EPA may only deny waiver requests based on the criteria in section 209(b), and inconsistency with EPCA is not one

^{359.} ATA v. EPA, 600 F.3d at 628.

^{360. 73} FR at 12159.

^{361.} Id.; 74 FR at 32783.

^{362. 74} FR at 32783.

^{363.} *Id.* (citing *MEMA I*, 627 F.2d at 1111, 1114-20, and *MEMA II*, 142 F.3d 449, 466-67 (D.C. Cir. 1998)).

of those criteria."³⁶⁴ A year later, EPA yet again declined to consider constitutionality claims, preemption under EPCA, and the implications of the Federal Aviation Administration Authorization Act of 1994 (FAAAA).³⁶⁵ EPA explained that section 209(b) limits the Agency's authority to deny California's requests for waivers to the three criteria therein and that the Agency has consistently refrained from denying California's requests for waivers based on any other criteria.³⁶⁶

In SAFE 1, EPA changed course, reasoning instead that the Agency pronouncement in the ACC program waiver decision on factors EPA could consider in denying a waiver request "was inappropriately broad, to the extent it suggested that EPA is categorically forbidden from ever determining that a waiver is inappropriate due to consideration of anything other than the 'criteria' or 'prongs' at section 209(b)(1)(B)(A)- (C)."³⁶⁷ EPA explained that this statement and EPA's historical practice of not

^{364. 78} FR at 2145.

^{365.} HD GHG Regulations for certain model year sleeper-cab tractors and dry-van and refrigerated-van trailers. 79 FR at 46256, 46264.

³⁶⁶. Id. In rejecting the commerce clause objection, the decision cited $MEMA\ I'$ s statement that "[t]he waiver proceeding produces a forum ill-suited to the resolution of constitutional claims." Id. (citing $MEMA\ I$, 627 F.2d at 1114-20). Thus, the decision concluded, "Constitutional challenges to the HD GHG Regulations [were] more appropriately addressed by a legal challenge directly against the state." Id.

^{367.} A complete discussion of preemption under EPCA in SAFE 1 can be found at 84 FR at 51337-38.

considering preemption under EPCA "were made in the context of EPA acting on its own to administer section 209(b) in considering such applications."368 Further, EPA distinguished these previous single-agency actions from its SAFE 1 joint action context by explaining that ignoring NHTSA's determination of preemption in the same action, "would place the United States Government in the untenable position of arguing that one federal agency can resurrect a State provision that, as another federal agency has concluded and codified, Congress has expressly preempted and therefore rendered void abinitio."369 At the same time, EPA expressed intentions not to consider factors outside the statutory criteria in future waiver proceedings. ³⁷⁰ EPA then concluded that NHTSA's determination of preemption in the same action "renders EPA's prior grant of a waiver for those aspects of California's regulations that EPCA preempts invalid, null, and void" because "California cannot enforce standards that are void ab initio."371

^{368.} Id.

^{369.} *Id.* Citing *Massachusetts* v. *EPA*, the Agency also asserted that the consideration of EPCA was supported by the Supreme Court's holding because it ensured consistency between NHTSA and EPA's programs. *Id.*

^{370. 84} FR at 51338.

^{371.} Id.

B. Notice of Reconsideration of SAFE 1 and Request for Comment

In its April 28, 2021, Notice of Reconsideration, EPA acknowledged that SAFE 1's consideration of NHTSA's finding of preemption under EPCA deviated from its historic practice of "declin[ing] to look beyond the waiver criteria in section 209(b) when deciding the merits of a waiver request from CARB."372 EPA sought comment on whether "EPA properly considered and withdrew portions of the ACC program waiver pertaining to GHG standards and the ZEV sales mandate based on NHTSA's EPCA preemption action, including whether EPA had the authority to withdraw an existing waiver based on a new action beyond the scope of section 209."373 Given EPA's reliance on NHTSA's preemption findings as a basis of waiver withdrawal in SAFE 1, EPA also sought comment on how the repeal of SAFE 1, should NHTSA take final action to do so, would affect its own reconsideration of SAFE 1.

C. Comments Received

EPA received comments in support of and against the consideration of preemption under EPCA in reviewing requests for waivers by California. Multiple comments related to the Agency's use of the joint action with NHTSA as a justification for deviating from the Agency's practice of reviewing waiver requests under the specific statutory

^{372. 86} FR at 22429.

^{373.} Id.

criteria. Some commenters agreed that the context of a joint action necessitated consideration of preemption under EPCA because NHTSA was the agency charged with interpreting and implementing EPCA and so EPA must consider its findings in the same action. The commenter also argued that the joint rulemaking of SAFE 1 would be consistent with pronouncements in *Massachusetts* v. *EPA* (2007) on the agencies' respective statutory obligations and the need to avoid inconsistency and so, "[o]nce NHTSA proposed to finalize a determination that EPCA preempts California's GHG motor vehicle standards, it would be unreasonable for the EPA to refuse to take NHTSA's action into account." The same action of preempts of the termination of th

Other commenters argued that the context of the rulemaking, whether joint or not, was irrelevant. One commenter stated emphatically that "what Congress directed EPA to consider when it wrote Section 209(b)(1) does not change depending on whether EPA acts alone or with another agency." Some commenters also argued that the context of the rulemaking was a particularly insufficient justification for revoking the waiver given language in SAFE 1 that allowed for inconsistent consideration of EPCA preemption. Several commenters noted that EPA constrained the future applicability of SAFE 1 by explaining that the Agency would not consider factors outside statutory criteria in future waiver reviews

^{374.} See, e.g., CEI at 11-12; AFPM at 2, 6.

^{375.} CEI at 11.

^{376.} States and Cities at 20. See also Twelve Public Interest Organizations app. 1 64-65.

in other subject areas.³⁷⁷ Another commenter also noted that "the action purported to be 'joint,' and yet as now acknowledged, SAFE Part 1 'is properly considered as two severable actions, a rulemaking by NHTSA and a final informal adjudication by EPA."³⁷⁸ These inconsistencies, they argued, made SAFE 1's distinction between single-agency and joint actions arbitrary and capricious.

Commenters also argued for and against consideration of factors outside the statutory criteria—including, but not limited to, preemption under EPCA—regardless of the kind of agency action, although EPA did not make this argument in SAFE 1. Commenters argued that EPA's authority to look outside the statutory criteria at EPCA was at least permissive, if not mandatory. According to one commenter, "EPA exaggerates the Court's position" in MEMA I in its Reconsideration notice: "[T]he court did not say that the EPA is *forbidden* to take constitutional ramifications into consideration, only that it is not required to do so."379 Another commenter agreed that MEMA I and MEMA II "do not preclude EPA from considering" preemption under EPCA but then went further, saying that "EPA is required to consider EPCA preemption." 380 The commenter argued that MEMA I rejected petitioners' constitutional objections to a waiver under an institutional

^{377.} NESCAUM at 3; Twelve Public Interest Organizations at app. 1 64-65; States and Cities at 20.

^{378.} SCAQMD at 7 (quoting 86 FR at 22439, n.40).

^{379.} CEI at 10 (original emphasis).

^{380.} AFPM at 5-6.

competence line of reasoning, concluding that "[t]he waiver proceeding produces a forum ill-suited to the resolution of constitutional claims."³⁸¹ In contrast, they continued, the waiver proceeding is an appropriate forum for determining whether emission standards "relate to" fuel economy because this issue is "within the agency's competence, as this relationship is mathematical and based in science rather than understandings of Constitutional law and precedent."³⁸² However, the other commenter, who agreed that EPA is not "forbidden" from considering preemption under EPCA, also noted that EPA "has no special competence to interpret EPCA."³⁸³

Several commenters also argued that EPA could not reinstate the waiver because NHTSA concluded that EPCA preempts the standards, such standards were void *ab initio*, and therefore "the state mandates referenced in CA's petition for reconsideration are not even eligible to be considered for a CAA waiver of preemption." ³⁸⁴ To

^{381.} Id. at 6 (quoting $MEMA\ I$, 627 F.2d 1095, 1114-15 (DC Cir. 1979)).

^{382.} Id.

^{383.} CEI at 11.

^{384.} NADA at 3-4; *See also* AFPM at 3 ("Since California's GHG tailpipe standards and ZEV mandate are related to fuel economy, they are not lawfully adopted and void *ab initio*—and there is nothing for EPA to reinstate."); Urban Air at 47-48; CEI at 2 ("But EPCA preemption is the proverbial elephant in the room. If SAFE 1's EPCA preemption argument is correct, the EPA could not grant a valid CAA preemption waiver for California's tailpipe CO 2 standards and ZEV mandates, because EPCA had already

ignore this, they claimed, would violate the Supremacy Clause of the Constitution. EPA, therefore, must look outside the statutory criteria to consider preemption under EPCA because it cannot "reasonably claim that the lawfulness and constitutionality of state actions over which it has supervision are issues outside the scope of its responsibility[.]"³⁸⁵

In contrast, other commenters pointed to EPA's historical practice of evaluating waiver requests under the section 209 statutory criteria, the text of the statute, and the policy implications of looking outside the statutory criteria, to support a return to EPA's traditional narrow approach. Most commenters argued that EPA's traditional interpretation was consistent with the text of section 209(b), which has no reference to preemption under EPCA or any other factors outside the three statutory criteria. Not only does EPA have "no grounds to read EPCA preemption considerations into

turned those policies into legal phantoms—mere proposals without legal force or effect.").

^{385.} CEI at 11.

^{386.} See, e.g., States and Cities at 20 ("EPA's traditional understanding of its limited role is entirely consistent with the text of Section 209(b)(1) and precedent interpreting it."); NCAT at 12 ("As EPA has stated in several prior waiver decisions, there is no reference in Section 209(b) to EPCA preemption nor anything that could be construed to address this issue. Section 209(b) is unambiguous in this regard, and EPA has no grounds to read EPCA preemption considerations into the statute.").

the statute,"³⁸⁷ these commenters argued, but to consider non-statutory criteria would actually be "arbitrary and capricious"³⁸⁸ and contrary to "precedent respecting separation of powers and federalism principles."³⁸⁹ Yet another commenter stated that the narrow interpretation "provides a safeguard from the capricious injection of outside-the-scope argumentation" because "[w]hen the adjudication is permitted to stray from the statutory criteria, prospects for a fair hearing can be derailed, and the EPA Administrator may be more prone to overstep and exert policy preferences that are impermissible."³⁹⁰

Additionally, in their petitions for reconsideration of SAFE 1, several states and cities asserted that EPA unlawfully changed course in SAFE 1 by considering (and relying on) the purported preemptive effect of

^{387.} NCAT at 12.

^{388.} NESCAUM at 7 ("As the D.C. Circuit has explained in the context of Section 209(b), 'there is no such thing as a general duty' on an administrative agency to make decisions based on factors other than those Congress expressly or impliedly intended the agency to consider.' It is a basic principle of administrative law that an agency action is 'arbitrary and capricious if the agency has relied on factors which Congress has not intended it to consider.").

^{389.} States and Cities at 20 ("It is likewise entirely consistent with precedent respecting separation of powers and federalism principles and holding that 'a federal agency may pre-empt state law only when and if it is acting within the scope of its congressionally delegated authority.' *Louisiana Pub. Serv. Comm'n* v. *FCC*, 476 U.S. 355, 374 (1986).").

^{390.} SCAQMD at 7.

EPCA, which is outside the confines of section 209(b) and argued that this rationale for withdrawing the waiver was flawed.³⁹¹

D. Analysis: EPA Is Rescinding Its SAFE 1 Actions Related to Preemption Under EPCA

Since SAFE 1, NHTSA has formally withdrawn its conclusions (and associated regulatory text) that state or local regulations of tailpipe carbon dioxide emissions are related to fuel economy standards and therefore preempted under EPCA. Thus the predicate for EPA's decision to withdraw the ACC waiver on that basis no longer exists. Furthermore, given the context of EPA's reconsideration of the ACC program waiver at the time of SAFE 1, the Agency believes it was inappropriate to reconsider the validity of the waiver against criteria such as preemption under EPCA. In this action, based on the two independent grounds noted above, the Agency is rescinding the portion of SAFE 1 that withdrew the ACC program waiver based on preemption under EPCA.

1. NHTSA Has Since Repealed Its Findings of Preemption Made in SAFE 1

In the Notice of Reconsideration, EPA sought comment on the Agency's reliance on NHTSA's preemption findings as a basis for its withdrawal of the ACC program waiver in SAFE 1. EPA also sought comment on how the repeal

^{391. 86} FR at 22428.

^{392. 86} FR 74236.

of SAFE 1, should NHTSA take final action to do so, would affect its own reconsideration of SAFE 1.393 NHTSA has since withdrawn its findings of preemption and the preemption basis of withdrawal is no longer applicable. Specifically, NHTSA has issued a new final rule that formally repeals the codified text and additional pronouncements regarding preemption under EPCA found in SAFE 1.394 In SAFE 1, EPA stated that it was appropriate to consider the effect of NHTSA's actions, including the view that California cannot enforce standards that are void ab initio and thus EPA stated that "to the extent that administrative action is necessary on EPA's part to reflect that state of affairs, EPA hereby withdraws that prior grant of a waiver on this basis." 395 Since this condition no longer exists, EPA believes it is appropriate to rescind the waiver withdrawal that was

^{393. 86} FR at 22429.

^{394. 86} FR 74236. NHTSA notes in this rulemaking that "the Agency is repealing all regulatory text and appendices promulgated in the SAFE I Rule. In doing so, the Agency underscores that any positions announced in preambulatory statements of prior NHTSA rulemakings, including in the SAFE I Rule, which purported to define the scope of preemption under the Energy Policy and Conservation Act (EPCA), do not reflect the Agency's reconsidered understanding of its proper role in matters of EPCA preemption."

^{395.} EPA distinguished these previous single-agency actions from its joint action context by explaining that ignoring NHTSA's determination of preemption in the same action, "would place the United States Government in the untenable position of arguing that one federal agency can resurrect a State provision that, as another federal agency has concluded and codified, Congress has expressly preempted and therefore rendered void *ab initio*." 84 FR at 51338.

based on preemption under EPCA. EPA believes that, to the extent it was ever appropriate for the Agency to base its action on NHTSA's finding of preemption under EPCA in SAFE 1, the repeal of the preemption rule makes it likewise appropriate to rescind the Agency's action in SAFE 1. This would also act to minimize regulatory uncertainty as to do otherwise would create further confusion that resulted from the joint action in SAFE 1 and would not appropriately reflect the current state of affairs under the circumstances of a unique federal regulation that had otherwise motivated EPA's actions in SAFE 1. NHTSA's recent action also supports EPA's belief that its practice of limiting its review of section 209(b) criteria, as explained below, remains appropriate in the context of preemption under EPCA.

2. EPA Improperly Deviated From Its Historical Practice of Limiting Its Review to Section 209(b) Criteria

Section 209(b)(1) of the Act limits the Agency's authority to deny California's requests for waivers to the three criteria contained therein and the Agency has consistently refrained from reviewing California's requests for waivers based on any other criteria. EPA acknowledges that California adopts its standards as a matter of law under its state police powers, that the Agency's task in reviewing waiver requests is limited to evaluating California's request according to the criteria in section 209(b), and that it is appropriate to defer to litigation brought by third parties in other courts, such as state or federal district court, for the resolution of any

constitutionality claims and assertions of inconsistency with other statutes. Considering the lack of statutory and precedential support as shown below, even if EPA were to have discretion to consider criteria outside section 209(b), EPA now views the joint-action context of SAFE 1 as an insufficient justification for deviating from its statutory authority and the Agency's historical practice and therefore the Agency rescinds its actions regarding preemption under EPCA in SAFE 1.

Withdrawal of the waiver was premised on NHTSA's preemption regulations in what EPA explained was a joint rulemaking action. But nothing in section 209(b) can be read as calling for consideration of preemption under EPCA in evaluating waiver requests regardless of whether EPA engaged in joint rulemaking with another agency or acted alone. Specifically, under section 209(b), EPA must grant California a waiver of the preemption contained in section 209(a) unless the Administrator makes a finding under any one of the listed criteria: "The Administrator shall . . . waive application of the preemption in section 209(a) if the Administrator finds any of the following: '(A) [California's] determination [that its standards in the aggregate will be at least as protective] is arbitrary and capricious, (B) [California] does not need such State standards to meet compelling and extraordinary conditions, or (C) such State standards and accompanying enforcement procedures are not consistent with section [202(a)].""396 Evaluation of preemption under EPCA is not a listed criterion.

^{396.} CAA section 209(b)(1)(A)-(C).

Nor did SAFE 1 premise preemption under EPCA on any of the three statutory criteria. In the ACC program waiver request, CARB made a protectiveness finding that, as a quantitative matter, its standards, in the aggregate, were as protective as the Federal standards and did not address preemption under EPCA.³⁹⁷ In fact, while California might opt to respond to comments on preemption under EPCA, California would not be expected to take it into account in any protectiveness finding made for a waiver request. It bears note that California's practice is not unusual because there are other factors and provisions of the CAA that California does not account for in making its protectiveness finding under section 209(b)(1).³⁹⁸ In granting the ACC program waiver request, EPA found that California's protectiveness finding was neither arbitrary nor capricious.³⁹⁹ EPA also responded to comments on the consideration of preemption under EPCA in granting the waiver but dismissed such objections as outside the scope of its review. 400 Historically, EPA draws a comparison between the numerical stringency of California and federal standards in making the requisite finding as to whether California's protectiveness determination is arbitrary and

^{397. 2012} Waiver Request at 15-17.

^{398.} For example, "California is not required to comply with section 207 to get a waiver." *MEMA II*, 142 F.3d 449, 467 (D.C. Cir. 1989).

^{399. 78} FR at 2125.

^{400.} Id. at 2145.

capricious.⁴⁰¹ Thus, neither California's initial request, nor EPA's waiver grant, considered preemption under EPCA and as previously explained in the ACC program waiver, EPA declined to consider preemption under EPCA viewing it as outside the scope of Agency review.

SAFE 1 made clear that consideration of and reliance on preemption under EPCA was the consequence of regulations promulgated by NHTSA. As SAFE 1 also acknowledged, however, EPA does not "administer" EPCA; that task falls to NHTSA. 402 Instead, "[i]f EPA concludes that California's standards [meet section 209(b)], it is obligated to approve California's waiver application." 403 EPA therefore disagrees with the comment that *Massachusetts* provides the Agency special duty to consider preemption under EPCA in a joint rulemaking action in reviewing waiver requests. In *Massachusetts*, the Supreme Court recognized the potential overlap

^{401.} Section 209(b)(2) provides that if each State [California] standard is at least as stringent as comparable applicable Federal standards then such standard shall be deemed to be as protective of public health and welfare as such federal standards for purposes of section 209(b)(1)(A). EPA acknowledges that in 1977 Congress amended the waiver provision to allow for California to address its unique combination of air quality problems and that California only be required to demonstrate stringency in the aggregate and that therefore some pollutant standards may not be as stringent.

^{402. 84} FR at 51338 ("EPA agrees with commenters that EPA is not the agency that Congress has tasked with administering and interpreting EPCA. This is especially so because '[t]he waiver proceeding produces a forum ill-suited to the resolution of constitutional claims.' *MEMA I*, 627 F.2d at 1115.").

^{403.} MEMA II, 142 F.3d at 463.

between NHTSA's and EPA's statutory obligations and concluded that "there is no reason to think the two agencies cannot both administer their obligations yet avoid inconsistency."404 As one commenter noted, EPA and NHTSA have previously engaged in joint actions that addressed fuel economy and GHG emissions. In those actions. NHTSA's role has been to set national fuel economy standards and EPA's role has been to set national GHG standards. 405 These roles are complementary, but distinct. The Court acknowledged the independence of these roles in Massachusetts: "EPA has been charged with protecting the public's 'health' and 'welfare,' 42 U.S.C. 7521(a)(1), a statutory obligation wholly independent of DOT's mandate to promote energy efficiency. See Energy Policy and Conservation Act, §2(5), 89 Stat. 874, 42 U.S.C. 6201(5)."406

Regarding the Agency's simultaneous pronouncement that reliance on preemption under EPCA would be a singular exercise that would not be repeated, statutory support or past precedent for this singular consideration was also lacking.⁴⁰⁷ In fact, this singular exercise would

^{404.} Massachusetts v. EPA, 549 U.S. 497, 532 (2007).

^{405.} In its most recent rulemaking addressing GHG emissions from light-duty vehicles, EPA extensively coordinated with NHTSA on details of the program but did not conduct it as a joint rulemaking. *See* 86 FR 74434, 74436 (December 30, 2021).

^{406.} Massachusetts, 549 U.S. at 497, 532.

^{407. &}quot;EPA does not intend in future waiver proceedings concerning submissions of California programs in other subject areas to consider factors outside the statutory criteria in section 209(b)(1) (A)-(C)." 84 FR at 51338.

allow for EPA to evaluate the same waiver request differently and depending on EPA's own choice—the choice to act with another agency or not—rather than on the merits of the waiver request itself within specified criteria in section 209(b). Again, the result of this unique application of EPA's authority is unsupported under section 209(b)(1).

As previously noted, EPCA is generally administered by NHTSA and consideration of preemption under EPCA in reviewing waiver requests would for instance call for EPA to resolve the much debated and differing views as to what is a "law or regulation related to fuel economy," as contemplated by 39 U.S.C. 32919(a). 408 Relevant judicial precedent would also appear to call into question whether California's GHG standards and ZEV sales mandates are indeed preempted under EPCA. 409 But as previously explained, EPA does not implement EPCA, and the Agency's review of waiver requests is highly deferential.

EPA also disagrees with comments that the Agency must generally consider factors outside the criteria listed in section 209(b), including preemption under EPCA, regardless of the joint- or single-agency nature of the action. EPA has never claimed that it has such broad authority to consider factors outside section 209(b) and the

^{408.} EPA takes no position on any role NHTSA might play under 42 U.S.C. 32919(a) and acknowledges that NHTSA discusses this in its recent final rulemaking. *See generally* 86 FR 74236.

^{409.} See, e.g., Cent. Valley Chrysler-Jeep, Inc. v. Goldstene, 529 F. Supp. 2d 1151, 1153-54 (E.D. Cal. 2007), as corrected Mar. 26, 2008; Green Mountain Chrysler Plymouth Dodge Jeep v. Crombie, 508 F. Supp. 2d 295, 300-01 (D. Vt. 2007).

decades of waiver practice, as well as judicial precedent, are indicative of the Agency's narrow scope of review for California waiver requests: "[T]he Administrator has consistently held since first vested with the waiver authority, [that] his inquiry under section 209 is modest in scope. He has no 'broad and impressive' authority to modify California regulations."410 Instead, EPA has consistently declined to consider factors outside the three statutory criteria listed in section 209(b).⁴¹¹ This limited scope of review has been repeatedly upheld by the courts. For example, in MEMA I, the D.C. Circuit stated that "there is no such thing as a "general duty" on an administrative agency to make decisions based on factors other than those Congress expressly or impliedly intended the agency to consider."412 In MEMA II, the D.C. Circuit again rejected consideration of a factor outside the 209(b) statutory criteria because doing so would restrict California's ability to "exercise broad discretion." 413

^{410.} MEMA I, 627 F.2d 1095, 1119 (D.C. Cir. 1979).

^{411.} See, e.g., 43 FR at 32184 (rejecting objections to the procedures at state level, objections that section 207(c)(3)(A) establishes field protection, and constitutional objections all as beyond the "narrow" scope of the Administrator's review); 74 FR at 32783 (declining to consider EPCA preemption, stating that "section 209(b) of the Clean Air Act limits our authority to deny California's requests for waivers to the three criteria therein."); 79 FR at 46264 (reiterating that EPA can only deny a waiver request based on the 209(b) statutory criteria, dismissing comments on preemption under EPCA, as well as the Constitution and the implications of the FAAAA).

^{412. 627} F.2d at 1116.

^{413. 142} F.3d at 464.

Commenters also claim that ignoring NHTSA's finding of preemption would violate the Supremacy Clause of the Constitution because the necessary consequence of NHTSA's conclusion in SAFE 1 is that certain standards were void *ab initio* as preempted under EPCA and as such that "the state mandates referenced in [California's] petition for reconsideration are not even eligible to be considered for a CAA waiver of preemption." EPA disagrees. As the D.C. Circuit has held, "[t]hat [the Administrator] like every other administrative officer owes allegiance to the Constitution does not mean that he is required to issue rulings of constitutional dimension." Thus, "[n]othing in section 209 requires [the Administrator] to consider the constitutional ramifications of the regulations for which California requests a waiver."

Moreover, consideration of factors beyond those set out in section 209(b)(1) would subject California and vehicle and engine manufacturers to changes in regulatory schemes by other federal agencies not acting under the authority of the CAA.⁴¹⁷ SAFE 1 and subsequent events

^{414.} NADA at 3.

^{415.} *MEMA I*, 627 F.2d at 1114-15.

^{416.} Id. at 1115.

^{417. &}quot;The manufacture of automobiles is a complex matter, requiring decisions to be made far in advance of their actual execution. The ability of those engaged in the manufacture of automobiles to obtain clear and consistent answers concerning emission controls and standards is of considerable importance so as to permit economies in production." S. Rep. No. 403, 90th Cong., at 730 1st Sess. (1967).

perfectly encapsulate this problem. For instance, NHTSA has since finalized the repeal of the regulatory provisions and pronouncements it made in SAFE 1 that were the underpinnings for EPA withdrawing certain aspects of the ACC program waiver and with that action the Agency's basis for revocation of the waiver under EPCA has now evanesced. Additionally, this is affirmation of EPA's long held view that waiver proceedings are not the appropriate venue for resolving these issues, and the joint-rulemaking context is not and should never have been justification for deviating from statutory authority and the Agency's historical practice.

It also bears note that consideration of factors beyond the criteria contained in section 209(b) would not be limited to preemption under EPCA. Commenters suggested, for instance, that EPA would not be able to "ignore the First Amendment," in the hypothetical situation where California impos[ed] standards on some manufacturers in retaliation for their voiced opposition to California's authority as well as criminality such as "bribery and extortion had been instrumental in assembling the legislative majorities." In short, under the commenter's view, factors for consideration in waiver proceedings would be innumerable. And yet these factors bear little or no relation to specific criteria in section 209(b) that would otherwise warrant the denial of a waiver request. The D.C. Circuit has already, several times, held that EPA is not required to consider factors outside of and unconnected

^{418.} See 86 FR 74236.

^{419.} CEI at 11.

to these statutory criteria, especially constitutional objections. In fact, regarding the commenter's example, the court has already specifically rejected consideration of the First Amendment in waiver evaluations. In MEMA I, the court considered and upheld EPA's decision declining to consider a First Amendment objection to a waiver as beyond the scope of agency review.⁴²⁰ Courts have also rejected objections based on the applicability of CAA section 207 to California waiver requests⁴²¹ and the Commerce Clause. 422 EPA is therefore not persuaded by these arguments. Additionally, courts have long held that administrative proceedings for California waiver requests are ill-suited for consideration of constitutional issues. Nothing precludes commenters from challenging California's standards themselves—whether under EPCA, another statute, or the Constitution—in other, bettersuited fora. According to the D.C. Circuit, for instance, [w]hile nothing in section 209 categorically forbids the Administrator from listening to constitutionality-based challenges, petitioners are assured through a petition of review . . . that their contentions will get a hearing."423 The D.C. Circuit has also repeatedly stated that challenges

^{420.} MEMA I, 627 F.2d 1095, 1115 (D.C. Cir. 1979).

^{421.} MEMA II, 142 F.3d 449, 467 (D.C. Cir. 1998).

^{422.} ATA v. EPA, 600 F.3d 624, 628 (D.C. Cir. 2010) ("EPA's only role is to review California's proposed rules under a narrowly defined set of statutory criteria."); OOIDA v. EPA, 622 Fed. Appx. 4, 5 (D.C. Cir. 2015) (rejecting a challenge for lack of jurisdiction because challengers objected to California's regulations themselves, not EPA's approval of them in a waiver under 209(b)).

^{423.} MEMA I, 627 F.2d at 1115.

which go to the legality of California's standards themselves, are better addressed directly by either courts or Congress.⁴²⁴ Challenges based on preemption under EPCA similarly go to the legality of California's standards themselves and are thus more appropriate in court or addressed to Congress.

E. Conclusion

Because the landscape of federal law has changed since SAFE 1 due to NHTSA's repeal of its regulatory text, appendix, and pronouncements regarding EPCA preemption in SAFE 1, EPA believes that it is appropriate to rescind its waiver withdrawal actions in SAFE 1 that were predicated on the federal law context created by NHTSA's SAFE 1 action. On separate grounds, EPA also believes that, based on the foregoing, EPA should not have deviated from its practice of limiting its waiver review to the criteria in section 209(b)(1). Thus, for the reasons stated above, EPA is rescinding those portions of SAFE 1 that withdrew the waiver of the ACC program on the basis of preemption under EPCA.

^{424.} *Id.* at 1105. In *ATA* v. *EPA*, the D.C. Circuit rejected a constitutional challenge to a California waiver, concluding that Congress made the decision to give California "the primary role in regulating certain mobile pollution sources" so the challenger's argument was best directed to Congress. 600 F.3d 624, 628 (D.C. Cir. 2010).

VII. EPA Inappropriately Set Forth an Interpretive View of Section 177 in SAFE 1

In SAFE 1, EPA provided an interpretive view of section 177 of the CAA, stating that states adopting California's new motor vehicle emission standards (section 177 states) could not adopt California's GHG standards. 425 In this action, EPA determines that it was both inappropriate and unnecessary within a waiver proceeding to provide an interpretive view of the authority of section 177 states to adopt California standards, as EPA plays no statutory approval role in connection with states' adoption of standards identical to those standards for which a waiver has been granted to California. 426 Rather, if a state chooses to submit such standards for inclusion in an SIP, EPA's role with regard to approval of these standards is to review them in the same way that EPA reviews all SIP revisions a state submits, via a notice and comment process, to ensure that the submission meets all statutory and regulatory requirements as part of the Agency's decision whether to approve or disapprove the submission. Therefore, the Agency is rescinding the interpretive views on section 177 set out in SAFE 1.

^{425. 84} FR at 51310, 51350.

^{426.} EPA is aware of instances of States adopting California new motor vehicle emission standards and not subsequently including such standards in their SIP. In these circumstances EPA has not played and would not play an approval role.

A. SAFE 1 Interpretation

In the SAFE proposal, EPA proposed to conclude that "States may not adopt California's GHG standards pursuant to section 177 because the text, context, and purpose of section 177 support the conclusion that this provision is limited to providing States the ability, under certain circumstances and with certain conditions, to adopt and enforce standards designed to control criteria pollutants to address NAAQS nonattainment."427 Additionally, the proposal noted the title of section 177 ("New motor vehicle emission standards in nonattainment areas") indicates a limited scope of application. 428 The proposal also suggested that, because "[a]reas are only designated nonattainment with respect to criteria pollutants," it would be "illogical" if states could use their 177 authority "to adopt California standards that addressed environmental problems other than nonattainment of criteria pollutant standards."429

In the SAFE 1 decision, EPA finalized its proposed interpretive view, reiterating that "the text (including both the title and main text), structural location, and purpose of the provision confirm that it does not apply to GHG standards." Because section 177's title references nonattainment areas, and because nonattainment designations only exist for criteria pollutants, EPA

^{427. 83} FR at 43240.

^{428.} Id.

^{429.} Id.

^{430. 84} FR at 51350.

claimed, states could not adopt standards for purposes of GHG control under section 177.431

As evidence for this interpretive view, EPA again pointed to the text and location of the section, which had been the basis for the Agency's interpretation in the SAFE proposal. EPA acknowledged commenters who argued that "CAA section 177 does not contain any text that could be read as limiting its applicability to certain pollutants only" and that EPA had "inappropriately relied on the heading for CAA section 177 to construe a statutory provision as well as arrogated authority to implement an otherwise self-implementing provision," but disagreed with these commenters. ⁴³² In addition to the evidence relied on in the proposal, EPA provided examples of legislative history from the 1977 amendments to support its interpretive view. ⁴³³

^{431.} Id.

^{432.} *Id*.

^{433.} In particular, EPA cited legislative history on section 172(b), which set forth the "requisite provisions" for state plans for nonattainment areas. *Id.* at 51350 n.286. According to the legislative history, one of the many factors that must be considered by a state plan is "actual emissions of such pollutant resulting from in-use motor vehicles." *Id.* (quoting H.R. Rep. No. 294, 95th Cong., 1st Sess. 212 (1977), 1977 U.S.C.C.A.N. 1077, 1291, 1997 WL 16034). Therefore, EPA claimed, this legislative history "identifies section 177 as a means of addressing the NAAQS attainment planning requirements of CAA section 172, including the specific SIP content and approvals criteria for EPA." *Id.* at 51351.

B. Notice of Reconsideration of SAFE 1 and Request for Comment

Acknowledging that "section 177 does not require States that adopt California emission standards to submit such regulations for EPA review" and that "California in previous waiver requests has addressed the benefits of GHG emissions reductions as it relates to ozone," EPA sought comment in the 2021 Notice of Reconsideration on whether EPA had the authority in the SAFE 1 context to interpret section 177 of the CAA and whether the interpretive view was appropriate. 434 Specifically, EPA sought comment on whether it was appropriate for EPA to provide an interpretive view of section 177 within the SAFE 1 proceeding.⁴³⁵ To the extent it was appropriate to provide an interpretation, EPA sought comment on whether section 177 was properly interpreted and whether California's motor vehicle emission standards adopted by states pursuant to section 177 may have both criteria emission and GHG emission benefits and purposes. 436

C. Comments Received

In response to SAFE 1, EPA received multiple petitions for reconsideration. One petition submitted by several states and cities asserted that, in adopting its interpretation of section 177, EPA "relie[d] on information

^{434. 86} FR at 22429.

^{435.} Id.

^{436.} Id.

and reasoning not presented in the SAFE Proposal," particularly the "superseded version of Section 172 . . . and legislative history for that outdated provision."⁴³⁷ The petition noted that the use of this information and reasoning was used in the SAFE 1 to conclude that "section 177 is in fact intended for NAAQS attainment planning and not to address global air pollution."⁴³⁸ Petitioners argued that because this information and reasoning was not presented in the proposal, "EPA should withdraw and reconsider its finalization of the Section 177 interpretation and allow for full and fair public comment before proceeding further."⁴³⁹

EPA also received many comments in response to the Notice of Reconsideration of SAFE 1, both supporting and opposing EPA's statements regarding section 177 in SAFE 1. Supporters of SAFE 1 reiterated the reasoning from the proposal and final action.⁴⁴⁰ For example, one commenter wrote, "In short, 'the text, context, and purpose of Section 177 suggest' that the provision is limited to motor vehicle standards 'designed to control criteria pollutants to address NAAQS nonattainment."⁴⁴¹ Like the SAFE proposal and final action, the commenter stated that in addition to the text and context of the

^{437.} See States and Cities' Petition at 27.

^{438.} Id. (quoting 84 FR at 51351).

^{439.} Id.

^{440.} CEI at 17-18; NADA at 6; AFPM at 12-13.

 $^{441.\;}$ CEI at 18 (quoting heavily from the SAFE proposal and SAFE final action).

section, there is "substantial legislative history showing that Congress's purpose in creating the Section 177 program was to address non-attainment with NAAQS for criteria pollutants, not to address any global atmospheric phenomenon."⁴⁴²

Opponents of SAFE 1 argued both that EPA had no authority to issue its 177 statement and that the merits of EPA's argument were wrong. On the issue of authority, opponents of SAFE 1 claimed that SAFE 1 failed to consider the reliance interests of the stakeholders, particularly section 177 states.⁴⁴³ SAFE 1, they argued,

442. Id.

443. States and Cities at 50-55; Institute for Policy Integrity Amicus Brief at 22-26 ("[T]he fact that California and many other states have detrimentally relied on this waiver to meet federal and state air-pollution mandates resolves any lingering doubt about the lawfulness of EPA's Action.... Revoking the preemption waiver... . jeopardizes the state's ability to meet federal standards for other harmful air pollutants, since the standards covered by the waiver would have reduced—directly and indirectly—nitrogen-oxide, ozone, and particulate-matter pollution. See 78 FR 2122, 2129, and 2134."); Tesla at 11-13; National Association of Clean Air Agencies (NACAA), Docket No. EPA-HQ-OAR-2021-0257-0096 at 3. Many of the 177 states had also provided comments, during the SAFE 1 comment period, explaining that they have adopted the ACC program standards to meet their public health goals. See, e.g., Maryland Department of the Environment, Docket No. EPA-HQ-OAR-2018-0283-5831 at 2-3; Delaware Department of Natural Resources and Environment Control, Docket No. EPA-HQ-OAR-2018-0283-5066 at 3-5; Massachusetts Department of Environmental Protection, Docket No. EPA-HQ-OAR-2018-0283-5476; State of California et al., Docket No. EPA-HQ-OAR-2018-0283-5481 at 130-31 (California was joined by the States of Connecticut, Delaware, Hawaii, Iowa, Illinois,

upset this reliance and created uncertainty.⁴⁴⁴ A substantial number of commentors also argued that EPA had no authority to make its statements on section 177 because "Congress gave EPA no role in implementing Section 177 and no authority to constrain States' decisions regarding adoption of California emissions standards."⁴⁴⁵

On the merits of EPA's SAFE 1 argument, opponents of the action commented that EPA misinterpreted section 177 and that, even if EPA's interpretive view were correct, EPA misapplied it. Multiple commenters wrote that the text of section 177 does not limit the

Maine, Maryland, Minnesota, New Jersey, New Mexico, New York, North Carolina, Oregon, Rhode Island, Vermont, Washington, the Commonwealths of Massachusetts, Pennsylvania, and Virginia, the District of Columbia, and the Cities of Los Angeles, New York, Oakland, San Francisco, and San Jose).

445. States and Cities at 51. See also Tesla at 11-13; Twelve Public Interest Organizations app. 1 at 2; NESCAUM at 8-9; Southern Environmental Law Center (SELC), Docket No. EPA-HQ-OAR-2021-0257-0125 at 2-3; NCAT at 12; Class of '85, Docket No. EPA-HQ-OAR-2021-0257-0454 (correction to an earlier comment by the same commenter, which can be found at Docket No. EPA-HQ-OAR-2021-0257-0388) at 5-6; Maine at 2; OTC at 2. Ironically, one supporter of SAFE 1, while arguing that EPA cannot consider GHG reductions from section 177 states in its second prong analysis, acknowledged EPA's lack of an oversight role under section 177: "EPA cannot consider GHG reductions, if any, attributable to 'opt-in' states under Section 177, as these are out of the scope of a waiver application. Indeed, EPA has no legal role in reviewing opt-in states, as the statute grants the agency no role in reviewing opt-in by other states." AFPM at 15.

^{444.} See, e.g., States and Cities at 50-55; Tesla at 11-13.

types of pollutants for which motor vehicle emission standards can be authorized. Commenters also noted that the title of section 177 refers to geographic *areas*, not pollutants, and argued that the restriction was therefore on which states could adopt California standards (states with plan provisions approved under Part D) not on the pollutants for which those states could adopt standards. A few commenters also argued that EPA's section 177 interpretive view would create a "third vehicle" scenario, in contradiction of section 177's identicality requirement. Even if EPA's interpretation were correct, opponents continued, California's standards have both criteria emission and GHG emission benefits and purposes. Commenters cited the factual record as well as EPA's own

^{446.} See, e.g., States and Cities at 53; NESCAUM at 9; NCAT at 12.

^{447.} See, e.g., States and Cities at 53 ("[T]he reference in the title to 'nonattainment areas' is not a limitation to 'nonattainment (i.e., criteria) pollutants' or standards that target them" but rather a limitation on the states that can adopt California's standards); NESCAUM at 9; SELC at 2; NCAT at 12.

^{448.} Commenters feared that EPA's interpretation, which "prevents Section 177 States from adopting California's GHG standards, but not any other California standards," could require states to "extract just the GHG portion of the Advanced Clean Cars rules from their programs, thus potentially creating type of "third vehicle" forbidden by Section 177 (*i.e.*, a vehicle subject to a hybrid combination of the other California standards and the (now weakened) federal GHG standards." States and Cities at 54. *See also* NESCAUM at 11-12; SELC at 5.

^{449.} States and Cities at 31-32, 50-55; NESCAUM at 12-13; SELC at 5; NCAT at 12; Class of '85 at 4-5.

past findings as evidence of the connection between GHG standards and NAAQS attainment.

D. Analysis: EPA Is Rescinding SAFE 1's Interpretive Views of Section 177

EPA is withdrawing its non-regulatory and non-binding interpretation of section 177 set forth in SAFE 1. EPA plays no statutory approval role in connection with states' adoption of standards identical to those standards for which the Agency has granted a waiver to California. At Epa review at the standards for inclusion in a SIP, EPA's role with regard to approval of these standards is to review them in the same way that EPA reviews all SIP revisions a state submits, via a notice and comment process, to ensure that the submission meets all statutory and regulatory requirements as part of the Agency's decision whether to approve or disapprove the submission.

In reconsidering SAFE 1, EPA now believes that it was inappropriate to offer an interpretive view of section

^{450.} EPA is aware of instances of States adopting California new motor vehicle emission standards and not subsequently including such standards in their SIP. In these circumstances EPA has not played and would not play an approval role.

^{451.} EPA notes that although section 177 states that "... any State which has plan provisions approved under this part may adopt and enforce for any model year standards relating to control of emissions from new motor vehicles ..." the language in section 177 does not require a state to submit its adopted motor vehicle emissions standards for SIP approval.

177 in the context of that action. EPA believes it acted inappropriately in providing an interpretive view in SAFE 1 and that such action was based on an inaccurate assessment of the factual record. EPA's interpretive view was not compelled by any petition, request, or legislative or judicial mandate and was otherwise not final agency action. EPA is therefore rescinding the interpretive views contained in SAFE 1.

As commenters have noted, section 177 does not describe a direct approval role for EPA. Section 177 says that "any State which has plan provisions approved under this part may adopt and enforce" identical California standards and delineates three specific criteria for adoption. Nothing in this language or in the text of the rest of the section requires or allows EPA to approve such adoption and enforcement or directs EPA to implement the section through regulation; EPA plays no statutory approval role in the adoption of California standards by other states other than action on a SIP revision, should those states include the standards in their plans. In fact, there are only three prerequisites to adoption and enforcement by a state: That the state has a federally approved SIP, that the standards are identical (thus the

^{452. 84} FR at 51338 n.256 ("EPA acknowledges that its actions in this document may have implications for certain prior and potential future EPA reviews of and actions on state SIPs.... EPA will consider whether and how to address those implications, to the that they exist, is separate actions."). EPA action on a state plan (including application of Section 177) is subject to judicial review. 42 U.S.C. 7607(b)(1).

^{453. 42} U.S.C. 7507.

state standards must not create or have the effect of creating a "third vehicle") to California standards for which California has received a waiver, and that California and the state adopt the standards with at least two years lead time. ⁴⁵⁴ This limited role has been acknowledged by courts and EPA alike. ⁴⁵⁵ Thus, it is well established that states have broad discretion to adopt California standards without being subject to EPA's approval. ⁴⁵⁶

States with approved SIPs that have adopted the waived California standard into state law may submit a

455. In 1979, for example, only two years after the adoption of section 177, the D.C. Circuit stated that the Act only requires the three listed prerequisites, "not... that the EPA administrator conduct a separate waiver proceeding for each state that chooses [to adopt California standards]." Ford Motor Co. v. EPA, 606 F.2d 1293, 1298 (D.C. Cir. 1979). Similarly, in 1994, while enacting rules implementing section 209(e)(2)(B), the parallel provision for the nonroad vehicle section of the California Waiver program, EPA noted that section 177 states had not "ask[ed] for EPA authorization before they adopted the California standards, nor did EPA or the automobile industry suggest that they needed such authorization." 56 FR 36969, 36983 (1994). See also 77 FR 62637 n.54 ("States are not required to seek EPA approval under the terms of section 177.").

456. EPA also notes that there are ample judicial avenues to directly challenge state adoption of California standards. For example, the First and Second Circuits have already addressed objections to the adoption of California standards under section 177. In both *Am. Auto. Mfrs. Ass'n v. Mass. DEP* and *Motor Vehicle Mfrs. Ass'n v. NYSDEC*, petitioners argued that the States' adoption of California's low emission vehicles standards without the associated clean fuels plan violated section 177. 31 F.3d 18 (1st Cir. 1994); 17 F.3d 521 (2d Cir. 1994).

^{454.} Id.

SIP revision that includes that adopted standard. In that proceeding, EPA could determine whether the statutory criteria for adoption are met for purposes of approving a SIP revision. Indeed, in the litigation following SAFE 1, EPA acknowledged that its interpretive view of section 177 would have no actual effect until applied in a future SIP context. ⁴⁵⁷ SIPs are a crucial planning tool in helping states reach attainment for NAAQS and California's standards are key components of many of these SIPs. ⁴⁵⁸ In a SIP proceeding, these states and other stakeholders are better able to provide specific and comprehensive

^{457.} Several commenters on the Notice of Reconsideration argued that SAFE 1 violated conformity rules by interfering with already approved SIPs. However, as EPA explained in the litigation over SAFE 1, the action had no actual effect on "either existing approvals of state plans or the plans themselves for criteria pollutants." Final Brief for Respondents at 106, *Union of Concerned Scientists* v. *NHTSA*, No. 19-1230 (D.C. Cir. Oct. 27, 2020). *See also* 84 FR 51338, n.256.

^{458.} Wisconsin at 1 ("These standards provide important and necessary reductions in both GHG and criteria pollutant emissions needed to meet state and local air quality goals and address federal CAA requirements."); Connecticut at 2 ("These programs enable long-term planning and yield critical emission reductions that are critical to meeting Connecticut's climate goals as well as our statutory obligations to reach attainment with the ozone NAAQS."); Delaware at 2 ("Delaware adopted the California LEV regulation and incorporated the LEV and GHG standards into the State Implementation Plan. . . . Delaware will not meet air quality goals without more protective vehicle emission standards. "); Maine at 1 ("[T]he LEV program was initially created to help attain and maintain the health-based National Ambient Air Quality Standards (NAAQS)... The California ZEV and GHG programs enable longterm planning for both the states and the regulated community and have been drivers of technological change across the industry.").

comments about the intent and effect of adopting California standards.⁴⁵⁹

For these reasons, EPA believes that it was inappropriate to provide an interpretive view of section 177 in SAFE 1.⁴⁶⁰ Therefore, EPA is withdrawing its SAFE 1 interpretive view of section 177.

E. Conclusion

EPA determines that it was both inappropriate and unnecessary, within the SAFE 1 waiver proceeding, to provide an interpretive view of the authority of section 177 states to adopt California standards. Therefore, EPA withdraws its interpretive views that had been set forth in SAFE 1.

VIII. Other Issues

A. Equal Sovereignty

As explained in Section VI, EPA must grant California's waiver request unless the Agency makes one of the specified findings in section 209(b)(1). In this

^{459.} The Agency has considered whether there may be any reliance interests on EPA's previous interpretive view of section 177 described in the SAFE 1 action. EPA is unaware of any such interests, and none were raised in comments.

^{460.} To the extent that EPA's reasoning in its SAFE 1 section 177 determination lacked fair notice, as the States and Cities' Petition claimed, such a contention is rendered moot by this action.

instance, Congress has made multiple determinations through its adoption of section 209 and subsequent amendments, dating from 1967 through the 1990 CAA Amendments, regarding California's role and its relation to federal standard setting for mobile sources. EPA's longstanding waiver practice, consistent with case law, has been to refrain from considering factors beyond section 209(b)(1) criteria as well as constitutional claims in the review of California waiver requests. EPA acknowledges that California adopts its standards as a matter of law under its police powers, that the Agency's task in reviewing waiver requests is properly limited to evaluating California's request according to

^{461.} EPA has declined to consider constitutional challenges to California Waivers since at least 1976. 41 FR 44212 (Oct. 7, 1976) ("An additional argument against granting the waiver was raised by the Motorcycle Industry Council and Yamaha, who contended that the CARB had violated due process when adopting their standards, by not allowing the manufacturers a fair and full opportunity to present their views at a State hearing. If this argument has any validity, the EPA waiver hearing is not the proper forum in which to raise it. Section 209(b) does not require that EPA insist on any particular procedures at the State level. Furthermore, a complete opportunity was provided at the EPA waiver hearing for the presentation of views."). See also, e.g., 43 FR at 32184 (July 25, 1978) (rejecting objections to the procedures at state level, objections that section 207(c)(3)(A) establishes field protection, and constitutional objections all as beyond the "narrow" scope of the Administrator's review).

^{462.} Central Valley Chrysler-Jeep, Inc. v. Goldstene, 529 F.Supp.2d 1151, 1174 ("The waiver provision of the Clean Air Act recognizes that California has exercised its police power to regulate pollution emissions from motor vehicles since before March 30, 1966; a date that predates both the Clean Air Act and EPCA.").

the criteria in section 209(b), and that it is appropriate to defer to litigation brought by third parties in other courts, such as state or federal court, for the resolution of constitutionality claims and inconsistency, if any, with other statutes. As further explained this practice flows from the statute and legislative history, which reflect a broad policy deference that is afforded to California to address its serious air quality problems (which are ongoing) as well as to drive emission control innovation. And so, EPA has historically declined to consider constitutional issues in evaluating and granting section 209 waivers. In MEMA I, the D.C. Circuit rejected a First Amendment challenge to a waiver as outside the scope of review. 463 In 2009, EPA approved a waiver (and authorization) under section 209(e), granting California authority to enforce its Airborne Toxic Control Measure, which established in-use emission performance standards for engines in transport refrigeration units (TRUs) and TRU generator sets. 464 Responding to comments that the waiver reached beyond California's borders in violation of the Dormant Commerce Clause, EPA stated that such considerations are not factors that EPA must consider under section 209(e) because "EPA's review of California's regulations is limited to the criteria that Congress directed EPA to review."465 This interpretation was upheld by the D.C. Circuit Court of Appeals. The Court agreed with EPA that the commenters had sought to "improperly . . . engraft

^{463.} MEMA I, 627 F.2d 1095, 1111, 1114-14 (D.C. Cir. 1979).

^{464. 74} FR 3030 (January 16, 2009).

^{465.} Decision Document, EPA-HQ-OAR-2005-0123-0049 at 67.

a type of constitutional Commerce Clause analysis onto EPA's Section 7543(e) waiver decisions that is neither present in nor authorized by the statute."⁴⁶⁶ Consistent with the Agency's long standing practice, the decision on whether to grant the ACC program waiver was based solely on criteria in section 209(b) and the Agency did not either interpret or apply the Equal Sovereignty Doctrine or any other constitutional or statutory provision in that waiver decision.⁴⁶⁷

Although EPA specified issues that it was seeking comment on within the Notice of Reconsideration, commenters nevertheless argued that the Equal Sovereignty Doctrine, which was not one of the identified aspects in that notice, preempts reinstitution of the relevant aspects of the ACC program waiver. According to these commenters, "Section 209, by allowing California and only California to retain a portion of its sovereign authority that the Clean Air Act takes from other States,

^{466.} ATA v. EPA, 600 F.3d 624, 628 (D.C. Cir. 2010) (quoting the U.S. brief). In a footnote to this statement, the Court said ATA could attempt to bring a constitutional challenge directly (which would argue that the waiver unconstitutionally burdens interstate commerce) but "express[ed] no view on that possibility." Id. at n.1. See also OOIDA v. EPA, 622 Fed. Appx. 4, 5 (D.C. Cir. 2015) (rejecting a challenge for lack of jurisdiction because challengers objected to the state regulations themselves, not EPA's approval of them in a waiver under 209(b)) ("To the extent there is any tension in our case law surrounding whether we might decide a constitutional claim brought within a broader challenge to an EPA waiver decision, OOIDA does not present us with such a challenge, and we have no occasion to resolve that question here.").

^{467. 78} FR at 2145.

is unconstitutional and thus unenforceable."⁴⁶⁸ Other commenters argued that the Equal Sovereignty doctrine does not apply to the California waiver program. One comment maintained that the holding in *Shelby County* v. *Holder* is distinguishable from the CAA.⁴⁶⁹ California disagreed with EPA's characterization of the relevance of the doctrine, commenting that the Supreme Court has only applied the "rarely invoked" doctrine of Equal

^{468.} Ohio and 15 States, Docket No. EPA-HQ-OAR-2021-0257-0124 at 1. This commenter also stated that "The waiver at issue here, allowing only California to regulate carbon emissions, is not sufficiently related to the problem that Section 209(a) targets, Congress enacted that section to permit California to address *local* air pollution. But California seeks special treatment for its proposed greenhouse gas targets . . . designed to mitigate climate change—an inherently global interest." *Id.* at 8-9. EPA notes that this characterization of CARB's standards is addressed in Section V.

^{469.} Twelve Public Interest Organizations at 5 (" Shelby County does not govern here. See Amicus Br. of Prof. Leah Litman 12-17, Union of Concerned Scientists v. NHTSA, No. 19-1230 (July 6, 2020) (A-0384). First, Clean Air Act Section 209(b) places no extraordinary burden or disadvantage on one or more States. Rather, the statute benefits California by allowing the exercise of its police power authority to address its particular pollution control needs. Second, the foundation for reserving California's authority has not waned over time. California had in 1967, and continues to have, the Nation's absolute worst air quality. For example, the South Coast air basin, home to 17 million people, typically leads the Nation in ozone (smog) pollution. The American Lung Association's 2021 'State of the Air' report on national air pollution shows that seven of the ten worst areas for ozone pollution in the country are in California, as are six of the worst ten for small particulate matter. Am. Lung Ass'n, Most Polluted Cities, https://www.lung.org/research/sota/city-rankings/ most-polluted-cities (last visited July 2, 2021) (A-0422).").

Sovereignty in the "rare instance where Congress undertook 'a drastic departure from basic principles of federalism' by authorizing 'federal intrusion into sensitive areas of state and local policymaking."⁴⁷⁰

As explained in the 2013 ACC program waiver decision, EPA continues to believe that waiver requests should be reviewed based solely on the criteria in section 209(b)(1) and specifically, that the Agency should not consider constitutional issues in evaluating waiver requests. As previously noted in Section VI, the constitutionality of section 209 is not one of the three statutory criteria for reviewing waiver requests, and such objections are better directed to either the courts or Congress. As the D.C. Circuit reasoned in *MEMA I*, "it is generally considered that the constitutionality of Congressional enactments is beyond the jurisdiction of administrative agencies." Although commenters here raise a new constitutional argument—that of Equal Sovereignty rather than the

^{470.} States and Cities at 41-42.

^{471. 78} FR at 2145.

^{472.} *MEMA I*, 627 F.2d 1095, 1114-15 (D.C. Cir. 1979) (holding that EPA did not need to consider whether California's standards "unconstitutionally burden[ed] [petitioners'] right to communicate with vehicle purchasers."). *See also* Twelve Public Interest Organizations at 7 ("As regulatory agencies are not free to declare an act of Congress unconstitutional,' *Springsteen-Abbott* v. *SEC*, 989 F.3d 4, 8 (D.C. Cir. 2021), EPA cannot determine whether a statute Congress directed it to implement contravenes the equal-sovereignty principle. Thus, EPA should proceed to rescind the Waiver Withdrawal and leave Ohio's argument for review by an appropriate court.").

First Amendment or the Dormant Commerce Clause—EPA is no more well-suited to resolve this constitutional objection than it is to resolve previous constitutional objections.⁴⁷³

EPA notes that Congress struck a deliberate balance in 1967 when it acknowledged California's serious air quality problems as well as it being a laboratory for the country, and once again in 1977 when Congress continued to acknowledge California's air quality problems as well as problems in other states and decided that California's new motor vehicle standards, once waived by EPA and subject to certain conditions, would be optionally available for all states under section 177 under specified criteria. 474 In striking a balance between one national standard and 51 different state standards, Congress chose to authorize two standards—the federal standard, and California's standards (which other states may adopt). EPA believes this balance reflected Congress's desire for California to serve as a laboratory of innovation and Congress's understanding of California's extraordinary pollution problems on the one hand, and its desire to ensure that

^{473.} See, e.g., Johnson v. Robison, 415 U.S. 361, 368, (1974) ("Adjudication of the constitutionality of congressional enactments has generally been thought beyond the jurisdiction of administrative agencies"); Springsteen-Abbott, 989 F.3d at 8; Meredith Corp. v. FCC, 809 F.2d 863, 872 (D.C. Cir. 1987).

^{474. &}quot;\$177... permitted other states to 'piggyback' onto California 's standards, if the state's standards 'are identical to the California standards for which a waiver has been granted for such model year." *Motor Vehicle Mfrs. Ass'n* v. *New York State Dep't of Envtl. Conservation*, 17 F.3d 521, 525 (2d Cir. 1994).

automakers were not subject to too many different standards on the other.

In reconsidering the SAFE 1 action and the appropriateness of reinstating the 2013 ACC program waiver, EPA has not considered whether section 209(a) and section 209(b) are unconstitutional under the Equal Sovereignty Doctrine. As in the 2013 ACC program waiver, the decision on whether to grant the waiver and the consequence of a reinstated waiver is based solely on the criteria in section 209(b) and this decision does not attempt to interpret or apply the Equal Sovereignty Doctrine or any other constitutional or statutory provision.

B. CARB's Deemed-To-Comply Provision

EPA received comments arguing that California's 2018 clarification to its deemed-to-comply provision "changed important underlying requirements of the original 2012 waiver application" and "EPA cannot reinstate a Clean Air Act waiver for a program that no longer exists." These commenters maintain that California has never sought a waiver for the 2018 amendments or a determination that the change is within the scope of the prior waiver. As such, commenters maintain that EPA lacks a necessary predicate to permit California's enforcement of its amended GHG standards.

Other commenters argued that the "deemed to comply" provision was always conditioned on the federal standards

^{475.} AFPM at 7; Urban Air at 2, 18-19; NADA at 6.

providing GHG reductions that were at least equal to or as protective as California's program and so the 2018 amendments did not substantively change the provision or affect any related reliance interests and instead were designed to clarify the provision. 476 Commenters maintain that CARB adopted "non-substantive amendments for its LEV III regulations to further clarify that the deemedto-comply provision would only apply if the federal GHG standards remained substantially as they were as of the date of the 2017 Final Determination."477 According to these commenters, California adopted these amendments after EPA's withdrawal of its 2017 Final Determination that had determined that its Federal GHG standards for model years 2022-2025 remained appropriate and instead concluded that the federal standards for model years 2022-2025 may be too stringent and should be revised. EPA notes that after the January 2017 MTE CARB subsequently found that compliance with those federal standards would result in equivalent or greater GHG benefits than originally projected for California. 478 These commenters further maintain that the clarification of the

^{476.} States and Cities at 58-61. ("California always intended its standards would 'remain an important backstop in the event the national program is weakened or terminated.' 78 FR at 2,128.").

^{477.} *Id.* at 60. "Final Determination on the Appropriateness of the Model Year 2022-2025 Light-Duty Vehicle Greenhouse Gas Emissions Standards under the Midterm Evaluation" (2017 Final Determination) at https://nepis.epa.gov/Exe/ZyPDF.cgi?Dockey=P100QQ91.pdf.

 $^{478.\ \ 82\} FR\ 14671$ (March 22, 2017) and 83 FR 16077 (April 13, 2018).

deemed-to-comply provision is immaterial to the reversal of the waiver withdrawal in SAFE 1 because the SAFE 1 action was expressly based on EPA's decision to rely on NHTSA's preemption findings and section 209(b)(1)(B) determination, neither of which was based on CARB's 2018 clarification rulemaking. As such, the commenters maintain that the clarification of the deemed-to-comply provision has no bearing on and does not preclude EPA's SAFE 1 waiver withdrawal.⁴⁷⁹

As previously explained, under section 209(b)(1) EPA is to grant a waiver of preemption for California to enforce its own standards that would otherwise be preempted under section 209(a). This preemption does not extend to federal standards that are adopted under section 202(a). EPA explained this in responding to comments on the deemed-to-comply provision in the ACC program waiver decision. "[T]he waiver decision affects only California's emission standards, not the federal standards that exist regardless of EPA's decision." This preemptive effect of section 209(a) does not change even when California chooses to allow for compliance with its standards through federal standards as envisaged by the deemed-to-comply provision.

It also bears note that in SAFE 1, EPA made clear that the 2018 amendment was not a "necessary part of the basis for the waiver withdrawal and other actions that

^{479.} States and Cities at 60-62.

^{480. 78} FR at 2124.

EPA finalizes in this [SAFE1] document. In the Notice of Reconsideration, EPA neither reopened nor reconsidered elements of the 2013 waiver that were not part of EPA's findings in SAFE 1. As noted in this decision, EPA has evaluated the factual and legal errors that occurred in SAFE 1. As part of this evaluation, EPA believes it has considered all appropriate and relevant information necessary to its review of issues associated with the second waiver prong or consideration of preemption under EPCA. The Agency also recognizes that it received comments from parties that raised non-germane issues to EPA's Notice of Reconsideration. EPA did not conduct an analysis of such comments in the context of reconsidering the specific actions taken in SAFE 1. EPA also makes clear that the result of rescinding its part of SAFE 1 is

^{481.} EPA declined to "take any position at this point on what effect California's December 2018 amendment to its "deemed to comply' provision . . . [may] have on the continued validity of the January 2013 waiver." 84 FR at 51329, n.208, 51334, n.230. Although EPA claimed in SAFE 1 that the deemed to comply clarification confirmed and provided further support for the SAFE 1 action, EPA no longer makes this claim to the extent it is relevant in its reconsideration and rescission of SAFE 1. The consequence of this action is the reinstatement of the ACC program waiver issued in 2013 and does not extend to other regulatory developments in California or by EPA that occurred subsequent to that waiver decision.

^{482. 86} FR at 22423. In addition to declining to take a position on the effect of California's 2018 amendments to its "deemed to comply" provision, SAFE 1 did not finalize the withdrawal of the waiver under the first or third waiver prongs. EPA also notes that it has previously responded twice to the comments suggesting that CARB's deemed-to-comply provision demonstrates that California does not have a need for its own standards. See 78 FR at 2124-25.

the automatic reinstatement of the waiver granted to California in 2013 for its ACC program. That is the result of the action taken herein.⁴⁸³

IX. Decision

After review of the information submitted by CARB and other public commenters, the SAFE 1 action, and the record pertaining to EPA's 2013 ACC program waiver, I find that EPA did not appropriately exercise its limited inherent authority to reconsider waiver grants in SAFE 1. SAFE 1 did not correct a clerical or factual error, nor did the factual circumstances and conditions related to the three statutory criteria change prior to SAFE 1, much less change so significantly as to cast the propriety of the waiver grant into doubt. On this basis, I am rescinding the SAFE 1 action.

^{483.} EPA acknowledges that motor vehicle emission standards in California as well as federally are periodically clarified, amended, or revised. For example, after California issued its first deemed-tocomply regulation, EPA determined that the state's GHG standards were within the scope of the 2009 waiver. While EPA believes that Congress intended regulatory certainty to be attached to the Agency's waivers issued under section 209, EPA acknowledges that conditions may change over time so significantly that it could merit a review of California's motor vehicle emission program and applicable standards therein or that would prompt California to submit a related waiver request to EPA. As explained in this decision, the conditions associated with the analysis of the three waiver criteria performed in the ACC waiver decision did not change so as to merit the SAFE 1 action. EPA recognizes that federal light-duty vehicle GHG emission standards have been modified twice since SAFE 1 was issued; the current standards do not change EPA's conclusion that SAFE 1 should be rescinded.

Furthermore, after review of both the 2013 ACC program waiver record as well as the SAFE 1 record, to the extent that EPA did have authority to reconsider the ACC program waiver, I have determined that the asserted bases were in error and did not justify the waiver withdrawal. With respect to the Agency's first purported basis—its discretionary decision to undertake a reinterpretation of the second waiver prong—I find that the statutory interpretation adopted in SAFE 1 is a flawed reading of the statute, and I hereby return to the traditional interpretation of the second waiver prong, which is, at least, the better interpretation. Under the traditional interpretation, which looks at the program as a whole, California clearly had a compelling need for the ACC program. Even if SAFE 1's statutory reinterpretation, which focuses on California's compelling need for the specific standards, were an appropriate reading, EPA did not perform a reasonable, accurate, and complete review of the factual record in its findings regarding the criteria emission benefits of CARB's ZEV sales mandate and GHG emission regulations. Upon review, I find that SAFE 1's predicate for concluding that California did not have a compelling need for these specific standards was not reasonable given the record at the time of the ACC program waiver and once again during the SAFE 1 proceeding. A reasonable, accurate, and complete review of the record supports the need for California's specific GHG emission standards and ZEV sales mandate to meet compelling and extraordinary conditions in California. This is true whether I look at how these standards reduce criteria pollution, GHG pollution, or both. In sum, although I am not adopting the interpretation of the second waiver prong set forth in SAFE 1, I find that the burden of proof necessary to demonstrate that CARB's ZEV sales mandate and GHG emission standards are not needed to meet compelling and extraordinary conditions has not been met under either interpretation of the second waiver prong. Therefore, I rescind the Agency's part of the SAFE 1 action to the extent it relied upon the second waiver prong to withdraw the ACC program waiver.

With regard to the applicability of preemption under EPCA, I find that, to the extent EPA's authority to reconsider the ACC program waiver rested upon NHTSA's joint action at the time as well as the applicability of its EPCA interpretation to EPA's review, this statute falls clearly outside the confines of section 209(b) where EPA's authority to grant, deny, and reconsider waivers resides. In any event, the grounds for such action under SAFE 1 no longer exist given NHTSA's recent final action withdrawing its EPCA preemption rule in its entirety.

Each of the decisions and justifications contained in this final action is severable.

This decision rescinds EPA's SAFE 1 action and therefore, as a result, the waiver of preemption EPA granted to California for its ACC program ZEV sales mandates and GHG emission standards issued in 2013, including for the 2017 through 2025 model years, comes back into force.

Judicial Review

Section 307(b)(1) of the CAA governs judicial review of final actions by EPA. This section provides, in part, that petitions for review must be filed in the Court of Appeals for the District of Columbia Circuit: (i) When the agency action consists of "nationally applicable regulations promulgated, or final actions taken, by the Administrator," or (ii) when such action is locally or regionally applicable, but "such action is based on a determination of nationwide scope or effect and if in taking such action the Administrator finds and publishes that such action is based on such a determination." For locally or regionally applicable final actions, the CAA reserves to EPA complete discretion whether to invoke the exception in (ii).

This final action is "nationally applicable" within the meaning of section 307(b)(1). In the alternative, to the extent a court finds this action to be locally or regionally applicable, the Administrator is exercising the complete discretion afforded to him under the CAA to make and publish a finding that this action is based on a determination of "nationwide scope or effect" within the meaning of section 307(b)(1).⁴⁸⁴ This action rescinds EPA's final action in SAFE 1, which withdrew a waiver

^{484.} In deciding whether to invoke the exception by making and publishing a finding that this final action is based on a determination of nationwide scope or effect, the Administrator has also taken into account a number of policy considerations, including his judgment balancing the benefit of obtaining the D.C. Circuit's authoritative centralized review versus allowing development of the issue in other contexts and the best use of agency resources.

for new motor vehicle greenhouse gas emission standards and ZEV sales mandate granted to California under section 209(b) of the CAA. In addition to California, sixteen other states and the District of Columbia have already adopted California's motor vehicle greenhouse gas standards. The other states are New York, Massachusetts, Vermont, Maine, Pennsylvania, Connecticut, Rhode Island, Washington, Oregon, Minnesota, New Jersey, Nevada, Maryland, Virginia, Colorado, and Delaware. 485 These jurisdictions represent a wide geographic area and fall within eight different judicial circuits. 486 In addition, this action will affect manufacturers nationwide who produce vehicles to meet the emissions standards of these states. For these reasons, this final action is nationally applicable or, alternatively, the Administrator is exercising the complete discretion afforded to him by the CAA and hereby finds that this final action is based on a determination of nationwide scope or effect for purposes of section 307(b)(1) and is hereby publishing that finding in the Federal Register.

Under CAA section 307(b)(1), petitions for judicial review of this action must be filed in the United States Court of Appeals for the District of Columbia Circuit

^{485.} The same states have adopted California's ZEV sales mandate regulation with the exception of Pennsylvania, Washington, and Delaware.

^{486.} In the report on the 1977 Amendments that revised CAA section 307(b)(1), Congress noted that the Administrator's determination that the "nationwide scope or effect" exception applies would be appropriate for any action that has a scope or effect beyond a single judicial circuit. See H.R. Rep. No. 95-294 at 323-24, reprinted in 1977 U.S.C.C.A.N. 1402-03.

within 60 days from the date this final action is published in the **Federal Register**.

X. Statutory and Executive Order Reviews

As with past waiver decisions, this action is not a rule as defined by Executive Order 12866. Therefore, it is exempt from review by the Office of Management and Budget as required for rules and regulations by Executive Order 12866.

In addition, this action is not a rule as defined in the Regulatory Flexibility Act, 5 U.S.C. 601(2). Therefore, EPA has not prepared a supporting regulatory flexibility analysis addressing the impact of this action on small business entities.

Further, Subtitle E of the Small Business Regulatory Enforcement Fairness Act of 1996, also known as the Congressional Review Act, 5 U.S.C. 801, et seq., does not apply because this action is not a rule for purposes of 5 U.S.C. 804(3).

Michael S. Regan,

Administrator.

[FR Doc. 2022-05227 Filed 3-11-22; 8:45 am]

BILLING CODE 6560-50-P

APPENDIX C

42 U.S.C. § 7507 provides:

New motor vehicle emission standards in nonattainment areas

Notwithstanding section 7543(a) of this title, any State which has plan provisions approved under this part may adopt and enforce for any model year standards relating to control of emissions from new motor vehicles or new motor vehicle engines and take such other actions as are referred to in section 7543(a) of this title respecting such vehicles if—

- (1) such standards are identical to the California standards for which a waiver has been granted for such model year, and
- (2) California and such State adopt such standards at least two years before commencement of such model year (as determined by regulations of the Administrator).

Nothing in this section or in subchapter II of this chapter shall be construed as authorizing any such State to prohibit or limit, directly or indirectly, the manufacture or sale of a new motor vehicle or motor vehicle engine that is certified in California as meeting California standards, or to take any action of any kind to create, or have the effect of creating, a motor vehicle or motor vehicle engine different than a motor vehicle or engine certified in California under California standards (a "third vehicle") or otherwise create such a "third vehicle". (July 14, 1955, ch. 360, title I, § 177, as added Pub. L. 95–95, title I, § 129(b), Aug. 7, 1977, 91 Stat. 750; amended Pub. L. 101–549, title II, § 232, Nov. 15, 1990, 104 Stat. 2529.)

42 U.S.C. § 7543 provides:

State standards.

(a) Prohibition

No State or any political subdivision thereof shall adopt or attempt to enforce any standard relating to the control of emissions from new motor vehicles or new motor vehicle engines subject to this part. No State shall require certification, inspection, or any other approval relating to the control of emissions from any new motor vehicle or new motor vehicle engine as condition precedent to the initial retail sale, titling (if any), or registration of such motor vehicle, motor vehicle engine, or equipment.

(b) Waiver

- (1) The Administrator shall, after notice and opportunity for public hearing, waive application of this section to any State which has adopted standards (other than crankcase emission standards) for the control of emissions from new motor vehicles or new motor vehicle engines prior to March 30, 1966, if the State determines that the State standards will be, in the aggregate, at least as protective of public health and welfare as applicable Federal standards. No such waiver shall be granted if the Administrator finds that—
- (A) the determination of the State is arbitrary and capricious,

- (B) such State does not need such State standards to meet compelling and extraordinary conditions, or
- (C) such State standards and accompanying enforcement procedures are not consistent with section 7521(a) of this title.
- (2) If each State standard is at least as stringent as the comparable applicable Federal standard, such State standard shall be deemed to be at least as protective of health and welfare as such Federal standards for purposes of paragraph (1).
- (3) In the case of any new motor vehicle or new motor vehicle engine to which State standards apply pursuant to a waiver granted under paragraph (1), compliance with such State standards shall be treated as compliance with applicable Federal standards for purposes of this subchapter.

(c) Certification of vehicle parts or engine parts

Whenever a regulation with respect to any motor vehicle part or motor vehicle engine part is in effect under section 7541(a)(2) of this title, no State or political subdivision thereof shall adopt or attempt to enforce any standard or any requirement of certification, inspection, or approval which relates to motor vehicle emissions and is applicable to the same aspect of such part. The preceding sentence shall not apply in the case of a State with respect to which a waiver is in effect under subsection (b).

(d) Control, regulation, or restrictions on registered or licensed motor vehicles

Nothing in this part shall preclude or deny to any State or political subdivision thereof the right otherwise to control, regulate, or restrict the use, operation, or movement of registered or licensed motor vehicles.

(e) Nonroad engines or vehicles

- (1) Prohibition on certain State standards No State or any political subdivision thereof shall adopt or attempt to enforce any standard or other requirement relating to the control of emissions from either of the following new nonroad engines or nonroad vehicles subject to regulation under this chapter—
- (A) New engines which are used in construction equipment or vehicles or used in farm equipment or vehicles and which are smaller than 175 horsepower.
- (B) New locomotives or new engines used in locomotives. Subsection (b) shall not apply for purposes of this paragraph.
- (2) Other nonroad engines or vehicles In the case of any nonroad vehicles or engines other than those referred to in subparagraph (A) or (B) of paragraph (1), the Administrator shall, after notice and opportunity for public hearing, authorize California to adopt and enforce standards and other requirements relating to the control of emissions from such vehicles or engines if California determines that California standards will be, in the aggregate, at least as

protective of public health and welfare as applicable Federal standards. No such authorization shall be granted if the Administrator finds that—

- (i) the determination of California is arbitrary and capricious,
- (ii) California does not need such California standards to meet compelling and extraordinary conditions, or
- (iii) California standards and accompanying enforcement procedures are not consistent with this section.
- (B) Any State other than California which has plan provisions approved under part D of subchapter I may adopt and enforce, after notice to the Administrator, for any period, standards relating to control of emissions from nonroad vehicles or engines (other than those referred to in subparagraph (A) or (B) of paragraph (1)) and take such other actions as are referred to in subparagraph (A) of this paragraph respecting such vehicles or engines if—
- (i) such standards and implementation and enforcement are identical, for the period concerned, to the California standards authorized by the Administrator under subparagraph (A), and
- (ii) California and such State adopt such standards at least 2 years before commencement of the period for which the standards take effect.

The Administrator shall issue regulations to implement this subsection.